KAR Auction Services, Inc.

Form 10-K

February 19, 2014

Table of Contents

UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended December 31, 2013

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission File Number: 001-34568

KAR Auction Services, Inc.

(Exact name of Registrant as specified in its charter)

Delaware

20-8744739

(State or other jurisdiction of

(I.R.S. Employer

incorporation or

Identification No.)

organization)

. .

13085 Hamilton Crossing Boulevard

Carmel, Indiana 46032

(Address of principal executive offices, including zip code)

Registrant's telephone number, including area code: (800) 923-3725

Securities registered pursuant to Section 12(b) of the Act:

Title of each class

Name of each exchange on which

registered

Common Stock, par value \$0.01 per

New York Stock Exchange

Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes ý No o

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes o No ý

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \circ No o Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\circ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \circ No o

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§ 229.405 of this chapter) is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. o Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \(\) Accelerated filer \(\) Non-accelerated filer \(\) Smaller reporting company \(\) Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes \(\) No \(\) The aggregate market value of the registrant's common stock held by stockholders who were not affiliates (as defined by regulations of the Securities and Exchange Commission) of the registrant was \$1,780,435,195 at June 30, 2013. As of February 7, 2014, 139,170,099 shares of the registrant's common stock, par value \$0.01 per share, were outstanding.

Documents Incorporated by Reference

Certain information required by Part III of this Annual Report on Form 10-K is incorporated by reference herein from the registrant's Definitive Proxy Statement for its 2014 Annual Meeting of Stockholders to be filed with the SEC within 120 days after the end of the registrant's fiscal year ended December 31, 2013.

Table of Contents

Index

Defined T		Page 3
<u>PART I</u>		
Item 1B. Item 2.	Business Risk Factors Unresolved Staff Comments Properties Legal Proceedings Mine Safety Disclosures	4 16 27 27 27 27 28
<u>PART II</u>		
Item 5. Item 6. Item 7. Item 7A. Item 8. Item 9. Item 9A. Item 9B.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities Selected Financial Data Management's Discussion and Analysis of Financial Condition and Results of Operations Quantitative and Qualitative Disclosures About Market Risk Financial Statements and Supplementary Data Changes in and Disagreements with Accountants on Accounting and Financial Disclosure Controls and Procedures Other Information	31 32 57 58 99 99
<u>Item 12.</u> <u>Item 13.</u>	Directors, Executive Officers and Corporate Governance Executive Compensation Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters Certain Relationships and Related Transactions, and Director Independence Principal Accounting Fees and Services	100 101 101 102 102
PART IV Item 15. Signatures 2	Exhibits, Financial Statement Schedules	103 104

DEFINED TERMS

Unless otherwise indicated or unless the context otherwise requires, the following terms used in this Annual Report on Form 10-K have the following meanings:

"we," "us," "our" and "the Company" refer, collectively, to KAR Auction Services, Inc. (formerly known as KAR Holdings, Inc.) and all of its subsidiaries;

"2007 Transactions" refers to the following events: On December 22, 2006, KAR LLC entered into a definitive merger agreement to acquire ADESA. The merger occurred on April 20, 2007. Concurrently with the merger, IAA was contributed by affiliates of Kelso & Company and Parthenon Capital and IAA's management to KAR Auction Services. Both ADESA and IAA became wholly-owned subsidiaries of KAR Auction Services, which was wholly-owned by KAR LLC prior to the initial public offering. KAR Auction Services is the accounting acquirer, and the assets and liabilities of both ADESA and IAA were recorded at fair value as of April 20, 2007;

"ADESA" refers, collectively, to ADESA, Inc., a wholly-owned subsidiary of KAR Auction Services, and ADESA, Inc.'s subsidiaries, including OPENLANE, Inc. (together with OPENLANE, Inc.'s subsidiaries, "OPENLANE");

"AFC" refers, collectively, to Automotive Finance Corporation, a wholly-owned subsidiary of ADESA, and Automotive Finance Corporation's subsidiaries and other related entities, including PWI Holdings, Inc.;

- "ALLETE" refers to ALLETE, Inc. the former parent company of ADESA;
- "AutoVIN" refers to AutoVIN, Inc., our wholly-owned subsidiary;
- "Axle LLC" refers to Axle Holdings, II, LLC, the former ultimate parent company of IAA and a holder of common equity interests in KAR LLC;
- "Credit Agreement" refers to the Credit Agreement, dated May 19, 2011, among KAR Auction Services, as the borrower, the several banks and other financial institutions or entities from time to time parties thereto and the administrative agent, as amended on November 29, 2012 and March 12, 2013;
- "2007 Credit Agreement" refers to the Credit Agreement, dated April 20, 2007, among KAR Auction Services, as the borrower, KAR LLC, as guarantor, the several lenders from time to time parties thereto and the administrative agent, the joint bookrunners, the co-documentation agents, the syndication agent and the joint lead arrangers named therein, as amended on June 10, 2009, October 23, 2009 and November 11, 2010. The 2007 Credit Agreement was terminated concurrently with our entry into the Credit Agreement described above;
- "Credit Facility" refers to the six year senior secured term loan facility ("Term Loan B") and the \$250 million, five year senior secured revolving credit facility, the terms of which are set forth in the Credit Agreement;
- "Equity Sponsors" refers, collectively, to Kelso Investment Associates VII, L.P., GS Capital Partners VI, L.P., ValueAct Capital Master Fund, L.P. and Parthenon Investors II, L.P.;
- "fixed senior notes" refers to KAR Auction Services' 8 3/4% Senior Notes due May 1, 2014. In June 2011, we prepaid the \$450.0 million aggregate principal amount outstanding on the fixed senior notes;
- "floating senior notes" refers to KAR Auction Services' Floating Rate Senior Notes due May 1, 2014. In April 2013, we prepaid the \$150.0 million aggregate principal amount outstanding on the floating rate senior notes;
- "IAA" refers, collectively, to Insurance Auto Auctions, Inc., a wholly-owned subsidiary of KAR Auction Services, and Insurance Auto Auctions, Inc.'s subsidiaries;
- *KAR Auction Services" refers to KAR Auction Services, Inc., and not to its subsidiaries;
- "KAR LLC" refers to KAR Holdings II, LLC, which is owned by affiliates of the Equity Sponsors, other equity co-investors and management of the Company;
- "notes" refers, collectively, to our senior notes and senior subordinated notes;
- "senior notes" refers, collectively, to the fixed senior notes and floating senior notes; and
- "senior subordinated notes" refers to KAR Auction Services' 10% Senior Subordinated Notes due May 1, 2015. In June 2011, we prepaid the \$131.1 million aggregate principal amount outstanding on the senior subordinated notes.

PART I

Item 1. Business

Overview

We are a leading provider of vehicle auction services in North America. We facilitate an efficient marketplace by providing auction services for sellers of used, or "whole car," vehicles and salvage vehicles through our 229 physical auction locations at December 31, 2013, and multiple proprietary Internet venues. In 2013, we facilitated the sale of approximately 3.7 million used and salvage vehicles. Our revenues are generated through auction fees from both vehicle buyers and sellers, as well as by providing value-added ancillary services, including transportation, reconditioning, inspections, marshalling, titling and floorplan financing. We facilitate the transfer of ownership directly from seller to buyer and generally we do not take title to or ownership of vehicles sold at our auctions. ADESA, our whole car auction services business, is the second largest provider of used vehicle auction services in North America. Vehicles at ADESA's auctions are typically sold by commercial fleet operators, financial institutions, rental car companies, used vehicle dealers and vehicle manufacturers and their captive finance companies to franchised and independent used vehicle dealers. In 2011, we acquired OPENLANE, the premier upstream platform focused on private label auction sales for automobile original equipment manufacturers, or OEMs, and their captive finance companies. Through ADESA.com, powered by OPENLANE technology, ADESA provides a comprehensive remarketing solution to automobile manufacturers, captive finance companies, lease and daily rental car companies, financial institutions and wholesale automobile auctions, IAA, our salvage auction services business, is one of the two largest providers of salvage auction services in North America. Vehicles at our salvage auctions are typically damaged or low-value vehicles that are predominantly sold by automobile insurance companies, non-profit organizations, automobile dealers, vehicle leasing companies and rental car companies to licensed dismantlers, rebuilders, scrap dealers or qualified public buyers. An important component of ADESA's and IAA's services to their buyers is providing short-term inventory-secured financing, known as floorplan financing, primarily to independent used vehicle dealers through our wholly-owned subsidiary, AFC.

At December 31, 2013, we had a network of 65 whole car auction locations and 164 salvage auction locations. Our auction locations are primarily standalone facilities dedicated to either whole car or salvage auctions; however, some of our sites are utilized to service both whole car and salvage customers at the same location. We believe our extensive geographic network and diverse product offerings enable us to leverage relationships with North American providers and buyers of used and salvage vehicles.

Our Corporate History

KAR Auction Services (formerly KAR Holdings, Inc.) was incorporated in 2006 and commenced operations in April 2007 upon the consummation of the 2007 Transactions. On November 3, 2009, we changed our name from KAR Holdings, Inc. to KAR Auction Services, Inc. ADESA entered the vehicle remarketing industry in 1989 and first became a public company in 1992. In 1994, ADESA acquired AFC. ADESA remained a public company until 1995 when ALLETE purchased a majority of its outstanding equity interests. In June 2004, ALLETE sold 20% of ADESA to the public and then spun off their remaining 80% interest to shareholders in September 2004. ADESA was acquired by the Company in April 2007. IAA entered the vehicle salvage business in 1982, and first became a public company in 1991. After growing through a series of acquisitions, IAA was acquired by affiliates of Kelso & Company and Parthenon Capital in 2005. Affiliates of Kelso & Company and Parthenon Capital and certain members of IAA management contributed IAA to KAR Auction Services in connection with the 2007 Transactions. In a series of transactions between December 2012 and November 2013, the Equity Sponsors sold all of their common stock in secondary offerings.

Our Industry

Auctions are the hub of the remarketing system for used and salvage vehicles, bringing professional sellers and buyers together and creating a marketplace for the sale of these vehicles. Whole car auction vehicles include vehicles from dealers turning their inventory, off-lease vehicles, vehicles repossessed by financial institutions and rental and other program fleet vehicles that have reached a predetermined age or mileage. The salvage vehicle auction industry provides a venue for sellers, primarily automobile insurance companies, to dispose or liquidate damaged or low value vehicles to dismantlers, rebuilders, scrap dealers or qualified public buyers. The following are key industry highlights:

Whole Car Auction Industry Volumes

During the period from 1999 to 2009, approximately 9 to 10 million used vehicles per year were sold in North America through whole car auctions. Whole car auction volumes, including online only volumes, were 8.0 million, 8.2 million and an estimated 8.7 million in 2011, 2012 and 2013, respectively. Data for the whole car auction industry is collected by the National Auto Auction Association ("NAAA") through an annual survey. NAAA industry volumes for 2013 have not yet been released; however, we estimate that used vehicle auction volumes in North America in 2013 will be approximately 8.7 million vehicles (including approximately 0.4 million vehicles sold online by ADESA prior to reaching a physical auction). The reduction in auction volumes since 2009 is attributable to supply shortages in the North American whole car auction industry caused principally by declines in new vehicle sales and lease originations from 2007 to 2009 and declines in repossessions from 2009 to 2012. We expect the industry to experience an increase in whole car auction volumes in 2014 as a result of increasing new vehicle sales and lease originations since 2009, as well as improving credit availability.

Salvage Auction Industry Volumes

We believe that the North American salvage vehicle auction industry volumes are affected primarily by accident rates, the age of the vehicle fleet on the road, miles driven, weather and the complexity of vehicles in operation. Vehicles deemed a total loss by automobile insurance companies represent the largest category of vehicles sold in the salvage vehicle auction industry. As vehicle design becomes more complex with additional enhancements, such as airbags and electrical components, vehicles are more costly to repair following an accident and insurance companies are more likely to declare a damaged vehicle a total loss. In addition, the utilization of recycled parts from salvage vehicles by the collision repair industry continues to increase as the quality of these parts gains wider acceptance and insurance companies attempt to reduce their repair claim costs. We believe that salvage volumes will continue to grow over time as the salvage auction industry expands.

Consolidated Whole Car and Salvage Auction Markets

The North American used vehicle auction market is largely consolidated. We estimate that Manheim, a subsidiary of Cox Enterprises, and ADESA together represent approximately 70% of the North American whole car auction market. We estimate that ADESA represents approximately 24% of the North American whole car auction market. The North American salvage vehicle auction market is also largely consolidated with the top two competitors, IAA and Copart, Inc., together representing over 80% of the market.

High Barriers to Entry

High barriers to entry make it difficult for new entrants to capture significant market share. The required investment in technology and related infrastructure in addition to ongoing maintenance costs required to meet customers' demands present challenges for new entrants. Large tracts of land and a significant investment in facilities and land improvements are required to build new physical auctions. In addition, the need to comply with regulatory requirements would pose a challenge for new entrants to build a large-scale operation. Larger participants are also able to better develop relationships with many of the major whole car and salvage sellers and buyers, which increases the sellers' flexibility to redistribute vehicles to markets where demand best matches supply in order to maximize proceeds, while also reducing the cost of disposition.

Our Business Strategy

We continue to focus on growing our revenues and profitability through the execution of the following key operating strategies:

Expand Opportunities for Customers to Buy and Sell Online

We are focused on enhancing our Internet solutions in all of the key channels we operate in, and we will continue to invest in our technology platforms to capitalize on new opportunities and attract new customers. Online vehicle remarketing solutions provide the opportunity to improve the customer experience, expand our volume of transactions and potentially increase proceeds for sellers through greater buyer participation at auctions. We acquired OPENLANE in order to better capitalize on the increasing use of the Internet as a means to purchase wholesale vehicles. Through its OPENLANE technology, ADESA offers comprehensive private label remarketing solutions to automobile manufacturers, captive finance companies, lease and daily rental car companies, financial institutions and wholesale automobile auctions throughout the United States and Canada. IAA is the only national salvage auction company that

offers buyers both live and Internet purchasing opportunities. ADESA provides online solutions to sell vehicles directly from a dealership or other interim storage location (upstream selling), online solutions to offer vehicles for sale while in transit to auction locations (midstream selling) and broadcasting video and audio of the physical auctions to online bidders (simulcast).

Grow Our Dealer Consignment Business

The dealer consignment business is a highly local market-specific business that requires local auction sales representatives who have experience in the used vehicle business and an intimate knowledge of their local market. We have augmented our local auction teams with the addition of corporate-level resources focused on growing the number of dealer vehicles sold at our physical and online auctions. The corporate team assists the local sales representatives in developing and implementing standard best practices for building and maintaining relationships with dealers to increase our market share. Our sales representatives also utilize proprietary technology solutions to maintain and grow the dealer consignment business by strategically matching the supply of vehicles with prospective buyers at auction. We believe this combination of a standard centralized approach with decentralized resources close to large populations of dealers will enhance our relationships with the dealer community and increase dealer volumes at our whole car auctions. On a same store basis, our dealer consignment sales volumes were up approximately 8% and 11% for the twelve months ended December 31, 2013 and 2012, respectively, compared with the same periods in prior years.

Continue to Grow Revenue per Vehicle

From 2009 through 2013, our whole car and salvage revenue per vehicle sold at our physical auction sites grew at compound annual growth rates of 3.0% and 4.7%, respectively. Revenue per vehicle generally consists of auction fees and fees from ancillary services. Increased utilization of ancillary services, selective fee increases, higher used vehicle prices and the introduction of new product offerings were key components of this growth. We believe these services provide economic benefits to our customers who are willing to utilize our products and services that improve their ability to manage their remarketing efforts and increase their returns. We plan to grow revenue by increasing customer utilization of these existing products and by enhancing our core auction services.

Leverage AFC's Products and Services at ADESA and IAA

We intend to selectively grow AFC while using enhanced credit analysis and risk management techniques to mitigate risk. We will continue to focus on expanding dealer coverage and improving coordination with ADESA and IAA to capitalize on cross-selling opportunities with AFC. By encouraging a collaborative marketing effort between AFC, ADESA and IAA, we believe we can market an enterprise solution more effectively to used vehicle dealers and tailor AFC's financing products to individual dealer needs. We will maintain our focus on generating additional revenues by expanding our suite of floorplan financing and related products and services and leveraging our market position, broad infrastructure and diversified business relationships to capitalize on current market opportunities.

Grow Our Non-Insurance Salvage Auction Customer Base

More than 12 million vehicles are de-registered annually, but only approximately 4 million are sold through salvage auctions, mostly by automobile insurance companies. In order to capture a greater portion of the total unit volume, we are increasingly focused on growing our vehicle supplier base, with a particular focus on non-insurance company customers, which includes charitable organizations, rental car, captive finance and fleet companies, as well as the general public. ADESA's strong customer relationships with used vehicle dealers as well as rental car, captive finance and fleet companies provide an advantage in accessing such segments, as these customers already use ADESA's whole car auction services.

Continue to Improve Operating Efficiency

We continue to focus on reducing costs by optimizing efficiency at each of our auction locations and consolidating certain management functions. Since 2007, a number of initiatives have been implemented, which have streamlined operations and improved operating efficiencies. As part of these initiatives, we introduced a management operating system to actively monitor and manage staffing levels and, as a result, have realized additional labor efficiency gains. Additional efficiencies have been gained through two of our wholly-owned subsidiaries, AuctionTrac, a vehicle tracking system at ADESA, and CarsArrive, an Internet-based system that allows customers to review instantly price quotes, delivery times, available transportation loads and also to receive instant notification of available shipments. In 2013, ADESA acquired High Tech Locksmiths ("HTL") and will integrate ADESA's key-cutting services, one of the many ancillary services provided by ADESA, with HTL.

Use Excess Cash Flow to Invest in Strategic Growth Initiatives, Return Capital to Shareholders and Reduce Debt

We generate strong cash flows as a result of our attractive gross margins, the ability to leverage our corporate infrastructure across our multiple auction locations, low levels of capital expenditures and limited working capital requirements. Management plans to utilize excess cash generated by the business to invest in strategic growth initiatives, return capital to shareholders and for debt reduction for the foreseeable future. We generated \$434.0 million and \$290.2 million of cash flow from operations for the twelve months ended December 31, 2013 and 2012, respectively. After paying any future dividends to shareholders (subject to prior declaration by our board of directors), we expect that significant cash flow will remain to support growth initiatives.

On April 4, 2013, July 3, 2013 and October 3, 2013, we paid a cash dividend of \$0.19 per share to stockholders of record at the close of business on March 25, 2013, June 24, 2013 and September 24, 2013, respectively. On January 3, 2014, we paid a cash dividend of \$0.25 per share to stockholders of record at the close of business on December 20, 2013. On February 18, 2014, we announced that our board of directors declared a cash dividend of \$0.25 per share payable on April 3, 2014 to stockholders of record at the close of business on March 26, 2014.

In April 2013, we made an excess cash flow payment of \$39.4 million on our senior secured term loan facility, or Term Loan B, and prepaid the entire \$150.0 million principal amount of the floating rate senior notes due 2014 with the additional proceeds received as part of the Second Amendment to the Credit Agreement.

Selective Acquisitions and Greenfield Expansion

Increased demand for single source solutions by our customers and other factors may increase our opportunities to acquire competitors. Both ADESA and IAA have a strong record of acquiring and integrating independent auction operations and improving profitability. We will continue to evaluate opportunities to open and acquire new sites in selected markets in order to effectively leverage our sales and marketing capabilities and expand our geographic presence for both ADESA and IAA. In addition, we may pursue opportunities to acquire additional product offerings in each of our business segments.

Increase Our International Presence

In both our whole car and salvage vehicle businesses, we have experience managing a global buyer base with relationships in over 100 countries. We believe we are well positioned to grow internationally through both technology-based and physical auction expansion. We continue to identify opportunities to expand certain of our service offerings globally and specifically plan to extend our technology. We expect that our ability to efficiently layer in our product and technology licensing will allow us to enter other mature auction markets.

Our Business Segments

We operate as three reportable business segments: ADESA Auctions, IAA and AFC. Our revenues for the year ended December 31, 2013 were distributed as follows: ADESA 52%, IAA 38% and AFC 10%. Geographic information as well as comparative segment revenues and related financial information pertaining to ADESA, IAA and AFC for the years ended December 31, 2013, 2012 and 2011 are presented in the tables in Note 20, Segment Information, to the Consolidated Financial Statements for KAR Auction Services, Inc., which are included under Item 8 in this Annual Report on Form 10-K.

ADESA

Overview

We are the second largest provider of whole car auctions and related services to the vehicle remarketing industry in North America. We serve our international customer base through online auctions and auction facilities that are developed and strategically located to draw professional sellers and buyers together and allow the buyers to inspect and compare vehicles remotely or in person. Our online service offerings include ADESA.com, LiveBlock and DealerBlock and allow us to offer vehicles for sale from any location.

Vehicles available at our auctions include vehicles from institutional customers such as off-lease vehicles, repossessed vehicles, rental vehicles and other program fleet vehicles that have reached a predetermined age or mileage and have been repurchased by the manufacturers, as well as vehicles from used vehicle dealers turning their inventory. The number of vehicles offered for sale at auction is the key driver of our costs incurred in the whole car auction process, and the number of vehicles sold is the key driver of the related fees generated by the remarketing process. We offer both online and physical auctions as well as value-enhancing ancillary services in an effective and efficient manner to maximize returns for the sellers of used vehicles. We quickly transfer the vehicles and ownership to the buyer and the net funds to the seller. Vehicles are typically offered for sale at the physical auctions on at least a weekly basis at most locations and the auctions are simulcast over the Internet with streaming audio and video (LiveBlock) so that remote bidders can participate via our online capabilities. Our online auctions (DealerBlock) function 24 hours a day, 7 days a week, providing our customers with maximum exposure for their vehicles and the flexibility to offer vehicles at buy now prices or in auctions that last for a few hours, days or even weeks. We also provide customized "private label" selling systems (including buy now functionality as well as online auctions) for our customers, primarily utilizing technology acquired with the purchase of OPENLANE.

We generate revenue primarily from auction fees paid by vehicle buyers and sellers. Generally, we do not take title to or bear the risk of loss for vehicles sold at whole car auctions. Our buyer fees and dealer seller fees are typically based on a tiered structure with fees increasing with the sale price of the vehicle, while institutional seller fees are typically fixed. We add buyer

fees to the gross sales price paid by buyers for each vehicle, and generally customers do not receive title or possession of vehicles after purchase until payment is received, proof of floorplan financing is provided or credit is approved. We generally deduct seller fees and other ancillary service fees to sellers from the gross sales price of each vehicle before remitting the net amount to the seller.

Customers

Suppliers of vehicles to our whole car auctions primarily include (i) large institutions, such as vehicle manufacturers and their captive finance arms, vehicle rental companies, financial institutions, and commercial fleets and fleet management companies (collectively "institutional customers"); and (ii) franchised and independent used vehicle dealers (collectively "dealer customers"). For the year ended December 31, 2013, no single supplier accounted for more than 5% of ADESA's revenues.

Buyers of vehicles at our whole car auctions primarily include franchised and independent used vehicle dealers. For the year ended December 31, 2013, no single buyer accounted for more than 2% of ADESA's revenues. Services

Our whole car auctions also provide a full range of innovative and value-added services to sellers and buyers that enable us to serve as a "one-stop shop." Many of these services may be provided or purchased independently from the auction process, including:

Services Description

Auction Related Services

ADESA provides marketing and advertising for the vehicles to be auctioned, dealer registration, storage of consigned and purchased inventory, clearing of funds, arbitration of disputes, auction vehicle registration, condition report processing, post-sale inspections, security for consigned inventory, title processing, sales results reports, pre-sale lineups and auctioning of vehicles by licensed auctioneers.

We provide both inbound (pickup) and outbound (delivery) transportation services utilizing our own equipment and personnel as well as licensed and insured third party carriers. Through our subsidiary, CarsArrive and its Internet-based system which provides automated vehicle shipping services, customers can instantly review price quotes and delivery times, and vehicle transporters can check available loads and also receive instant notification of available shipments. The same system is utilized at our whole car auction locations.

Transportation Services

Our auctions provide detailing, body work, paintless dent repair ("PDR"), light mechanical work, glass repair, tire and key replacement and upholstery repair.

Reconditioning Services

AutoVIN provides vehicle condition reporting, inventory verification auditing, program compliance auditing and facility inspections. Field managers are equipped with handheld computers and digital cameras to record all inspection and audit data on-site. The same technology is utilized at our whole car auction locations and we believe that the expanded utilization of comprehensive vehicle condition reports with pictures facilitates dealers sourcing vehicles via the Internet.

Inspection Services Provided By AutoVIN

Title and Repossession Administration and Remarketing Services Provided By PAR

PAR provides end-to-end management of the remarketing process including titling, repossession administration, inventory management, auction selection, pricing and representation of the vehicles at auction for those customers seeking to outsource all or just a portion of their

remarketing needs.

ADESA Analytical Services

ADESA Analytical Services provides value-added market analysis to our customers and the media. These services include access to publications and custom analysis of wholesale market trends for ADESA's customers, including peer group and market benchmarking studies, analysis of the benefits of reconditioning, site selection for optimized remarketing of vehicles, portfolio analysis of auction sales and computer-generated mapping and buyer analysis.

Sales and Marketing

Our sales and marketing approach at ADESA is to develop strong relationships and interactive dialogue with our customers. We have relationship managers for the various categories of institutional customers, including vehicle manufacturers, rental car companies, finance companies and others. These relationship managers focus on current trends and customer needs for their respective seller group in order to better coordinate our sales effort and service offerings.

Managers of individual auction locations are ultimately responsible for providing services to the institutional customers whose vehicles are directed to the auctions by the corporate sales team. Developing and servicing the largest possible population of buying dealers for the vehicles consigned for sale at each auction is integral to maximizing value for our vehicle suppliers.

We have local auction sales representatives who have experience in the used vehicle business and an intimate knowledge of local markets. These local representatives focus on the dealer segment and are complemented by local telesales representatives and are managed by a corporate-level team focused on developing and implementing standard best practices. We believe this combination of a centralized structure with decentralized resources enhances relationships with the dealer community and may further increase dealer consignment business at our auctions. Through our ADESA Analytical Services department, we also provide market analysis to our customers, as they use analytical techniques in making their remarketing decisions.

Online Solutions

Our current ADESA online solutions include:

Proprietary ADESA Technology

Description

ADESA.com and ADESA DealerBlock® This platform provides for either real-time or "bulletin-board" online auctions of consigned inventory at physical auction locations and is powered by the technology we acquired from OPENLANE in 2011. We also utilize this platform to provide upstream and midstream selling capabilities for our consignors, which facilitate the sale of vehicles prior to their arrival at a physical auction site. Auctions can be either closed (restricted to certain eligible dealers) or open (available to all eligible dealers) and inventory feeds of vehicles are automated with many customers' systems as well as third party providers that are integrated with various dealer management systems.

ADESA LiveBlock®

Our live auction Internet bidding solution, ADESA LiveBlock®, operates in concert with our physical auctions and provides registered buyers with the opportunity to participate in live auctions. Potential buyers bid online in real time along with the live local bidders and other Internet bidders via a simple, web-based interface. ADESA LiveBlock™ provides real-time streaming audio and video from the live auction and still images of vehicles and other data. Buyers inspect and evaluate the vehicle and listen to the live call of the auctioneer while viewing the physical auction that is underway.

ADESA Run List®

Provides a summary of consigned vehicles offered for auction sale, allowing dealers to preview inventory and vehicle condition reports prior to an auction event.

ADESA Market Guide®

Provides wholesale auction prices, auction sales results, market data and vehicle condition information.

ADESA Virtual Inventory

Subscription-based service to allow dealers to embed ADESA's search technology into a dealer's Web site to increase the number of vehicles advertised by the dealer.

Competition

In the North American whole car auction industry, we compete with Manheim, a subsidiary of Cox Enterprises, Inc., OVE.com (Manheim's "Online Vehicle Exchange"), SmartAuction, as well as several smaller chains of auctions and independent auctions, some of which are affiliated through their membership in industry associations. In the United States, competition is strongest with Manheim for the supply of used vehicles from national institutional customers. In Canada, we are the largest provider of whole car vehicle auction services. The supply of vehicles from dealers is dispersed among all of the auctions in the used vehicle market.

Due to the increased viability of the Internet as a marketing and distribution channel, new competition has arisen from Internet-based companies and our own customers who have historically remarketed vehicles through various channels, including auctions. Direct sales of vehicles by institutional customers and large dealer groups through internally developed or third-party online platforms have largely replaced telephonic and other non-auction methods, becoming a significant portion of overall used vehicle remarketing. The extent of use of direct, online systems varies by customer. In addition, we and some of our competitors offer online auctions in connection with physical auctions, and other online companies now include used vehicles among the products offered at their auctions.

IAA

Overview

As one of the leading providers of salvage vehicle auctions and related services in North America, we operate as Insurance Auto Auctions ("IAA") in the United States and Impact Auto Auctions in Canada and serve our customer base through salvage auction locations throughout North America. We facilitate the remarketing of vehicles for a variety of sellers including insurance companies, dealerships, rental car companies, fleet lease companies and charitable organizations. Our auctions provide buyers from around the globe with the salvage vehicles they need to fulfill their scrap demand, replacement part inventory or vehicle rebuild requirements. Fees for our services are earned from both sellers and buyers of salvage vehicles.

We process salvage vehicles primarily on a consignment basis. In return for agreed-upon fees, vehicles are sold on behalf of our sellers, which continue to own the vehicle until it is sold to buyers at auction. Other services available to vehicle sellers, for which fees may be charged, include towing, title processing, marketing and other administrative services. Under all methods of sale, we also charge the buyer of each vehicle fees based on a tiered structure that increases with the sale price of the vehicle as well as fixed fees for other services.

Auctions are typically held weekly at most locations. Vehicles are marketed at each respective auction site to live bidders as well as via an online auction list, allowing prospective bidders to preview and bid on vehicles prior to the actual auction event. IAA's Auction Center feature provides Internet buyers with an open, competitive bidding environment that reflects the dynamics of the live salvage auction. The Auction Center includes such services as comprehensive auction lists featuring links to digital images of vehicles available for sale, an "Auto Locator" function that promotes the search for specific vehicles within the auction system and special "Flood" or other catastrophe auction notifications. Higher prices at auction are generally driven by broader market exposure and increased competitive bidding. Our mobile device applications provide great flexibility for buyers to interact with our auctions. In 2013, we further enhanced these mobile applications with payment features allowing buyers to pay for their vehicles virtually anywhere. Our mobile applications are designed for the latest handheld devices, including Apple and Android, and are optimized for the most recent operating systems.

Online tools have also been developed to assist consignors in remarketing their vehicles and establishing salvage vehicle values. In 2013, we launched the IAA Market Value tool app via our CSAToday salvage management platform. The Market Value tool app allows customers to estimate the value of their vehicle anytime, anywhere. Through IAA's auction model, vehicles are offered simultaneously to live and online buyers in a live auction format utilizing i-Bid LIVESM. We believe the capability of the auction models maximizes auction proceeds and returns to our customers. First, the physical auctions allow buyers to inspect and compare the vehicles, thus enabling them to make fully-informed bidding decisions. These physical auction abilities are an important part of the bidding process. Second, our Internet auction capabilities allow buyers to participate in a greater number of auctions than if physical attendance was required. Online inventory browsing and digital alerts (via email or through buyer app) reduce the time required to acquire vehicles.

Services

IAA also offers a comprehensive suite of auction, logistics and vehicle selling services, aimed at maximizing the price vehicles sell at auction, lowering administrative costs, shortening the selling cycle and increasing the predictability of returns to vehicle sellers, while simultaneously expanding our ability to handle an increasing proportion of the vehicle-processing function as a "one-stop shop" for sellers. Some of the services provided by IAA include:

Services Description

Vehicles are offered simultaneously to live and online buyers in a live auction format utilizing i-Bid LIVESM technology. We believe this Live and Live Online Auction Model

exposes the vehicles to the maximum number of potential buyers.

inventory and cannot be auctioned until IAA receives and processes its transferable title. We process title documents in order to comply with the IAA Title Management Department of Motor Vehicles ("DMV") requirements for all vehicles.

Wherever possible, we maintain working relationships with each state's

After a vehicle is received at one of our facilities, it remains in storage

respective DMV to supplement an established electronic interface.

IAA Title Management services range from aging inventory reports to complete oversight of the title process. Our titling expertise results in faster cycle time, which decreases expenses for the vehicle suppliers like

insurance companies.

We maintain vehicle inspection centers ("VIC") at many of our facilities. A VIC is a temporary storage and inspection facility located at one of our

sites that is operated by the insurance company. Some of these sites are formalized through temporary license agreements with the insurance companies that supply the vehicles. Having a VIC minimizes vehicle

storage charges incurred by insurance company suppliers at the temporary storage facility or repair shop and also improves service time for the

policyholder.

Inbound logistics administration with actual services typically provided by Transportation and Towing

third-party carriers.

Focuses on vehicles, rental sellers, fleet and leasing companies, banks and Remarketing Division

dealer trade-in inventory.

Processes vehicles for a variety of charitable organizations across the **Donation Division**

United States and Canada, assisting them in turning donated vehicles into

cash to support their respective cause.

Customers

Vehicle Inspection Centers

We obtain IAA's supply of vehicles from insurance companies, non-profit organizations, automobile dealers and vehicle leasing and rental car companies, as well as the general public. We have established long-term relationships with virtually all of the major automobile insurance companies. The vast majority of the vehicles we process are on a consignment basis. For the year ended December 31, 2013, no single seller accounted for more than 4% of IAA's revenues.

The buyers of salvage vehicles include automotive body shops, rebuilders, used car dealers, automotive wholesalers, exporters, dismantlers, recyclers, brokers, and where allowed, non-licensed (public) buyers. For the year ended December 31, 2013, no single buyer accounted for more than 4% of IAA's revenues.

Sales and Marketing

The IAA sales force solicits prospective vehicle sellers and buyers at the national, regional and local levels. Branch managers address customer needs at the local level. We also participate in a number of local, regional and national trade show events that further promote the benefits of our products and services.

In addition to providing a variety of sellers with a means of processing and selling vehicles, IAA offers a comprehensive suite of services to help maximize returns and shorten the vehicle selling and processing time. We

work to establish workflow integration within our sellers' processes, and view such mutually beneficial relationships as an essential component of our effort to attract and retain suppliers.

By analyzing historical industry and customer data, we provide sellers with a detailed analysis of their current selling prices and returns, and a proposal detailing methods to improve selling prices and returns, reduce administrative costs and provide proprietary turn-key selling and processing services.

We also seek to expand our seller relationships through recommendations from individual customers at the local level to other local offices of the same company. Our broad and industry leading geographic coverage allows us to service sellers on a national basis.

Table of Contents

Online Solutions

Our current IAA online solutions include:

Proprietary IAA Technology

Description

i-Bid LIVESM

Our live auction Internet bidding solution, i-Bid LIVE, operates in concert with our physical auctions and provides registered buyers with the opportunity to participate in live auctions. Potential buyers bid online in real time along with the live local bidders and other Internet bidders via a simple, web-based interface. In addition, i-Bid LIVE provides real-time streaming audio from the live auction and images of salvage vehicles and other data. Buyers inspect and evaluate the salvage vehicle and listen to the auction while it is underway.

I-Buy FastSM

I-Buy Fast is an immediate buying option that allows qualified buyers to purchase vehicles between auctions for a fixed price. Each I-Buy Fast vehicle first runs at a previous auction where an established reserve price was not met.

CSAToday®

The process of salvage disposition through our system begins when a vehicle seller first consigns the vehicle to be sold through IAA via a variety of factors including a total loss, a recovered theft, a vehicle donation, a fleet vehicle retired, a vehicle repossessed, etc. A seller representative consigns the vehicle to us, either by phone, facsimile or electronically through CSAToday, our online proprietary salvage inventory management system.

With CSAToday, vehicle sellers enter vehicle data electronically and then track and manage the progress of vehicles in terms of both time and sales price. With this tool, they have 24-hour access to their vehicles. The information provided through this system ranges from the details associated with a specific vehicle, to comprehensive management reports for an entire area or geographic region. Additional features of this system include inventory management tools and a powerful new IAA Market ValueTM tool that helps customers determine the approximate value of a potential vehicle. This tool is helpful to adjusters when evaluating the "repair vs. total" decision. The management tools provided by CSAToday enable seller personnel to monitor and manage their vehicles more effectively. For example, insurance company sellers can also use CSAToday to view original garage receipts, verify ignition key availability, view settlement documents and images of the vehicles and receive updates of other current meaningful data.

Automated Salvage Auction Processing (ASAP)

We have developed a proprietary web-based information system, Automated Salvage Auction Processing system, or ASAP, to streamline all aspects of our operations and centralize operational data collection. The system provides sellers with 24-hour online access to powerful tools to manage the salvage disposition process, including inventory management, sales price analysis and electronic data interchange of titling information.

Our other information systems, including i-Bid LIVE and CSAToday systems, are integrated with our ASAP product, facilitating seamless auction processes and information flow with internal operational systems. Our technology platform is a significant competitive advantage that allows us to efficiently manage our business, improve customer selling prices, shorten customers' selling cycle and lower our customers' administration costs.

Competition

In the salvage sector, we compete with: Copart; Total Resource Auctions (a Manheim company); independent auctions, some of which are affiliated through their membership in industry organizations to provide broader coverage through network relationships; and a limited number of used vehicle auctions that regularly remarket salvage vehicles. Additionally, some dismantlers of salvage vehicles such as Greenleaf and LKQ Corporation and Internet-based companies have entered the market, thus providing alternate avenues for sellers to remarket vehicles. While most insurance companies have abandoned or reduced efforts to sell salvage vehicles without the use of service providers such as us, they may in the future decide to dispose of their vehicles directly to end users. In Canada, we are the largest provider of salvage vehicle auction services. Our competitors include Copart, independent vehicle auctions, brokers, online auction companies, and vehicle recyclers and dismantlers.

AFC

Overview

We are a leading provider of floorplan financing to independent used vehicle dealers. We provide short-term inventory-secured financing, known as floorplan financing, to independent used vehicle dealers through branches throughout North America. In 2013, AFC serviced over 1.3 million loan transactions, which includes both loans paid off and loans extended, or curtailed. We sell the majority of our U.S. dollar-denominated finance receivables without recourse to a wholly-owned bankruptcy remote special purpose entity, which sells an undivided participation interest in such finance receivables to a group of bank purchasers on a revolving basis. We also securitize the majority of our Canadian dollar denominated finance receivables through a separate third-party facility. We generate a significant portion of our revenues from fees. These fees include origination, floorplan, curtailment and other related program fees. When the loan is extended or paid in full, AFC collects all accrued fees and interest.

In June 2013, AFC acquired Preferred Warranties, Inc., a vehicle service contract business, as part of its strategy to provide new services to independent used vehicle dealers. We receive advance payments for the warranty contracts and unearned revenue is deferred and recognized over the terms of the contracts, which range from 3 months to 7 years, on an individual contract basis. The average term of these contracts originated in 2013 was approximately 1.5 years. We currently purchase program insurance which provides for satisfaction of certain of the Company's warranty and extended service contract related liabilities in the event the Company is unable to perform under the terms of specific warranty contracts covered by program insurance.

Customers and Locations

Floorplan financing supports independent used vehicle dealers in North America who purchase vehicles from our auctions, other auctions and non-auction purchases. In 2013, over 83% of the vehicles floorplanned by AFC were vehicles purchased by dealers at auction. Our ability to provide floorplan financing facilitates the growth of vehicle sales at auction. As of December 31, 2013, we serviced auctions through 105 locations which are conveniently located at or within close proximity of auctions held by ADESA and other auctions, which allows dealers to reduce transaction time by providing immediate payment for vehicles purchased at auction. We provide availability lists on behalf of our customers to auction representatives regarding the financing capacity of our customers, thereby increasing the purchasing potential at auctions. In addition, we have the ability to send finance representatives on-site to most approved independent auctions during auction sale-days, as well as maintaining a presence at the ADESA auctions. Geographic proximity to the customers gives our employees the ability to stay in close contact with outstanding accounts, thereby better enabling them to manage credit risk.

As of December 31, 2013, AFC had approximately 9,300 active dealers with an average line of credit of approximately \$185,000 and no one dealer representing greater than 1.4% of our portfolio. An average of approximately 14 vehicles per active dealer was floorplanned with an approximate average value outstanding of \$8,360 per vehicle as of December 31, 2013.

Sales and Marketing

AFC approaches and seeks to expand its share of the independent dealer floorplan market through a number of methods and channels. We target and solicit new dealers through both direct sales efforts at the dealer's place of business as well as auction-based sales and customer service representatives, who service our dealers at auctions where they replenish and rotate vehicle inventory. These largely local efforts are handled by branch managers, branch personnel and area sales managers. AFC's corporate-level team and Business Development Center provide sales and marketing support to AFC field personnel by helping to identify target dealers and coordinating promotional activity with auctions and other vehicle supply sources.

Credit

Our procedures and proprietary computer-based system enable us to manage our credit risk by tracking each vehicle from origination to payoff, while expediting services through our branch network. Typically, we assess a floorplan fee at the inception of a loan and we collect all accrued fees and interest when the loan is extended or repaid in full. In addition, AFC generally holds the title or other evidence of ownership to all vehicles which are floorplanned. Typical loan terms are 30 to 60 days, each with a possible loan extension. For an additional fee, this loan extension allows the dealer to extend the duration of the loan beyond the original term for another 30 to 60 days if the dealer makes

payment towards principal and pays accrued fees and interest.

The extension of a credit line to a dealer starts with the underwriting process. Credit lines up to \$300,000 are extended using a proprietary scoring model developed internally by AFC. Credit lines in excess of \$300,000 may be extended using underwriting guidelines which require dealership and personal financial statements and tax returns. The underwriting of each line of credit requires an analysis, write-up and recommendation by the credit department and, in case of credit lines in excess of \$300,000, final review by a credit committee.

Collateral Management

Collateral management is an integral part of daily operations at each AFC branch and our corporate headquarters. AFC's proprietary computer-based system facilitates this daily collateral management by providing real-time access to dealer information and enables branch and corporate personnel to assess and manage potential collection issues. Restrictions are automatically placed on customer accounts in the event of a delinquency, payments by dealers from bank accounts with insufficient funds or poor audit results. Branch personnel are proactive in managing collateral by monitoring loans and notifying dealers that payments are coming due. In addition, almost 70,000 routine audits, or lot checks, are performed annually on the dealers' lots through our AutoVIN subsidiary. Poor results from lot checks typically require branch personnel to take actions to determine the status of missing collateral, including visiting the dealer personally, verifying units held off-site and collecting payments for units sold. Audits also identify troubled accounts, triggering the involvement of AFC's collections department.

AFC operates two divisions which are organized into twelve regions in North America. Each division and region is monitored by managers who oversee daily operations. At the corporate level, AFC employs full-time collection specialists and collection attorneys who are assigned to specific regions and monitor collection activity for these areas. Collection specialists work closely with the branches to track trends before an account becomes a troubled account and to determine, together with collection attorneys, the best strategy to secure the collateral once a troubled account is identified.

Securitization

AFC sells the majority of its U.S. dollar denominated finance receivables on a revolving basis and without recourse to a wholly-owned, bankruptcy remote, consolidated, special purpose subsidiary ("AFC Funding Corporation"), established for the purpose of purchasing AFC's finance receivables. A securitization agreement allows for the revolving sale by AFC Funding Corporation to a group of bank purchasers of undivided interests in certain eligible finance receivables subject to committed liquidity. AFC's securitization facility has been in place since 1996. AFC Funding Corporation had committed liquidity of \$800 million from a third party facility for U.S. finance receivables at December 31, 2013. The agreement expires on June 30, 2016.

Also, we have an agreement in place for the securitization of Automotive Finance Canada, Inc.'s ("AFCI") receivables. This securitization facility provides up to C\$100 million in financing for eligible finance receivables through a third party conduit (separate from the U.S. facility). The agreement expires on June 30, 2016. The receivables sold pursuant to both the U.S. and Canadian securitization agreements are accounted for as secured borrowings.

Competition

AFC primarily provides short-term dealer floorplan financing of wholesale vehicles to independent vehicle dealers in North America. At the national level, AFC's competition includes NextGear Capital, a subsidiary of Manheim, other specialty lenders, banks and financial institutions. At the local level, AFC faces competition from banks, credit unions and independent auctions who may offer floorplan financing to local auction customers. Such entities typically service only one or a small number of auctions.

Some of our industry competitors who operate whole car auctions on a national scale may endeavor to capture a larger portion of the floorplan financing market. AFC competes primarily on the basis of quality of service, convenience of payment, scope of services offered and historical and consistent commitment to the sector. Our long-term relationships with customers have been established over time and act as a competitive strength for us. Seasonality

The volume of vehicles sold at our auctions generally fluctuates from quarter to quarter. This seasonality is caused by several factors including weather, the timing of used vehicles available for sale from selling customers, the availability and quality of salvage vehicles, holidays, and the seasonality of the retail market for used vehicles, which affects the demand side of the auction industry. Used vehicle auction volumes tend to decline during prolonged periods of winter weather conditions. In addition, mild weather conditions and decreases in traffic volume can each lead to a decline in the available supply of salvage vehicles because fewer traffic accidents occur, resulting in fewer damaged vehicles overall. As a result, revenues and operating expenses related to volume will fluctuate accordingly on a quarterly basis. The fourth calendar quarter typically experiences lower used vehicle auction volume as well as additional costs

associated with the holidays and winter weather.

Vehicle and Lending Regulation

Our operations are subject to regulation, supervision and licensing under various U.S., Canadian and Mexican federal, state, provincial and local authorities, agencies, statutes and ordinances, which, among other things, require us to obtain and maintain certain licenses, permits and qualifications, provide certain disclosures and notices and limit interest rates, fees and other charges. Some examples of the regulations and laws that impact our company are included in Item 1A "Risk Factors"

under the risk: "We are subject to extensive governmental regulations, including vehicle brokerage and auction laws and currency reporting obligations. Our business is subject to risks related to litigation and regulatory actions." Environmental Regulation

Our operations are subject to various foreign, federal, state and local environmental, health and safety laws and regulations, including those governing the emission or discharge of pollutants into the air or water, the generation, treatment, storage and release of hazardous materials and wastes and the investigation and remediation of contamination. Our failure to comply with current or future environmental, health or safety laws or to obtain and comply with permits required under such laws, could subject us to significant liability or require costly investigative, remedial or corrective actions.

In the used vehicle remarketing industry, large numbers of vehicles, including wrecked vehicles at salvage auctions, are stored and/or refurbished at auction facilities and during that time minor releases of fuel, motor oil and other materials may occur. We have investigated or remediated, or are currently investigating or remediating, contamination resulting from various sources, including gasoline, fuel additives (such as methyl tertiary butyl ether, or MTBE), motor oil, petroleum products and other hazardous materials released from aboveground or underground storage tanks or in connection with current or former operations conducted at our facilities. We have incurred, and may in the future incur, expenditures relating to releases of hazardous materials, investigative, remedial or corrective actions, claims by third parties and other environmental issues, and such expenditures, individually or in the aggregate, could be significant.

Federal and state environmental authorities are currently investigating IAA's role in contributing to contamination at the Lower Duwamish Waterway Superfund Site in Seattle, Washington. IAA's potential liability, if any, at this site cannot be estimated at this time. See Item 3, "Legal Proceedings" for a further discussion of this matter. Management considers the likelihood of loss or the incurrence of a liability, as well as the ability to reasonably estimate the amount of loss, in determining loss contingencies. We accrue an estimated loss contingency when it is probable that a liability has been incurred and the amount of loss (or range of possible losses) can be reasonably estimated. Management regularly evaluates current information available to determine whether accrual amounts should be adjusted. Accruals for contingencies including environmental matters are included in "Other accrued expenses" at undiscounted amounts and exclude claims for recoveries from insurance or other third parties. These accruals are adjusted periodically as assessment and remediation efforts progress, or as additional technical or legal information becomes available. If the amount of an actual loss is greater than the amount accrued, this could have an adverse impact on our operating results in that period.

Employees

At December 31, 2013, we had a total of 12,300 employees, of which 9,700 were located in the U.S. and 2,600 were located in Canada and Mexico. Approximately 70% of our workforce consists of full-time employees. Currently, none of our employees participate in collective bargaining agreements.

In addition to the employee workforce, we also utilize temporary labor services to assist in handling the vehicles consigned to us and to provide certain other services. Nearly all of our auctioneers are independent contractors. Some of the services we provide are outsourced to third party providers that perform the services either on-site or off-site. The use of third party providers depends upon the resources available at each auction facility as well as peaks in the volume of vehicles offered at auction.

Available Information

Our Web address is www.karauctionservices.com. Our electronic filings with the Securities and Exchange Commission ("SEC") (including all Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K, and if applicable, amendments to those reports) are available free of charge on the Web site as soon as reasonably practicable after they are electronically filed with, or furnished to, the SEC. In addition, our Corporate Governance Guidelines, Code of Conduct and Ethics, Code of Ethics for Principal Executive and Senior Financial Officers and charters of the audit committee, the nominating and corporate governance committee and the compensation committee of our board of directors are available on our Web site and available in print to any shareholder who requests it. The information posted on our Web site is not incorporated into this Annual Report.

Any materials that we file with the SEC may be read and copied at the SEC's Public Reference Room at 100 F Street, N.E., Washington, D.C. 20549. Information on the operation of the Public Reference Room may be obtained by calling the SEC at 1-800-SEC-0330. In addition, the SEC maintains an Internet Web site that contains reports, proxy statements and other information about issuers, like us, that file electronically with the SEC. The address of that site is www.sec.gov.

Item 1A. Risk Factors

Investing in our Company involves a high degree of risk. You should carefully consider the following risk factors, as well as all of the other information contained in this Annual Report on Form 10-K, before deciding to invest in our Company. The occurrence of any of the following risks could materially and adversely affect our business, financial condition, prospects, results of operations and cash flows. In such case, the trading price of our common stock could decline and you could lose all or part of your investment. These risks are not the only ones we face. Additional risks and uncertainties not presently known to us or that we currently deem immaterial may also materially affect our business, financial condition, results of operations and prospects.

Risks Related to Our Business

Adverse economic conditions may negatively affect our business and results of operations.

Future adverse economic conditions could increase our exposure to several risks, including:

Fluctuations in the supply of used vehicles. We are dependent on the supply of used vehicles coming to auction, and our financial performance depends, in part, on conditions in the automotive industry. During the past global economic downturn and credit crisis, there was an erosion of retail demand for new and used vehicles that led many lenders to cut back on originations of new loans and leases and led to significant manufacturing capacity reductions by automakers selling vehicles in the United States and Canada. Capacity reductions could depress the number of vehicles received at auction in the future and could lead to reduced program vehicles and rental fleet sales, negatively impacting auction volumes. In addition, weak growth in or declining new vehicle sales negatively impacts used vehicle trade-ins to dealers and auction volumes. These factors could adversely affect our revenues and profitability.

Decline in the demand for used vehicles. We may experience a decrease in demand for used vehicles from buyers due to factors including the lack of availability of consumer credit and declines in consumer spending and consumer confidence. Adverse credit conditions also affect the ability of dealers to secure financing to purchase used vehicles at auction, which further negatively affects buyer demand. In addition, a reduction in the number of franchised and independent used car dealers may reduce dealer demand for used vehicles.

Decrease in the supply and demand of salvage vehicles. If the number of miles driven decreases, the number of salvage vehicles received at auction may also decrease. In addition, decreases in commodity prices, such as steel and platinum, may negatively affect vehicle values and demand at salvage auctions. In addition, if consumers eliminate their automotive collision coverage, this could result in fewer vehicles being declared a total loss. Additionally, government regulations on the standards for producing vehicles could affect the supply of vehicles at salvage auctions.

Decrease in consumer spending. Consumer purchases of new and used vehicles may be adversely affected by economic conditions such as employment levels, wage and salary levels, trends in consumer confidence and spending, reductions in consumer net worth, interest rates, inflation, the availability of consumer credit and taxation policies. Consumer purchases in general may decline during recessions, periods of prolonged declines in the equity markets or housing markets and periods when disposable income and perceptions of consumer wealth are lower. Changes to U.S. federal tax policy may negatively affect consumer spending. To the extent retail demand for new and used vehicles decreases, negatively impacting our auction volumes, our results of operations and financial position could be materially and adversely affected.

Volatility in the asset-backed securities market. Volatility and disruption in the asset-backed commercial paper market could lead to a narrowing of interest rate spreads at AFC in certain periods. In addition, any volatility and disruption has affected, and could affect, AFC's cost of financing related to its securitization facility.

•

Ability to service and refinance indebtedness. Uncertainty in the financial markets may negatively affect our ability to service our existing debt, access additional financing or to refinance our existing indebtedness on favorable terms or at all. If economic weakness exists, it may affect our cash flow from operations and results of operations, which may affect our ability to service payment obligations on our debt or to comply with our debt covenants.

Increased counterparty credit risk. Any market deterioration could increase the risk of the failure of financial institutions party to our Credit Agreement and other counterparties with which we do business to honor their obligations to us. Our ability to replace any such obligations on the same or similar terms may be limited if challenging credit and general economic conditions exist.

Our business is dependent on information and technology systems. Failure to effectively maintain or update these systems could result in us losing customers and materially adversely affect our operating results and financial condition.

Robust information systems are critical to our operating environment and competitive position, including with respect to our online auctions. We may not be successful in structuring our information system infrastructure or developing, acquiring or implementing information systems which are competitive and responsive to the needs of our customers and we might lack sufficient resources to continue to make the significant necessary investments in information systems to compete with our competitors. Certain information systems initiatives that management considers important to our long-term success will require capital investment, have significant risks associated with their execution, and could take several years to implement. We may not be able to develop/implement these initiatives in a cost-effective, timely manner or at all.

Our information and technology systems are vulnerable to damage or interruption from computer viruses, network failures, computer and telecommunications failures, infiltration by unauthorized persons and security breaches, usage errors by our employees, power outages and catastrophic events such as fires, tornadoes, floods, hurricanes and earthquakes. If these systems were compromised, not operable for extended periods of time or ceased to function properly, we may have to make a significant investment to fix or replace them and our ability to provide many of our electronic and online solutions to our customers may be impaired. If that were to occur, it could have a material adverse effect on our operating results and financial condition.

Aspects of our operations and business are subject to privacy regulation in the United States and elsewhere. Many U.S. states have enacted data breach regulations and laws requiring varying levels of consumer notification in the event of a security breach. Increased regulation and enforcement activity throughout the world in the areas of data privacy and data security/breach may materially increase our costs, which could have a material adverse effect on our operating results. Our failure to comply with the privacy and data security/breach laws to which we are subject could also result in fines, sanctions and damage to our reputation and trade names or the loss of significant customers.

Used vehicle prices have a significant effect on fee revenue per unit at IAA and loan losses at AFC and may impact the supply of used vehicles at ADESA.

The volume of new vehicle production, accuracy of lease residual estimates, interest rates, customer demand and changes in regulations, among other things, all potentially affect the pricing of used vehicles. Used vehicle prices may affect the volume of vehicles entered for sale at our used vehicle auctions and the demand for those used vehicles, the fee revenue per unit at our salvage auctions, loan losses for our dealer financing business and our ability to retain customers. Throughout 2011 and 2012, used vehicle prices remained high, which led many used vehicle dealers to retail more of their trade-in vehicles on their own rather than selling them at auction. The high used vehicle prices in 2011 and 2012 also contributed to strong proceeds in the salvage auction industry. Throughout 2013, used vehicle prices decreased moderately. A sustained reduction in used vehicle pricing could result in lower proceeds from the sale of salvage vehicles and a related reduction in revenue per vehicle, a potential loss of consignors, an increase in loan losses at AFC and decreased profitability.

Decreases in the supply of used vehicles coming to auction have impacted and may continue to impact auction sales volumes, which may adversely affect our revenues and profitability.

The number of new and used vehicles that are leased by consumers affects the supply of vehicles coming to auction in future periods as the leases mature. If manufacturers and other lenders decrease the number of new vehicle lease originations and extend the terms of some of the existing leases, the number of off-lease vehicles available at auction

for the industry would decline. Based on our estimates, off-lease vehicles available at auction for the industry declined slightly in 2010 and declined by over 40% in 2011, reflecting declines in lease origination volume and new vehicle sales approximately three years earlier during the global economic downturn and credit crisis. If the supply of off-lease vehicles coming to auction declines, our revenues and profitability may be adversely affected. Our expectation is that the decline in off-lease vehicles will continue to impact Canada, where we have a significant auction market presence.

Volumes of off-lease vehicles in subsequent periods will be affected by total new vehicle sales and the future leasing behavior of manufacturers and lenders; therefore, we may not be able to accurately predict the volume of vehicles coming to auction. The supply of off-lease vehicles coming to auction is also affected by the market value of used vehicles compared to the residual value of those vehicles per the lease terms. In most cases, the lessee and the dealer have the ability to purchase the vehicle at the residual price at the end of the lease term. Generally, as market values of used vehicles rise, the number of vehicles purchased at residual value by the lessees and dealers increases, thus decreasing the number of off-lease vehicles available at auction.

In 2009, the auction industry sales volume was over 9 million units. However, auction sales volumes, including online only sales, declined to approximately 8.0 million units in 2011, 8.2 million units in 2012 and an estimated 8.7 million units in 2013. We believe that auction sales volumes will continue to recover over the next several years, and we estimate volumes, including online only volumes, will be approximately 9.1 million, 9.5 million and 9.7 million units in 2014, 2015 and 2016, respectively. We believe that an increase in the volume of off-lease and repossessed vehicles, among others, that are remarketed through whole car auctions are a significant contributor to this growth, and our performance could be adversely impacted if volumes do not increase and we are not able to reduce our costs permanently to compensate for the lower industry auction volumes.

An increase in the number of used and salvage vehicles purchased on virtual auction platforms could adversely affect our revenue per vehicle, operating results and financial condition.

We acquired OPENLANE in recognition of the increasing use of the Internet as a means to purchase wholesale vehicles. If sellers and buyers increase the number of vehicles transacted on virtual auction platforms, our revenue per vehicle will likely decline. In connection with online auctions, ADESA and IAA offer physical auctions, which allow buyers to physically inspect and compare vehicles. In addition, our cost structure includes a significant fixed cost component, including occupancy costs, that cannot be readily reduced if revenue per vehicle declines. If a shift in the percentage of used and salvage vehicles sold online as compared with used and salvage vehicles sold at physical auctions occurs, and we are unable to generate new sources of revenue, our operating results and financial condition could be adversely affected.

We face significant competition and may not successfully adapt to industry changes, which may adversely affect our business and results of operations.

We face significant competition for the supply of used and salvage vehicles, the buyers of those vehicles and the floorplan financing of these vehicles. Our principal sources of competition historically have come from: (1) direct competitors (e.g., Manheim, Copart and NextGear Capital), (2) new entrants, including new vehicle remarketing venues and dealer financing services, and (3) existing alternative vehicle remarketing venues. Due to the increasing use of the Internet and other technology as marketing and distribution channels, we also face increasing competition from online wholesale and retail vehicle selling platforms (generally without any meaningful physical presence) and from our own customers when they sell directly to end users through such platforms rather than remarket vehicles through our auctions and other channels. Increased competition could result in price reductions, reduced margins or loss of market share.

Our future success also depends on our ability to respond to evolving industry trends, changes in customer requirements and new technologies. One potentially adverse trend would be a market shift towards the simultaneous listing of vehicles on multiple online sales platforms. Were such trend to take hold, the vehicle remarketing industry's economics could change. For example, we might need to incur additional costs or otherwise alter our business model to adapt to these changes. In such case, the volume of vehicles supplied to us and our overall revenues and fees per vehicle sold could decrease. We are discussing with one or more other auction houses and industry participants the development of a multiple platform listing system. Any such collaboration may be unsuccessful.

Some of our competitors may have greater financial and marketing resources than we do, may be able to respond more quickly to evolving industry dynamics and changes in customer requirements, or may be able to devote greater resources to the development, promotion and sale of new or emerging services and technologies. If we are unable to compete successfully or to successfully adapt to industry changes, our business, revenues and profitability could be materially adversely affected.

ADESA currently competes with online wholesale and retail vehicle selling platforms, including OVE.com (affiliated with Manheim), SmartAuction, eBay Motors and others. With the exception of OVE.com, these online selling platforms generally do not have any meaningful physical presence and may cause the volume of vehicles sold through our online and physical auctions to decrease. If the number of vehicles sold at our auctions decreases due to these competitors or other industry changes such as the use of a multiple platform listing system, our revenue and profitability may be negatively impacted.

In our salvage auction business, potential competitors include used vehicle auctions, providers of claims software to insurance companies and certain salvage buyer groups and automobile insurance companies, some of which currently supply salvage vehicles to us. Insurance companies may in the future decide to dispose of their salvage vehicles directly to end users, which would negatively affect our volumes, revenue and profitability.

ADESA and IAA's agreements with its largest institutional suppliers of used and salvage vehicles are generally subject to cancellation by either party upon 30 to 90 days' notice. In addition, it is common that institutional suppliers regularly review their relationships with whole car and salvage auctions through written requests for proposals. Such suppliers may from time to time require us to make changes to the way we do business as part of the request for proposal process. There can be no assurance that our existing agreements will not be canceled or that we will be able to enter into future agreements with these or other suppliers on similar terms, or at all, and our ability to grow and sustain profitability could be impaired.

Fluctuations in the supply of and demand for salvage vehicles impact auction sales volumes, which may adversely affect our revenues and profitability.

We are dependent upon receiving a sufficient number of total loss vehicles to sustain profit margins in our salvage auction business. Factors that can adversely affect the number of vehicles received include, but are not limited to, a decrease in the number of vehicles in operation or miles driven, mild weather conditions that cause fewer traffic accidents, reduction of policy writing by insurance providers that would affect the number of claims over a period of time, a decrease in the percentage of claims resulting in a total loss, delays or changes in state title processing, and changes in direct repair procedures that would reduce the number of newer, less damaged total loss vehicles, which tend to have higher salvage values. In addition, our salvage auction business depends on a limited number of automobile insurance companies to supply the salvage vehicles we sell at auction. Our agreements with these insurance company suppliers are generally subject to cancellation by either party upon 30 to 90 days' notice. There can be no assurance that our existing agreements will not be canceled or that we will be able to enter into future agreements with these suppliers. Future decreases in the quality and quantity of vehicle inventory, and in particular the availability of newer and less damaged vehicles, could have a material adverse effect on our operating results and financial condition. If the supply of salvage vehicles coming to auction declines significantly, our revenues and profitability may be adversely affected. Furthermore, in periods when the supply of vehicles from the insurance sector declines, salvage operators have acquired and in the future may acquire vehicles on their own. If we purchase vehicles, the increased costs associated with acquiring the vehicles could have a material adverse effect on our gross profit and operating results.

We may not successfully implement our business strategies or maintain gross profit margins.

We are pursuing strategic initiatives that management considers critical to our long-term success, including but not limited to growing market share and volume, increasing revenue per vehicle and improving customer experiences through Internet initiatives, leveraging AFC's products and services at ADESA and IAA and continuing to improve operating efficiency. There are significant risks involved with the execution of these initiatives, including significant business, economic and competitive uncertainties, many of which are outside of our control. Accordingly, we cannot predict whether we will succeed in implementing these strategic initiatives. For example, if we are unsuccessful in continuing to generate significant cash provided by operations (we generated \$434.0 million and \$290.2 million of cash flow from operations for the years ended December 31, 2013 and 2012, respectively), we may be unable to reduce our outstanding indebtedness, which could negatively affect our financial position and results of operations and our ability to execute our other strategies. It could take several years to realize any direct financial benefits from these initiatives if any direct financial benefits from these initiatives are achieved at all. Additionally, our business strategy may change from time to time, which could delay our ability to implement initiatives that we believe are important to our business.

If we are unable to successfully acquire and integrate other businesses, our growth prospects could be adversely affected.

Acquisitions have been a part of our historical growth and have enabled us to further broaden and diversify our service offerings. Our strategy generally involves the acquisition and integration of additional physical auction sites, technologies and personnel. Acquisition of businesses requires substantial time and attention of management personnel and may also require additional equity or debt financings. Further, integration of newly established or acquired businesses is often disruptive. Since we have acquired or in the future may acquire one or more businesses, there can be no assurance that we will identify appropriate targets, will acquire such businesses on favorable terms, or will be able to successfully integrate such organizations into our business. Failure to do so could materially adversely affect our business, financial condition and results of operations. In addition, we expect to compete against other auction groups or new industry consolidators for suitable acquisitions. If we are able to consummate acquisitions, such acquisitions could be dilutive to earnings, and we could overpay for such acquisitions.

In pursuing a strategy of acquiring other businesses, we face other risks including, but not limited to:

*ncurring significantly higher capital expenditures and operating expenses;

entering new markets with which we are unfamiliar;

Table of Contents

incurring potential undiscovered liabilities at acquired businesses;

failing to maintain uniform standards, controls and policies;

impairing relationships with employees and customers as a result of management changes; and

increasing expenses for accounting and computer systems, as well as integration difficulties.

Changes in interest rates or market conditions could adversely impact the profitability and business of AFC.

Rising interest rates may have the effect of depressing the sales of used vehicles because many consumers finance their vehicle purchases. In addition, AFC securitizes a majority of its finance receivables on a revolving basis. Volatility and/or market disruption in the asset-backed securities market in the United States or Canada can impact AFC's cost of financing related to, or its ability to arrange financing on acceptable terms through, its securitization facility, which could negatively affect AFC's business and our financial condition and operations.

In addition, AFC typically assesses its U.S. dealer customers with an interest rate comprised of a minimum prime rate of 5% plus an interest spread. The U.S. prime rate was 3.25% at December 31, 2013. Any increase above 3.25% up to 5% will likely compress AFC's margins as the variable borrowing cost of securitizing receivables will rise without a corresponding increase in the financing rate AFC charges to its U.S. dealer customers.

Increases in the value of the U.S. dollar relative to certain foreign currencies may negatively impact foreign buyer participation at our auctions.

We have a significant number of non-U.S. based buyers who participate in our auctions. Increases in the value of the U.S. dollar relative to these buyers' local currencies may reduce the prices they are willing to pay at auction, which may negatively affect our revenues.

Weather-related and other events beyond our control may adversely impact operations.

Extreme weather or other events, such as hurricanes, tornadoes, earthquakes, forest fires, floods, terrorist attacks or war, may adversely affect the overall economic environment, the markets in which we compete, our operations and profitability. These events may impact our physical auction facilities, causing a material increase in costs, or delays or cancellation of auction sales, which could have a material adverse impact on our revenues and profitability. In some instances, for example with the severe storm in October 2012, known as "Superstorm Sandy," these events may result in a sharp influx in the available supply of salvage vehicles and there can be no assurance that our salvage auction business will have sufficient resources to handle such extreme increases in supply. Our failure to meet our customers' demands in such situations could negatively affect our relationships with such customers and result in a loss of future business, which would adversely affect our operating results and financial condition. In addition, salvage revenues generated as a result of the total loss of vehicles associated with such a catastrophe are typically recognized subsequent to the incurrence of incremental costs and such revenues may not be sufficient to offset the costs incurred.

Mild weather conditions tend to result in a decrease in the available supply of salvage vehicles because traffic accidents decrease and fewer vehicles are damaged. Accordingly, mild weather can have an adverse effect on our salvage vehicle inventories, which would be expected to have an adverse effect on our revenue and operating results and related growth rates.

High fuel prices could lead to a reduction in miles driven and may have an adverse effect on our revenues and operating results, as well as our earnings growth rates.

High fuel prices could lead to a reduction in the miles driven per vehicle, which may reduce accident rates. High fuel prices may also disproportionately affect the demand for sports cars, luxury vehicles, sport utility and full-sized vehicles which are generally not as fuel-efficient as smaller vehicles. Retail sales and accident rates are factors that affect the number of used and salvage vehicles sold at auction, wholesale prices of those vehicles and the conversion rates at used vehicle auctions. Additionally, high fuel costs increase the cost of transportation and towing of vehicles and we may not be able to pass on such higher costs to our customers.

Table of Contents

A portion of our net income is derived from our international operations, primarily Canada, which exposes us to foreign exchange risks that may impact our financial statements.

Fluctuations between U.S. and foreign currency values may adversely affect our results of operations and financial position, particularly fluctuations with Canadian currency values. In addition, there may be tax inefficiencies in repatriating cash from Canada. Approximately 15% of our revenues were attributable to our Canadian operations for the year ended December 31, 2013. A decrease in the value of the Canadian currency relative to the U.S. dollar would reduce our profits from Canadian operations and the value of the net assets of our Canadian operations when reported in U.S. dollars in our financial statements. This could have a material adverse effect on our business, financial condition or results of operations as reported in U.S. dollars.

In addition, fluctuations in exchange rates may make it more difficult to perform period-to-period comparisons of our reported results of operations. For purposes of accounting, the assets and liabilities of our Canadian operations are translated using period-end exchange rates; such translation gains and losses are reported in "Accumulated other comprehensive income/loss" as a component of stockholders' equity. The revenues and expenses of our Canadian operations are translated using average exchange rates during each period.

Environmental, health and safety risks could adversely affect our operating results and financial condition.

Our operations are subject to various foreign, federal, state and local environmental, health and safety laws and regulations, including those governing the emission or discharge of pollutants into the air or water, the generation, treatment, storage and release of hazardous materials and wastes and the investigation and remediation of contamination. Our failure to comply with current or future environmental, health or safety laws or to obtain and comply with permits required under such laws, could subject us to significant liability or require costly investigative, remedial or corrective actions.

In the used vehicle remarketing industry, large numbers of vehicles, including wrecked vehicles at salvage auctions, are stored and/or refurbished at auction facilities and during that time minor releases of fuel, motor oil and other materials may occur. We have investigated or remediated, or are currently investigating or remediating, contamination resulting from various sources, including gasoline, fuel additives (such as methyl tertiary butyl ether, or MTBE), motor oil, petroleum products and other hazardous materials released from aboveground or underground storage tanks or in connection with current or former operations conducted at our facilities. We have incurred and may in the future incur expenditures relating to releases of hazardous materials, investigative, remedial or corrective actions, claims by third parties and other environmental issues, and such expenditures, individually or in the aggregate, could be significant.

Federal and state environmental authorities are currently investigating IAA's role in contributing to contamination at the Lower Duwamish Waterway Superfund Site in Seattle, Washington. IAA's potential liability at this site cannot be estimated at this time. See Item 3, "Legal Proceedings" for a further discussion of this matter.

We have a substantial amount of debt, which could impair our financial condition and adversely affect our ability to react to changes in our business.

As of December 31, 2013, our total debt was approximately \$1.8 billion, exclusive of liabilities related to our securitization facilities, and we had \$250.0 million of borrowing capacity under our senior secured credit facilities. In addition, we had related outstanding letters of credit in the aggregate amount of \$26.3 million at December 31, 2013, which reduce the amount available for borrowings under the credit facilities.

Our substantial indebtedness could have important consequences including:

limiting our ability to borrow additional amounts to fund working capital, capital expenditures, debt service requirements, execution of our business strategy, acquisitions and other purposes;

requiring us to dedicate a substantial portion of our cash flow from operations to pay principal and interest on debt, which would reduce the funds available for other purposes, including funding future expansion;

making us more vulnerable to adverse changes in general economic, industry and competitive conditions, in government regulation and in our business by limiting our flexibility in planning for, and making it more difficult to react quickly to, changing conditions; and

Table of Contents

exposing us to risks inherent in interest rate fluctuations because the majority of our indebtedness is at variable rates of interest, which could result in higher interest expenses in the event of increases in interest rates.

In addition, if we are unable to generate sufficient cash from operations to service our debt and meet other cash needs, we may be forced to reduce or delay capital expenditures, suspend or eliminate dividends, sell assets or operations, seek additional capital or restructure or refinance our indebtedness. We may not be able to refinance our debt or sell additional debt or equity securities or our assets on favorable terms, if at all, particularly because of our high levels of debt and the restrictions imposed by the agreement governing our Credit Facility on our ability to incur additional debt and use the proceeds from asset sales. If we must sell certain of our assets, it may negatively affect our ability to generate revenue. The inability to obtain additional financing could have a material adverse effect on our financial condition.

If we cannot make scheduled payments on our debt, we would be in default and, as a result:

our debt holders could declare all outstanding principal and interest to be due and payable;

the lenders under our senior secured credit facilities could terminate their commitments to lend us money and foreclose against the assets securing their borrowings; and

we could be forced into bankruptcy or liquidation.

Restrictive covenants in agreements governing our debt may adversely affect our ability to operate our business.

The agreement governing our Credit Facility contains, and future debt instruments may contain, various provisions that limit our ability and the ability of our subsidiaries, including ADESA and IAA, to, among other things:

incur additional debt:

provide guarantees in respect of obligations of other persons;

issue redeemable stock and preferred stock;

pay dividends or distributions or redeem or repurchase capital stock;

prepay, redeem or repurchase certain debt;

make loans, investments and capital expenditures;

incur liens;

pay dividends or make other payments by our restricted subsidiaries;

enter into certain transactions with affiliates;

sell assets and capital stock of our subsidiaries; and

consolidate or merge with or into, or sell substantially all of our assets to, another person.

We assume the settlement risk for all vehicles sold through our auctions.

We do not have recourse against sellers for any buyer's failure to satisfy its payment obligations. Since revenue for most vehicles does not include the gross sales proceeds, failure to collect the receivables in full may result in a net loss up to the gross sales proceeds on a per vehicle basis in addition to any expenses incurred to collect the receivables and to provide the services associated with the vehicle. If we are unable to collect payments on a large number of vehicles, the resulting payment obligations to the seller and decreased fee revenues may have a material adverse effect on our results of operations and financial condition.

Table of Contents

Changes in laws affecting the importation of salvage vehicles may have an adverse effect on our business and financial condition.

Our Internet-based auction services have allowed us to offer our products and services to international markets and has increased our international buyer base. As a result, foreign importers of salvage vehicles now represent a significant part of our total buyer base. Changes in laws and regulations that restrict the importation of salvage vehicles into foreign countries may reduce the demand for salvage vehicles and impact our ability to maintain or increase our international buyer base. For example, in March 2008, a decree issued by the president of Mexico became effective that placed restrictions on the types of vehicles that can be imported into Mexico from the United States. The adoption of similar laws or regulations in other jurisdictions that have the effect of reducing or curtailing our activities abroad could have a material adverse effect on our results of operations and financial condition by reducing the demand for our products and services.

We are partially self-insured for certain losses.

We self-insure a portion of employee medical benefits under the terms of our employee health insurance program, as well as a portion of our automobile, general liability and workers' compensation claims. We record an accrual for the claims expense related to our employee medical benefits, automobile, general liability and workers' compensation claims based upon the expected amount of all such claims. If actual trends, including the severity of claims and medical cost inflation above expectations were to occur, our self-insured costs would increase, which could have an adverse impact on the operating results in that period.

If we fail to attract and retain key personnel, we may not be able to execute our business strategy and our financial results could be negatively affected.

Our success depends in large part on the performance of our executive management team and other key employees, including key field personnel. If we lose the services of one or more of our executive officers or key employees, or if one or more of them decides to join a competitor or otherwise compete with us, we may not be able to effectively implement our business strategies, our business could suffer and the value of our common stock could be materially adversely affected. Our auction business is directly impacted by the business relationships our employees have established with customers and suppliers and, as a result, if we lose key personnel, we may have difficulty in retaining and attracting customers, developing new services, negotiating favorable agreements with customers and providing acceptable levels of customer service. Leadership changes will occur from time to time and we cannot predict whether significant resignations will occur or whether we will be able to recruit additional qualified personnel. We do not have nor do we currently expect to obtain key person insurance on any of our executive officers.

We are subject to extensive governmental regulations, including vehicle brokerage and auction laws and currency reporting obligations. Our business is subject to risks related to litigation and regulatory actions.

Our operations are subject to regulation, supervision and licensing under various U.S., Canadian and Mexican federal, state, provincial and local authorities, agencies, statutes and ordinances, which, among other things, require us to obtain and maintain certain licenses, permits and qualifications, provide certain disclosures and notices and limit interest rates, fees and other charges. The regulations and laws that impact our company include, without limitation, the following:

The acquisition and sale of used, leased, totaled and recovered theft vehicles are regulated by state or other local motor vehicle departments in each of the locations in which we operate.

•

Some of the transport vehicles used at our auctions are regulated by the U.S. Department of Transportation or similar regulatory agencies in Canada and Mexico.

In many states and provinces, regulations require that a salvage vehicle be forever "branded" with a salvage notice in order to notify prospective purchasers of the vehicle's previous salvage status.

Some state, provincial and local regulations limit who can purchase salvage vehicles, as well as determine whether a salvage vehicle can be sold as rebuildable or must be sold for parts or scrap only.

AFC is subject to laws in certain states and in Canada which regulate commercial lending activities and interest rates and, in certain jurisdictions, require AFC or one of its subsidiaries to be licensed.

Table of Contents

We are subject to various local zoning requirements with regard to the location of our auction and storage facilities, which requirements vary from location to location.

Certain of the Company's subsidiaries, including ADESA, are indirectly subject to the regulations of the Consumer Financial Protection Act of 2010, or the CFPA, due to their vendor relationships with financial institutions.

Changes in law or governmental regulations or interpretations of existing law or regulations could result in increased costs, reduced vehicle prices and decreased profitability for us. In addition, failure to comply with present or future laws and regulations or changes in existing laws or regulations or in their interpretation could have a material adverse effect on our operating results and financial condition.

We are also subject from time to time to a variety of legal actions relating to our current and past business operations, including litigation relating to employment-related issues, the environment and insurance claims. There is no guarantee that we will be successful in defending ourselves in legal and administrative actions or in asserting our rights under various laws. In addition, we could incur substantial costs in defending ourselves or in asserting our rights in such actions. The costs and other effects of pending litigation and administrative actions against us cannot be determined with certainty. Although we currently believe that no such proceedings will have a material adverse effect, there can be no assurance that the outcome of such proceedings will be as expected.

If we are unable to protect our intellectual property, the value of our brand and other intangible assets may be diminished, and our business may be adversely affected.

We rely and expect to continue to rely on a combination of confidentiality and license agreements with our employees, consultants and third parties with whom we have relationships, as well as trademark, copyright, patent, trade secret, and domain name protection laws, to protect our proprietary rights. In the United States and internationally, we have filed various applications for protection of certain aspects of our intellectual property, and we currently hold issued patents in the United States. However, third parties may knowingly or unknowingly infringe our proprietary rights, third parties may challenge proprietary rights held by us, and pending and future trademark and patent applications may not be approved. In addition, effective intellectual property protection may not be available in every country in which we operate or intend to operate our business. In any or all of these cases, we may be required to expend significant time and expense in order to prevent infringement or to enforce our rights. Although we have taken measures to protect our proprietary rights, there can be no assurance that others will not offer products or concepts that are substantially similar to ours and compete with our business. If the protection of our proprietary rights is inadequate to prevent unauthorized use or appropriation by third parties, the value of our brand and other intangible assets may be diminished and competitors may be able to more effectively mimic our service and methods of operations. Any of these events could have an adverse effect on our business and financial results.

We may be subject to patent or other intellectual property infringement claims, which could have an impact on our business or operating results due to a disruption in our business operations, the incurrence of significant costs and other factors.

From time to time, we may receive notices from others claiming that we infringed or otherwise violated their patent or intellectual property rights, and the number of these claims could increase in the future. Claims of intellectual property infringement or other intellectual property violations could require us to enter into licensing agreements on unfavorable terms, incur substantial monetary liability or be enjoined preliminarily or permanently from further use of the intellectual property in question, which could require us to change business practices and limit our ability to compete effectively. Even if we believe that the claims are without merit, the claims can be time-consuming and costly to defend and may divert management's attention and resources away from our businesses. If we are required to take any of these actions, it could have an adverse impact on our business and operating results.

We are dependent on the continued and uninterrupted service from our workforce.

Currently, none of our employees are covered by collective bargaining agreements. If we negotiate a first-time collective bargaining agreement, we could be subject to a substantial increase in labor and benefits expenses that we may be unable to pass through to customers for some period of time, if at all. While the current Congressional political deadlock is unlikely to produce any new legislation expanding the rights of labor unions to organize, the National Labor Relations Board ("NLRB" or "the Board") is moving unilaterally via its rulemaking authority to push several amendments to its rules and regulations that could adversely affect our operations. On February 4, 2014 the Board issued for public comment a series of new rules that, if adopted, will make union organizing drives much easier to start and much more difficult for the company to defeat. Some of the changes include shortening the time for an election from 30 days to 10 (reducing the opportunity for the company to campaign against organization), strictly limit the time and scope of employer objections to union efforts and reduce the possibility for

Table of Contents

Board review of election results. Additionally, the NLRB is granting organization rights to smaller and smaller groups of employees, meaning that it may allow an election to take place involving only detailers, or only mechanics, thereby introducing a union foothold of 5-10 employees into the midst of an otherwise non-union facility. This could start a chain reaction leading to organization of entire auctions.

We have a material amount of goodwill which, if it becomes impaired, would result in a reduction in our net income.

Goodwill represents the amount by which the cost of an acquisition accounted for using the purchase method exceeds the fair value of the net assets acquired. Current accounting standards require that goodwill no longer be amortized but instead be periodically evaluated for impairment based on the fair value of the reporting unit. A significant percentage of our total assets represents goodwill primarily associated with the 2007 Transactions. Declines in our profitability or the value of comparable companies may impact the fair value of our reporting units, which could result in a write-down of goodwill and a reduction in net income.

New accounting pronouncements or new interpretations of existing standards could require us to make adjustments to accounting policies that could adversely affect the financial statements.

The Financial Accounting Standards Board, or the FASB, the Public Company Accounting Oversight Board, the SEC, and other accounting organizations or governmental entities from time to time issue new pronouncements or new interpretations of existing accounting and auditing standards that require changes to our accounting policies and procedures and could cause us to incur additional costs. To date, we do not believe any new pronouncements or interpretations have had a material adverse effect on our financial condition or results of operations, but future pronouncements or interpretations could require the change of policies or procedures.

Risks Related to Ownership of Our Common Stock

The market price and trading volume of our common stock may be volatile, which could result in rapid and substantial losses for our stockholders.

You should consider an investment in our common stock to be risky, and you should invest in our common stock only if you can withstand a significant loss and wide fluctuations in the market value of your investment. Many factors could cause the market price of our common stock to rise and fall, including the following:

our announcements or our competitors' announcements regarding new products or services, enhancements, significant contracts, acquisitions or strategic investments;

changes in earnings estimates or recommendations by securities analysts, if any, who cover our common stock;

results of operations that are below our announced guidance or below securities analysts' or consensus estimates or expectations;

fluctuations in our quarterly financial results or the quarterly financial results of companies perceived to be similar to us;

changes in our capital structure, such as future issuances of securities, sales of large blocks of common stock by our stockholders or our incurrence of additional debt;

investors' general perception of us and our industry;

changes in general economic and market conditions in North America;

changes in industry conditions; and

changes in regulatory and other dynamics.

In addition, if the market for stocks in our industry, or the stock market in general, experiences a loss of investor confidence, the trading price of our common stock could decline for reasons unrelated to our business, financial condition or results of operations. If any of the foregoing occurs, it could cause our stock price to fall and may expose us to lawsuits that, even if successfully defended, could be costly to defend and a distraction to management.

Table of Contents

Future offerings of debt or equity securities, which would rank senior to our common stock, may adversely affect the market price of our common stock.

If, in the future, we decide to issue debt or equity securities that rank senior to our common stock, it is likely that such securities will be governed by an indenture or other instrument containing covenants restricting our operating flexibility. Additionally, any convertible or exchangeable securities that we issue in the future may have rights, preferences and privileges more favorable than those of our common stock and may result in dilution to owners of our common stock. We and, indirectly, our stockholders, will bear the cost of issuing and servicing such securities. Because our decision to issue debt or equity securities in any future offering will depend on market conditions and other factors beyond our control, we cannot predict or estimate the amount, timing or nature of our future offerings. Thus, holders of our common stock will bear the risk of our future offerings reducing the market price of our common stock and diluting the value of their stock holdings in us.

The market price of our common stock could be negatively affected by sales of substantial amounts of our common stock in the public market.

Future sales by us or by our existing stockholders of substantial amounts of our common stock in the public market, or the perception that these sales may occur, could cause the market price of our common stock to decline. These sales also could impede our ability to raise future capital. Under our amended and restated certificate of incorporation, we are authorized to issue up to 400,000,000 shares of common stock, of which 139,027,581 shares of common stock were outstanding as of December 31, 2013. In addition, pursuant to a registration statement under the Securities Act, we have registered shares of common stock reserved for issuance in respect of stock options and other incentive awards granted to our officers and certain of our employees. If any of these holders cause a large number of securities to be sold in the public market, the sales could reduce the trading price of our common stock. We cannot predict the size of future sales of shares of our common stock or the effect, if any, that future sales, or the perception that such sales may occur, would have on the market price of our common stock.

Provisions in our amended and restated certificate of incorporation and by-laws, and of Delaware law, may prevent or delay an acquisition of us, which could decrease the trading price of our common stock.

Our amended and restated certificate of incorporation and by-laws contain provisions that may be considered to have an anti-takeover effect and may delay or prevent a tender offer or other corporate transaction that a stockholder might consider to be in its best interest, including those transactions that might result in a premium over the market price for our shares.

These provisions include:

limiting the right of stockholders to call special meetings of stockholders to holders of at least 35% of our outstanding common stock:

rules regarding how our stockholders may present proposals or nominate directors for election at stockholder meetings;

permitting our board of directors to issue preferred stock without stockholder approval;

granting to the board of directors, and not the stockholders, the sole power to set the number of directors;

authorizing vacancies on our board of directors to be filled only by a vote of the majority of the directors then in office and specifically denying our stockholders the right to fill vacancies in the board;

authorizing the removal of directors only for cause and only upon the affirmative vote of holders of a majority of the outstanding shares of our common stock entitled to vote for the election of directors; and

prohibiting stockholder action by written consent.

These provisions apply even if an offer may be considered beneficial by some stockholders.

You may not receive any future dividends on our common stock.

On November 30, 2012, we announced that our board of directors approved the initiation of a quarterly cash dividend on our common stock. Holders of our common stock are only entitled to receive such dividends as our board of directors may declare out of funds legally available for such payments. We are not required to declare cash dividends on our common stock. Future dividend decisions will be based on and affected by a variety of factors, including our financial condition and results of operations, contractual restrictions, including restrictive covenants contained in our Credit Agreement and AFC's securitization facilities, capital requirements and other factors that our board of directors deems relevant. Therefore, no assurance can be given as to whether any future dividends may be declared by our board of directors or the amount thereof.

Item 1b. Unresolved Staff Comments

None.

Item 2. Properties

The corporate headquarters of KAR Auction Services, ADESA and AFC are located in Carmel, Indiana. The facilities are leased properties, with office space being leased in each case through 2019. At December 31, 2013, properties utilized by the ADESA business segment include 65 used vehicle auction facilities in North America, which are either owned or leased. Each auction is generally a multi-lane, drive-through facility, and may have additional buildings for reconditioning, registration, maintenance, bodywork, and other ancillary and administrative services. Each auction also has secure parking areas to store vehicles. The ADESA auction facilities vary in size based on the market demographics and offer anywhere from 1 to 16 auction lanes, with an average of approximately 7 lanes per location. IAA is headquartered in Westchester, Illinois, with office space being leased through 2016. At December 31, 2013, properties utilized by the IAA business segment include 164 salvage vehicle auction facilities in the United States and Canada, most of which are leased. Salvage auctions are generally smaller than used vehicle auctions in terms of acreage and building size and some locations share facilities with ADESA. The IAA properties are used primarily for auction and storage purposes consisting on average of approximately 28 acres of land per site.

Of AFC's 105 locations in North America at December 31, 2013, 73 are physically located at auction facilities (including 53 at ADESA and 4 at IAA). Each of the remaining AFC offices is strategically located in close proximity to at least one of the auctions that it serves. AFC generally leases its branches.

We believe our existing properties are adequate to meet current needs and that suitable additional space will be available as needed to accommodate any expansion of operations and additional offices on commercially acceptable terms.

Item 3. Legal Proceedings

We are involved in litigation and disputes arising in the ordinary course of business, such as actions related to injuries; property damage; handling, storage or disposal of vehicles; environmental laws and regulations; and other litigation incidental to the business such as employment matters and dealer disputes. Such litigation is generally not, in the opinion of management, likely to have a material adverse effect on our financial condition, results of operations or cash flows. Legal and regulatory proceedings which could be material are discussed below.

IAA—Lower Duwamish Waterway

On March 25, 2008, the United States Environmental Protection Agency, or the "EPA," issued a General Notice of Potential Liability, or "General Notice," pursuant to Section 107(a), and a Request for Information pursuant to Section 104(e) of the Comprehensive Environmental Response, Compensation, and Liability Act, or "CERCLA" to IAA for a Superfund site known as the Lower Duwamish Waterway Superfund Site ("LDW Site") in Seattle, Washington. On November 7, 2012, the EPA issued a Second General Notice of Potential Liability, or "Second General Notice," to IAA for the LDW Site. Since 2004, IAA has operated a branch on property it leases in Tukwila, Washington, which is located adjacent to the LDW Site. The LDW Site was identified as a Superfund site in 2001, three years before IAA began leasing the branch in Tukwila. At this time, the EPA has not demanded that IAA pay any funds or take any action apart from responding to the Section 104(e) Information Request. The EPA's website indicates that the EPA has issued notice letters to approximately 116 entities, and has issued Section 104(e) Requests to more than 300 entities. Four Potentially Responsible Parties, or "PRPs," The Boeing Company, the City of Seattle,

the Port of Seattle and King County, have funded a remedial investigation and feasibility study of the LDW Site, but the EPA has not yet issued a final plan for remediating the site. IAA is aware that certain authorities may bring natural resource damage claims against PRPs. In the General Notice and Second General Notice, the EPA informed IAA that the EPA believes IAA may be a PRP, but the EPA has not specified the factual basis for this assertion. At this time, the Company does

not have adequate information to determine IAA's responsibility for contamination at this site, if any, or to estimate IAA's loss as a result of this potential liability.

In addition, the Washington State Department of Ecology is working with the EPA in relation to the LDW Site, primarily to investigate and address sources of potential contamination contributing to the LDW Site. The current Tukwila property owner, the former Tukwila property owner and IAA have had discussions with the Washington State Department of Ecology concerning possible source control obligations, including an investigation of the water and soils entering the stormwater system, an analysis of the source of any contamination identified within the system and possible repairs and upgrades to the stormwater capture and filtration system. In 2011, IAA submitted results of its stormwater system investigation to comply with the Washington State Department of Ecology source control requirements. Additional source control obligations, if any, are not expected to have a material adverse effect on future recurring operating costs.

Item 4. Mine Safety Disclosures Not applicable.

PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

Market Information and Holders of Record

KAR Auction Services' common stock is traded on the New York Stock Exchange ("NYSE") under the symbol "KAR" and has been traded on the NYSE since December 11, 2009. As of February 7, 2014, there were 10 stockholders of record. Because many shares of our common stock are held by brokers and other institutions on behalf of stockholders, we are unable to estimate the total number of stockholders represented by these holders of record. The following table sets forth the range of high and low intraday sales prices per share of common stock for each quarter during fiscal years 2013 and 2012:

	2013	
	High	Low
4th Quarter (October 1 - December 31)	\$30.32	\$26.97
3rd Quarter (July 1 - September 30)	\$28.67	\$22.96
2nd Quarter (April 1 - June 30)	\$24.37	\$19.27
1st Quarter (January 1 - March 31)	\$22.32	\$19.06
	2012	
	2012 High	Low
4th Quarter (October 1 - December 31)		Low \$17.00
4th Quarter (October 1 - December 31) 3rd Quarter (July 1 - September 30)	High	
	High \$20.49	\$17.00
3rd Quarter (July 1 - September 30)	High \$20.49 \$20.85	\$17.00 \$14.10

On November 30, 2012, we announced that our board of directors approved the initiation of a quarterly cash dividend on our common stock. On April 4, 2013, July 3, 2013 and October 3, 2013, we paid a cash dividend of \$0.19 per share to stockholders of record at the close of business on March 25, 2013, June 24, 2013 and September 24, 2013, respectively. On January 3, 2014, we paid a cash dividend of \$0.25 per share to stockholders of record at the close of business on December 20, 2013. On February 18, 2014, the board of directors announced a cash dividend of \$0.25 per share payable on April 3, 2014, to stockholders of record at the close of business on March 26, 2014, representing an annualized dividend of \$1.00 per share. Future dividend decisions will be based on and affected by a variety of factors, including our financial condition and results of operations, contractual restrictions, including restrictive covenants contained in our Credit Agreement and AFC's securitization facilities, capital requirements and other factors that our board of directors deems relevant. No assurance can be given as to whether any future dividends may be declared by our board of directors or the amount thereof.

29

Stock Price Performance Graph

The graph below shows the cumulative total stockholder return, assuming the investment of \$100, for the period beginning on December 11, 2009, the first trading day of KAR Auction Services' common stock, and ending on December 31, 2013, on each of KAR Auction Services' common stock, the Standard & Poor's 400 Midcap Index and the Standard and Poor's Smallcap 600 Index. For the year ended December 31, 2013, we have also added the S&P 500 Index to the comparison of cumulative total return. For the years following 2013, we will no longer use the Standard and Poor's Smallcap 600 Index as a comparable index because our market capitalization is no longer comparable to the average market capitalizations included in the index. We believe the S&P 500 provides a better comparison in terms of comparable market capitalization. In addition, the vesting of certain of our stock-based compensation awards are tied to the Company's total shareholder return relative to that of companies within the S&P 500 Index. Our stock price performance shown in the following graph is not indicative of future stock price performance.

Company/Index	Base Period 12/11/2009	12/31/2009	12/31/2010	12/31/2011	12/31/2012	12/31/2013
KAR Auction Services, Inc.	\$100	\$114.63	\$114.71	\$112.22	\$168.25	\$245.64
S&P 400 Midcap Index	\$100	\$102.94	\$128.52	\$124.54	\$144.55	\$190.18
S&P Smallcap 600 Index	\$100	\$104.86	\$131.05	\$130.85	\$150.23	\$209.80
S&P 500 Index	\$100	\$100.79	\$113.67	\$113.66	\$128.90	\$167.06

Item 6. Selected Financial Data

The following selected financial data should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations," the audited consolidated financial statements and related notes thereto of KAR Auction Services, Inc. and other financial information included elsewhere in this Annual Report on Form 10-K.

Selected Financial Data of KAR Auction Services

For the Years Ended December 31, 2013, 2012, 2011, 2010 and 2009

The following consolidated financial data for the years ended December 31, 2013, 2012, 2011, 2010 and 2009 is based on our audited financial statements.

based on our addred infancial statements.					
	Year Ended December 31,				
(Dollars in millions except per share amounts)	2013	2012	2011	2010	2009
Operations:					
Operating revenues					
ADESA	\$1,118.6	\$1,053.5	\$1,017.4	\$1,075.9	\$1,088.5
IAA	830.0	716.1	700.1	610.4	553.1
AFC	224.7	193.8	168.8	136.3	93.9
Total operating revenues	\$2,173.3	\$1,963.4	\$1,886.3	\$1,822.6	\$1,735.5
Operating expenses (exclusive of depreciation and amortization)	1,722.2	1,506.2	1,424.6	1,382.5	1,367.8
Operating profit	256.7	267.0	281.9	268.8	195.3
Interest expense	104.7	119.4	143.1	141.4	172.6
Income from continuing operations	67.7	92.0	72.2	69.6	23.2
Net income	67.7	92.0	72.2	69.6	23.2
Net income per share					
Basic	0.49	0.67	0.53	0.52	0.21
Diluted	0.48	0.66	0.52	0.51	0.21
Weighted average shares outstanding					
Basic	137.9	136.5	136.0	134.9	108.0
Diluted	140.8	139.0	137.8	135.9	108.1
Cash dividends declared per common share	0.82	0.19	_	_	
·	As of December 31,				
	2013	2012	2011	2010	2009
Financial Position:					
Working capital (1)	\$356.9	\$294.5	\$177.0	\$287.9	\$299.5
Total assets	5,127.2	4,922.3	4,779.1	4,525.0	4,251.3
Total debt, net of unamortized debt discount	1,767.2	1,818.3	1,902.8	1,875.7	2,272.9
Total stockholders' equity	1,481.8	1,443.7	1,343.2	1,244.6	1,141.5
Total Stockholdels equity	Year Ended December 31,				1,11110
					2009
Other Financial Data:	2013	2012	2011	2010	2007
Net cash provided by operating activities	\$434.0	\$290.2	\$305.8	\$467.6	\$250.8
Capital expenditures	96.6	102.0	85.8	78.9	65.6
Depreciation and amortization	194.4	190.2	179.8	171.3	172.4
Depreciation and amortization	1/7.7	170.2	1/7.0	1/1.5	1/4,7

⁽¹⁾ Working capital is defined as current assets less current liabilities.

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with the "Selected Financial Data" and the consolidated financial statements and notes thereto included elsewhere in this Annual Report on Form 10-K.

Forward-Looking Statements

This Annual Report on Form 10-K contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 and which are subject to certain risks, trends and uncertainties. In particular, statements made in this report on Form 10-K that are not historical facts (including, but not limited to, expectations, estimates, assumptions and projections regarding the industry, business, future operating results, potential acquisitions and anticipated cash requirements) may be forward-looking statements. Words such as "should," "may," "will," "anticipates," "expects," "intends," "plans," "believes," "seeks," "estimates," and similar expressions identify forward-looking statements. Such statements, including statements regarding our future growth; anticipated cost savings, revenue increases and capital expenditures; dividend declarations and payments; strategic initiatives, greenfields and acquisitions; our competitive position and retention of customers; and our continued investment in information technology, are not guarantees of future performance and are subject to risks and uncertainties that could cause actual results to differ materially from the results projected, expressed or implied by these forward-looking statements. Factors that could cause or contribute to such differences include, but are not limited to, those discussed in Item 1A "Risk Factors" of this Annual Report on Form 10-K. Some of these factors include:

fluctuations in consumer demand for and in the supply of used, leased and salvage vehicles and the resulting impact on auction sales volumes, conversion rates and loan transaction volumes;

trends in new and used vehicle sales and incentives, including wholesale used vehicle pricing;

the ability of consumers to lease or finance the purchase of new and/or used vehicles;

•he ability to recover or collect from delinquent or bankrupt customers;

economic conditions including fuel prices, foreign exchange rates and interest rate fluctuations;

trends in the vehicle remarketing industry;

trends in the number of commercial vehicles being brought to auction, in-particular off-lease volumes;

changes in the volume of vehicle production, including capacity reductions at the major original equipment manufacturers;

increases in the number of used vehicles purchased on virtual auction platforms;

significant current competition and the introduction of new competitors;

laws, regulations and industry standards, including changes in regulations governing the sale of used vehicles, the processing of salvage vehicles and commercial lending activities;

•changes in the market value of vehicles auctioned, including changes in the actual cash value of salvage vehicles; •competitive pricing pressures;

costs associated with the acquisition of businesses or technologies;

ditigation developments;

our ability to successfully implement our business strategies or realize expected cost savings and revenue enhancements;

our ability to maintain our brand and protect our intellectual property;

our ability to develop and implement information systems responsive to customer needs:

business development activities, including acquisitions and integration of acquired businesses;

the costs of environmental compliance and/or the imposition of liabilities under environmental laws and regulations; weather, including increased expenses as a result of catastrophic events; general business conditions;

our substantial amount of debt:

restrictive covenants in our debt agreements;

our assumption of the settlement risk for vehicles sold;

any impairment to our goodwill or other intangible assets;

our self-insurance for certain risks;

any losses of key personnel;

interruptions to service from our

workforce;

changes in effective tax rates;

changes to accounting standards; and

other risks described from time to time in our filings with the SEC, including the Quarterly Reports on Form 10-Q to be filed by us in 2014.

Many of these risk factors are outside of our control, and as such, they involve risks which are not currently known that could cause actual results to differ materially from those discussed or implied herein. The forward-looking statements in this document are made as of the date on which they are made and we do not undertake to update our forward-looking statements.

Our future growth depends on a variety of factors, including our ability to increase vehicle sold volumes and loan transaction volumes, expand our product and service offerings, including information systems development, acquire and integrate additional business entities, manage expansion, control costs in our operations, introduce fee increases, and retain our executive officers and key employees. We cannot predict whether our growth strategy will be successful. In addition, we cannot predict what portion of overall sales will be conducted through online auctions or other remarketing methods in the future and what impact this may have on our auction business.

Overview

We provide whole car and salvage auction services in North America. Our business is divided into three reportable business segments, each of which is an integral part of the vehicle remarketing industry: ADESA Auctions, IAA and AFC.

The ADESA Auctions segment serves a domestic and international customer base through live and online auctions and through 65 whole car auction facilities in North America, that are developed and strategically located to draw professional sellers and buyers together and allow the buyers to inspect and compare vehicles remotely or in person. Through ADESA.com, powered by OPENLANE technology, ADESA offers comprehensive private label remarketing solutions to automobile manufacturers, captive finance companies and other institutions to offer vehicles via the Internet prior to arrival at the physical auction. Vehicles at ADESA's auctions are typically sold by commercial fleet operators, financial institutions, rental car companies, new and used vehicle dealers and vehicle manufacturers and their captive finance companies to franchise and independent used vehicle dealers. ADESA also provides value-added ancillary services including inspections, storage, transportation, reconditioning, titling and other administrative services.

The IAA segment serves a domestic and international customer base through live and online auctions and through 164 salvage vehicle auction sites in the United States and Canada at December 31, 2013. The salvage auctions facilitate the remarketing of damaged vehicles designated as total losses by insurance companies, charity donation vehicles, recovered stolen (or theft) vehicles and low value used vehicles. The salvage auction business specializes in providing services such as inbound transportation, titling, salvage recovery and claims settlement administrative services. The AFC segment provides short-term, inventory-secured financing, known as floorplan financing, primarily to independent used vehicle dealers. At December 31, 2013, AFC conducted business at 105 locations in the United States and Canada.

The holding company is maintained separately from the three reportable segments and includes expenses associated with the corporate office, such as salaries, benefits, and travel costs for our management team, certain human resources, information technology and accounting costs, and certain insurance, treasury, legal and risk management costs. Holding company interest expense includes the interest expense incurred on the corporate debt structure. Intercompany charges relate primarily to interest on intercompany debt or receivables and certain administrative costs

allocated by the holding company.

Industry Trends

Whole Car

Used vehicles sold in North America through whole car auctions, including online only sales, were 8.0 million, 8.2 million and an estimated 8.7 million in 2011, 2012 and 2013, respectively. The reduction in auction volumes since 2009, when industry volumes exceeded 9 million used vehicles sold, is attributable to supply shortages in the North American whole car auction industry caused principally by declines in new vehicle sales and lease originations from 2007 to 2009 and declines in repossessions from 2009 to 2012. The supply shortages and resulting decline in industry auction volumes reflected a reduction in units sold by institutional consignors, which was partially offset by dealer consignment units sold.

Other reasons for the fluctuations in industry volumes, which may also impact future volumes include: New car sales in the U.S. declined from 17.0 million in 2005 to 10.4 million in 2009. Although new vehicle sales have elimbed over the last four years to approximately 15.6 million units in 2013, the recovery in new vehicles sales has been gradual and may continue to be so for the foreseeable future.

The previous decline in total new vehicle sales, coupled with a tightening of consumer credit and changing policies regarding delinquent loans by the major lenders, resulted in a decline in repossessed vehicles coming to auction.

When lease residuals are below wholesale vehicle values, more vehicles are purchased by either the consumer

• or the "grounding dealer" (the dealer that the lessee returned the vehicle to) or by dealers in online auctions, prior to the vehicles being transported to a physical auction.

As used vehicle prices have increased over the last few years, new vehicle dealers have shifted to selling more used vehicles, which can offset lower new vehicle sales. Dealers can also offer financing, warranties and insurance services on their used vehicle sales. Despite changes in the number of trade-ins dealers keep for retail sale as compared to periods prior to 2010, the number of dealer consignment vehicles remarketed through whole car auctions has increased since 2010.

There was a reduction in the number of lease originations in 2008 and 2009, as new vehicle sales fell and lease financing was scaled back. This decline negatively impacted the supply of off-lease vehicles available at auction through 2012.

We estimate that used vehicle auction volumes in North America, including online only volumes, will be approximately 9.1 million units in 2014, approximately 9.5 million units in 2015 and approximately 9.7 million units in 2016.

Salvage

Vehicles deemed a total loss by automobile insurance companies represent the largest category of vehicles sold in the salvage vehicle auction industry. The percentage of claims resulting in total losses was approximately 14% in both 2013 and 2012, up from approximately 13% in 2011. There is no central reporting system that tracks the number of total loss vehicles in any given year which makes estimating industry volumes very difficult.

Fluctuations in used vehicle and commodity pricing have an impact on proceeds received in the salvage vehicle auction industry. In times of rising prices, as the industry has experienced over the last few years, revenue and gross profit are positively impacted. If used vehicle and commodity prices continue to moderate in 2014, proceeds, revenue and gross profit at salvage auctions may be negatively impacted, which could adversely affect the level of profitability.

Automotive Finance

AFC works with independent used vehicle dealers to improve their results by providing a comprehensive set of business and financial solutions that leverages its local branches, industry experience and scale, as well as KAR affiliations. Over the last few years, the U.S. independent used vehicle dealer base has rebounded from approximately 36,000 dealers in 2009 to about 37,000 dealers in 2013. During this time, AFC's dealer base has grown from over 9,700 dealers in 2009 to almost 12,000 dealers in 2013 and loan transactions, which includes both loans paid off and loans curtailed, have grown from approximately 800,000 in 2009 to approximately 1,355,000 in 2013.

Key challenges for the independent used vehicle dealer include used vehicle sales volume demand, disruptions in pricing of used vehicle inventory and lack of access to consumer financing. These same challenges, to the extent they occur, could result in a material negative impact on AFC's results of operations. A significant decline in used vehicle

sales would result in a decrease in consumer auto loan originations and an increased number of dealers defaulting on their loans. In addition, volatility in wholesale vehicle pricing impacts the value of recovered collateral on defaulted loans and the resulting severity of credit losses at AFC.

AFC implemented a number of strategic initiatives in recent years to enhance credit standards, improve portfolio risk management and enhance the customer experience. Additionally, in June 2013, AFC acquired Preferred Warranties, Inc., a service contract business, as part of its strategy to provide new services to independent used vehicle dealers. These initiatives, along with the current industry environment, have enabled AFC to increase its penetration of the independent dealer base while maintaining a high level of portfolio quality evidenced by low levels of net credit losses and a managed portfolio which was 99 percent current at the end of 2013. Seasonality

The volume of vehicles sold at our auctions generally fluctuates from quarter to quarter. This seasonality is caused by several factors including weather, the timing of used vehicles available for sale from selling customers, the availability and quality of salvage vehicles, holidays, and the seasonality of the retail market for used vehicles, which affects the demand side of the auction industry. Used vehicle auction volumes tend to decline during prolonged periods of winter weather conditions. In addition, mild weather conditions and decreases in traffic volume can each lead to a decline in the available supply of salvage vehicles because fewer traffic accidents occur, resulting in fewer damaged vehicles overall. As a result, revenues and operating expenses related to volume will fluctuate accordingly on a quarterly basis. The fourth calendar quarter typically experiences lower used vehicle auction volume as well as additional costs associated with the holidays and winter weather.

Sources of Revenues and Expenses

Our revenue is derived from auction fees and related services at our whole car and salvage auctions, and from dealer financing fees, interest income and other service revenue at AFC. Although auction revenues primarily include the auction services and related fees, our related receivables and payables include the gross value of the vehicles sold. Our operating expenses consist of cost of services, selling, general and administrative and depreciation and amortization. Cost of services is composed of payroll and related costs, subcontract services, supplies, insurance, property taxes, utilities, service contract claims, maintenance and lease expense related to the auction sites and loan offices. Cost of services excludes depreciation and amortization. Selling, general and administrative expenses are composed of payroll and related costs, sales and marketing, information technology services and professional fees.

Gross profit*

Results of Operations

Overview of Results of KAR Auction Services, Inc. for the Years Ended December 31, 2013 and 2012:

941.1

Year Ended

	December 31,			
(Dollars in millions except per share amounts)	2013	2012		
Revenues				
ADESA	\$1,118.6	\$	1,053.5	
IAA	830.0	716.1		
AFC	224.7	193.8		
Total revenues	2,173.3	1,963.4		
Cost of services*	1,232.2	1,087.1		

CRP will have performed or complied in all material respects with all agreements and covenants required by the merger agreement to be performed or complied with by it on or prior to the effective time of the merger;

CRP will have delivered to HCP a certificate executed by its Chief **Executive Officer or Chief Financial** Officer certifying as to the satisfaction of the preceding two conditions;

there will not have occurred, since May 1, 2006, any event, circumstance, change or effect that, individually or in the aggregate, has had, or is reasonably likely to have, a "material adverse effect" with respect to CRP;

HCP will have received an opinion of Greenberg Traurig, LLP, outside counsel to CRP, dated the closing date, regarding CRP's tax status as a REIT; and

the acquisition of the Advisor will have simultaneously closed.

CRP's obligation to effect the merger is subject to the satisfaction or waiver, where permissible, of the following conditions:

> the representations and warranties of HCP must be true and correct as of May 1, 2006 and as of the effective time of the merger as though made on and as of the effective time (except to the extent those

representations and warranties speak as of an earlier date, in which case as of such earlier date), except for such failures to be true and correct that individually or in the aggregate would not reasonably be likely to have a material adverse effect with respect to HCP, provided that for purposes of determining whether this condition is satisfied, references to "material adverse effect" and any other materiality qualification will be ignored;

HCP will have performed or complied in all material respects with all agreements and covenants required by the merger agreement to be performed or complied with by it on or prior to the effective time;

HCP will have delivered to CRP a certificate executed by its Chief Executive Officer or Chief Financial Officer certifying as to the satisfaction of the preceding two conditions:

there will not have occurred, since the date of the merger agreement, any event, circumstance, change or effect that, individually or in the aggregate, has had, or is reasonably likely to have, a "material adverse effect" with respect to HCP; and

CRP will have received the opinion of Latham & Watkins LLP, dated the closing date, regarding HCP's tax status as a REIT.

Neither party may rely on the failure of any of the foregoing conditions to be satisfied if such party's failure to comply with any provision of the merger agreement in a material respect has been the proximate cause of, or resulted in, the failure of the condition.

Amendment, Waiver and Extension of the Merger Agreement

At any time prior to the effective time of the merger, each of HCP and CRP may:

mutually amend the merger agreement in writing by action taken by or on behalf of its respective board of directors;

waive any inaccuracy in the representations and warranties of any other party contained in the merger agreement; and

in writing, waive compliance with any agreement of any other party or any condition to its obligations contained in the merger agreement and extend the time for the performance of any obligation or other act of any other party to the merger agreement.

However, after approval of the merger by the stockholders of CRP, no amendment to the merger agreement that would require further approval of CRP's stockholders may be made without first obtaining such approval.

Termination of the Merger Agreement

The parties may terminate the merger agreement at any time prior to the closing date only as follows:

by mutual written consent of HCP and CRP;

by either of CRP or HCP by written notice to the other:

if CRP's stockholders do not approve the merger at the special meeting (or at any adjournment or postponement thereof);

if any governmental authority issues an order that restrains, enjoins or otherwise prohibits or makes illegal the consummation of the merger and such order becomes final and non-appealable (provided the terminating party has used its reasonable

best efforts to have the order vacated); and

if the consummation of the merger does not occur on or before October 31, 2006 (provided the terminating party has not breached any of its representations, warranties or covenants in a material respect that was the proximate cause of, or resulted in, the failure of the merger to occur on or before such date).

HCP may unilaterally terminate the merger agreement at any time prior to the closing date as follows:

if CRP breaches or fails to perform in any material respect any of its representations, warranties or covenants contained in the merger agreement, which breach would entitle HCP not to close the merger based on the failure of the corresponding closing condition, for its benefit, and such condition is incapable of being satisfied by October 31, 2006 or such breach has not been cured by CRP within 30 business days after CRP's receipt of written notice of such breach from HCP;

if, within 15 business days after the date on which HCP's right to terminate first accrued, prior to the approval of the merger by CRP's stockholders, the board of directors of CRP and the special committee (i) publicly withdraw or knowingly modify, in a manner adverse to HCP, their recommendation of the merger, (ii) approve or recommend an alternative acquisition proposal, (iii) at any time after the end of 15 business days following

receipt of an acquisition proposal, fail to reaffirm such recommendation within five business days after receipt of any request to do so from HCP (provided the five day period may be extended for an additional five business days if prior thereto CRP certifies to HCP that the board of directors and special committee are in good faith seeking to obtain additional information regarding the reaffirmation decision) or (iv) recommend that CRP's stockholders tender their shares in a publicly announced offer for more than 20% of CRP's outstanding shares (not commenced by HCP or an affiliate of HCP); and

if the acquisition of the Advisor is terminated in accordance with its terms.

CRP may unilaterally terminate the merger agreement at any time prior to the closing date as follows:

if HCP breaches or fails to perform in any material respect any of its representations, warranties or covenants contained in the merger agreement, which breach would entitle CRP not to close the merger based on the failure of the corresponding closing condition for its benefit, and such condition is incapable of being satisfied by October 31, 2006 or such breach has not been cured by HCP within 30 business days after HCP's receipt of written notice of such breach from CRP; and

if CRP enters into a binding written agreement to effect a "superior proposal," provided that CRP first complies with its obligations under the merger agreement, including allowing HCP to submit a revised proposal at least as favorable from a financial point of view to CRP's stockholders as the superior proposal and paying HCP the termination fee described below.

If the merger agreement is validly terminated, it will become void and have no effect, without any liability or obligation on the part of any party unless the party has committed fraud or is in willful breach of any representations, warranties, covenants or other agreements contained in the merger agreement. However, the provisions of the merger agreement relating to confidentiality and the payment of termination fees and expenses will continue in effect notwithstanding termination of the merger agreement.

Fee and Expenses

Termination Fee

CRP has agreed to pay HCP an amount in cash equal to \$107 million if:

CRP terminates the merger agreement to enter into a

definitive agreement to effect a "superior proposal" in accordance with the terms of the merger agreement;

HCP terminates the merger agreement because CRP's board of directors and the special committee (i) publicly withdraw or knowingly modify, in a manner adverse to HCP, their recommendation of the merger, (ii) approve or recommend an alternative acquisition proposal, (iii) fail to reaffirm such recommendation within five business days after receipt of any request to do so from HCP or (iv) recommend that CRP's stockholders tender their shares in a publicly announced offer for more than 20% of CRP's outstanding shares; or

CRP or HCP terminates the merger agreement due to the failure of the CRP stockholders to approve the merger at the special meeting after an acquisition proposal has been publicly announced and remains outstanding, and within 12 months after such termination CRP enters into a definitive agreement with respect to or consummates an alternative "acquisition proposal" for more than 50% of the consolidated assets or equity securities of CRP.

Expenses

The merger agreement provides that each of HCP and CRP will pay its own costs and expenses incurred in connection with the merger agreement and the transactions contemplated by the merger agreement. If, however, the merger agreement is properly terminated by either party due to the material breach by the other party of its representations, warranties or covenants then the breaching party will pay the reasonable out-of-pocket costs and expenses incurred by the non-breaching party up to \$3 million.

MATERIAL UNITED STATES FEDERAL INCOME TAX CONSIDERATIONS

Introduction

The following is a general discussion of the federal income tax consequences of the merger reasonably anticipated to be material to "U.S. stockholders" and "non-U.S. stockholders" (both as defined below), the consequences of owning and disposing of HCP common stock, and the material federal income tax considerations related to HCP's REIT election. This summary is based on current law, is for general information only and is not tax advice. This summary is based on the Internal Revenue Code of 1986, as amended (the "Code"), applicable Treasury Regulations, and administrative and judicial interpretations thereof, each as in effect as of the date hereof, all of which are subject to change or different interpretations, possibly with retroactive effect. This summary assumes that shares of CRP common stock and HCP common stock are held as capital assets. HCP has not requested, and does not plan to request, any rulings from the IRS concerning HCP's tax treatment or the tax treatment of the merger, and the statements in this proxy statement/prospectus are not binding on the IRS or any court. HCP can provide no assurance that the tax consequences contained in this discussion will not be challenged by the IRS, or if challenged, will be sustained by a court. This summary does not address all of the tax consequences that may be relevant to particular holders of CRP common stock or HCP common stock in light of their personal circumstances, except to the extent discussed below under the headings "Tax Consequences of the Merger Taxation of U.S. Stockholders", "Tax Consequences of the Merger Taxation of Non-U.S. Stockholders," "Taxation of Holders of HCP's Common Stock Taxation of Tax-Exempt Stockholders" and " Taxation of Non-U.S. Stockholders," or to other types of holders, including, without limitation:

banks, insurance companies or other financial institutions;	
broker-dealers;	
traders;	
expatriates;	

tax-exempt organizations;

persons who are subject to alternative minimum tax;

persons who hold their shares of CRP or HCP common stock as a position in a "straddle" or as part of a "hedging", "conversion" or other risk reduction transaction:

persons deemed to sell their shares of CRP common stock or HCP common stock under the constructive sale provisions of the Code;

United States persons that have a functional currency other than the United States dollar;

except to the extent specifically discussed below, non-U.S. stockholders (as defined below);

partnerships or other entities treated as partnerships for United States federal income tax purposes and partners in such partnerships; or

persons who acquired their shares of CRP common stock or HCP common stock upon the exercise of stock options or otherwise as compensation.

In addition, this discussion does not address any state, local or foreign tax consequences of the merger, ownership of HCP common stock or HCP's election to be taxed as a REIT.

For purposes of this discussion, a "U.S. stockholder" means a holder of CRP common stock or HCP common stock, as applicable, that is:

a citizen or resident of the United States;

a corporation, a partnership or an entity treated as a corporation or a partnership for United States federal income tax purposes created or organized in or under the laws of the United States or any State or the District of Columbia;

an estate, the income of which is subject to United States federal income taxation regardless of its source; or

a trust (a) the administration over which a United States court can exercise primary supervision and (b) all of the substantial decisions of which one or more United States persons have the authority to control, and certain other trusts considered U.S. stockholders for federal income tax purposes.

A "non-U.S. stockholder" is a holder of CRP common stock or HCP common stock, as applicable, other than a U.S. stockholder.

You are urged to consult your tax advisor regarding the specific tax consequences to you of:

the merger;

the acquisition, ownership and sale or other disposition of HCP common stock, including the federal, state, local, foreign and other tax consequences;

CRP's and HCP's elections to be taxed as a REIT for federal income tax purposes; and

potential changes in applicable tax laws.

ALL HOLDERS OF CRP COMMON STOCK AND HCP COMMON STOCK RECEIVED IN THE MERGER ARE URGED TO CONSULT THEIR OWN TAX ADVISORS

REGARDING THE SPECIFIC TAX CONSEQUENCES TO THEM OF THE MERGER AND OF HOLDING AND DISPOSING OF HCP COMMON STOCK, INCLUDING THE APPLICABILITY AND EFFECT OF FEDERAL, STATE AND LOCAL OR FOREIGN INCOME AND OTHER TAX LAWS TO THEIR PARTICULAR CIRCUMSTANCES.

Tax Consequences of the Merger

Tax Treatment of the Merger

The merger of CRP with and into Merger Sub will be treated for United States federal income tax purposes as a taxable sale by CRP of all of its assets to Merger Sub in return for the merger consideration and the assumption of CRP's liabilities, followed by a taxable liquidation of CRP under Section 331 of the Code. CRP will be deemed to have distributed the merger consideration to the stockholders of CRP in connection with the deemed liquidation. The liquidating distribution will be treated as a dividend for purposes of the dividend paid deduction to the extent of CRP's earnings and profits and thus will be sufficient to satisfy CRP's requirement to distribute 90% of its REIT taxable income for the taxable year of the merger. Merger Sub will have a basis in the assets acquired from CRP equal to the aggregate of the fair market value of the merger consideration and the amount of any liabilities that are assumed by Merger Sub in the merger.

Taxation of U.S Stockholders

U.S. stockholders who exchange CRP common shares for the merger consideration will be treated for United States federal income tax purposes as having exchanged their CRP common shares for the HCP common shares and cash in a taxable transaction. As a result, each such stockholder will recognize capital gain or loss in the merger equal to the difference between (A) the aggregate of the fair market value of the HCP common shares and cash that such stockholder receives in the exchange, and (B) the tax basis in the CRP shares that are surrendered by such stockholder in the exchange. Capital gain of a noncorporate U.S. stockholder arising from the sale of CRP common shares will

generally be taxed at a maximum rate of 15% where the stockholder has a holding period greater than one year.

A U.S. stockholder who has held his or her CRP common shares for six months or less at the effective time of the merger, taking into account the holding period rules of Sections 246(c)(3) and (4) of the Code, and who recognizes a capital loss with respect to those shares will be treated as recognizing a long-term, rather than short-term, capital loss to the extent of any capital gain dividends previously received by such U.S. stockholder from CRP with respect to those shares. The deductibility of capital losses, in general, is subject to limitations. In the case of stockholders that hold multiple blocks of CRP common shares (i.e., the shares were acquired separately at different times and/or different prices), gain or loss must be calculated and accounted for separately for each block of shares.

The tax basis in the HCP common shares that are received by the stockholders of CRP in the merger will be equal, for United States federal income tax purposes, to the fair market value of the HCP common shares on the date of the merger. The holding period with respect to such shares shall commence on the day after the effective date of the merger.

Taxation of Non-U.S. stockholders

In order to determine the tax consequences of the merger to a non-U.S. stockholder, one must first allocate the merger consideration that is received by the stockholder between (i) such stockholder's ratable share of the gain that is recognized by CRP in respect of the merger that is attributable to United States real property interests that were held by CRP immediately prior to the merger plus such stockholder's ratable share of any undistributed gain that was recognized by CRP prior to the merger and attributable to the sale of United States real property interests (the "Capital Gain Consideration") and (ii) the remainder of the consideration that is received by such stockholder in the Merger (the "Residual Consideration").

Dividends that are paid by a REIT to a non-U.S. stockholder are generally subject to a 35% FIRPTA withholding tax to the extent that the dividend is attributable to gain that is recognized by the REIT in respect of a sale of United States real property interests. It is not entirely clear whether such withholding tax should apply to the Capital Gain Consideration that is deemed to be paid to a non-U.S. stockholder in connection with CRP's deemed liquidation upon the closing for the merger. In light of this uncertainty, HCP and CRP intend to withhold 35% of the portion of the

merger consideration that is paid to a non-U.S. stockholder that consists of Capital Gain Consideration. Non-U.S. stockholders should consult their tax advisors regarding the possibility of obtaining a refund of such withholding tax.

A non-U.S. stockholder will generally not be subject to U.S. tax in respect of the portion of the merger consideration that consists of Residual Consideration if CRP qualifies as a "domestically controlled" REIT for tax purposes. CRP will qualify as a domestically controlled REIT if at all times during the five-year period preceding the closing for the merger less than 50% in value of the shares of CRP have been directly or indirectly held by non-U.S. stockholders. CRP believes, based on current information, that CRP qualifies as a domestically controlled REIT and based on that belief HCP does not intend to impose FIRPTA withholding in respect of the portion of the merger consideration that consists of Residual Consideration. If HCP and CRP believe as of the closing of the merger that there is material risk that CRP does not qualify as a domestically controlled REIT, they intend to withhold 10% of the portion of the merger consideration that consists of Residual Consideration. In addition, if CRP does not qualify as a domestically controlled REIT as of the closing of the merger, then a non-U.S. stockholder will generally be subject to FIRPTA tax at the same marginal rates that are applicable to U.S. stockholders in respect of the gain that is recognized by the non-U.S. stockholder in respect of its receipt of the merger consideration and any tax withheld by HCP or CRP will be credited against this tax liability.

Notwithstanding the foregoing, capital gain that is not subject to FIRPTA under the rules described above will be taxable to a non-U.S. stockholder if, among other conditions, the non-U.S. stockholder is a nonresident alien individual who is present in the United States for 183 days or more during the taxable year or is a former citizen or long-term resident of the United States subject to special rules that apply to expatriates. In addition, a non-U.S. stockholder will generally be taxed in the manner described above with respect to U.S. stockholders if its ownership of CRP common shares is effectively connected with a U.S. trade or business. The branch profits tax may also apply to a foreign corporation's share of any effectively connected income.

Information Reporting and Backup Withholding

Under United States federal income tax laws, consideration to be received in the merger may be subject to information reporting and a 28% backup withholding tax. Backup withholding generally will not apply to payments made to certain exempt recipients, such as a corporation or financial institution or to a stockholder who certifies such stockholder's taxpayer identification number and certain other required information or provides a certificate of foreign status. Backup withholding is not an additional tax. If backup withholding applies, the amount withheld will be allowed as a refund or a credit against such stockholder's United States federal income tax liability, provided the required information is furnished by the stockholder to the IRS on a timely basis.

Taxation of HCP as a Real Estate Investment Trust

General

HCP elected to be taxed as a REIT under Sections 856 through 860 of the Code, commencing with its taxable year ending December 31, 1985. HCP believes it has been organized and has operated in a manner which allowed it to qualify for taxation as a REIT under the Code, commencing with its taxable year ending December 31, 1985. HCP currently intends to continue to operate in this manner. However, qualification and taxation as a REIT depend upon HCP's ability to meet the various qualification tests imposed under the Code, including through actual annual operating results, asset composition, distribution levels and diversity of stock ownership. Accordingly, no assurance can be given that HCP has operated or will continue to

operate in a manner so as to qualify or remain qualified as a REIT. See " Failure to Qualify."

It is a condition to the closing of the merger that Latham & Watkins LLP render an opinion to CRP, dated as of the closing date, to the effect that, commencing with HCP's taxable year ended December 31, 1985, HCP has been organized in conformity with the requirements for qualification and taxation as a REIT, and its proposed method of operation will enable it to continue to meet the requirements for qualification and taxation as a REIT under the Code. It must be emphasized that this opinion will be based on various assumptions and representations as to factual matters, including representations made by HCP in a factual certificate provided by one of its officers. In addition, this opinion will be based upon HCP's factual representations set forth in this proxy statement/prospectus. Moreover, HCP's qualification and taxation as a REIT depend upon HCP's ability to meet the various qualification tests imposed under the Code which are discussed below, including through actual annual operating results, asset composition, distribution levels and diversity of stock ownership, the results of which have not been and will not be reviewed by Latham & Watkins LLP. Accordingly, no assurance can be given that HCP's actual results of operation for any particular taxable year have satisfied or will satisfy those requirements. See Failure to Qualify." Further, the anticipated income tax treatment described in this proxy statement/prospectus may be changed, perhaps retroactively, by legislative, administrative or judicial action at any time. Latham & Watkins LLP will have no obligation to update its opinion subsequent to its date.

The sections of the Code and the corresponding Treasury Regulations that relate to the qualification and operation of a REIT are highly technical and complex. The following sets forth the material aspects of the sections of the Code that govern the United States federal income tax treatment of a REIT and its stockholders. This summary is qualified in its entirety by the applicable Code provisions, Treasury Regulations, and related administrative and judicial interpretations thereof.

Provided HCP qualifies for taxation as a REIT, HCP generally will not be required to pay United States federal corporate income taxes on its net income that is currently distributed to its stockholders. This treatment substantially eliminates the "double taxation" that ordinarily results from investment in a C corporation. A C corporation generally is required to pay tax at the corporate level. Double taxation means taxation once at the corporate level when income is earned and once again at the stockholder level when the income is distributed. HCP will be required to pay federal income tax, however, as follows:

HCP will be required to pay tax at regular corporate tax rates on any undistributed REIT taxable income, including undistributed net capital gains.

HCP may be required to pay the "alternative minimum tax" on its items of tax preference under certain circumstances.

If HCP has: (a) net income from the sale or other disposition of "foreclosure property" which is held primarily for sale to customers in the ordinary course of business; or (b) other nonqualifying income from foreclosure property, HCP will be required to pay tax at the highest corporate rate on this income. Foreclosure property generally is defined as property HCP acquired through foreclosure or after a default on a loan secured by the property or a lease of the property.

HCP will be required to pay a 100% tax on any net income from prohibited transactions. Prohibited transactions are, in general, sales or other taxable dispositions of property, other than foreclosure property, held as inventory or primarily for sale to customers in the ordinary course of business.

If HCP fails to satisfy the 75% gross income test or the 95% gross income test discussed below, but has otherwise maintained its qualification as a REIT because certain other requirements are met, HCP will be required to pay a tax equal to (a) the greater of (i) the amount by which 75% of its gross income exceeds the amount qualifying under the 75% gross income test and (ii) the amount by which 95% of its gross income (90% for its taxable years ended on or prior to December 31, 2004) exceeds the amount qualifying under the 95% gross income test, multiplied by (b) a fraction intended to reflect its profitability.

If HCP fails to satisfy any of the REIT asset tests (other than a de minimis failure of the 5% and 10% asset tests), as described below, due to reasonable cause and not due to willful neglect, and HCP nonetheless maintains its REIT qualification because of specified cure provisions, HCP will be required to pay a tax equal to the greater of \$50,000 or the highest corporate tax rate multiplied by the net income generated by the nonqualifying assets that caused it to fail such test.

If HCP fails to satisfy any provision of the Code that would result in its failure to qualify as a REIT (other than a violation of the REIT gross

income tests or certain violations of the asset tests described below) and the violation is due to reasonable cause and not due to willful neglect, HCP may retain its REIT qualification but will be required to pay a penalty of \$50,000 for each such failure.

HCP will be required to pay a 4% excise tax to the extent HCP fails to distribute during each calendar year at least the sum of (a) 85% of its REIT ordinary income for the year, (b) 95% of

its REIT capital gain net income for the year, and (c) any undistributed taxable income from prior periods.

If HCP acquires any asset from a corporation which is or has been a C corporation in a transaction in which the basis of the asset in its hands is determined by reference to the basis of the asset in the hands of the C corporation, and HCP subsequently recognizes gain on the disposition of the asset during the ten-year period beginning on the date on which HCP acquired the asset, then HCP will be required to pay tax at the highest regular corporate tax rate on this gain to the extent of the excess of (a) the fair market value of the asset over (b) its adjusted basis in the asset, in each case determined as of the date on which HCP acquired the asset. The results described in this paragraph with respect to the recognition of gain assume that certain elections specified in applicable Treasury Regulations are either made or forgone, by HCP or by the entity from which the assets are acquired, in each case, depending upon the date such acquisition occurred.

HCP will be required to pay a 100% tax on any "redetermined rents," "redetermined deductions" or "excess interest." In general, redetermined rents are rents from real property that are overstated as a result of services furnished to any of HCP's tenants by a "taxable REIT subsidiary" of HCP. Redetermined deductions and excess interest generally represent amounts that are deducted by a taxable REIT subsidiary of HCP for amounts paid to HCP that are in excess of the amounts that would have been deducted based on arm's-length negotiations.

HCP's taxable REIT subsidiaries are subject to federal and state taxation. Requirements for Qualification as a

REIT

The Code defines a REIT as a corporation, trust or association:

- (1) that is managed by one or more trustees or directors;
- (2) that issues transferable shares or transferable certificates to evidence its beneficial ownership;
- (3) that would be taxable as a domestic corporation but for Sections 856 through 860 of the Code;
- (4) that is not a financial institution or an insurance company within the meaning of the Code;
- (5) that is beneficially owned by 100 or more persons;
- (6) not more than 50% in value of the outstanding stock of which is owned, actually or constructively, by five or fewer individuals, including certain specified entities, during the last half of each taxable year; and
- (7) that meets other tests, described below, regarding the nature of its income and assets and the amount of its distributions.

The Code provides that conditions (1) to (4), inclusive, must be met during the entire taxable year and that condition (5) must be met during at least 335 days of a taxable year of twelve months, or during a proportionate part of a taxable year of less than twelve months. Conditions (5) and (6) do not apply until after the first taxable year for which an election is made to be taxed as a REIT. For

purposes of condition (6), pension funds and other specified tax-exempt entities generally are treated as individuals, except that a "look-through" exception applies with respect to pension funds.

HCP believes that it has been organized, has operated and has issued sufficient shares of capital stock with sufficient diversity of ownership to allow it to satisfy conditions (1) through (7) inclusive, during the relevant time periods. In addition, HCP's charter documents provide for restrictions regarding ownership and transfer of its shares which are intended to assist it in continuing to satisfy the ownership requirements described in conditions (5) and (6) above. The stock ownership and transfer

restrictions relating to HCP's common stock are described in "Description of HCP Capital Stock Transfer and Ownership Restrictions Relating to HCP Common Stock" and " Business Combination Provisions" in this proxy statement/prospectus. Similar restrictions are provided with respect to HCP's preferred stock. These restrictions, however, may not ensure that HCP will, in all cases, be able to satisfy the share ownership requirements described in conditions (5) and (6) above. If HCP fails to satisfy these share ownership requirements, except as provided in the next two sentences, its status as a REIT will terminate. See " Failure to Qualify." If, however, HCP complies with the rules contained in applicable Treasury Regulations that require it to ascertain the actual ownership of its shares and HCP does not know, or would not have known through the exercise of reasonable diligence, that HCP failed to meet the requirement described in condition (6) above, HCP will be treated as having met this requirement.

In addition, HCP may not maintain its status as a REIT unless its taxable year is the calendar year. HCP has had and will continue to have a calendar taxable year.

Ownership of Interests in Partnerships and Limited Liability Companies

HCP owns and operates one or more properties through partnerships and limited liability companies. Treasury Regulations provide that if HCP is a partner in a partnership, HCP will be deemed to own its proportionate share of the assets of the partnership based on its interest in partnership capital, subject to special rules relating to the 10% REIT asset test described below. Also, HCP will be deemed to be entitled to its proportionate share of the income of the partnership. The assets and gross income of the partnership retains the same character in HCP's hands for purposes of Section 856 of the Code, including satisfying the gross income tests and the asset tests. In addition, for these purposes, the assets and items of income of any partnership in which HCP owns a direct or indirect interest include such partnership's share of assets and items of income of any partnership in which it owns an interest. A brief summary of the rules governing the federal income taxation of partnerships and their partners is included below in " Tax Aspects of the Partnerships." The treatment described above also applies with respect to the ownership of interests in limited liability companies or other entities that are treated as partnerships for tax purposes.

HCP has direct or indirect control of certain partnerships and limited liability companies and intends to continue to operate them in a manner consistent with the requirements for HCP's qualification as a REIT. HCP is a limited partner or non-managing member in certain partnerships and limited liability companies. If such a partnership or limited liability company were to take actions that could jeopardize HCP's status as a REIT or require it to pay tax (such as tax on prohibited transaction income), HCP may be forced to dispose of its interest in such entity. In addition, it is possible that a partnership or limited liability company could take an action which could cause it to fail a REIT income or asset test, and that HCP would not become aware of such action in a time frame which would allow it to dispose of its interest in the applicable entity or take other corrective action on a timely basis. In that case, unless HCP were entitled to relief, as described below, HCP would fail to qualify as a REIT.

Ownership of Interests in Qualified REIT Subsidiaries

HCP owns and operates a number of properties through its wholly-owned subsidiaries that HCP believes will be treated as "qualified REIT subsidiaries" under the Code. A corporation will qualify as HCP's qualified REIT subsidiary if HCP owns 100% of its outstanding stock and if HCP does not elect with the subsidiary to treat it as a "taxable REIT subsidiary," as described below. A corporation that is a qualified REIT subsidiary is not treated as a separate corporation for federal income tax purposes, and all assets, liabilities and items of income, deduction and credit of a qualified REIT subsidiary are treated as assets, liabilities and items of income, deduction and credit (as the case may be) of the parent REIT for all purposes under the Code (including all REIT qualification tests). Thus, in applying

the federal tax requirements described in this proxy statement/prospectus, the subsidiaries in which HCP owns a 100% interest (other than any taxable REIT subsidiaries) are ignored, and all assets, liabilities and items of income, deduction and credit of such subsidiaries are treated as HCP's assets, liabilities and items of income, deduction and credit. A qualified REIT subsidiary is not required to pay federal income tax, and HCP's ownership of the stock of a qualified REIT subsidiary does not violate the restrictions on ownership of securities of any one issuer which constitute more than 10% of the voting power or value of such issuer's securities or more than 5% of the value of HCP's total assets, as described below in " Asset Tests."

Ownership of Interests in Taxable REIT Subsidiaries

A taxable REIT subsidiary is a corporation other than a REIT in which a REIT directly or indirectly holds stock, and that has made a joint election with the REIT to be treated as a taxable REIT subsidiary. A taxable REIT subsidiary also includes any corporation other than a REIT with respect to which a taxable REIT subsidiary owns securities possessing more than 35% of the total voting power or value of the outstanding securities of such corporation. A taxable REIT subsidiary generally may engage in any business, including the provision of customary or non-customary services to tenants of its parent REIT, except that a taxable REIT subsidiary may not directly or indirectly operate or manage a lodging or healthcare facility or directly or indirectly provide to any other person (under a franchise, license or otherwise) rights to any brand name under which any lodging or healthcare facility is operated. A taxable REIT subsidiary is subject to federal income tax, and state and local income tax where applicable, as a regular C corporation. In addition, a taxable REIT subsidiary may be prevented from deducting interest on debt funded directly or indirectly by its parent REIT if certain tests regarding the taxable REIT subsidiary's debt to equity ratio and interest expense are not satisfied. HCP currently owns interests in several taxable REIT subsidiaries, and may acquire interests in additional taxable REIT subsidiaries in the future. HCP's ownership of securities of its taxable REIT subsidiaries will not be subject to the 5% or 10% asset tests described below but will be subject to the 20% asset test limitation. See " Asset Tests."

Income Tests

HCP must satisfy two gross income requirements annually to maintain its qualification as a REIT:

First, each taxable year, HCP must derive directly or indirectly at least 75% of its gross income, excluding gross income from prohibited transactions, from (a) certain investments relating to real property or mortgages on real property, including "rents from real property" and, in some circumstances, interest, or (b) some types of temporary investments; and

Second, each taxable year, HCP must derive at least 95% of its gross income, excluding gross income from prohibited transactions or certain hedging transactions, from the real property investments described above, dividends, interest and gain from the sale or disposition of stock or securities, or from any combination of the foregoing.

For these purposes, the term "interest" generally does not include any amount received or accrued, directly or indirectly, if the determination of all or some of the amount depends in any way on the income or profits of any person. However, an amount received or accrued generally will not be excluded from the term "interest" solely by reason of being based on a fixed percentage or percentages of receipts or sales.

Rents HCP receives from a tenant will qualify as "rents from real property" for the purpose of satisfying the gross income requirements for a REIT described above only if all of the following conditions are met:

The amount of rent must not be based in any way on the income or profits of any person from the property. However, an amount HCP receives or accrues generally will not be excluded from the term "rents from real property" solely because it is based on a fixed percentage or percentages of receipts or sales;

HCP, or an actual or constructive owner of 10% or more of its capital stock, must not actually or constructively own 10% or more of the interests in the assets or net profits of the tenant, or, if the tenant is a corporation, 10% or more of the voting power or value of all classes of stock of the tenant. Rents HCP receives from such a tenant that also is its taxable REIT subsidiary, however, will not be excluded from the definition of "rents from real property" as a result of this condition if at least 90% of the space at the property to which the rents relate is leased to third parties, and the rents paid by the taxable REIT subsidiary are substantially comparable to rents paid by its other tenants for comparable space. Whether rents paid by a taxable REIT subsidiary are substantially comparable to rents paid by other tenants is determined at the time the lease with the taxable REIT subsidiary is entered into, extended, and modified, if such modification increases the rents due under such lease. Notwithstanding the foregoing, however, if a lease with a "controlled taxable REIT subsidiary" is modified and such modification results in an increase in the rents payable by such taxable REIT

subsidiary, any such increase will not qualify as "rents from real property." For purposes of this rule, a "controlled taxable REIT subsidiary" is a taxable REIT subsidiary in which HCP owns stock possessing more than 50% of the voting power or more than 50% of the total value of the outstanding stock;

Rent attributable to personal property, leased in connection with a lease of real property, must not be greater than 15% of the total rent HCP receives under the lease. If this condition is not met, then the portion of rent attributable to the personal property will not qualify as "rents from real property;" and

HCP generally must not operate or manage the property or furnish or render services to its tenants, subject to a 1% de minimis exception and except as provided below. HCP may, however, perform services that are "usually or customarily rendered" in connection with the rental of space for occupancy only and are not otherwise considered "rendered to the occupant" of the property. Examples of such services include the provision of light, heat, or other utilities, trash removal and general maintenance of common areas. In addition, HCP may employ an independent contractor from whom HCP derives no revenue to provide customary services, or a taxable REIT subsidiary, which may be wholly or partially owned by HCP, to provide both customary and non-customary services to its tenants without causing the rent HCP receives from those tenants to fail to qualify as "rents from real property." Any amounts HCP receives from a taxable REIT subsidiary with respect to the taxable REIT subsidiary's provision of non-customary services will, however, be

nonqualifying income under the 75% gross income test and, except to the extent received through the payment of dividends, the 95% gross income test.

HCP generally does not intend to receive rent which fails to satisfy any of the above conditions. Notwithstanding the foregoing, HCP may have taken and may continue to take actions which fail to satisfy one or more of the above conditions to the extent that HCP determines, based on the advice of its tax counsel, that those actions will not jeopardize its tax status as a REIT.

Income HCP receives that is attributable to the rental of parking spaces at its properties will constitute rents from real property for purposes of the REIT gross income tests if any services provided with respect to the parking facilities are performed by independent contractors from whom HCP derives no revenue, either directly or indirectly, or by a taxable REIT subsidiary, and certain other

requirements are met. With the exception of two parking facilities HCP operates, HCP believes that the income HCP receives that is attributable to parking facilities meets these tests and, accordingly, will constitute rents from real property for purposes of the REIT gross income tests.

From time to time, HCP enters into hedging transactions with respect to one or more of its assets or liabilities. HCP's hedging activities may include entering into interest rate swaps, caps, and floors, options to purchase these items, and futures and forward contracts. Any income HCP derives from a hedging transaction will be nonqualifying income for purposes of the 75% gross income test. Except to the extent provided by Treasury Regulations, however, income from a hedging transaction, including gain from the sale or disposition of such a transaction, entered into prior to January 1, 2005 will be qualifying income for purposes of the 95% gross income test, but only to the extent that the transaction hedges indebtedness incurred or to be incurred by it to acquire or carry real estate. Income from such a hedging transaction entered into on or after January 1, 2005 that is clearly identified as such as specified in the Code will not constitute gross income for purposes of the 95% gross income test, and therefore will be exempt from this test. The term "hedging transaction," as used above, generally means any transaction HCP enters into in the normal course of its business primarily to manage risk of interest rate changes or fluctuations with respect to borrowings made or to be made by it. To the extent that HCP hedges with other types of financial instruments, the income from those transactions is not likely to be treated as qualifying income for purposes of the gross income tests. HCP intends to structure any hedging transactions in a manner that does not jeopardize its status as a REIT.

HCP believes that the aggregate amount of its nonqualifying income, from all sources, in any taxable year will not exceed the limit on nonqualifying income under the gross income tests. If HCP fails to satisfy one or both of the 75% or 95% gross income tests for any taxable year, HCP may nevertheless qualify as a REIT for the year if HCP is entitled to relief under certain provisions of the Code. Commencing with its taxable year beginning January 1, 2005, HCP generally may make use of the relief provisions if:

following HCP's identification of the failure to meet the 75% or 95% gross income tests for any taxable year, HCP files a

schedule with the IRS setting forth each item of its gross income for purposes of the 75% or 95% gross income tests for such taxable year in accordance with Treasury Regulations to be issued; and

HCP's failure to meet these tests was due to reasonable cause and not due to willful neglect.

It is not possible, however, to state whether in all circumstances HCP would be entitled to the benefit of these relief provisions. For example, if HCP fails to satisfy the gross income tests because nonqualifying income that HCP intentionally accrues or receives exceeds the limits on nonqualifying income, the IRS could conclude that its failure to satisfy the tests was not due to reasonable cause. If these relief provisions do not apply to a particular set of circumstances, HCP will not qualify as a REIT. As discussed above in " Taxation of HCP as a Real Estate Investment Trust General," even if these relief provisions apply, and HCP retains its status as a REIT, a tax would be imposed with respect to its nonqualifying income. HCP may not always be able to comply with the gross income tests for REIT qualification despite its periodic monitoring of its income.

Prohibited Transaction Income

Any gain that HCP realizes on the sale of property held as inventory or otherwise held primarily for sale to customers in the ordinary course of business will be treated as income from a prohibited transaction that is subject to a 100% penalty tax. HCP's gain would include any gain realized by its qualified REIT subsidiaries and its share of any gain realized by any of the partnerships or limited liability companies in which it owns an interest. This prohibited transaction income may also adversely affect HCP's ability to satisfy the income tests for qualification as a REIT. Under existing law, whether property is held as inventory or primarily for sale to customers in the ordinary course of a trade or

business is a question of fact that depends on all the facts and circumstances surrounding the particular transaction. HCP intends to hold its properties for investment with a view to long-term appreciation and to engage in the business of acquiring, developing and owning its properties. HCP has made, and may in the future make, occasional sales of properties consistent with its investment objectives. HCP does not intend to enter into any sales that are prohibited transactions. The IRS may contend, however, that one or more of these sales is subject to the 100% penalty tax.

Like-Kind Exchanges

HCP has in the past disposed of properties in transactions intended to qualify as like-kind exchanges under the Code, and may continue this practice in the future. Such like-kind exchanges are intended to result in the deferral of gain for federal income tax purposes. The failure of any such transaction to qualify as a like-kind exchange could subject it to federal income tax, possibly including the 100% prohibited transaction tax, depending on the facts and circumstances surrounding the particular transaction.

Penalty Tax

Any redetermined rents, redetermined deductions or excess interest HCP generates will be subject to a 100% penalty tax. In general, redetermined rents are rents from real property that are overstated as a result of any services furnished by one of HCP's taxable REIT subsidiaries to any of HCP's tenants, and redetermined deductions and excess interest represent any amounts that are deducted by a taxable REIT subsidiary of HCP for amounts paid to HCP that are in excess of the amounts that would have been deducted based on arm's-length negotiations. Rents HCP receives will not constitute redetermined rents if they qualify for certain safe harbor provisions contained in the Code.

HCP believes that, in all instances in which its taxable REIT subsidiaries provide services to its tenants, the fees paid to such taxable REIT subsidiaries for such services are at arm's-length rates, although the fees paid may not satisfy the safe-harbor provisions referenced above. These determinations are inherently factual, and the IRS has broad discretion to assert that amounts paid between related parties should be reallocated to clearly reflect their respective incomes. If the IRS successfully made such an assertion, HCP would be required to pay a 100% penalty tax on the excess of an arm's-length fee for tenant services over the amount actually paid.

Asset Tests

At the close of each quarter of its taxable year, HCP also must satisfy four tests relating to the nature and diversification of its assets.

First, at least 75% of the value of HCP's total assets, including assets held by HCP's qualified REIT subsidiaries and HCP's allocable share of the assets held by the partnerships and limited liability companies in which HCP owns an interest, must be represented by real estate assets, cash, cash items and government securities. For purposes of this test, the term "real estate assets" generally means real property (including interests in real property and interests in mortgages on real property) and shares (or transferable certificates of beneficial interest) in other REITs, as well as any stock or debt instrument attributable to the investment of the proceeds of a stock offering or a public debt offering with a term of at least five years, but only for the one-year period beginning on the date the REIT receives such proceeds.

Second, not more than 25% of the value of HCP's total assets may be represented by securities other than those securities included in the 75% asset test.

Third, of the investments included in the 25% asset class and except for certain investments in other REITs and HCP's qualified REIT subsidiaries and taxable REIT subsidiaries, the value of any

one issuer's securities may not exceed 5% of the value of HCP's total assets, and HCP may not own more than 10% of the total vote or value of the outstanding securities of any one issuer. Solely for the purposes of the 10% value test, however, certain securities including, but not limited to "straight debt" securities having specified characteristics, and loans to an individual or an estate, obligations to pay rents from real property and securities issued by a REIT, are disregarded as securities. In addition, commencing with HCP's taxable year beginning January 1, 2005, solely for purposes of the 10% value test, the determination of its interest in the assets of a partnership or limited liability company in which it owns an interest will be based on HCP's proportionate interest in any securities issued by the partnership or limited liability company, excluding for this purpose certain securities described in the Code. For years prior to 2001, the 10% limit applies only with respect to voting securities of any issuer and not to the value of the securities of any issuer.

Fourth, not more than 20% of the value of HCP's total assets may be represented by the securities of one or more taxable REIT subsidiaries. The 20% asset test is effective for taxable years ending after December 31, 2000.

HCP currently owns 100% of the outstanding stock of several subsidiaries that have elected, together with it, to be treated as taxable REIT subsidiaries. So long as these subsidiaries qualify as taxable REIT subsidiaries, HCP will not be subject to the 5% asset test, the 10% voting securities limitation or the 10% value limitation with respect to HCP's ownership of their securities. HCP may acquire securities in other taxable REIT subsidiaries in the future. HCP believes that the aggregate value of HCP's taxable REIT subsidiaries will not exceed 20% of the value of its gross assets. With respect to each issuer in which HCP currently owns an interest that does not qualify as a REIT, a qualified REIT subsidiary or a taxable REIT subsidiary, HCP believes that HCP's ownership of the securities of any such issuer has complied with the 5% value limitation, the 10% voting securities limitation and the 10% value limitation. No independent appraisals have been obtained to support these conclusions. In addition, there can be no assurance that the IRS will not disagree with HCP's determinations of value. HCP also owns, and may continue to make, loans which must qualify under the "straight debt safe harbor" in order to satisfy the 10% value limitation described above. HCP believes, based on the advice of its tax counsel, that all of these loans have qualified under this safe harbor.

The asset tests described above must be satisfied at the close of each quarter of HCP's taxable year in which it (directly or through its partnerships or limited liability companies) acquires securities in the applicable issuer, increases its ownership of securities of such issuer (including as a result of increasing its interest in a partnership or limited liability company which owns such securities), or acquires other assets. For example, HCP's indirect ownership of securities of an issuer may increase as a result of its capital contributions to a partnership or limited liability company. After initially meeting the asset tests at the close of any quarter, HCP will not lose its status as a REIT for failure to satisfy the asset tests at the end of a later quarter solely by reason of changes in asset values. If HCP fails to satisfy an asset test because HCP acquires securities or other property during a quarter (including as a result of an increase in its interests in a partnership or limited liability company), HCP can cure this failure by disposing of sufficient nonqualifying assets within 30 days after the close of that quarter. HCP believes that HCP has maintained and intends to maintain adequate records of the value of its assets to ensure compliance with the asset tests. In addition, HCP intends to take such actions within 30 days after the close of any quarter as may be required to cure any noncompliance.

Certain relief provisions may be available to HCP if it discovers a failure to satisfy the asset tests described above after the 30 day cure period. Under these provisions, HCP will be deemed to have met the 5% and 10% REIT asset tests if the value of its nonqualifying assets (i) does not exceed the lesser of (a) 1% of the total value of HCP's assets at the end of the applicable quarter or (b) \$10,000,000, and (ii) HCP disposes of the nonqualifying assets or otherwise satisfies such asset tests within (a) six months after the last day of the quarter in which the failure to satisfy the asset tests is discovered or

(b) the period of time prescribed by Treasury Regulations to be issued. For violations of any of the asset tests due to reasonable cause and not due to willful neglect and that are, in the case of the 5% and 10% asset tests, in excess of the deminimis exception described above, HCP may avoid disqualification as a REIT after the 30 day cure period by taking steps including (i) the disposition of sufficient nonqualifying assets, or the taking of other actions, which allow it to meet the asset tests within (a) six months after the last day of the quarter in which the failure to satisfy the asset tests is discovered or (b) the period of time prescribed by Treasury Regulations to be issued, (ii) paying a tax equal to the greater of (a) \$50,000 or (b) the highest corporate tax rate multiplied by the net income generated by the nonqualifying assets, and (iii) disclosing certain information to the IRS.

Although HCP believes that it has satisfied the asset tests described above and plans to take steps to ensure that it satisfies such tests for any quarter with respect to which retesting is to occur, there can be no assurance that HCP will always be successful or will not require a reduction in its overall interest in an issuer (including in a taxable REIT subsidiary). If HCP fails to cure any noncompliance with the asset tests in a timely manner and the relief provisions described above are not available, HCP would cease to qualify as a REIT. See "Failure to Qualify" below.

Annual Distribution Requirements

To maintain its qualification as a REIT, HCP is required to distribute dividends, other than capital gain dividends, to its stockholders in an amount at least equal to the sum of:

90% of HCP's "REIT taxable income"; and

90% of HCP's after tax net income, if any, from foreclosure property; minus

the excess of the sum of specified items of HCP's non-cash income items over 5% of HCP's "REIT taxable income" as described below.

For these purposes, HCP's "REIT taxable income" is computed without regard to the

dividends paid deduction and its net capital gain. In addition, for purposes of this test, non-cash income means income attributable to leveling of stepped rents, original issue discount on purchase money debt, cancellation of indebtedness, and like-kind exchanges that are later determined to be taxable.

In addition, if HCP disposes of any asset it acquired from a corporation which is or has been a C corporation in a transaction in which HCP's basis in the asset is determined by reference to the basis of the asset in the hands of that C corporation within the ten-year period following its acquisition of such asset, HCP would be required to distribute at least 90% of the after-tax gain, if any, HCP recognized on the disposition of the asset, to the extent that gain does not exceed the excess of (a) the fair market value of the asset, over (b) its adjusted basis in the asset, in each case, on the date HCP acquired the asset.

HCP generally must pay, or be treated as paying, the distributions described above in the taxable year to which they relate. At HCP's election, a distribution will be treated as paid in a taxable year if it is declared before HCP timely files its tax return for such year and paid on or before the first regular dividend payment after such declaration, provided such payment is made during the twelve-month period following the close of such year. These distributions generally are taxable to HCP's stockholders, other than tax-exempt entities, in the year in which paid. See " Taxation of Holders of HCP Common Stock." This is so even though these distributions relate to the prior year for purposes of HCP's 90% distribution requirement.

However, such distributions will be treated as paid by HCP in the year actually paid, for purposes of calculating the 4% excise tax.

The amount distributed must not be preferential. To avoid being preferential, every stockholder of the class of stock to which a distribution is made must be treated the same as every other stockholder

of that class, and no class of stock may be treated other than according to its dividend rights as a class. To the extent that HCP does not distribute all of its net capital gain, or distributes at least 90%, but less than 100%, of its "REIT taxable income," as adjusted, HCP will be required to pay tax on the undistributed amount at regular corporate tax rates. HCP believes it has made, and intends to continue to make, timely distributions sufficient to satisfy these annual distribution requirements and to minimize its corporate tax obligations.

HCP expects that its REIT taxable income will be less than its cash flow because of depreciation and other non-cash charges included in computing REIT taxable income. Accordingly, HCP anticipates that it generally will have sufficient cash or liquid assets to enable it to satisfy the distribution requirements described above. However, from time to time, HCP may not have sufficient cash or other liquid assets to meet these distribution requirements due to timing differences between the actual receipt of income and payment of deductible expenses, and the inclusion in income and deduction of expenses in determining its taxable income. If these timing differences occur, HCP may be required to borrow funds or pay dividends in the form of taxable stock dividends in order to meet the distribution requirements.

Under certain circumstances, HCP may be able to rectify an inadvertent failure to meet the 90% distribution requirement for a year by paying "deficiency dividends" to stockholders in a later year, which may be included in its deduction for dividends paid for the earlier year. Thus, HCP may be able to avoid being taxed on amounts distributed as deficiency dividends. However, HCP will be required to pay interest to the IRS based upon the amount of any deduction claimed for deficiency dividends.

In addition, HCP will be required to pay a 4% excise tax to the extent HCP fails to distribute during each calendar year, or in the case of distributions with declaration and record dates falling in the last three months of the calendar year, by the end of January immediately following such year, at least the sum of 85% of HCP's ordinary income for such year, 95% of HCP's capital gain net income for the year and any undistributed taxable income from prior periods. Any ordinary income and net capital gain on which this excise tax is imposed for any year is treated as an amount distributed during that year for purposes of calculating such tax.

For purposes of the 90% distribution requirement and excise tax described above, distributions declared during the last three months of the taxable year, payable to HCP's stockholders of record on a specified date during such period, and paid during January of the following year, will be treated as paid by HCP and received by its stockholders on December 31 of the year in which they are declared.

Failure to Qualify

Specified cure provisions may be available to HCP in the event that HCP discovers a violation of a provision of the Code that would otherwise result in its failure to qualify as a REIT. Except with respect to violations of the REIT income tests and assets tests (for which the cure provisions are described above), and provided the violation is due to reasonable cause and not due to willful neglect, these cure provisions generally impose a \$50,000 penalty for each violation in lieu of a loss of REIT status. If HCP fails to qualify for taxation as a REIT in any taxable year, and the relief provisions of the Code do not apply, HCP will be required to pay tax, including any applicable alternative minimum tax, on its taxable income at regular corporate tax rates. Distributions to HCP's stockholders in any year in which HCP fails to qualify as a REIT will not be deductible by HCP, and HCP will not be required to distribute any amounts to its stockholders. As a result, HCP anticipates that its failure to qualify as a REIT would reduce the cash available for distribution by it to its stockholders. In addition, if HCP fails to qualify as a REIT, all distributions to HCP's stockholders will be taxable as regular corporate dividends to the extent of HCP's current and accumulated earnings and profits. In this event, subject to certain limitations under the Code, corporate distributees may be eligible for the dividends-received deduction. Unless entitled to relief under specific statutory provisions, HCP will also be

disqualified from taxation as a REIT for the four taxable years following the year in which HCP lost its qualification. It is not possible to state whether in all circumstances HCP would be entitled to this statutory relief.

Tax Aspects of the Partnerships

General

HCP owns, directly or indirectly, interests in various partnerships and limited liability companies which are treated as partnerships or disregarded entities for federal income tax purposes and may own interests in additional partnerships and limited liability companies in the future. HCP's ownership interests in such partnerships and limited liability companies involve special tax considerations. These special tax considerations include, for example, the possibility that the IRS might challenge the status of one or more of the partnerships or limited liability companies in which HCP owns an interest as partnerships or disregarded entities, as opposed to associations taxable as corporations, for federal income tax purposes. If a partnership or limited liability company in which HCP owns an interest, or one or more of its subsidiary partnerships or limited liability companies, were treated as an association, it would be taxable as a corporation and would therefore be subject to an entity-level tax on its income. In this situation, the character of HCP's assets and items of gross income would change, and could prevent it from satisfying the REIT asset tests and possibly the REIT income tests. See " Taxation of HCP as a Real Estate Investment Trust Asset Tests" and " Taxation of HCP as a Real Estate Investment Trust Income Tests." This, in turn, could prevent it from qualifying as a REIT. See " Failure to Qualify" for a discussion of the effect of HCP's failure to meet these tests. In addition, a change in the tax status of one or more of the partnerships or limited liability companies in which HCP owns an interest might be treated as a taxable event. If so, HCP might incur a tax liability without any related cash distributions.

Treasury Regulations that apply for tax periods beginning on or after January 1, 1997, provide that a domestic business entity not otherwise organized as a corporation (an "eligible entity") may elect to be taxed as a partnership or, if owned by a single person, as a disregarded entity for federal income tax purposes. Unless it elects otherwise, an eligible entity in existence prior to January 1, 1997, will have the same classification for federal income tax purposes that it claimed under the entity classification Treasury

Regulations in effect prior to this date. In addition, an eligible entity which did not exist or did not claim a classification prior to January 1, 1997, will be classified as a partnership or disregarded entity, if owned by a single person, as a for federal income tax purposes unless it elects otherwise. All of the partnerships and limited liability companies in which HCP owns an interest intend to claim classification as partnerships or disregarded entities under these Treasury Regulations. As a result, HCP believes that these partnerships and limited liability companies will be classified as partnerships or disregarded entities for federal income tax purposes.

Allocations of Income, Gain, Loss and Deduction

A partnership or limited liability company agreement generally will determine the allocation of income and losses among partners or members. These allocations, however, will be disregarded for tax purposes if they do not comply with the provisions of Section 704(b) of the Code and the related Treasury Regulations. Generally, Section 704(b) of the Code and the related Treasury Regulations require that partnership and limited liability company allocations respect the economic arrangement of the partners and members. If an allocation is not recognized for federal income tax purposes, the relevant item will be reallocated according to the partners' or members' interests in the partnership or limited liability company, as the case may be. This reallocation will be determined by taking into account all of the facts and circumstances relating to the economic arrangement of the partners or members with respect to such item. The allocations of taxable income and loss in each of the partnerships and limited liability companies in which HCP owns an interest are intended to comply with the requirements of Section 704(b) of the Code and the applicable Treasury Regulations.

Tax Allocations with Respect to the Properties

Under Section 704(c) of the Code, income, gain, loss and deduction attributable to appreciated or depreciated property that is contributed to a partnership or limited liability company in exchange for an interest in the partnership or limited liability company must be allocated in a manner so that the contributing partner or member is charged with the unrealized gain or benefits from the unrealized loss associated with the property at the time of the contribution. The amount of the unrealized gain or loss generally is equal to the difference between the fair market value and the adjusted tax basis of the contributed property at the time of contribution. These allocations are solely for federal income tax purposes and do not affect the book capital accounts or other economic or legal arrangements among the partners or members. Some of the partnerships and/or limited liability companies in which HCP owns an interest were formed by way of contributions of appreciated property. The relevant partnership and/or limited liability company agreements require that allocations be made in a manner consistent with Section 704(c) of the Code. This could cause HCP to be allocated lower amounts of depreciation deductions for tax purposes than would be allocated to it if the contributed properties were acquired in a cash purchase, and could cause it to be allocated taxable gain upon a sale of the contributed properties in excess of the economic or book income allocated to it as a result of such sale. These adjustments could make it more difficult for HCP to satisfy the REIT distribution requirements.

Other Tax Consequences

HCP may be required to pay tax in various state or local jurisdictions, including those in which HCP transacts business. In addition, HCP's state and local tax treatment may not conform to the federal income tax consequences discussed above.

Tax Liabilities and Attributes Inherited From CRP

If CRP failed to qualify as a REIT for any of its taxable years, it would be required to pay federal income tax (including any applicable alternative minimum tax) on its taxable income at regular corporate rates. Unless statutory relief provisions apply, CRP would be disqualified from treatment as a REIT for the four taxable years following the year during which it lost qualification. Because the merger will be treated for income tax purposes as if CRP sold all of its assets in a taxable transaction, if CRP did not

qualify as a REIT for the tax year of the merger, it would be subject to tax in respect of the built-in gain in all of its assets because it would not be eligible for the dividends paid deduction that is available to a REIT. "Built-in gain" generally means the excess of the fair market value of an asset over its adjusted tax basis. HCP, as successor-in-interest to CRP, would be required to pay these taxes. Furthermore, after the merger, the asset and income tests described in "Taxation of HCP as a Real Estate Investment Trust Asset Tests" and "Taxation of HCP as a Real Estate Investment Trust Income Tests" will apply to all of HCP's assets, including the assets HCP acquires from CRP, and to all of HCP's income, including the income derived from the assets HCP acquires from CRP. As a result, the nature of the assets that HCP acquires from CRP and the income HCP derives from those assets may have an effect on HCP's tax status as a REIT.

Qualification as a REIT requires CRP to satisfy numerous requirements, some on an annual and others on a quarterly basis, as described above with respect to HCP. There are only limited judicial and administrative interpretations of these requirements and qualification as a REIT involves the determination of various factual matters and circumstances which were not entirely within CRP's control.

It is a condition to the closing of the merger that Greenberg Traurig, LLP, CRP's REIT counsel, render an opinion to HCP, dated as of the closing date of the merger, to the effect that CRP qualified as a REIT under the Code for the taxable years ending December 31, 1999 through December 31, 2005, CRP is organized in conformity with the requirements for qualification as a REIT, and CRP's method of operation has enabled CRP to satisfy the requirements for qualification as a REIT under

the Code for the taxable years ending on or prior to the closing date of the merger. This opinion will be based on various assumptions and representations as to factual matters, including representations made by CRP in a factual certificate provided by one of its officers, as well as other oral and written statements of officers and other representatives of CRP and others as to the existence and consequence of certain factual and other matters. CRP's ability to achieve and maintain qualification as a REIT through the closing date of the merger depends upon its ability to achieve and maintain certain diversity of stock ownership requirements and, through actual annual operating results, certain requirements under the Code regarding its income, assets and distribution levels. No assurance can be given that the actual ownership of CRP's stock and its actual operating results and distributions for any taxable year satisfy the tests necessary for CRP to have achieved and maintained its status as a REIT.

Tax Liabilities and Attributes Inherited from the Advisor

Concurrently with the merger, the Advisor will merge with and into Ocean Acquisition 2, LLC, or Ocean 2, a wholly owned subsidiary of HCP, pursuant to the Agreement and Plan of Merger, dated as of May 1, 2006, by and among HCP, the Advisor, the stockholders of the Advisor and Ocean 2, which we refer to as the Advisor merger. As a result of the Advisor merger, HCP will succeed to the assets and the liabilities of the Advisor, including any liabilities for unpaid taxes and any tax liabilities created in connection with the Advisor merger. It is a condition to the closing of the Advisor merger that HCP receive an opinion of its counsel, Sullivan & Cromwell LLP, and that the Advisor and the Advisor stockholders receive an opinion of their counsel, Arnold & Porter LLP, substantially to the effect that, on the basis of the facts, representations and assumptions set forth or referred to in such opinions, for federal income tax purposes the Advisor merger will qualify as a reorganization within the meaning of Section 368(a) of the Code. To the extent that the Advisor merger so qualifies, no gain or loss will be recognized by the Advisor or HCP in the Advisor merger. HCP's tax basis in the Advisor's assets acquired in merger will be equal to the Advisor's tax basis in such assets immediately prior to the merger. If HCP disposes of such assets in a taxable transaction during the ten-year period beginning on the date of the Advisor merger, then HCP will be required to pay tax at the highest regular corporate tax rate on the gain recognized to the extent of the excess of (a) the fair market value of the asset over (b) HCP's adjusted basis in the asset, in each case

determined as of the date of the Advisor merger. The opinions of counsel to be delivered in connection with the Advisor merger represent the best legal judgment of counsel to HCP and counsel to the Advisor and the Advisor stockholders and are not binding on the IRS or the courts. Neither HCP nor the Advisor has requested nor will request a ruling from the IRS as to the tax consequences of the Advisor merger, and there can be no assurance that the IRS will agree with the conclusions in the above-described opinions.

If the Advisor merger does not qualify as a reorganization within the meaning of Section 368(a) of the Code, the Advisor merger would be treated as a sale of the Advisor's assets to HCP in a taxable transaction, and the Advisor would recognize taxable gain. In such a case, HCP, as the Advisor's successor-in-interest, would be required to pay the tax on any such gain. In addition, HCP's tax basis in the Advisor's assets would be equal to the merger consideration paid to the Advisor stockholders, which could be higher than the tax basis of such assets if the Advisor merger qualified as a reorganization. This higher tax basis would cause HCP to have higher depreciation deductions and lower gain on the sale of the Advisor assets.

As a result of the Advisor merger and assuming it qualifies as a reorganization under the Code, HCP will succeed to the tax attributes and earnings and profits of the Advisor. To qualify as a REIT, HCP must distribute any such earnings and profits by the close of the taxable year in which the Advisor merger occurs. Any adjustments of the Advisor's income for taxable years ending on or before the Advisor merger, including as a result of an examination of the Advisor's tax returns by the IRS, could affect the calculation of the Advisor's earnings and profits. If the IRS were to determine that HCP acquired earnings and profits from the Advisor that it failed to distribute prior to the end of the taxable year in which the Advisor merger occurs, HCP could avoid disqualification as a REIT by using

"deficiency dividend" procedures. Under these procedures, HCP generally would be required to distribute any such earnings and profits to its stockholders within 90 days of the determination and pay a statutory interest charge at a specified rate to the IRS.

Taxation of Holders of HCP's Common Stock

Taxation of Taxable U.S. Stockholders Generally

Distributions Generally. Distributions out of HCP's current or accumulated earnings and profits will be treated as dividends and, other than capital gain dividends, and certain amounts that have previously been subject to corporate level tax, discussed below, will be taxable to HCP's taxable U.S. stockholders as ordinary income. See " Tax Rates" below. As long as HCP qualifies as a REIT, these distributions will not be eligible for the dividends-received deduction in the case of U.S. stockholders that are corporations. For purposes of determining whether distributions to holders of HCP common stock are out of current or accumulated earnings and profits, HCP's earnings and profits will be allocated first to HCP's outstanding preferred stock and then to HCP's outstanding common stock.

To the extent that HCP makes distributions on its common stock in excess of its current and accumulated earnings and profits, these distributions will be treated first as a tax-free return of capital to a U.S. stockholder. This treatment will reduce the U.S. stockholder's adjusted tax basis in its shares of HCP common stock by the amount of the distribution, but not below zero. Distributions in excess of HCP's current and accumulated earnings and profits and in excess of a U.S. stockholder's adjusted tax basis in its shares will be taxable as capital gain. Such gain will be taxable as long-term capital gain if the shares have been held for more than one year. Dividends HCP declares in October, November, or December of any year and which are payable to a stockholder of record on a specified date in any of these months will be treated as both paid by it and received by the stockholder on December 31 of that year, provided HCP actually pays the dividend on or before January 31 of the following year. U.S. stockholders may not include in their own income tax returns any of HCP's net operating losses or capital losses.

Capital Gain Dividends. Dividends that HCP properly designates as capital gain dividends will be taxable to HCP's taxable U.S. stockholders as a gain from the sale or disposition of a capital

asset, to the extent that such gain does not exceed HCP's actual net capital gain for the taxable year. These gains may be taxable to non-corporate U.S. stockholders at a 15% or 25% rate. U.S. stockholders that are corporations may, however, be required to treat up to 20% of some capital gain dividends as ordinary income. If HCP properly designates any portion of a dividend as a capital gain dividend then, except as otherwise required by law, HCP is required by the terms of its corporate charter to allocate a portion of the total capital gain dividends paid or made available to holders of all classes of HCP's stock for the year to the holders of HCP preferred stock in proportion to the amount that HCP's total dividends, as determined for United States federal income tax purposes, paid or made available to the holders of such stock for the year bears to the total dividends, as determined for United States federal income tax purposes, paid or made available to holders of all classes of HCP's stock for the year.

Retention of Net Capital Gains. HCP may elect to retain, rather than distribute as a capital gain dividend, all or a portion of HCP's net capital gains. If HCP makes this election, HCP would pay tax on HCP's retained net capital gains. In addition, to the extent HCP so elects, a U.S. stockholder generally would:

include its pro rata share of HCP's undistributed net capital gains in computing its long-term capital gains in its return for its taxable year in which the last day of HCP's taxable year falls, subject to certain limitations as to the amount that is includable;

be deemed to have paid the capital gains tax imposed on it on the designated amounts included in the U.S. stockholder's long-term capital gains;

receive a credit or refund for the amount of tax deemed paid by it;

increase the adjusted basis of its common stock by the difference between the amount of includable gains and the tax deemed to have been paid by it; and

in the case of a U.S. stockholder that is a corporation, appropriately adjust its earnings and profits for the retained capital gains in accordance with Treasury Regulations to be promulgated by the IRS.

Passive Activity Losses and Investment Interest Limitations. Distributions HCP makes and gain arising from the sale or exchange by a U.S. stockholder of HCP's shares will not be treated as passive activity income. As a result, U.S. stockholders generally will not be able to apply any "passive losses" against this income or gain. A U.S. stockholder may elect to treat capital gain dividends, capital gains from the disposition of stock and qualified dividend income as investment income for purposes of computing the investment interest limitation, but in such case, the stockholder will be taxed at ordinary income rates on such amount. Other distributions made by HCP, to the extent they do not constitute a return of capital, generally will be treated as investment income for purposes of computing the investment interest limitation.

Dispositions of HCP's Common Stock. If a U.S. stockholder sells or disposes of shares of HCP common stock to a person other than HCP (or to HCP in a manner that qualifies as a redemption for federal income tax purposes), it will recognize gain or loss for federal income tax purposes in an amount equal to the difference between the amount of cash and the fair market value of any property received on the sale or other disposition and the holder's adjusted basis in the shares for tax purposes. This gain or loss, except as provided below, will be long-term capital gain or loss if the holder has held the common stock for more than one year. If, however, a U.S. stockholder recognizes loss upon the sale or other disposition of HCP common stock that it has held for six months or less, after applying certain holding period rules, the loss recognized will be treated as a long-term capital loss to the extent the U.S. stockholder received distributions from it which

were required to be treated as long-term capital gains.

Backup Withholding

HCP reports to HCP's U.S. stockholders and the IRS the amount of dividends paid during each calendar year, and the amount of any tax withheld. Under the backup withholding rules, a stockholder may be subject to backup withholding with respect to dividends paid unless the holder is a corporation or comes within certain other exempt categories and, when required, demonstrates this fact, or provides a taxpayer identification number, certifies as to no loss of exemption from backup withholding, and otherwise complies with applicable requirements of the backup withholding rules. A U.S. stockholder that does not provide HCP with its correct taxpayer identification number may also be subject to penalties imposed by the IRS. Backup withholding is not an additional tax. Any amount paid as backup withholding will be creditable against the stockholder's federal income tax liability. In addition, HCP may be required to withhold a portion of capital gain distributions to any stockholders who fail to certify their non-foreign status. See " Taxation of Non-U.S. Stockholders."

Tax Rates

The maximum tax rate for non-corporate taxpayers for (1) capital gains, including certain "capital gain dividends," has generally been reduced to 15% (although depending on the characteristics of the assets which produced these gains and on designations which HCP may make, certain capital gain dividends may be taxed at a 25% rate) and (2) "qualified dividend income" has generally been reduced to 15%. In general, dividends payable by REITs are not eligible for the reduced tax rate on corporate dividends, except to the extent that certain holding requirements have been met and the REIT's dividends are attributable to dividends received from taxable corporations (such as its taxable REIT

subsidiaries), to income that was subject to tax at the corporate/REIT level (for example, if it distributed taxable income that it retained and paid tax on in the prior taxable year) or to dividends properly designated by the REIT as "capital gain dividends." The currently applicable provisions of the United States federal income tax laws relating to the 15% tax rate are currently scheduled to "sunset" or revert to the provisions of prior law effective for taxable years beginning after December 31, 2010, at which time the capital gains tax rate will be increased to 20% and the rate applicable to dividends will be increased to the tax rate then applicable to ordinary income.

Taxation of Tax-Exempt Stockholders

Dividend income from HCP and gain arising upon a sale of HCP common stock generally will not be unrelated business taxable income to a tax-exempt stockholder, except as described below. This income or gain will be unrelated business taxable income, however, if a tax-exempt stockholder holds its shares as "debt-financed property" within the meaning of the Code or if the shares are used in a trade or business of the tax-exempt stockholder. Generally, "debt-financed property" is property, the acquisition or holding of which was financed through a borrowing by the tax-exempt stockholder.

For tax-exempt stockholders which are social clubs, voluntary employee benefit associations, supplemental unemployment benefit trusts, or qualified group legal services plans exempt from federal income taxation under Sections 501(c)(7), (c)(9), (c)(17) or (c)(20) of the Code, respectively, income from an investment in HCP's shares will constitute unrelated business taxable income unless the organization is able to properly claim a deduction for amounts set aside or placed in reserve for specific purposes so as to offset the income generated by its investment in HCP's shares. These prospective investors should consult their tax advisors concerning these "set aside" and reserve requirements.

Notwithstanding the above, however, a portion of the dividends paid by a "pension-held REIT" may be treated as unrelated business taxable income as to certain trusts that hold more than 10%, by value, of the interests in the REIT. A REIT will not be a "pension-held REIT" if it is able to satisfy the "not closely held" requirement without relying on the "look-through" exception with respect to certain trusts or if such REIT is not "predominantly held" by "qualified trusts." As a result of limitations on the transfer and ownership of stock contained in HCP's charter, HCP does not expect to be classified as a "pension-held REIT," and as a result, the tax treatment described in this

paragraph should be inapplicable to HCP's stockholders. However, because HCP's stock is publicly traded, HCP cannot guarantee that this will always be the case.

Taxation of Non-U.S. Stockholders

The following discussion addresses the rules governing United States federal income taxation of the ownership and disposition of HCP common stock by non-U.S. stockholders. These rules are complex, and no attempt is made herein to provide more than a brief summary of such rules. Accordingly, the discussion does not address all aspects of United States federal income taxation that may be relevant to a non-U.S. stockholder in light of its particular circumstances and does not address any state, local or foreign tax consequences. HCP urges non-U.S. stockholders to consult their tax advisors to determine the impact of federal, state, local and foreign income tax laws on the purchase, ownership, and disposition of shares of HCP common stock, including any reporting requirements.

Distributions Generally. Distributions that are neither attributable to gain from HCP's sale or exchange of United States real property interests nor designated by it as capital gain dividends will be treated as dividends of ordinary income to the extent that they are made out of HCP's current or accumulated earnings and profits. Such distributions ordinarily will be subject to withholding of United States federal income tax at a 30% rate or such lower rate as may be specified by an applicable income tax treaty unless the distributions are treated as effectively connected with the conduct by the non-U.S. stockholder of a United States trade or business. Under certain treaties, however, lower withholding

rates generally applicable to dividends do not apply to dividends from a REIT. Certain certification and disclosure requirements must be satisfied to be exempt from withholding under the effectively connected income exemption. Dividends that are treated as effectively connected with such a trade or business will be subject to tax on a net basis at graduated rates, in the same manner as dividends paid to U.S. stockholders are subject to tax, and are generally not subject to withholding. Any such dividends received by a non-U.S. stockholder that is a corporation may also be subject to an additional branch profits tax at a 30% rate or such lower rate as may be specified by an applicable income tax treaty.

HCP expects to withhold United States income tax at the rate of 30% on any distributions made to a non-U.S. stockholder unless:

a lower treaty rate applies and the non-U.S. stockholder files with it an IRS Form W-8BEN evidencing eligibility for that reduced treaty rate; or

the non-U.S. stockholder files an IRS Form W-8ECI with it claiming that the distribution is income effectively connected with the non-U.S. stockholder's trade or business.

Distributions in excess of HCP's current and accumulated earnings and profits will not be taxable to a non-U.S. stockholder to the extent that such distributions do not exceed the non-U.S. stockholder's adjusted basis in HCP common stock, but rather will reduce the adjusted basis of such common stock. To the extent that these distributions exceed a non-U.S. stockholder's adjusted basis in HCP common stock, they will give rise to gain from the sale or exchange of such stock. The tax treatment of this gain is described below.

For withholding purposes, HCP expects to treat all distributions as made out of HCP's current or accumulated earnings and profits. However, amounts withheld should generally be refundable if it is subsequently determined that the distribution was, in fact, in excess of HCP's current and accumulated earnings and profits.

Capital Gain Dividends and Distributions
Attributable to a Sale or Exchange of United States
Real Property Interests. Distributions to a

non-U.S. stockholder that HCP properly designates as capital gain dividends, other than those arising from the disposition of a United States real property interest, generally should not be subject to United States federal income taxation, unless:

(1) the investment in HCP common stock is treated as effectively connected with the non-U.S. stockholder's United States trade or business, in which case the non-U.S. stockholder will be subject to the same treatment as U.S. stockholders with respect to such gain, except that a non-U.S. stockholder that is a foreign corporation may also be subject to the 30% branch profits tax, as discussed above; or

the non-U.S. stockholder is a nonresident alien individual who is present in the United States for 183 days or more during the taxable year and certain other conditions are met, in which case the nonresident alien individual will be subject to a 30% tax on the individual's capital gains.

Pursuant to the Foreign Investment in Real Property Tax Act, which is referred to as "FIRPTA," distributions to a non-U.S. stockholder that are attributable to gain from HCP's sale or exchange of United States real property interests (whether or not designated as capital gain dividends) will cause the non-U.S. stockholder to be treated as recognizing such gain as income effectively connected with a United States trade or business. Non-U.S. stockholders would generally be taxed at the same rates applicable to U.S. stockholders, subject to a special alternative minimum tax in the case of nonresident alien individuals. HCP also will be required to withhold and to remit to the IRS 35% of any distribution to a non-U.S. stockholder that is designated as a capital gain dividend, or, if greater, 35% of a distribution to the non-U.S. stockholder that could have been designated as a capital gain dividend. The amount withheld is creditable against the non-U.S. stockholder's United States federal

income tax liability. However, any distribution with respect to any class of stock which is regularly traded on an established securities market located in the United States is not subject to FIRPTA, and therefore, not subject to the 35% U.S. withholding tax described above, if the non-United States stockholder did not own more than 5% of such class of stock at any time during the one-year period ending on the date of the distribution. Instead, such distributions will be treated as ordinary dividend distributions.

Retention of Net Capital Gains. Although the law is not clear on the matter, it appears that amounts designated by HCP as retained capital gains in respect of the common stock held by U.S. stockholders generally should be treated with respect to non-U.S. stockholders in the same manner as actual distributions by HCP of capital gain dividends. Under this approach, a non-U.S. stockholder would be able to offset as a credit against its United States federal income tax liability resulting from its proportionate share of the tax paid by it on such retained capital gains, and to receive from the IRS a refund to the extent its proportionate share of such tax paid by it exceeds their actual United States federal income tax liability.

Sale of HCP's Common Stock. Gain recognized by a non-U.S. stockholder upon the sale or exchange of HCP common stock generally will not be subject to United States federal income taxation unless such stock constitutes a "United States real property interest" within the meaning of FIRPTA. HCP common stock will not constitute a "United States real property interest" so long as HCP is a "domestically-controlled qualified investment entity." A "domestically-controlled qualified investment entity" includes a REIT in which at all times during a specified testing period less than 50% in value of its stock is held directly or indirectly by non-U.S. stockholders. HCP believes, but cannot guarantee, that HCP has been a "domestically-controlled qualified investment entity." Even if HCP has been a "domestically-controlled qualified investment entity," because HCP's capital stock is publicly traded, no assurance can be given that HCP will continue to be a "domestically-controlled qualified investment entity."

Notwithstanding the foregoing, gain from the sale or exchange of HCP common stock not otherwise subject to FIRPTA will be taxable to a non-U.S. stockholder if either (1) the investment in HCP common stock is treated as effectively connected with the non-U.S. stockholder's United States trade or business or (2) the non-U.S.

stockholder is a nonresident alien individual who is present in the United States for 183 days or more during the taxable year and certain other conditions are met. In addition, even if HCP is a domestically controlled qualified investment entity, upon disposition of HCP common stock (subject to the 5% exception applicable to "regularly traded" stock described below), a non-U.S. stockholder may be treated as having gain from the sale or exchange of United States real property interest if the non-U.S. stockholder (1) disposes of HCP common stock within a 30-day period preceding the ex-dividend date of a distribution, any portion of which, but for the disposition, would have been treated as gain from the sale or exchange of a United States real property interest and (2) acquires, or enters into a contract or option to acquire, other shares of HCP's common stock within 30 days after such ex-dividend date.

Even if HCP does not qualify as a "domestically-controlled qualified investment entity" at the time a non-U.S. stockholder sells or exchanges HCP common stock, gain arising from such a sale or exchange would not be subject to United States taxation under FIRPTA as a sale of a "United States real property interest" if:

- (1)

 HCP common stock is "regularly traded," as defined by applicable Treasury regulations, on an established securities market such as the NYSE; and
- such non-U.S. stockholder owned, actually and constructively, 5% or less of HCP common stock throughout the five-year period ending on the date of the sale or exchange.

If gain on the sale or exchange of HCP common stock were subject to taxation under FIRPTA, the non-U.S. stockholder would be subject to regular United States federal income tax with respect to such

gain in the same manner as a taxable U.S. stockholder (subject to any applicable alternative minimum tax and a special alternative minimum tax in the case of nonresident alien individuals) and the purchaser of the common stock would be required to withhold and remit to the IRS 10% of the purchase price.

Backup Withholding Tax and Information Reporting. Generally, HCP must report annually to the IRS the amount of dividends paid to a non-U.S. stockholder, such holder's name and address, and the amount of tax withheld, if any. A similar report is sent to the non-U.S. stockholder. Pursuant to tax treaties or other agreements, the IRS may make its reports available to tax authorities in the non-U.S. stockholder's country of residence.

Payments of dividends or of proceeds from the disposition of stock made to a non-U.S. stockholder may be subject to information reporting and backup withholding unless such holder establishes an exemption, for example, by properly certifying its non-United States status on an IRS Form W-8BEN or another appropriate version of IRS Form W-8. Notwithstanding the foregoing, backup withholding and information reporting may apply if either HCP or its paying agent has actual knowledge, or reason to know, that a stockholder is a United States person.

Backup withholding is not an additional tax. Rather, the United States income tax liability of persons subject to backup withholding will be reduced by the amount of tax withheld. If withholding results in an overpayment of taxes, a refund or credit may be obtained, provided that the required information is furnished to the IRS.

Other Tax Consequences

State, local and foreign income tax laws may differ substantially from the corresponding U.S. federal income tax laws, and this discussion does not purport to describe any aspect of the tax laws of any state, local or foreign jurisdiction. You should consult your tax advisor regarding the effect of state, local and foreign tax laws with respect to HCP's tax treatment as a REIT and on an investment in HCP common stock.

REIT Qualification of HCP and CRP

As a condition to the completion of the merger, Latham & Watkins, LLP, special tax counsel to HCP, will deliver an opinion to CRP that HCP was organized and has operated in conformity with the requirements for qualification as a REIT under the Internal Revenue Code of 1954 for its taxable years ending December 31,

1985 and 1986 and in conformity with the requirements for qualifications as a REIT under the Code, commencing with its taxable year ending December 31, 1987, and that its proposed method of operation will enable HCP to continue to meet the requirements for qualification and taxation as a REIT under the Code.

The opinion of Latham & Watkins, LLP relies upon customary representations made by HCP about factual matters relating to the organization and operation of HCP and its subsidiaries.

It is a condition to the closing of the merger that Greenberg Traurig, LLP, CRP's REIT counsel, render an opinion to HCP, dated as of the closing date of the merger, to the effect that, CRP qualified as a REIT under the Code for the taxable years ending December 31, 1999 through December 31, 2005, CRP is organized in conformity with the requirements for qualification as a REIT, and CRP's method of operations has enabled CRP to satisfy the requirements for qualification as a REIT under the Code for the taxable years ending on or prior to the closing date of the merger. This opinion will be based on various assumptions and representations as to factual matters, including representations made by CRP in a factual certificate provided by one of its officers, as well as other oral and written statements of officers and other representatives of CRP and others as to the existence and consequence of certain factual and other matters.

INFORMATION ABOUT HCP AND MERGER SUB

Health Care Property Investors, Inc.

Health Care Property Investors, Inc., or HCP, together with its consolidated subsidiaries and joint ventures, invests primarily in real estate serving the healthcare industry in the United States. HCP is a Maryland corporation organized to qualify as a real estate investment trust, or REIT, in 1985. HCP is headquartered in Long Beach, California, with operations in Nashville, Tennessee, and its portfolio includes, as of March 31, 2006, interests in 534 properties in 42 states and consists of 138 senior housing facilities, 185 medical office buildings, 29 hospitals, 156 skilled nursing facilities and 26 other healthcare facilities. HCP acquires healthcare facilities and leases them to healthcare providers and provides mortgage financing secured by healthcare facilities. HCP's portfolio includes: (1) senior housing, including independent living facilities, assisted living facilities, and continuing care retirement communities; (2) medical office buildings; (3) hospitals; (4) skilled nursing facilities; and (5) other healthcare facilities, including laboratory and office buildings.

HCP's executive offices are located at 3760 Kilroy Airport Way, Suite 300, Long Beach, CA 90806 and its telephone number is (562) 733-5100. For additional information on HCP, see "Where You Can Find More Information" on page 181.

Ocean Acquisition 1, Inc.

Ocean Acquisition 1, Inc., or Merger Sub, is a Maryland corporation, recently organized as a wholly-owned subsidiary of HCP solely for the purpose of effecting the merger. It has no material assets and has not engaged in any activities except in connection with the merger. Merger Sub's executive offices are located at 3760 Kilroy Airport Way, Suite 300, Long Beach, CA 90806 and its telephone number is (562) 733-5100.

INFORMATION ABOUT CRP

CRP's Business

General

CNL Retirement Properties, Inc., or CRP, was organized on December 22, 1997 as a Maryland corporation, and elected to be taxed as a REIT commencing with its taxable year ending December 31, 1999. Through its wholly-owned subsidiaries, consolidated partnerships and joint ventures, CRP primarily acquires, develops, manages and owns real estate properties. CRP is one of the nation's largest investors in healthcare-related real estate, investing primarily in properties related to senior housing and healthcare facilities located across the United States. The properties include independent living, assisted living and skilled nursing facilities, continuing care retirement communities and life care communities, medical office buildings, walk-in clinics, free standing ambulatory surgery centers, specialty or general hospitals and other types of healthcare-related facilities. As of March 31, 2006, CRP had invested \$3.7 billion in 271 properties located in 33 states, consisting of 185 senior housing facilities and 86 medical facilities, including two specialty hospitals and two walk-in clinics. As of March 31, 2006, CRP also owned two senior housing facilities and a parcel of land that CRP held for sale. CRP leases its senior housing properties on a long-term (generally 15 years), triple-net basis and its medical facilities on either a triple-net or gross basis generally over 5 to 15 years.

CRP's executive offices are located at the CNL Center II at City Commons, 420 South Orange Avenue, Suite 500, Orlando, Florida 32801 and its telephone number is (407) 650-1000.

CNL Retirement Corp., or the Advisor, is CRP's advisor and provides management, acquisition, advisory and administrative services to CRP.

Since CRP's inception it has made five best efforts public offerings of common stock and received aggregate subscriptions of \$2.7 billion. CRP's fifth public offering closed on March 26, 2006.

CRP operates in one business segment which is the ownership, development, management and leasing of healthcare-related real estate.

CRP's Investment Objectives and Policies

CRP's charter provides that CRP's primary investment objectives are to preserve, protect, and enhance CRP's assets while (i) making quarterly distributions; (ii) obtaining fixed income through the receipt of base rent, and increasing CRP's income (and distributions) and providing protection against inflation through automatic fixed increases in base rent or increases in base rent based on increases in consumer price indices over the terms of the leases, and obtaining fixed income through the receipt of payments on mortgage loans and secured equipment leases; (iii) continuing to qualify as a REIT for federal income tax purposes; and (iv) providing stockholders of CRP with liquidity of their investment, either in whole or in part, on or before December 31, 2008, through (a) listing, or, (b) if listing does not occur by December 31, 2008, the commencement of orderly sales of CRP's assets, outside the ordinary course of business and consistent with its objective of qualifying as a REIT, and distribution of the proceeds thereof. The sheltering from tax of income from other sources is not an objective of CRP. If CRP is successful in achieving its investment and operating objectives, the stockholders (other than tax-exempt entities) are likely to recognize taxable income in each year. While there is no order of priority intended in the listing of CRP's objectives, stockholders should realize that the ability of CRP to meet these objectives may be severely handicapped by any lack of diversification of CRP's investments and the terms of the leases.

In the event the merger is not consummated, CRP intends to meet its objectives through its investment policies by purchasing interests in carefully selected, well-located properties either directly, or indirectly through the acquisition of interests in entities which own such properties or interests

therein, and leasing the properties primarily on a "triple-net" basis (which means that the tenant will be responsible for paying the cost of all repairs, maintenance, property taxes, and insurance) to tenants under leases generally requiring the tenant to pay base annual rent with automatic fixed increases in base rent or increases in base rent based on increases in consumer price indices over the term of the lease. CRP may also invest in medical office buildings and expects leases relating to medical office buildings will be on a "gross" basis. In addition, if the merger is not consummated, CRP may offer mortgage loans and secured equipment leases to operators, and CRP may also invest a small portion of its total assets in equity interests in businesses that provide services to or are otherwise ancillary to the retirement industry.

In accordance with its investment policies, CRP intends, unless the merger is consummated, to invest in properties whose tenants are operators to be selected by CRP, or whose tenants have contracted with third-party operators approved by CRP, based upon recommendations by the Advisor. With respect to investments in medical office buildings, CRP intends, unless the merger is consummated, to contract with third-party property managers. Although there is no limit on the number of properties of a particular tenant or operator which CRP may acquire, CRP's board of directors, including a majority of the independent directors, will review CRP's properties and potential investments in terms of geographic, facility type, tenant, operator and brand diversification. Potential mortgage loan borrowers and secured equipment lease lessees or borrowers will similarly be operators selected or approved by CRP, following the Advisor's recommendations. CRP has undertaken, consistent with its objective of qualifying as a REIT for federal income tax purposes, to ensure that the value of all secured equipment leases, in the aggregate, will not exceed 25% of CRP's total assets, while secured equipment leases to any single lessee or borrower, in the aggregate, will not exceed 5% of CRP's total assets. It is intended that any future investments will be made in properties, mortgage loans, other permitted investments and secured equipment leases in various locations in an attempt to achieve diversification and thereby minimize the effect of changes in local economic conditions and certain other risks.

The investment objectives of CRP may not be changed without the approval of stockholders owning a majority of the shares of outstanding common stock. CRP's bylaws require the independent directors to review CRP's investment

policies at least annually to determine that the policies are in the best interests of the stockholders. The determination must be set forth in the minutes of CRP's board of directors along with the basis for such determination. CRP's directors (including a majority of the independent directors) have the right, without a stockholder vote, to alter CRP's investment policies but only to the extent consistent with CRP's investment objectives and investment limitations. See "Certain Investment Limitations" below.

Certain Investment Limitations

In addition to other investment restrictions imposed by CRP's board of directors from time to time, consistent with CRP's objective of qualifying as a REIT, CRP's charter or bylaws provide for the following limitations on CRP's investments:

- 1. Not more than 10% of CRP's total assets can be invested in unimproved real property or mortgage loans on unimproved real property. For purposes of this paragraph, "unimproved real property" does not include any property under construction, under contract for development or planned for development within one year.
- 2. CRP cannot invest in commodities or commodity future contracts. This limitation is not intended to apply to interest rate futures, when used solely for hedging purposes.
- 3. CRP cannot invest in or make mortgage loans unless an appraisal is obtained concerning the underlying property. Mortgage indebtedness on any property cannot exceed such property's appraised value. In cases in which a majority of the independent directors so determine, and in all

cases in which the mortgage loan involves the Advisor, CRP's directors, or CRP's affiliates, such appraisal must be obtained from an independent expert concerning the underlying property. Such appraisal must be maintained in CRP's records for at least five years, and must be available for inspection and duplication by any stockholder. In addition to the appraisal, a mortgagee's or owner's title insurance policy or commitment as to the priority of the mortgage or condition of the title must be obtained.

- 4. CRP may not make or invest in mortgage loans, including construction loans, on any one property if the aggregate amount of all mortgage loans outstanding on the property, including the loans of CRP, would exceed an amount equal to 85% of the appraised value of the property as determined by appraisal unless substantial justification exists because of the presence of other underwriting criteria. For purposes of this paragraph, the "aggregate amount of all mortgage loans outstanding on the property, including the loans of CRP" includes all interest (excluding contingent participation in income and/or appreciation in value of the mortgaged property), the current payment of which may be deferred pursuant to the terms of such loans, to the extent that deferred interest on each loan exceeds 5% per annum of the principal balance of the loan.
- 5. CRP may not invest in indebtedness, or junior debt, secured by a mortgage on real property which is subordinate to the lien or other indebtedness, or senior debt, except where the amount of such junior debt, plus the outstanding amount of the senior debt, does not exceed 90% of the appraised value of such property, if after giving effect thereto, the value of all such investments of CRP (as shown on the books of CRP in accordance with generally accepted accounting principles after all reasonable reserves but before provision for depreciation) would not then exceed 25% of CRP's net assets. The value of all investments in junior debt of CRP which does not meet the aforementioned requirements is limited to 10% of CRP's tangible assets (which

is included within the 25% limitation).

- 6. CRP may not engage in any short sale, or borrow, on an unsecured basis, if such borrowing will result in an asset coverage of less than 300%, except that such borrowing limitation shall not apply to a first mortgage trust. "Asset coverage," for the purpose of this paragraph, means the ratio which the value of the total assets of an issuer, less all liabilities and indebtedness except indebtedness for unsecured borrowings, bears to the aggregate amount of all unsecured borrowings of such issuer.
- 7. CRP may not incur any indebtedness which would result in an aggregate amount of leverage in excess of 300% of net assets.
- 8. CRP may not make or invest in any mortgage loans that are subordinate to any mortgage, other indebtedness or equity interest of the Advisor, CRP's directors or CRP's affiliates.
- 9. CRP may not invest in equity securities unless a majority of CRP's directors (including a majority of independent directors) not otherwise interested in such transaction approve the transaction as being fair, competitive, and commercially reasonable and determine that the transaction will not jeopardize CRP's ability to qualify and remain qualified as a REIT. Investments in entities affiliated with the Advisor, a director of CRP, CRP, or CRP's affiliates thereof are subject to the restrictions on joint venture investments. In addition, CRP cannot invest in any security of any entity holding investments or engaging in activities prohibited by CRP's charter.
- 10. CRP may not issue (i) equity securities redeemable solely at the option of the holder (except that stockholders may offer their shares to CRP under certain circumstances); (ii) debt securities unless the historical debt service coverage (in the most recently completed fiscal year), as adjusted for known charges, is sufficient to service that higher level of debt properly; (iii) shares on a deferred payment basis or under similar arrangements; (iv) non-voting or assessable securities; or

- (v) options, warrants, or similar evidences of a right to buy its securities (collectively, "options"); provided however that options may be issued (1) to all of its stockholders ratably, (2) as part of a financing arrangement, or (3) as part of a stock option plan available to CRP's directors, officers, or employees of CRP or the Advisor. Options may not be issued to the Advisor, CRP's directors or any affiliate of CRP thereof except on the same terms as such options are sold to the general public. Options may be issued to persons other than the Advisor, CRP's directors or any affiliate thereof but not at exercise prices less than the fair market value of the underlying securities on the date of grant and not for consideration that, in the judgment of the independent directors, has a market value less than the value of such option on the date of grant. Options issuable to the Advisor, CRP's directors or any affiliate thereof cannot exceed 10% of CRP's outstanding shares on the date of grant.
- 11. A majority of CRP's directors must authorize the consideration to be paid for each property, based on the fair market value of the property. If a majority of the independent directors determine, or if the property is acquired from the Advisor, a director of CRP, or affiliates thereof, such fair market value must be determined by an independent expert selected by the independent directors.
- 12. CRP does not engage in underwriting or the agency distribution of securities issued by others or in trading, as compared to investment activities.
- 13. CRP does not invest in real estate contracts of sale unless such contracts of sale are in recordable form and appropriately recorded in the chain of title.
- 14. CRP does not invest in any foreign currency or bullion or engage in short sales.
- 15. CRP does not issue senior securities except notes to banks and other lenders and preferred shares.

- 16. CRP does not make loans to the Advisor or its affiliates, except (A) mortgage loans subject to the restrictions governing mortgage loans in CRP's charter (including the requirement to obtain an appraisal from an independent expert) or (B) to wholly owned subsidiaries of CRP.
- 17. CRP does not operate so as to be classified as an "investment company" under the Investment Company Act of 1940, as amended.
- 18. CRP does not make any investments that CRP believes will be inconsistent with its objective of qualifying as a REIT.

The foregoing limitations may not be modified or eliminated without the approval of a majority of the shares of CRP's outstanding common stock.

Standards for Investment in Individual Properties

Selection of Tenants, Operators and Property Managers. The selection of tenants, operators and property managers by the Advisor, as approved by CRP's board of directors, will be based on a number of factors which may include: an evaluation of the operations of their properties, the number of properties operated or managed, the relationship of average revenue per available unit (or bed) to the average capital cost per unit (or bed) for each retirement facility operated, the relative competitive position among the same types of properties offering similar services, market penetration, the relative financial success of the operator in the geographic area in which the property is located, overall historical financial performance of the tenant and operator, the management capability of the operator or property manager, and the prior experience of the tenant, operator or property manager in leasing, operating and managing similar properties.

Selection of Properties. In making investments in properties, the Advisor will consider relevant real property and financial factors, including the condition, use, and location of the property, income-producing capacity, and the prospects for long-term appreciation. The proper location, design and amenities are important to the success of a property.

In selecting specific properties, the Advisor, as approved by CRP's board of directors, will apply the following minimum standards:

- 1. Each property will be in what the Advisor believes is a prime location for that type of property.
- 2. Base (or minimum) annual rent will provide a specified minimum return on CRP's cost of purchasing, and if applicable, developing the property, and the lease also may provide for automatic fixed increases in base rent at specified times or increases in the base rent based on increases in consumer price indices over the term of the lease. In addition, certain leases related to retirement facilities may provide for the payment of additional rent based on achieving specified operating performance thresholds.
- 3. The initial lease term typically will be at least ten to 20 years, or in the case of direct financing leases, up to 35 years. With respect to medical office buildings, the initial lease term typically will be five to 15 years.
- 4. In general, CRP will not acquire a property if its board of directors, including a majority of the independent directors, determines that the acquisition would adversely affect CRP in terms of geographic, property type or chain diversification.

Joint Venture Arrangements

Subject to certain restrictions contained in the merger agreement on the conduct of CRP's business prior to the merger, CRP has and may continue to enter into joint ventures, partnerships and limited liability companies as an alternative method to investing directly in real estate. If for any reason CRP is unable to consummate the proposed merger, it may pursue one or more strategic transactions and/or continued implementation of its primary business strategy. There can be no assurance as to the completion, timing or terms of any other strategic transaction.

CRP may enter into a joint venture to purchase and hold for investment a property with various unaffiliated persons or entities or with another program formed by the principals of CRP or the Advisor or their affiliates, if a majority of CRP's directors, including a majority of the independent directors, not otherwise interested in the transaction determine that the investment in the joint venture is fair and reasonable to CRP and on substantially the same terms and conditions as those to be received by the co-venturer or co-venturers. CRP may take more or less than a 50% interest in any joint venture, subject to obtaining the requisite approval of CRP's directors.

Under the terms of each joint venture agreement, it is anticipated that CRP and each joint venture partner would be jointly and severally liable for all debts, obligations, and other liabilities of the joint venture, and CRP and each joint venture partner would have the power to bind each other with any actions they take within the scope of the joint venture's business. In addition, it is expected that the Advisor or its affiliates would be entitled to reimbursement, at cost, for actual expenses incurred by the Advisor or its affiliates on behalf of the joint venture. Events of dissolution will include the bankruptcy, insolvency, or termination of any co-venturer, sale of the property owned by the joint venture, mutual agreement of CRP and its joint venture partner to dissolve the joint venture, and the expiration of the term of the joint venture. The joint venture agreement typically restricts each venturer's ability to sell, transfer, or assign its joint venture interest without first offering it for sale to its co-venturer. In addition, in any joint venture with another program sponsored by the Advisor or its affiliates, where such arrangements are entered into for the purpose of purchasing and holding properties for

investment, in the event that one party desires to sell the property and the other party does not desire to sell, either party will have the right to trigger dissolution of the joint venture by sending a notice to the other party. The notice will establish the price and terms for the sale or purchase of the other party's interest in the joint venture to the other party. The joint venture agreement will grant the receiving party the right to elect either to purchase the other party's interest on the terms set forth in the notice or to sell its own interest on such terms.

The following paragraphs describe the allocations and distributions under the expected terms of the joint venture agreement for any joint venture in which CRP and its co-venturer each have a 50% ownership interest. In any other case, the allocations and distributions are expected to be similar to those described below, except that allocations and distributions which are described below as being made 50% to each co-venturer will instead be made in proportion to each co-venturer's respective ownership interest.

Under the terms of each joint venture agreement, operating profits and losses generally will be allocated 50% to each co-venturer. Profits from the sale or other disposition of joint venture property first will be allocated to any co-venturers with negative capital account balances in proportion to such balances until such capital accounts equal zero, and thereafter 50% to each co-venturer. Similarly, losses from the sale or other disposition of joint venture property first will be allocated to joint venture partners with positive capital account balances in proportion to such balances until such capital accounts equal zero, and thereafter 50% to each co-venturer. Notwithstanding any other provisions in the joint venture agreement, income, gain, loss, and deductions with respect to any contributed property will be shared in a manner which takes into account the variation between the basis of such property and its fair market value at the time of contribution in accordance with section 704(c) of the Code.

Net cash flow from operations of the joint venture generally will be distributed 50% to each joint venture partner. Any liquidation proceeds, after paying joint venture debts and liabilities and funding reserves for contingent liabilities, generally will be distributed first to the joint venture partners with positive capital account balances in proportion to such balances until such balances equal zero, and thereafter 50% to each joint venture partner. Nevertheless, there may be some transactions in which CRP gets a preferred

return so that CRP receives distributions before the co-venturer receives its distributions; and in some of these situations, the co-venturer may then get a larger share of the remaining proceeds. In addition, there may be some transactions in which the co-venturer gets a preferred return so that it receives distributions before CRP receives its distributions; and in some of these situations, CRP may then get a larger share of the remaining proceeds.

In order that the allocations of joint venture income, gain, loss, and deduction provided in joint venture agreements may be respected for federal income tax purposes, it is expected that any joint venture agreement either (i) (a) will contain a "qualified income offset" provision, (b) will prohibit allocations of loss or deductions to the extent such allocation would cause or increase an "Adjusted Capital Account Deficit," and (c) will require (1) that capital accounts be maintained for each joint venture partner in a manner which complies with Treasury Regulation Section 1.704-1(b)(2)(iv) and (2) that distributions of proceeds from the liquidation of a partner's interest in the joint venture (whether or not in connection with the liquidation of the joint venture) be made in accordance with the partner's positive capital account balance, or (ii) otherwise will provide for allocations of income, gain, deduction and loss which are deemed to have economic effect under the provisions of Treasury Regulation Section 1.704-1(b)(2)(ii)(i).

Prior to entering into any joint venture arrangement with any unaffiliated co-venturer (or the principals of any unaffiliated co-venturer), CRP will confirm that such person or entity has demonstrated to CRP's satisfaction that requisite financial qualifications are met.

Strategic Alliances

The Cirrus Group, LLC. In August 2005, CRP entered into an agreement with The Cirrus Group, LLC, or Cirrus, a development and property management company, to acquire, at CRP's election, medical facilities, some of which have yet to be developed. The acquisitions contemplated under the Cirrus agreement are expected to occur over a five-year term, subject to certain conditions, or until \$1.0 billion is invested in medical facilities, including specialty hospitals. CRP will have minority interest partners in connection with the ownership of each of these properties, including Cirrus principals, physicians and other investors associated with Cirrus principals. As of March 31, 2006, CRP had acquired a majority equity interest in two medical facilities for \$52.6 million under the Cirrus agreement, for which Cirrus and its affiliates made \$0.9 million in minority interest contributions.

At March 31, 2006, Cirrus managed 23 of CRP's medical facilities, including two properties that CRP acquired from third parties.

In 2005, CRP entered into an agreement to provide a Cirrus affiliate with an interest only, five-year senior secured term loan under which up to \$85.0 million (plus capitalized interest) may be borrowed to finance the acquisition, development, syndication and operation of new and existing surgical partnerships. At March 31, 2006, the balance outstanding under the senior secured term loan was \$34.0 million. In connection with the senior secured term loan, CRP received stock warrants which are exercisable into a 10% to 15% ownership interest of the borrower. The stock warrants are exercisable at the earlier of an event of default or the full repayment of the senior secured term loan and expire in September 2015.

The DASCO Companies, LLC. CRP owns a 55% controlling interest in The DASCO Companies, LLC, or DASCO, a development and property management company. CRP's relationship with DASCO has provided and may, in the event the merger is not consummated, continue to provide opportunities for CRP to participate in new medical facility development and acquisition opportunities as well as medical facilities management. DASCO may also provide development and property management services to third parties. At March 31, 2006, DASCO managed 54 of CRP's medical facilities, including two walk-in clinics and was developing five of CRP's medical facilities.

Other Permitted Investments

In the event the merger is not consummated, CRP may also provide (i) mortgage financing to operators to enable them to acquire properties that would secure the loan, (ii) furniture, fixtures and equipment financing, (iii) other loans to entities in which CRP holds an interest, and (iv) CRP may invest up to a maximum of 5% of its total assets in equity interests in businesses, including those that provide services to or are otherwise ancillary to the retirement and healthcare industries.

If CRP provides such financing, CRP expects that the interest rates and terms of the mortgage loans CRP provides will be similar to those of CRP's leases. However, because CRP prefers to focus on investing in properties which have the potential to appreciate, CRP currently expects to provide mortgage loans in the aggregate principal amount of no more than 5% to 10% of its total assets. CRP had \$4.8 million invested in mortgage loans at March 31, 2006.

To a lesser extent, CRP also may provide secured equipment leases to operators, pursuant to which CRP will finance the equipment through loans or direct financing leases. It is expected that the leases or loans will have a term of no more than seven years, will be secured by the personal property and include an option for the lessee to acquire the subject equipment at the end of the term. The aggregate outstanding principal amount of secured equipment leases is not expected to exceed 10% of CRP's total assets. CRP had no secured equipment leases at March 31, 2006.

In the event the merger is not consummated, CRP may make other loans to operators or developers of senior housing or other healthcare-related facilities collateralized by real estate owned by the borrower.

During the three months ending March 31, 2006, CRP advanced \$18.0 million under the senior secured term loan. The balance outstanding under the senior secured term loan was \$34.0 million at March 31, 2006.

Advisory Services

Pursuant to an advisory agreement, the Advisor provides management services relating to CRP's administration, the properties, the mortgage loans, the secured equipment lease program and other loans. Under this agreement, the Advisor is responsible for assisting CRP in negotiating leases, other permitted investments, lines of credit and permanent financing; collecting rental, mortgage loan, secured equipment lease and other loan payments; inspecting the properties and the tenants' books and records; and responding to tenants' inquiries and notices. The Advisor is also responsible for providing information to CRP about the status of the leases, properties, other permitted investments, any lines of credit and any permanent financing. In exchange for these services, the Advisor is entitled to receive certain fees from CRP. For supervision of the properties and the mortgage loans, the Advisor receives an asset management fee, which is payable monthly, in an amount equal to 0.05% of the total amount invested in the properties, exclusive of acquisition fees and acquisition expenses, plus 0.05% of the outstanding principal amount of any mortgage loans, as of the end of the preceding month. For negotiating secured equipment leases and supervising the secured equipment lease program, the Advisor will receive, upon entering into each lease, a secured equipment lease servicing fee, payable out of the proceeds of borrowings, equal to 2% of the purchase price of the equipment subject to each secured equipment lease. For identifying the properties, structuring the terms of the acquisition and leases of the properties and structuring the terms of the mortgage loans, the Advisor receives an acquisition fee on the gross proceeds from the offerings and loan proceeds from permanent financing, excluding that portion of the permanent financing used to finance secured equipment leases, equal to 3% for the period from May 3, 2005 through December 31, 2005, 4% for the period from May 14, 2004 to May 2, 2005 and 4.5% with respect to the offerings prior to the 2004 offering. In addition, if there is a listing, the Advisor will receive an acquisition fee of 3% of

amounts outstanding on the line of credit, if any, at the time of listing.

In accordance with the advisory agreement, the Advisor is required to reimburse CRP the amount by which the total operating expenses incurred by CRP in any four consecutive fiscal quarters exceed the greater of 2% of average invested assets or 25% of net income.

On May 2, 2005, CRP entered into a renewal agreement with the Advisor with respect to the advisory agreement, pursuant to which the advisory agreement was renewed for an additional one-year term commencing on May 3, 2005, and ending at 12:00 a.m. on May 3, 2006. On July 13, 2005, CRP and the Advisor amended the renewal agreement to reduce the percentage rate of total proceeds to be used in determining acquisition fees payable to the Advisor under the advisory agreement from 4% to 3%. This reduction is deemed to be effective as of May 3, 2005.

On May 1, 2006, CRP entered into another renewal agreement with the Advisor with respect to the advisory agreement, pursuant to which the advisory agreement was renewed for an additional one-year term commencing on May 3, 2006, and ending on May 3, 2007. Such renewal was approved by CRP's board of directors (after recommendation of the independent directors) at a meeting on April 28, 2006.

Borrowings

CRP has historically and, unless the merger is consummated, may continue to borrow money to acquire properties, make mortgage loans, other loans and pay certain fees, and CRP encumbers properties in connection with these borrowings. CRP may also borrow money to enter into secured equipment leases. CRP has a \$320.0 million revolving line of credit that may be amended to increase the revolving line of credit to \$400.0 million. The amount available for use under the revolving line of credit is subject to certain limitations based on the pledged collateral. As of March 31, 2006, the revolving line of credit was collateralized by 36 properties with a carrying value of \$389.4 million that, in the aggregate, allowed CRP to draw up to \$283.0 million. Per the terms of the merger agreement, CRP is only permitted to draw up to an additional \$25.0 million under the revolving line of credit without HCP's prior written consent.

Competition

CRP historically has competed with other REITs, real estate partnerships, healthcare providers and other investors, including, but not limited to, banks and insurance companies, many of which may have greater financial resources than CRP's, in the acquisition, leasing and financing of senior housing and medical facilities. Further, non-profit entities are particularly suited to make investments in healthcare facilities because of their ability to finance acquisitions through the issuance of tax-exempt bonds, providing non-profit entities with a relatively lower cost of capital as compared to for-profit purchasers. In addition, in certain states, facilities owned by non-profit entities are exempt from taxes on real property. Competition to acquire senior housing and medical facilities has continued to increase due, in part, to the continued interest in the sector from private equity sources, including foreign investors. In some cases, this competition has caused acquisition prices to increase, making it more challenging for CRP to be competitive in the acquisition of new investments.

During 2005, CRP continued to focus its investments in the acquisition of existing senior housing and medical facilities, as well as in the development of such new properties through strategic alliances with new and existing business partners. The development of new properties allowed CRP to avoid the pricing pressures in the open market and to develop facilities that met CRP's investment requirements. However, although successful in 2005, there can be no

assurance that this investment strategy will be followed in upcoming periods or, if followed, that it will generate the same results as in 2005. Further, there can be no assurance that CRP will be able to secure business partners to develop the properties.

Employees

CRP has no employees, other than CRP's executive officers who are not compensated by CRP. CRP has retained the Advisor to provide management, acquisition, advisory and certain administrative services and have retained certain other affiliates of the Advisor to provide additional administrative services.

CRP's Properties

Generally, properties acquired by CRP consist of both land and building, although in some cases CRP acquires only the land underlying the building with the building owned by the tenant or a third party, and also acquires the building only with the land owned by a third party. In general, the properties will be freestanding and surrounded by paved parking areas and landscaping. Although, buildings may be suitable for conversion to various uses through modifications, some properties may not be economically convertible to other uses.

Either before or after construction or renovation, the properties acquired by CRP are one of a tenant's approved designs. Prior to purchase of all properties, other than those purchased prior to completion of construction, CRP receives a copy of the certificate of occupancy issued by the local

building inspector or other governmental authority and all other governmental certificates or permits which permit the use of the property as a retirement facility, and shall receive a certificate from the tenant to the effect that (i) the property is operational and in compliance with all required governmental permits and certificates, and (ii) the property is in compliance with all of the tenant's requirements, including, but not limited to, building plans and specifications approved by the tenant. CRP also receives a certificate of occupancy and all other required governmental permits or certificates for each property for which construction has not been completed at the time of purchase, prior to CRP's payment of the final installment of the purchase price for the property.

Except in the case of gross leases related to certain medical office buildings, a tenant generally is required by the lease agreement to make such capital expenditures as may be reasonably necessary to refurbish buildings, premises, signs, and equipment and maintain the leasehold in a manner that allows operation for its intended purpose. These capital expenditures generally will be paid by the tenant during the term of the lease up to a certain capped amount and CRP will be responsible for the balance.

Characteristics of CRP's Senior Housing Leases. Senior housing properties are leased on a long-term (generally 15 years), triple-net basis, whereby the tenants are generally responsible for all operating expenses relating to the property, including property taxes, maintenance, repairs, utilities and insurance as well as capital expenditures that may be reasonably necessary to maintain the leasehold in a manner that allows operation for its intended purpose. Substantially all of the leases provide options that allow the tenants to renew the leases for 5 to 20 successive years subject to the same terms and conditions as the initial leases. These leases provide for minimum annual base rent payments, generally payable monthly in arrears, that increase at predetermined intervals (typically on an annual basis) during the terms of the leases. In addition to minimum annual base rent, many tenants are subject to contingent rent if the properties achieve specified operating performance thresholds. The amount of contingent rent payable is based on factors such as percentage of gross revenues, occupancy rates of the properties or a percentage of CRP's investment in the property. The majority of the leases also provide for the tenant to fund, in addition to its lease payments, a furniture, fixture and equipment, or FF&E, reserve fund. In such cases, the tenant deposits funds into the FF&E reserve account and periodically uses these funds to cover the cost of

the replacement, renewal and additions to FF&E. CRP may be responsible for capital expenditures or repairs in excess of the amounts in the reserve fund, and the tenant generally is responsible for replenishing the reserve fund and for paying a specified return on the amount of capital expenditures or repairs paid for by CRP in excess of amounts in the reserve fund.

At December 31, 2005, 33 of CRP's senior housing properties were accounted for as direct financing leases with terms that range from 10 to 35 years (expiring between 2013 and 2038). Certain of these direct financing leases contain provisions that allow the lessee to elect to purchase the property during or at the end of the lease term for CRP's initial investment amount. Certain of the leases also permit CRP to require the tenants to purchase the properties at the end of the lease terms for CRP's initial investment amount.

The senior housing lessees' ability to satisfy the lease obligations depends primarily on the properties' operating results. CRP selects properties for investment based on a credit underwriting process designed to identify those properties that management believes will be able to fund such lease obligations. To mitigate credit risk, certain leases are combined into portfolios that contain cross-default terms, meaning that if a tenant of any of the properties in a portfolio defaults on its obligations under its lease, CRP may pursue remedies under the lease with respect to any of the tenant's properties in the portfolio. In addition, certain portfolios contain terms whereby the net operating profits of the properties are combined for the purpose of funding rental payments due under each lease. For certain properties, CRP requires security deposits, tenant or operator guarantees or additional types of income support. Guarantees or other forms of credit support may be necessary if a senior housing facility is

still in the process of achieving a stable occupancy rate, in which case the property would not be able to generate minimum rent until reaching occupancy stabilization. In order to determine the amount of the guarantee that would be needed to fund minimum rent, CRP estimates future cash flows available to the tenant to pay minimum rent based on projected occupancies and an analysis of the surrounding real estate market, including demographic information and industry standards, to predict operating expenses. CRP's estimates are based on assumptions and there can be no assurances as to the actual amounts that will need to be paid under the guarantees.

Characteristics of CRP's Medical Facilities
Leases. CRP owns both single-tenant and
multi-tenant medical office buildings, specialty
hospitals and walk-in clinics that are leased on
either a triple-net or gross basis, primarily to
tenants in the healthcare industry. The leases have
initial terms of 5 to 15 years, provide for minimum
rent and are generally subject to renewal options.
Substantially all leases require minimum annual
rents to increase at predetermined intervals during
the lease terms. Under CRP's gross leases, tenants
generally will be responsible for a certain capped
amount of repairs, maintenance, property taxes,
utilities and insurance and CRP will be responsible
for the balance.

Major Tenants and Operators. As of December 31, 2005, CRP leased its senior housing properties to 22 tenants. Two tenants affiliated with Horizon Bay Management, LLC, or Horizon Bay, contributed 21% of CRP's total revenues for the year ended December 31, 2005. As of December 31, 2005, 10 of CRP's tenants, each of which is a subsidiary or affiliate of Harbor Retirement Associates, LLC, or HRA, contributed 22% of total revenues. No other senior housing tenant contributed more than 10% of total revenues. CRP's other tenants include other affiliates of Horizon Bay and subsidiaries or affiliates of: American Retirement Corporation, or ARC; Aureus Group, LLC, or Aureus; Eby Realty Group, LLC, or Eby; Encore Senior Living, LLC, or Encore; Erickson Retirement Communities, LLC; Greenwalt Corporation, or Greenwalt; Prime Care properties, LLC; Summit Companies, Incorporated; Solomon Senior Living Holdings, LLC; and Sunrise Senior Living Services, Inc., or Sunrise. Several of CRP's senior housing tenants, including the HRA tenants, are thinly capitalized corporations that rely on the cash flow generated from the senior housing facilities to fund rent obligations under their leases. CRP's medical facilities are leased to more than 700 tenants. Tenancy in the medical facilities is generally a mix

of physician practices and CRP also leases space to several large hospital systems and other healthcare providers. See "Management's Discussion and Analysis of Financial Condition and Results of Operations as of December 31, 2005 Accounts and Other Receivables" below for a description of past due rent receivables and the reserve for doubtful accounts attributable to certain of CRP's major tenants and operators.

Although CRP acquires properties located in various states and regions and screens its tenants in order to reduce risks of default, failure of Horizon Bay, the HRA tenants or CRP's other tenants, their guarantors or the Sunrise or Horizon Bay brands would significantly impact CRP's results of operations. It is expected that the percentage of total rental income contributed by Horizon Bay and the HRA tenants will decrease as additional properties are acquired and leased to diversified tenants.

On September 1, 2004, a company which is owned by CRP's chairman of the board sold its 30% voting membership interest in a limited liability company which is affiliated with the HRA tenants to the remaining members of the limited liability company. The HRA tenants contributed 30% and 35% of CRP's total revenues for the years ended December 31, 2004 and 2003, respectively.

The following table summarizes information about CRP's operator and manager concentration, excluding five properties that were held for sale as of December 31, 2005 (dollars in thousands):

Operator or Manager	Number of Facilities	Total Investment	Annualized Revenue(1)	
Senior				
Housing:				
Sunrise Senior Living Services, Inc.	107	\$ 1,480,990	\$ 152,531	42%
Horizon Bay Management,	27	768,799	82,925	22%
LLC Encore Senior Living, LLC	17	143,177	15,157	4%
American Retirement Corporation	8	154,623	18,103	5%
Harbor Assisted Living, LLC	7	65,811	6,551	2%
Erickson Retirement Communities, LLC(2)	6	131,880	19,351	5%
Eby Realty Group, LLC	6	33,994	4,033	1%
CateredLife Communities, Inc.	5	32,846	4,155	1%
Other	1	46,207		%
	184	2,858,327	302,806	82%
Medical Facilities:				
The DASCO Company	53	377,452	38,436	10%
The Cirrus Group, LLC	10	117,488	10,504	3%
Four third-party managers	10	149,844	17,789	5%
	73	644,784	66,729	18%
	257	\$ 3,503,111	\$ 369,535	100%

(1)
In 2005, CRP added properties to its investment portfolio throughout the year.
In order to evaluate the ongoing effect of operator or manager concentrations CRP

has calculated an annualized revenue amount based on each property's actual rental income from operating leases or earned income from direct financing leases during the year ended December 31, 2005.

(2) Land only leases.

As of March 31, 2006, CRP had invested \$3.7 billion in 274 properties located in 33 states. Generally, CRP's properties conform to the following specifications of size and type of land and buildings:

Independent Living Facilities. Independent living facilities offer a lifestyle choice, including residential accommodations with access to services, such as housekeeping, transportation, dining and social activities, for those who wish to maintain their lifestyles independently. The independent living facilities are primarily apartment buildings which contain a significant amount of common space to accommodate dining, recreation, activities and other support services for senior citizens. These properties range in size from 100 to 500 units with an average size of approximately 225 units. Units include studios and one and two bedroom units ranging in size from 450 square feet to over 1,500 square feet. Residents generally pay the operator of the facilities a base rent for their housing, which may include a meal program. In addition, a menu of other services is provided at an additional charge. The cost of independent living facilities generally ranges from \$10 million to \$60 million.

Assisted Living Facilities. Assisted living facilities provide a combination of housing, supportive services, personalized assistance and healthcare to their residents in a manner which is designed to respond to individual needs. These facilities generally offer a lower-cost alternative to skilled nursing facilities for those who do not require intensive nursing care. Assisted living facilities may include units for residents with Alzheimer's and related memory disorders. Current industry practice generally is to

build freestanding assisted living facilities with an average of between 40 and 150 units, depending on such factors as market forces, site constraints and program orientation. Current economics place the size of the private living space of a unit in the range of 300 square feet for an efficiency unit to 750 square feet for a large one bedroom unit. Units are typically private, allowing residents the same general level of control over their units as residents of a rental apartment would typically have.

Common areas of the most recently developed assisted living facilities may total as much as 30% to 40% of the square footage of a facility. The cost of assisted living facilities generally ranges from \$5 million to \$25 million.

Skilled Nursing Facilities. In addition to housing, meals, transportation and housekeeping, skilled nursing facilities provide comprehensive nursing and long-term care to their residents. Skilled nursing facilities may be freestanding or attached to a larger facility. The facilities are designed to meet institutional standards for safety. The rooms in skilled nursing facilities are equipped with patient monitoring devices and emergency call systems. Oxygen systems may also be present. Both multiple floor and single floor designs are common. Individual rooms in skilled nursing facilities may be as small as 100 square feet, with common areas varying greatly in size. The cost of skilled nursing facilities generally ranges from \$5 million to \$10 million.

Continuing Care Retirement Communities. Independent living facilities sometimes have assisted living and/or skilled nursing facilities attached or adjacent to their locations. When this occurs, the projects are often referred to as continuing care retirement communities, life care communities or CCRCs. The intent of CCRCs or life care communities is to provide a continuum of care to the residents. As residents age and their healthcare needs increase, they can receive the care they need without having to move away from the "community" which has become their home. CCRCs typically operate on a fee-for-service basis and the units are rented on a monthly basis to residents, while life care communities generally charge an entrance fee that may be partially refundable, plus a monthly maintenance fee. CCRCs and life care communities are the most expensive type of senior housing with prices for each facility generally ranging from \$40 million to over \$200 million.

Medical Office Buildings. Medical office buildings, including physicians' offices, special purpose facilities such as laboratories, diagnostic, cancer treatment and outpatient centers, and

walk-in clinics are conventional office buildings with additional plumbing, mechanical and electrical service amenities which facilitate physicians and medical delivery companies in the practice of medicine, laboratory research and delivery of healthcare services. These facilities can range in size from 3,000 square feet (walk-in clinic) to up to 150,000 square feet (medical office building) with costs generally ranging from \$1 million to \$10 million. It is common for medical office buildings to be located in close proximity to hospitals where physicians have practice privileges.

Specialty Hospitals. Specialty hospitals are facilities that provide specialized procedures, usually cardiac, orthopedic or surgical, on an inpatient or outpatient basis. Specialty hospitals are licensed as acute care hospitals, but they are typically smaller and more specialized. They usually do not have emergency rooms and can range from 20,000 to over 100,000 square feet, depending on the number of beds and operating rooms.

Generally, properties acquired consist of land, building and equipment, however, in certain cases CRP has only acquired the land underlying the building with the building owned by the tenant or a third party, and in other cases CRP has only acquired the building with the land owned by a third party. CRP owns fee title to all properties, except for properties which are owned by certain partnerships and joint ventures in which case the partnerships or joint ventures have fee title ownership and properties which are subject to ground leases. In general, the properties are freestanding and surrounded by paved parking areas and landscaping. Although buildings may be suitable for conversion to various uses through modifications, some properties may not be economically convertible to other uses.

Independent Living Facilities:

The following table summarizes the facility type, location, number of units or square footage, CRP's investment amount at December 31, 2005, and the annualized rental income and rental income for the year ended December 31, 2005, excluding five properties then held for sale. Senior housing facilities are apartment-like facilities and are therefore stated in units. Medical facilities are measured in square feet (dollars in thousands):

						Rental I	ncome
Facility Type and Location	Number of Facilities	Number of Units	Square Feet	In	Total ivestmentAi	Donnualized(1	For Year Ended ecember 3) 2005
Assisted							
Living							
Facilities:	2	100		ф	16.500	h 1.050 i	1.050
Alabama	2	198		\$	16,529		
Arizona	3	180			24,165	2,559	1,919
California	16	1,860			197,193	19,606	19,402
Colorado	4	635			70,639	8,409	8,409
Connecticut	2	227			23,019	2,693	2,693
Florida	18	1,538			133,352	13,954	12,795
Georgia	9	736			71,187	8,150	8,150
Illinois Indiana	9	666 198			87,038	10,305	10,305
	3 2	-, -			17,490	2,164	2,164
Iowa		80			11,742	1,402	1,402
Kansas	1 2	152			20,476	2,359	2,359
Kentucky	_	188			15,005	1,860	1,860
Maryland	8	776			116,947	13,553	13,553
Massachusetts	4	396			61,397	6,815	6,815
Michigan	2	330			44,085	5,055	5,055
Missouri		152			37,428	3,246	3,246
Nebraska	1	150			12,723	1,462	1,462
New Jersey	7	756			115,450	13,122	13,122
New York	2	202			49,451	5,354	5,354
North		504			50.005	5.047	5.045
Carolina	6	584			50,005	5,847	5,847
Ohio	5	419			37,353	4,345	4,345
Oklahoma	2	212			8,235	999	999
Oregon	1	96			11,495	1,225	919
Rhode Island South	1	128			19,202	2,105	2,105
	=	426			21 062	2.011	2 01 1
Carolina Tennessee	5 2	205			31,863	3,911	3,911
Tennessee	6	486			16,054 64,437	1,999	1,999
Utah	1	486 158				8,143	8,143
	5				14,633	1,929	1,929
Virginia Washington	5	382 367			44,983 46,551	4,840	4,840
Washington		30/			40,331	4,522	4,522
	138	12,883		\$ 1	1,470,127	\$ 163,883	\$ 161,574

				Rental Inc	come
Alabama	2	540	\$ 82,285 \$	9,295 \$	9,295
Arizona	1	211	46,562	5,125	5,125
Arkansas	1	163	10,823	1,365	1,365
California	3	558	85,197	9,234	8,392
Florida	5	1,362	251,139	22,478	22,498
Illinois	5	1,016	177,163	19,296	18,316
Kentucky	1	120	10,496	1,322	1,322

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Oregon	1	265			18,697		1,994		1,495
Rhode Island	6	725			104,066		11,360		11,360
Texas	9	2,635			190,564		20,103		20,103
Utah	1	75			4,969		530		397
Virginia	2	275			42,537		4,423		4,423
v ii giiii u					12,337		1,123		1,123
	_								
	37	7,945		\$	1,024,498	\$	106,525	\$	104,091
	_			_		_		_	
CCDC									
CCRCs:		5.40		Φ.	100.045	Φ.	5 (02	ф	5 600
Pennsylvania	1	542		\$	109,845	\$	7,682	\$	7,682
Virginia	1	487			84,308		5,364		5,364
	_			_		_		_	
	2	1,029		\$	194,153	\$	13,046	\$	13,046
	_	1,027		Ψ	174,133	Ψ	13,040	Ψ	13,040
	_					_			
Land Only									
Leases:									
Illinois	1			\$	28,493	\$	4,103	\$	4,103
Massachusetts	2			Ė	40,266	Ė	6,007		6,007
Michigan	1				17,909		2,568		2,568
Pennsylvania	1				21,088		3,173		
-									3,173
Texas	1				24,124		3,501		3,501
	_			_		_		_	
	6			\$	131,880	\$	19,352	\$	19,352
	_				,	_	· ·		· ·
~ .									
Senior									
Housing									
Property									
Under									
Construction:									
New Jersey	1			\$	37,669	\$		\$	1,023
				-					
	_					_			
	_				•	_		_	
Medical	_				<u> </u>				
	_				·			_	
Medical	_				·				
Medical Office	3		286,686	\$	37,894	\$	4,737	\$	4,737
Medical Office Buildings: Arizona	3 4			\$	37,894	\$		\$	4,737 5,982
Medical Office Buildings: Arizona California	4		272,386	\$	37,894 61,415	\$	5,982	\$	5,982
Medical Office Buildings: Arizona California Colorado	4 4		272,386 215,644	\$	37,894 61,415 41,800	\$	5,982 4,709	\$	5,982 4,709
Medical Office Buildings: Arizona California Colorado Florida	4 4 11		272,386 215,644 550,357	\$	37,894 61,415 41,800 85,016	\$	5,982 4,709 9,808	\$	5,982 4,709 9,808
Medical Office Buildings: Arizona California Colorado Florida Georgia	4 4 11 1		272,386 215,644 550,357 14,680	\$	37,894 61,415 41,800 85,016 2,605	\$	5,982 4,709 9,808 286	\$	5,982 4,709 9,808 286
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois	4 4 11 1 6		272,386 215,644 550,357 14,680 274,391	\$	37,894 61,415 41,800 85,016 2,605 47,028	\$	5,982 4,709 9,808 286 5,749	\$	5,982 4,709 9,808 286 5,548
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana	4 11 1 6		272,386 215,644 550,357 14,680 274,391 79,762	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700	\$	5,982 4,709 9,808 286 5,749 251	\$	5,982 4,709 9,808 286 5,548 209
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky	4 11 1 6 1 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431	\$	5,982 4,709 9,808 286 5,749 251 1,432	\$	5,982 4,709 9,808 286 5,548 209 1,432
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland	4 11 1 6		272,386 215,644 550,357 14,680 274,391 79,762	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700	\$	5,982 4,709 9,808 286 5,749 251	\$	5,982 4,709 9,808 286 5,548 209
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky	4 11 1 6 1 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431	\$	5,982 4,709 9,808 286 5,749 251 1,432	\$	5,982 4,709 9,808 286 5,548 209 1,432
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland	4 11 1 6 1 2 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi	4 11 1 6 1 2 2 3		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North	4 4 11 1 6 1 2 2 3 1		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina	4 4 11 1 6 1 2 2 3 1		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma	4 4 11 6 1 2 2 3 1		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee	4 11 1 6 1 2 2 3 1 4 3 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas	4 4 11 1 6 1 2 2 3 1 4 3 2 16		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee	4 11 1 6 1 2 2 3 1 4 3 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas	4 4 11 1 6 1 2 2 3 1 4 3 2 16		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas	4 4 11 1 6 1 2 2 3 1 4 3 2 16		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355		37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523		5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691		5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792		37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190		5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978		5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas Virginia	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792		37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190		5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978		5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas Virginia	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792		37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190		5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978		5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas Virginia Specialty Hospital:	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978 63,938	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas Virginia	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978
Medical Office Buildings: Arizona California Colorado Florida Georgia Illinois Indiana Kentucky Maryland Mississippi Nebraska North Carolina Oklahoma Tennessee Texas Virginia Specialty Hospital:	4 4 11 1 6 1 2 2 3 1 1 4 3 2 16 2		272,386 215,644 550,357 14,680 274,391 79,762 115,975 78,940 132,204 97,262 104,889 128,910 108,077 885,355 147,792	\$	37,894 61,415 41,800 85,016 2,605 47,028 6,700 14,431 14,161 21,484 11,824 20,692 20,667 16,096 180,523 34,190	\$	5,982 4,709 9,808 286 5,749 251 1,432 1,433 1,621 481 1,531 1,826 2,423 17,691 3,978 63,938	\$	5,982 4,709 9,808 286 5,548 209 1,432 1,433 1,621 481 1,531 1,111 1,211 14,063 3,978

Walk-in Clinics:								
Arkansas	2		42,781	\$	4,479	\$ 502	\$	251
	_			_				_
Medical Facilities Under								
Constructi				\$		\$	¢	
Colorado Maryland	1			Þ	995	3	\$	
Texas	2				538			
Virginia	1				4,078			93
	_							
	5			\$	5,611	\$	\$	93
	257	21,857	3,593,675	\$ 3,	503,111	\$ 369,535	\$ 358	,142
	_							_

In 2005, CRP added properties to its investment portfolio throughout the year. In order to evaluate the ongoing effect of operator or manager concentrations CRP has calculated an annualized revenue amount based on each property's actual rental income from operating leases or earned income from direct financing leases during the year ended December 31, 2005.

CRP's Legal Proceedings

From time to time, CRP is exposed to litigation arising from an unidentified pre-acquisition contingency or from the operation of its business. Although currently exposed to certain such litigation, CRP does not believe that resolution of these matters will have a material adverse effect on its financial condition or results of operations.

CRP's Management's Discussion and Analysis of Financial Condition and Results of Operations as of March 31, 2006

Business Overview as of March 31, 2006

During the first quarter of 2006, CRP acquired 14 properties, 13 from its relationship with the Cirrus Group, LLC, or Cirrus, worked with the operators of its properties to enhance the operations of the properties, disposed of two properties that were held for sale at December 31, 2005, and explored strategic alternatives to raise capital. As of March 31, 2006, CRP had also entered into discussions with the lenders of CRP's \$320.0 million two-year senior secured revolving

line of credit to increase the amount available under the facility.

As of March 31, 2006, CRP held real estate assets located in 33 states consisting of (dollars in thousands):

	Number of Properties	Investment at March 31, 2006
Senior		
housing		
facilities:		
Operating	184	\$ 2,854,697
Under	1	1,045
construction		
Medical		
facilities:		
Operating	81	826,753
Under	5	11,757
construction		
	271	\$ 3,694,252
Real estate held for sale	3	\$ 6,921

Liquidity and Capital Resources as of March 31, 2006

CRP has primarily invested in or developed properties and CRP also provides (i) mortgage financing to operators to enable them to acquire properties that would secure the loan, (ii) furniture, fixtures and equipment financing, (iii) other loans to entities in which CRP holds an interest, and (iv) other permitted investments. CRP believes that over the short term, which is less than 12 months, net funds from operations, borrowings under new permanent or construction financing, the placement of permanent debt to replace maturing construction loans, advances under CRP's revolving line of credit and cash on hand at March 31, 2006, will be sufficient to meet CRP's forecasted capital requirements for property investments, senior secured term loan funding (an agreement to provide a Cirrus affiliate with an interest-only, five-year senior secured term loan under which up to \$85.0 million (plus capitalized interest) may be borrowed to finance the acquisition, development, syndication and operation of new and existing surgical partnerships), capital expenditures, the re-tenanting of CRP's medical facilities (medical office buildings, specialty and walk-in clinics, free standing ambulatory surgery centers, specialty or general hospitals and other types of healthcare-related facilities are collectively "medical facilities") and the scheduled maturities of permanent financings. In the event the merger is

not consummated, CRP also expects to continue to be able to pay distributions at current levels, or at a minimum, to maintain CRP's REIT qualification.

Unless the merger is consummated, over the long term, which is 12 months or more, CRP may raise capital by encumbering properties, entering into joint venture agreements with respect to CRP's investments in new or existing properties, issuing preferred stock, selling existing properties, or CRP may stop investing in properties. CRP is subject to certain limitations on CRP's conduct of business pursuant to the terms of CRP's merger agreement with HCP, as described below. CRP has relied on the sale of CRP's common stock and borrowings under permanent or construction financing to fund CRP's property investments. CRP's fifth public offering closed on March 26, 2006, and CRP does not presently intend to commence a new offering.

Operating cash flow for the year ended December 31, 2005, was \$188.3 million and is expected to increase with a full year of operations for the 40 properties that CRP acquired during 2005, the 14 properties acquired during the three months ended March 31, 2006, and in the event the merger is not consummated, those forecasted to be acquired during the remainder of 2006. As of March 31, 2006, CRP has adequate construction funding to complete and open the properties under construction, which CRP expects will generate additional operating cash flow.

The merger agreement provides, among other things, that CRP cannot incur additional indebtedness other than an additional \$25.0 million in draw downs under CRP's revolving line of credit or issue equity or convertible securities without the prior written consent of HCP. In addition, the

merger agreement limits CRP's ability to invest in and encumber assets, make loans and dispose of assets without the prior written consent of HCP. The restriction on CRP's ability to utilize sources of liquidity could have a material adverse affect on CRP's liquidity position. However, although there can be no assurances that the merger will be consummated, CRP expects to have sufficient liquidity from cash on hand, cash from operations and availability of funds from the revolving line of credit to fund all liquidity needs through the expected closing of the merger in the third quarter of 2006.

Cash from Operating Activities as of March 31, 2006

Net cash provided by operating activities was \$55.5 million and \$52.8 million for the three months ended March 31, 2006 and 2005, respectively. The increase was due, in part, to an increase in operating income and collection of past due accounts receivable, offset by higher interest costs due to an increase in the average amount of debt outstanding, reductions in accounts payable and inflows from security deposits from property acquisitions. Net cash from operating activities included draws on tenant and operator rent guarantees of \$1.9 million and \$0.8 million, respectively.

Accounts Receivable. The ability to collect rents from CRP's tenants when contractually due is critical to CRP's ability to meet short- and long-term cash obligations. CRP closely monitors rent collections and works closely with the tenants and operators of properties that are unable to pay full rent.

Accounts and other receivables, net decreased \$3.7 million to \$19.8 million at March 31, 2006, from \$23.5 million at December 31, 2005, consisting of a \$2.1 million decrease in accounts and other receivables and a \$1.6 million increase in the reserve for doubtful accounts. The \$2.1 million decrease in accounts and other receivables was due to (i) a decrease in other receivables of \$1.3 million and (ii) a \$1.6 million reduction due to the reclassification of rental revenues receivable from current receivables to deferred receivables, partially offset by (iii) an increase in rental revenues receivable of \$0.9 million. The \$1.6 million increase in the reserve for doubtful accounts included \$1.2 million for a portfolio of 19 properties whose operator guarantee had expired in December 31, 2005. Past due rents receivable were \$14.4 million and \$14.8 million at March 31, 2006, and December 31, 2005, respectively. At March 31, 2006, \$7.3 million of the \$8.8 million allowance for doubtful accounts was attributable to HRA tenants (HRA tenants consist of 10 of CRP's

tenants, each of which is thinly capitalized and is a subsidiary or affiliate of Harbor Retirement Associates, LLC).

In January 2006, CRP amended and restated the leases of a 14-property direct finance lease portfolio by extending the termination date by 5 years; all other lease terms were unchanged. The effective return on the leases increased to 12.7% from 11.9%.

Based on CRP's analysis of estimated future cash flows to be generated by certain properties for which CRP currently has reserves, CRP expects that certain delinquent amounts will be collected in 2006.

Cash from Investing Activities as of March 31, 2006

Net cash used in investing activities was \$218.2 million and \$201.5 million for the three months ended March 31, 2006 and 2005, respectively. The increase was due primarily to advances under the senior secured term loan offset by a reduction in the number of properties acquired and the payment of acquisition fees related to the close of CRP's equity offering.

Property Acquisitions. In January 2006, CRP acquired majority equity interests in seven medical facilities for \$84.5 million which CRP funded, in part, with proceeds from a new \$56.3 million, ten-year mortgage loan.

In February 2006, CRP acquired a senior housing property (types of CRP's properties which include independent living, assisted living and skilled nursing facilities, continuing care retirement

communities and life care communities are collectively "senior housing") that is being developed. The project is expected to be completed in the fourth quarter of 2006 with an estimated cost of \$5.7 million.

In March 2006, CRP acquired majority equity interests in five medical facilities for \$72.6 million which CRP funded, in part, with proceeds from a new \$47.2 million, ten-year mortgage loan.

Also in March 2006, CRP acquired a majority equity interest in another medical facility for \$24.5 million, which CRP funded with cash on hand.

During the first quarter of 2006, one senior housing facility that was under development as of December 31, 2005, commenced operations.

Other Investments. During the three months ending March 31, 2006, CRP advanced \$18.0 million under the senior secured term loan. The balance outstanding under the senior secured term loan was \$34.0 million at March 31, 2006.

Cash from Financing Activities as of March 31, 2006

Net cash provided by financing activities was \$170.0 million and \$198.6 million for the three months ended March 31, 2006 and 2005, respectively. The decrease was due primarily to reduced proceeds from permanent and construction financing as a result of the reduction in the number of properties acquired during the period, the repayment of a construction facility in 2006 and increased aggregate distributions to stockholders (due to the higher number of shares outstanding), offset by increased proceeds from CRP's revolving line of credit, net proceeds from CRP's equity offering and minority interest contributions.

Common Stock Offerings. CRP completed the 2004 offering on March 26, 2006. During the three months ended March 31, 2006, CRP raised \$103.2 million in subscription proceeds from the 2004 offering. Total subscription proceeds received from the 2004 offering and the four prior public offerings amount to \$2.7 billion at March 31, 2006.

During each of the quarters ended March 31, 2006 and 2005, CRP incurred \$8.2 million in offering costs, including \$7.6 million and \$6.5 million, respectively, in selling commissions and marketing support fees. These amounts are treated as stock issuance costs and charged to

stockholders' equity.

Redemptions. CRP has a redemption plan under which CRP may elect to redeem shares, subject to certain conditions and limitations. During the quarters ended March 31, 2006 and 2005, 1,724,246 shares and 847,143 shares, respectively, were redeemed and retired for \$16.4 million and \$8.0 million, respectively. CRP's board of directors has determined that it is in the best interest of CRP to suspend CRP's redemption plan, beginning with the second quarter of 2006. The suspension of CRP's redemption plan is effective as of June 15, 2006, and therefore no shares of CRP's common stock will be redeemed for the second quarter of 2006.

Distributions. CRP's board of directors authorized distributions to CRP's stockholders of \$45.5 million and \$42.6 million during the quarters ended March 31, 2006 and 2005, respectively. In addition, on April 1 and May 1, 2006, CRP's board of directors authorized distributions to stockholders of record on those dates, totaling \$31.3 million, or \$0.0592 per share of common stock at each record date, payable by June 30, 2006. During 2006, CRP intends to maintain CRP's quarterly distribution payment rate to stockholders of \$0.1776 per share. During the three months ended March 31, 2006, cash flow generated from operating activities was sufficient to fund the distribution to stockholders. CRP expects that cash flow generated from operations will continue to be sufficient to fund distribution payments; however, if cash flow generated from operations is not sufficient, CRP may use borrowings

under CRP's revolving line of credit to cover such shortage, subject to limitations imposed by the merger agreement, as described above.

Borrowings as of March 31, 2006

Revolving Line of Credit. At March 31, 2006, \$105.0 million was outstanding under CRP's \$320.0 million revolving line of credit. The revolving line of credit requires interest-only payments at LIBOR plus a percentage that fluctuates depending on CRP's aggregate amount of debt outstanding in relation to CRP's total assets (6.52% all-in rate at March 31, 2006, which represents a pricing of LIBOR plus 170 basis points). The amount available for use under the revolving line of credit is subject to certain limitations based on the pledged collateral. As of March 31, 2006, the revolving line of credit was collateralized by 36 properties with a carrying value of \$389.4 million that, in the aggregate, allowed us to draw up to \$283.0 million. Per the terms of the merger agreement, CRP is only permitted to draw up to an additional \$25.0 million under the revolving line of credit without HCP's prior written consent.

Mortgages Payable. At March 31, 2006, CRP had \$1.4 billion in mortgage debt secured by properties with an aggregate carrying value of \$2.4 billion. Interest rates on the mortgage notes ranged from 4.85% to 8.42% with a weighted-average rate of 5.87% at March 31, 2006. In the event the merger is not consummated, CRP expects to refinance loans as they mature, or CRP may use borrowings under its revolving line of credit to pay down maturities, subject to limitations imposed by the merger agreement as described above.

In January 2006, CRP entered into a \$56.3 million, ten-year mortgage loan that bears fixed-rate interest at 5.59%. Payments for the first five years are interest only, with principal payments beginning in March 2011.

In February 2006, CRP entered into a \$33.0 million mortgage loan and used the proceeds and cash on hand to pre-pay a \$48.0 million construction loan facility with a principal balance of \$41.9 million. The new interest-only, five-year loan bears interest at a rate equal to LIBOR plus 150 basis points (6.13% all-in rate at March 31, 2006).

In March 2006, CRP entered into a \$47.2 million, ten-year mortgage loan that bears fixed-rate interest at 5.81%. Payments for the first

five years are interest only, with principal payments beginning in May 2011.

Approximately 29% of the aggregate of CRP's mortgage notes payable, construction loans payable and amount outstanding under CRP's revolving line of credit at March 31, 2006, was subject to variable interest rates; therefore, CRP is exposed to market changes in interest rates as explained in "Quantitative and Qualitative Disclosures About Market Risk." Some of CRP's variable-rate loans contain provisions that allow us to convert the variable interest rates to fixed interest rates based on U.S. Treasury rates plus a premium at the time the conversion option is exercised. Fixed interest rates range from 4.85% to 8.42% with a weighted-average rate of 5.95%. Certain fixed-rate loans assumed by CRP contain substantial prepayment penalties and/or defeasance provisions that may make it economically unfavorable to prepay the loans prior to their maturity dates. Many of the loans have financial covenants which are typically found in commercial loans and which are primarily based on the operations of the properties. Certain loans contain extension options with terms similar to the initial loan terms.

During the first three months of 2006, CRP incurred \$1.5 million in loan costs in connection with the placement and assumption of permanent financing facilities. CRP was reimbursed \$2.7 million from a lender for previously paid rate locks and expense deposits.

Construction Loans Payable. Total construction loans outstanding at March 31, 2006, were \$108.5 million, and total liquidity remaining under CRP's construction loans was \$33.7 million. During

the three months ended March 31, 2006, CRP prepaid a construction loan facility with a \$41.9 million balance, entered into a new construction loan facility of \$7.7 million and collectively drew a net of \$10.1 million under all of CRP's construction loans related to certain properties in various stages of development. The loans are variable interest rate loans and mature from November 2006 through December 2013. CRP anticipates that CRP will obtain permanent financing to pay the construction loans as they become due or CRP may use borrowings under CRP's revolving line of credit, subject to limitations imposed by the merger agreement, as described above.

Bonds Payable. At March 31, 2006 CRP had \$101.2 million of non-interest bearing life care bonds at CRP's two CCRCs and non-interest bearing occupancy fee deposits at another of CRP's senior housing facilities, all of which were payable to certain residents of the facilities (collectively "bonds"). During the first quarter of 2006, the tenants of the facilities issued new bonds to new residents of the facilities totaling \$5.4 million and used the proceeds from the bonds issued in the current period and prior periods to retire \$2.2 million of bonds on CRP's behalf. At March 31, 2006, \$66.5 million of the bonds were refundable to the residents upon the resident moving out or to a resident's estate upon the resident's death and \$34.7 million of the bonds were refundable after the unit has been successfully remarketed to a new resident. Excess bond redemptions over bond issuance, if any, will be funded from prior net bonds issuance reserves (to the extent available) or from available operating cash flow.

Contractual Obligations as of March 31, 2006

The following table presents CRP's contractual cash obligations and related payment periods as of March 31, 2006 (in thousands):

	 ess than I Year	2 - 3 Years	4 - 5 Years	Thereafter	Total
Mortgages payable	\$ 76,517	\$ 227,394	\$ 515,537	\$ 533,381	\$ 1,352,829
Construction loans payable	80,546		16,109	11,817	108,472
Ground leases	761	1,722	1,731	42,746	46,960
DASCO office lease	153	418	437	245	1,253

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	Less than 1 Year	2 - 3 Years	4 - 5 Years	Thereafter	Total
Revolving line of credit		105,000			105,000
Bonds payable(1)				101,188	101,188
Security deposits and				23,386	23,386
rent support					
	\$ 157,977	\$ 334,534	\$ 533,814	\$ 712,763	\$ 1,739,088

Of this amount, \$66.5 million was due upon the resident moving out or the resident's death and \$34.7 million was due upon the unit being successfully remarketed to a new resident. It is expected that the proceeds from the issuance of new refundable life care bonds will be used to retire the existing bonds; therefore, bond redemptions are not expected to create a current net cash

Results of Operations as of March 31, 2006

obligation.

Comparison of the three months ended March 31, 2006 to the three months ended March 31, 2005

Net income for the three months ended March 31, 2006, totaled \$35.8 million or \$0.14 per share of common stock (\$36.2 million or \$0.14 per share of common stock from continuing operations), as compared to net income of \$32.6 million or \$0.14 per share of common stock (\$38.5 million or \$0.16 per share of common stock from continuing operations) for the three months ended March 31, 2005. The 5.8% decrease in income from continuing operations was primarily due to increases in operating expenses and interest and loan cost amortization expenses which more than offset the increase in revenue. The 9.6% increase in net income in the first quarter of 2006 was primarily due to the

recognition of an impairment charge on discontinued operations in the first quarter of 2005. These changes are discussed in further detail below.

Revenues

Rental and earned income from leases. At March 31, 2006, CRP owned 274 properties, including 14 properties acquired in 2006, compared to 244 properties owned at March 31, 2005, of which 22 properties were acquired during the first quarter 2005. As a result of the increase in the number of properties, rental and earned income from leases from properties from continuing operations increased 15.8% to \$97.3 million, including \$11.1 million as a result of straight-lining rent escalations throughout the lease terms for the three months ended March 31, 2006, compared to \$84.0 million, including \$11.8 million of straight-line rent revenue, for the three months ended March 31, 2005. The \$13.3 million increase in rental and earned income from leases was comprised of \$10.7 million from operations of the properties that were acquired or construction properties that commenced operations during 2005 and 2006 and \$2.6 million from properties owned as of January 1, 2005.

FF&E reserve income. FF&E reserve income from continuing operations increased 25.9% to \$2.0 million for the three months ended March 31, 2006, from \$1.6 million for the three months ended March 31, 2005. The increase of \$0.4 million was primarily due to contractual increases in FF&E reserve funding from tenants.

Tenant expense reimbursement revenue. Tenant expense reimbursement revenue from continuing operations increased 69.1% to \$4.6 million for the three months ended March 31, 2006, from \$2.7 million for the three months ended March 31, 2005. These revenue increases in 2006 reflect an increase of \$0.7 million in additional revenues from medical facilities that were owned as of January 1, 2005, \$0.9 million from properties acquired during 2005 and \$0.3 million from properties that were acquired or properties that commenced operations during 2006. Contractual recoveries from tenants represented 61% and 49% of CRP's medical facilities operating expenses for the three months ended March 31, 2006 and 2005, respectively.

Property management and development fees and loan interest income. Property management and development fees from The DASCO Companies, LLC decreased revenue by \$1.0 million and loan interest income from the senior secured term loan increased revenue by \$0.6 million for the three months ended March 31,

2006.

Expenses

Senior housing property expenses. Senior housing property expenses from continuing operations was \$0.3 million for each of the three months ended March 31, 2006 and 2005.

Medical facilities operating expenses. Medical facilities operating expenses from continuing operations increased 36.4% to \$7.5 million for the three months ended March 31, 2006, from \$5.5 million for the three months ended March 31, 2005. The increase was comprised of \$0.2 million related to the operations of the medical facilities that were owned as of January 1, 2005, \$1.3 million for properties acquired during 2005 and \$0.5 million related to the operations of the medical facilities that were acquired during 2006. CRP is generally responsible for the medical facilities' property operating expenses; however, under the terms of the leases, CRP recovers a portion of the expenses from the tenants.

General and administrative. General and administrative expenses from continuing operations increased 17.8% to \$4.7 million from \$4.0 million for the three months ended March 31, 2006 and 2005, respectively. The increase was due to the increased number of properties owned during 2006 and increased legal and consulting fees.

Asset management fees to related party. Asset management fees from continuing operations increased 17.9% to \$5.1 million for the three months ended March 31, 2006, from \$4.3 million for the three months ended March 31, 2005. The increase was primarily related to new operating properties that were acquired, or newly constructed and commenced operations during 2005 and 2006.

Provision for doubtful accounts. CRP recognized a provision for doubtful accounts from continuing operations for the three months ended March 31, 2006 and 2005, of \$1.5 million and \$0.8 million, respectively. The first-quarter 2006 provision was primarily related to a \$1.2 million reserve on a portfolio of 19 properties for which the operator guarantee had expired on December 31, 2005. The first-quarter 2005 provision was related to three senior housing portfolios and various medical facility tenants.

Depreciation and

amortization. Depreciation and amortization expense increased 18.5% to \$27.0 million for the three months ended March 31, 2006, from \$22.7 million for the three months ended March 31, 2005, as a result of the increase in properties subject to operating leases during 2005. The \$4.3 million depreciation and amortization expense increase was comprised of \$0.3 million from operations of the properties owned as of January 1, 2005, and \$4.0 million from operations of the properties that were acquired or had been under construction and commenced operations during 2005 and 2006.

Interest and other income

During the three months ended March 31, 2006 and 2005, CRP earned \$0.7 million and \$0.6 million, respectively, in interest income primarily from investments in money market accounts and other short-term, highly liquid investments.

Interest and loan cost amortization expense

Interest and loan cost amortization expense increased 49.2% to \$23.2 million for the three months ended March 31, 2006 from \$15.5 million for the three months ended March 31, 2005. The increase was primarily due to an increase in the average amount of debt outstanding as CRP continues to shift its reliance away from equity-offering proceeds to fund CRP's property acquisitions and other capital needs. The weighted-average interest rate was 6.0% for the three months ended March 31, 2006 as compared to 5.1% for the three months ended March 31,

2005.

Discontinued operations

Loss from discontinued operations for the three months ended March 31, 2006 was \$0.5 million compared to \$5.8 million for the three months ended March 31, 2005. The change was primarily due to an impairment charge of \$6.2 million recorded in the first quarter of 2005, partially offset by a first-quarter 2006 net loss of \$0.5 million on the sale of two properties and a provision for doubtful accounts of \$0.1 million recorded in the first quarter of 2006.

Inflation and Trends as of March 31, 2006

CRP's senior housing leases are triple-net leases and contain provisions that CRP believes will mitigate the effect of inflation. These provisions include clauses requiring automatic increases in base rent at specified times during the term of the lease (generally on an annual basis) and the payment of contingent rent if properties achieve specified operating thresholds (based on factors such as a percentage of gross revenue above a specified level). CRP has also invested in medical facilities, which include both triple-net and gross basis leases. These leases also contain provisions that mitigate the effect of inflation, such as scheduled base rent increases during the lease terms and, with respect to gross leases, the reimbursement of future increases in operating expenses (including real estate taxes, insurance, repairs, maintenance and utilities) over a specified base amount. Inflation and changing

prices may have an adverse impact on the potential disposition of the properties and on appreciation of the properties.

CRP believes that changes and trends in the healthcare industry will continue to create opportunities for growth of senior housing and other healthcare facilities, including (i) the growth of operators serving specific healthcare niches, (ii) the consolidation of providers and facilities through mergers, integration of physician practices, and elimination of duplicative services, (iii) the pressures to reduce the cost of providing quality healthcare, (iv) more dual-income and single-parent households leaving fewer family members available for in-home care of aging parents and necessitating more senior care facilities, and (v) an anticipated increase in the number of insurance companies and healthcare networks offering privately funded long-term care insurance. Additionally, CRP believes that demographic trends are significant when looking at the potential for future growth in the healthcare industry. Today's baby boomers (those born between 1946 and 1964) will begin reaching age 65 as early as 2011. According to the U.S. Census Bureau, the age 65 plus population is projected to more than double between now and the year 2050, to 82 million. Most of this growth is expected to occur between 2010 and 2030 when the number of older adults is projected to grow by an average of 2.8% annually.

CRP believes that during 2005, the senior housing industry experienced increased occupancies and average daily rates, and generally the facilities operated at a higher level of efficiency. The success of the future operations of CRP's properties will depend largely on each tenant's and operator's ability to adapt to dominant trends in the industry in each specific region, including, among others, greater competitive pressures, increased consolidation and changing demographics.

CRP is not aware of any material trends, favorable or unfavorable, in either capital resources or the outlook for long-term cash generation, nor does CRP expect any material changes in the availability and relative cost of such capital resources. Assuming the inflation rate remains low and long-term interest rates do not increase significantly, CRP believes that inflation will not impact the availability of debt financings.

Funds from Operations as of March 31, 2006

CRP considers funds from operations, or FFO, to be an indicative measure of operating performance due to the significant effect of

depreciation of real estate assets on net income. FFO, based on the revised definition adopted by the Board of Governors of the National Association of Real Estate Investment Trusts ("NAREIT") and as used herein, means net income determined in accordance with accounting principles generally accepted in the United States of America ("GAAP"), excluding gains or losses from sales of property, plus depreciation and amortization of real estate assets and after adjustments for unconsolidated partnerships and joint ventures. (Net income determined in accordance with GAAP includes the non-cash effect of straight-lining rent increases throughout the lease terms. This straight-lining is a GAAP convention requiring real estate companies to report rental revenue based on the average rent per year over the life of the leases. During the three months ended March 31, 2006 and 2005, net income included \$11.1 million and \$11.8 million, respectively, of these amounts.) CRP believes that by excluding the effect of depreciation, amortization and gains or losses from sales of real estate, all of which are based on historical costs and which may be of limited relevance in evaluating current performance, FFO can facilitate comparisons of operating performance between periods and between other equity REITs. FFO was developed by NAREIT as a relative measure of performance and liquidity of an equity REIT in order to recognize that income-producing real estate historically has not depreciated on the basis determined under GAAP. However, FFO (i) does not represent cash generated from operating activities determined in accordance with GAAP (which, unlike FFO, generally reflects all cash effects of transactions and other events that enter into the determination of net income), (ii) is not necessarily indicative of cash flow available to fund cash needs and (iii) should not be considered as an alternative to net income determined in accordance with GAAP as an indication of CRP's operating performance, or to cash flow from operating activities

determined in accordance with GAAP as a measure of either liquidity or CRP's ability to make distributions. FFO as presented may not be comparable to amounts calculated by other companies. Accordingly, CRP believes that in order to facilitate a clear understanding of the consolidated historical operating results, FFO should be considered in conjunction with CRP's net income and cash flows as reported in the accompanying consolidated financial statements and notes thereto.

The following is a reconciliation of net income to FFO (in thousands):

	Three Months Ended March 31,				
	2006	2005			
Net income	\$ 35,752	\$ 32,635			
Adjustments:					
Depreciation					
of real estate					
assets					
Continuing	23,217	19,392			
operations					
Discontinued		110			
operations					
Amortization of lease intangibles					
Continuing	3,618	3,318			
operations					
Discontinued		10			
operations					
Amortization of deferred leasing costs Continuing	114	25			
operations					
Effect of unconsolidated entity	101	61			
Effect of minority interests	(194)	(105)			
	\$ 62 608	\$ 55,446			
	Ψ 02,000	Ψ 33,770			
FFO per share (basic and diluted)	\$ 0.24	\$ 0.23			

Related Party Transactions as of March 31, 2006

CRP retained the Advisor as CRP's advisor to provide management, acquisition, advisory and administrative services relating to CRP's properties, mortgage loans, secured equipment lease program, other loans and other permitted investments pursuant to an advisory agreement dated May 14, 2004 (the "Advisory Agreement") that was renewed pursuant to a renewal agreement effective May 3, 2005 for a one-year term (the "2005 Renewal Agreement") and was amended by an amendment to the 2005 Renewal Agreement on July 13, 2005 (the "2005 Renewal Amendment" together with the 2005 Renewal Agreement, the "2005 Renewal Agreements"). On May 1, 2006, CRP entered into a renewal agreement (the "2006 Renewal Agreement") with the Advisor, pursuant to which the Advisory Agreement was renewed, as amended by the 2005 Renewal Agreements, for an additional one-year term commencing on May 3, 2006 and ending on May 3, 2007. The Advisory Agreement may be terminated at an earlier date upon 60 days prior written notice by either party or by mutual consent of the parties. Certain of CRP's directors and officers hold similar positions with the Advisor, the parent company of the Advisor and the managing dealer of CRP's public offerings, CNL Securities Corp. CRP's chairman of the board indirectly owns a controlling interest in the parent company of the Advisor.

Pursuant to the Advisory Agreement, as amended and renewed, the Advisor and its affiliates earn certain fees and are entitled to receive reimbursement of certain expenses. During the three months ended March 31, 2006 and 2005, CRP incurred acquisition fees of \$7.3 million and \$13.5 million, respectively, for, among other things, identifying properties and structuring the terms of the leases (equal to 3.0% of gross offering proceeds and loan proceeds from permanent financing from May 3,

2005 until the present and equal to 4.0% of gross offering proceeds and loan proceeds from May 14, 2004 through May 2, 2005). These fees are included in other assets in the accompanying consolidated balance sheets prior to being allocated to individual properties or intangible lease costs.

CRP incurred monthly asset management fees totaling \$5.1 million and \$4.5 million during the three months ended March 31, 2006 and 2005, respectively, (0.05% of CRP's real estate asset value, as defined in the Advisory Agreement, and the outstanding principal balance of any Mortgage Loans as of the end of the preceding month).

The Advisor and its affiliates also provide various administrative services, including, but not limited to, accounting; financial, tax, insurance administration and regulatory compliance reporting; stockholder distributions and reporting; due diligence and marketing; and investor relations. During the three months ended March 31, 2006 and 2005, CRP incurred \$0.6 million and \$2.0 million for these services, respectively.

CNL Securities Corp. received fees based on the amounts raised from CRP's offerings equal to: (i) selling commissions of 6.5% of gross proceeds under the 2004 offering and 7.5% under the prior offerings, (ii) a marketing support fee of 2.0% of gross proceeds under the 2004 Offering and 0.5% under the prior offerings and (iii) beginning on December 31, 2003, an annual soliciting dealer servicing fee equal to 0.2% of the aggregate proceeds raised in a prior offering. Affiliates of the Advisor are reimbursed for certain offering expenses incurred on CRP's behalf. Offering expenses incurred by the Advisor and its affiliates on CRP's behalf, together with selling commissions, the marketing support fee and due diligence expense reimbursements do not exceed 13% of the proceeds raised in connection with the offerings. During each of the quarters ended March 31, 2006 and 2005, CRP incurred \$8.2 million for these fees and costs. These amounts are treated as stock issuance costs and charged to stockholders' equity.

CRP owns a 9.90% interest in CNL Plaza, Ltd. (the "Owner"), a limited partnership that owns an office building located in Orlando, Florida, in which the Advisor and certain affiliates of CNL Financial Group ("CFG") lease office space. CFG owns a controlling interest in the parent company of the Advisor and is indirectly wholly owned by James M. Seneff, Jr., CRP's chairman of the board, and his wife. Robert A.

Bourne, CRP's vice-chairman of the board and treasurer, is an officer of CFG. The remaining interests in the Owner are held by several entities with present or former affiliations with CFG, including: CNL Plaza Venture, Ltd., which has a 1% interest as general partner of the Owner and whose general partner is indirectly wholly owned by Mr. Seneff and his wife; CNL Corporate Investors, Ltd., which is indirectly wholly owned by Messrs. Seneff and Bourne, and which has a 54.45% interest, as a limited partner, in the Owner; CNL Hotels & Resorts, Inc. which has a 9.90% interest, as a limited partner, in the Owner; and Commercial Net Lease Realty, Inc., which has a 24.75% interest, as a limited partner, in the Owner. CRP also owns a 9.90% interest in CNL Plaza Venture, Ltd. (the "Borrower"), a Florida limited partnership, which is the general partner of the Owner. The remaining interests in the Borrower are held by the same entities in the same proportion described above with respect to the Owner.

In 2004, the Owner conveyed a small portion of the premises underlying the parking structure adjacent to its office building, valued by the parties at \$0.6 million, to CNL Plaza II, Ltd., a limited partnership in which Messrs. Seneff and Bourne own a 60% interest and 40% interest, respectively, as part of the development of the premises surrounding the building. The purpose of the conveyance was to adjust the percentage fee simple ownership under the parking structure so as to allow joint parking privileges for a new office building that was developed in 2005 and is owned by CNL Plaza II, Ltd. In connection with this transaction, the Owner received an ownership interest in a cross-bridge that was constructed and an anticipated benefit from a reduction in the allocation of its operating expenses for the parking structure. In addition, the Owner received additional consideration pursuant to a purchase price adjustment.

On September 30, 2005, CRP executed a pro rata, several guarantee limited to 16.67%, or \$2.3 million, of a \$14.0 million uncollateralized promissory note of the Borrower that matures December 31, 2010.

CRP maintains bank accounts in a bank in which certain of CRP's officers and directors, including Messrs. Seneff and Bourne, serve as directors and are principal stockholders. The amounts deposited with this bank were \$4.9 million and \$3.1 million at March 31, 2006 and December 31, 2005, respectively.

CRP's chairman of the board is a director in a hospital that leases office space in seven of the medical facilities that CRP acquired in August 2004. Additionally, one of CRP's independent directors is a director in a health system that leases office space in one of the medical facilities that CRP acquired in April 2004. During the three months ended March 31, 2006 and 2005, these hospitals contributed less than 1% of CRP's total revenues.

Subsequent Events

Reinvestment Plan. CRP's board of directors has determined that it is in CRP's best interest to terminate CRP's distribution reinvestment plan, beginning with the second quarter of 2006. The termination of CRP's distribution reinvestment plan is effective as of June 15, 2006, and therefore no distributions to CRP's stockholders will be reinvested in shares of CRP's common stock pursuant to CRP's distribution reinvestment plan for the second quarter of 2006.

CRP's Management's Discussion and Analysis of Financial Condition and Results of Operations as of December 31, 2005

Business Overview as of December 31, 2005

CRP's continuing focus during 2005 was on (i) stabilizing the operating performance of its properties, (ii) developing new relationships and strengthening existing relationships with CRP's tenants, operators and managers (iii) strengthening existing relationships with lenders, and (iv) raising capital through CRP's equity offering and investing the proceeds in properties and other permitted investments.

(i)

CRP worked closely with

CRP's tenants and operators to

identify cash flow enhancing strategies where it identified deficiencies. Where necessary, CRP restructured lease terms and made capital expenditures to upgrade the properties, on which CRP will receive a return from the tenant. CRP also determined to hold for sale and recognized impairment losses on three properties that were identified as under-performing when acquired with their respective portfolios.

(ii)

CRP developed new relationships with senior housing tenants and operators. In addition, CRP entered into a new relationship with Cirrus to acquire, manage and develop new and existing medical facilities. As of March 15, 2006, CRP has acquired 14 medical facilities and two specialty hospitals through this relationship. CRP continued to foster its relationship with DASCO which, as of December 31, 2005, managed forty-eight of CRP's medical facilities and was developing five of CRP's medical facilities.

(iii)

CRP strengthened its relationship with lenders as CRP obtained or assumed \$348.6 million of permanent financing, entered into interest rate swap agreements tied to debt with an aggregate notional amount of \$233.8 million that reduced CRP's aggregate variable-interest-rate debt, drew \$63.4 million under CRP's construction loan facilities and amended CRP's revolving line of credit to increase the borrowing capacity to \$400.0 million under more favorable terms. Additionally, CRP prepaid \$58.3 million of above-market variable-interest-rate mortgage debt and repaid \$60.0 million under a term loan.

(iv)

During the year ended December 31, 2005, CRP received gross offering proceeds net of redemptions of \$178.4 million. CRP invested \$435.4 million in the acquisition of 40 properties, the funding of CRP's development projects and ongoing capital improvements of CRP's properties. CRP also funded a \$16.0 million loan advance to a Cirrus affiliate under a five-year \$85.0 million, senior secured term loan to finance the acquisition, development, syndication and operation of new and existing surgical partnerships.

As of December 31, 2005, CRP held real estate assets located in 33 states consisting of (dollars in thousands):

	Number of properties	Investment at December 31, 2005
Senior		
Housing		
facilities:		
Operating	183	\$ 2,820,657
Under	1	37,669
construction		
Medical		
Facilities:		
Operating	68	639,174
Under	5	5,611
construction		
	257	\$ 3,503,111
Real estate held for sale	5	\$ 12,692

Liquidity and Capital Resources as of December 31, 2005

CRP primarily invests in or develops properties and may invest in other permitted investments. As of March 15, 2006, CRP has relied on the sale of CRP's common stock to fund a significant portion of CRP's property investments. CRP also obtained funds through borrowings under permanent or construction financing, operating activities and draws on CRP's revolving

line of credit. The 2004 offering will close on or before March 26, 2006, and CRP does not presently intend to commence a new offering. Therefore, CRP will need to rely on other sources of capital or debt to fund property acquisitions and development. CRP believes that over the short term, which is less than 12 months, borrowings under permanent or construction financing, draws on CRP's revolving line of credit and cash on hand at December 31, 2005, will be sufficient to meet CRP's forecasted capital requirements for property investments, senior secured term loan funding, capital expenditures and the re-tenanting of CRP's medical facilities. Over the long term, which is 12 months or more, and subject to the terms of the merger agreement, CRP may raise capital by encumbering properties, entering into joint venture agreements with respect to CRP's investments in new or existing properties, issuing preferred stock, selling existing properties, or CRP may stop investing in properties.

Operating cash flow for the year ended December 31, 2005, was \$188.3 million and is expected to increase with a full year of operations for the 40 properties that CRP acquired during 2005 and those forecasted to be acquired during 2006. As of December 31, 2005, CRP has adequate construction funding to complete and open the properties under construction which CRP expects to generate additional operating cash flow.

CRP expects to continue to be able to pay distributions to maintain its REIT status, which requires that CRP distribute at least 90% of its taxable income to stockholders. Operating cash flow is expected to provide a significant portion of CRP's distributions, and to the extent necessary, CRP may borrow funds from its revolving line of credit to make distributions to stockholders. During 2006 until the closing of the merger, CRP intends to maintain its quarterly distribution payment rate to stockholders of \$0.1776 per share. CRP expects that cash flow generated from operations will be sufficient to fund distribution payments; however, if cash flow generated from operations is not sufficient, CRP may use borrowings under its revolving line of credit to cover such shortage.

Common Stock Offerings as of December 31, 2005

Upon formation in December 1997, CRP received an initial capital contribution of \$200,000 for 20,000 shares of common stock from the Advisor. From CRP's inception through December 31, 2005, CRP has made five public offerings of CRP's common stock and received subscriptions as follows (in thousands):

		Off	fering	Subso	criptions
Offering	Date Completed	Shares(a)	Amount	Shares(b)	Amount
Initial Offering	September 2000	15,500	\$ 155,000	972	\$ 9,719
2000 Offering	May 2002	15,500	155,000	15,500	155,000
2002 Offering	April 2003	45,000	450,000	45,000	450,000
2003 Offering	April 2004	175,000	1,750,000	156,793	1,567,925
2004 Offering	March 26, 2006	400,000	4,000,000	41,548	415,485
		651,000	\$ 6,510,000	259,813	\$ 2,598,129

- (a) Includes reinvestment plan shares of 500 in each of the initial and 2000 offerings, 5,000 in the 2002 offering, 25,000 in the 2003 offering and 15,000 in the 2004 offering.
- (b)
 Includes reinvestment plan shares of 5 in the initial offering, 42 in the 2000 offering, 129 in the 2002 offering, 1,728 in the 2003 offering and 8,749 in the 2004 offering.

The price per share of all of the equity offerings of CRP's common stock has been \$10.00 per share with the exception of (i) shares purchased pursuant to volume or other discounts and (ii) shares purchased through CRP's reinvestment plan which are currently priced at \$9.50 per share. Selling commissions, marketing support fees, due diligence expense reimbursements and other offering expenses have

not exceeded 13% of gross proceeds.

For the year ended December 31, 2005, net proceeds received from CRP's offering of shares, after deduction of selling commissions, marketing support fees, due diligence expense reimbursements, offering expenses and redemptions, totaled approximately \$160.0 million.

During the period January 1, 2006 through March 15, 2006, CRP received additional net offering proceeds of \$24.6 million, proceeds from new permanent financing of \$89.3 million and incurred acquisition fees and costs of \$3.5 million, including \$2.7 million related to acquisition fees on the new permanent financing.

If the merger is not consummated and if CRP does not list its shares by December 31, 2008, CRP will commence an orderly liquidation of its assets and the distribution of net proceeds to CRP's stockholders. CRP continues to monitor the market to determine if or when to list or pursue other strategic alternatives.

Redemptions as of December 31, 2005

CRP has a redemption plan under which CRP may elect to redeem shares, subject to certain conditions and limitations. Under the redemption plan, prior to such time, if any, as a listing occurs, any stockholder who has held shares for at least one year may present all or any portion equal to at least 25% of their shares to CRP for redemption in accordance with the procedures outlined in the redemption plan. Upon presentation, CRP may, at its option, redeem the shares for cash, subject to certain conditions and limitations. Redemptions are limited to the extent sufficient funds are available, however, at no time during any 12-month period may the number of shares CRP redeems exceed 5% of the number of shares of CRP's outstanding common stock at the beginning of the 12-month period. The full amount of proceeds from CRP's distribution reinvestment plan is available for redemptions. In

addition, CRP may, at its discretion, use up to \$100,000 per calendar quarter of the proceeds of any public offering of CRP's common stock for redemptions. In the second quarter of 2004, CRP amended its redemption plan to change CRP's redemption price from \$9.20 per share to \$9.50 per share. During the years ended December 31, 2005, 2004 and 2003, 3,904,039 shares, 685,396 shares and 131,781 shares, respectively, were redeemed and retired for \$37.1 million, \$6.5 million and \$1.2 million, respectively.

Property Acquisitions as of December 31, 2005

At December 31, 2005, CRP's investment portfolio consisted of 262 properties located in 33 states with an aggregate investment amount of approximately \$3.5 billion compared to 222 properties located in 32 states with an aggregate investment amount of approximately \$3.2 billion at December 31, 2004. During 2005, CRP invested \$435.4 million in 40 properties, including the payout of \$9.5 million in earnouts to the seller of two properties acquired in 2003. The properties acquired were (i) 18 senior housing facilities, consisting primarily of assisted living and independent living facilities, (ii) 21 medical facilities consisting of 18 medical office buildings, 5 of which were under construction, one specialty hospital and 2 walk-in clinics and (iii) a 10.4 acre parcel of land which CRP intends to sell. Two senior housing facilities and four medical facilities that were under construction at December 31, 2004 or when acquired during 2005 commenced operations during 2005. With the exception of one senior housing facility, CRP, as lessor, entered into long-term, triple-net lease agreements relating to the senior housing facilities and shorter-term, gross or triple-net lease agreements relating to the medical facilities. As of December 31, 2005, four of CRP's senior housing facilities and a parcel of land were held for sale.

The 40 properties acquired during 2005 are subject to operating leases. Operating leases related to CRP's senior housing facilities generally provide for initial terms of 15 years with options that allow the tenants to renew the leases for 5 to 20 successive years subject to the same terms and conditions as the initial leases. In addition to minimum annual base rent, a number of the senior housing leases require contingent rent if operating performance or occupancy rate thresholds, as defined in the lease agreements, are achieved. The leases generally also provide for the tenant to fund, in addition to minimum rent payments, an FF&E reserve fund. The tenant deposits funds into the FF&E reserve account and periodically uses these funds to cover the cost of the replacement, renewal and additions to furniture, fixtures and equipment.

Operating leases related to CRP's medical facilities include both triple-net and gross basis leases and generally have initial terms of 5 to 15 years. These leases provide for minimum rent and are generally subject to renewal options. The gross basis leases allow CRP to recover a portion of the medical facility operating expenses from the tenants, as specified in the lease agreements. Substantially all property leases require minimum annual base rent to be paid in monthly installments and to increase at predetermined intervals (typically on an annual basis) during the terms of the leases.

In accordance with Statement of Financial Accounting Standards No. 141, "Business Combinations" ("SFAS 141"), CRP allocates the value associated with having in-place operating leases at the date of acquisition to an intangible lease asset or liability considering factors associated with lease origination costs, customer relationships and above- or below-market leases. During 2005, CRP allocated \$15.7 million of acquired real estate value to in-place lease origination costs and customer relationships which are amortized over the remaining terms of the leases acquired with each property, \$1.3 million to an intangible lease asset related to above-market lease values which are amortized to rental income from operating leases. CRP also allocated \$1.9 million to an intangible lease liability acquired with each property related to below-market lease values which are accreted to rental income from operating leases over the remaining terms of the leases, including below-market lease extension, if any.

Other Investments as of December 31, 2005

In August 2005, CRP entered into an agreement to provide a Cirrus affiliate with an interest only, five-year senior secured term loan under which up to \$85.0 million (plus capitalized interest) may be borrowed to finance the acquisition, development, syndication and operation of new and existing surgical partnerships. During the first 48 months of the term, interest at the rate of 14.0% will accrue, of which 9.5% will be payable monthly and the balance of 4.5% will be capitalized; thereafter, interest at the greater of 14.0% or LIBOR plus 9.0% will be payable monthly. The loan is subject to equity contribution requirements and borrower financial covenants that will dictate draw down availability, is collateralized by all of the assets of the borrower (comprised primarily of interests in partnerships operating surgical facilities in premises leased from a Cirrus affiliate) and is guaranteed up to \$50.0 million through a combination of (i) a personal guarantee of up to \$13.0 million by a principal of Cirrus and (ii) a guarantee of the balance by other principals of Cirrus under arrangements for recourse limited only to their interests in certain entities owning real estate. At December 31, 2005, the balance outstanding under the senior secured term loan was \$16.0 million.

In connection with the senior secured term loan, CRP received stock warrants which are exercisable into a 10% to 15% ownership interest of the borrower. The stock warrants are exercisable at the earlier of an event of default or the full repayment of the senior secured term loan and expire in September 2015.

In August 2004, CRP acquired a 55% interest in DASCO for \$6.0 million including closing costs. CRP allocated \$5.8 million to goodwill, which represents the excess of the purchase price paid plus closing costs over the fair market value of the tangible assets (office furniture and equipment) acquired in the business acquisition. The purchase of the 55% interest in DASCO has provided and may continue to provide opportunities for CRP to participate in new medical office development and acquisition opportunities, as well as enter the business of managing medical facilities. As of December 31, 2005, DASCO managed forty-eight of CRP's medical facilities and was developing five of CRP's medical facilities.

CRP owns a 9.90% interest in CNL Plaza, Ltd., a limited partnership that owns an

office building located in Orlando, Florida, in which the Advisor and certain affiliates of the Advisor's parent company lease office space. CRP's initial equity investment in the partnership was \$0.3 million. CRP's share in the limited partnership's distributions is equivalent to CRP's equity interest in the limited partnership. The remaining interests in the limited partnership are owned by several entities with present or former affiliations with the Advisor's parent company. On September 30, 2005, CRP severally guaranteed 16.67%, or \$2.3 million, of a \$14.0 million uncollateralized promissory note of the limited partnership that matures December 31, 2010. As of December 31, 2005, the uncollateralized promissory note had an outstanding balance of \$13.9 million. CRP has not been required to fund any amount under this guarantee. In the event CRP is required to fund amounts under the guarantee, CRP believes that such amounts would be recoverable either from operations of the related asset or proceeds upon liquidation.

> Investments Subsequent to December 31, 2005 and Pending Investments

Investments Subsequent to December 31, 2005. In January 2006, CRP acquired seven medical facilities from Cirrus for \$84.5 million which CRP funded, in part, with proceeds from a new \$56.3 million, ten-year mortgage loan that bears fixed-rate interest at 5.59%. Four of the acquired properties are located in Texas, two are in Arizona and one is in Missouri, and in aggregate they contain approximately 255,000 square feet. Cirrus will manage the properties.

Pending Investments. As of March 15, 2006, CRP had initial commitments to acquire from Cirrus majority equity interests in five medical facilities for an aggregate price of \$72.6 million and for which CRP has posted a \$4.6 million non-refundable deposit as of December 31, 2005. Four of the medical facilities are located in Texas, and one medical facility is located in Oklahoma. CRP expects that Cirrus

will manage the five medical facilities. The acquisition of each of these investments is subject to the fulfillment of certain conditions. There can be no assurance that any or all of the conditions will be satisfied or, if satisfied, that one or more of these investments will be acquired by CRP.

Off-Balance Sheet Arrangements as of December 31, 2005

Interest Rate Swaps. On May 5, 2005, CRP entered into two interest rate swap agreements effective June 1, 2005, and one interest rate swap agreement effective July 1, 2005, for an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in interest rates on CRP's variable interest rate mortgage notes payable. The hedges have a 4.19% weighted-average fixed rate plus a 1.26% weighted-average spread, resulting in an all-in fixed interest rate of 5.45% until 2010.

Borrowings as of December 31, 2005

Revolving Line of Credit. On August 23, 2005, CRP amended and restated its \$85.0 million credit agreement and closed on a \$320.0 million amended and restated senior secured revolving line of credit, which permits CRP to expand the borrowing capacity up to \$400.0 million and extended the initial maturity date to August 23, 2007. The amount available for use under the revolving line of credit is subject to certain limitations based on the pledged collateral. The revolving line of credit is collateralized by 36 properties with a carrying value of approximately \$390.4 million at December 31, 2005, that in the aggregate, currently allows CRP to draw up to \$283.0 million. The revolving line of credit contains two one-year extension options and may be used to fund the acquisition and development of properties, purchase other permitted investments and for general corporate purposes. At closing, pricing was reduced from LIBOR plus 300 basis points to LIBOR plus 150 basis points. The revolving line of credit requires interest only payments at LIBOR plus a percentage that fluctuates depending on CRP's aggregate amount of debt outstanding in relation to CRP's total assets (6.20% all-in rate at December 31, 2005, which represents a pricing of LIBOR plus 170 basis points). At December 31, 2005, \$75.0 million was outstanding under the revolving line of credit.

Term Loan. In January 2005, CRP repaid and terminated a \$60.0 million, 14-day term loan used for the acquisition of certain properties until permanent financing was obtained in January 2005.

Permanent Financing. During 2005, CRP obtained \$348.6 million in permanent financing by assuming existing debt or securing new debt on various properties acquired during the period and by encumbering certain existing properties with new debt. As of December 31, 2005, CRP's aggregate permanent financing was \$1.2 billion and was collateralized by properties with an aggregate net book value of \$2.2 billion.

In July 2005, CRP prepaid a \$10.5 million mortgage note payable using available cash at June 30, 2005. In August 2005, CRP prepaid \$47.8 million in mortgage notes payable using available cash and proceeds from CRP's revolving line of credit.

Approximately 30% of the mortgage notes payable, construction loans payable and revolving line of credit at December 31, 2005, was subject to variable interest rates; therefore, CRP is exposed to market changes in interest rates as explained in "Quantitative and Qualitative Disclosures About Market Risk" below. Some of CRP's variable-rate loans contain provisions that allow CRP to convert the variable interest rates to fixed interest rates based on U.S. Treasury rates plus a premium at the time the conversion option is exercised. Fixed interest rates range from 4.85% to 8.42% with a weighted-average rate of 5.97%. Certain fixed-rate loans assumed by us contain substantial prepayment penalties and/or defeasance provisions that may make it economically unfavorable to prepay the loans prior to their maturity dates. Many of the loans have financial covenants which are typically found in commercial loans and which are primarily based on the operations of the properties. Certain loans contain extension options with terms similar to the initial loan terms.

During 2005, CRP incurred \$11.7 million in loan costs in connection with the placement and assumption of permanent financing facilities and the amended revolving line of credit.

The table below summarizes permanent financing that CRP obtained during the year ended December 31, 2005 (dollars in thousands):

Date Funded/A	Mortgage Ass Panyad le	Maturity Date	Interest Rate
Fixed-Ra	ite		
Debt: January 2005	\$ 7,108	June 2010	8.41%(1
March 2005	39,010	April 2012	4.85%
March 2005	34,299	January 2011 - April 2013	5.69% - 7.15%
June 2005	9,500	September 2012	5.67%
June 2005	1,669	May 2008	7.51%
October 2005	57,655	November 2015	5.39%
	149,241		
Variable-	-Rate		
Debt:			
Debt: January 2005	100,000	•	LIBOR 1.25%
January	100,000	2010 March	
January 2005 March	ĺ	2010 March	1.25% LIBOR
2005 March	50,000	2010 March	1.25% LIBOR
January 2005 March 2005	50,000	2010 March	1.25% LIBOR 1.50%
January 2005 March 2005 Other October	50,000	2010 March 2010	1.25% LIBOR 1.50%

(1)

The stated interest rate of 8.41% on this loan was greater than that available to CRP in the open capital market for comparable debt at the time of assumption. Consequently, CRP recognized \$0.7 million in debt premium that will be amortized over the period of the loan which reduces the effective interest rate to 5.67%. During 2005, CRP

recognized \$0.1 million in debt premium amortization related to this loan that is included in interest and loan cost amortization expense in the accompanying consolidated statements of income.

(2)

On October 3, 2005, CRP (i) exercised an extension option available under the \$140.4 million mortgage notes that were to mature in October 2005, (ii) negotiated the inclusion of an \$82.2 million variable-rate mortgage loan due to mature in April 2008 and (iii) drew an additional \$19.4 million under the facility, all with a new maturity date of October 2013. The facility contains provisions that will allow CRP to draw an additional \$58.0 million upon providing additional collateral. Of the new \$242.0 million mortgage note payable, \$121.0 million bears fixed-rate interest at 5.63% requiring principal and interest payments through maturity and \$121.0 million bears variable-rate interest based on the 3 to 9 month Fannie Mae Discount MBS rate plus 0.95% (5.16% at December 31, 2005) requiring interest only payments through maturity. CRP also has the option to convert the variable-rate debt component to fixed-rate debt.

Construction Financing. During 2005, CRP entered into new construction loan facilities of \$37.0 million and collectively drew a net of \$62.1 million under all of our construction loans related to certain properties in various stages of development. Total construction loans outstanding at December 31, 2005, were \$143.6 million, and total liquidity remaining was \$38.9 million. The loans are variable interest rate loans and mature from November 2006 through December 2013. CRP anticipates that it will obtain permanent financing to pay the construction loans as they become due.

Bonds Payable. At December 31, 2005 CRP had \$98.0 million of non-interest bearing life care bonds at our two CCRCs and non-interest bearing occupancy fee deposits at another of CRP's senior

housing facilities, all of which were payable to certain residents of the facilities (collectively "bonds"). During 2005, the tenants of the facilities issued new bonds to new residents of the facilities totaling \$12.6 million and used the proceeds from the bonds issued in the current period and prior periods to retire \$9.1 million of bonds on CRP's behalf. At December 31, 2005, \$68.7 million of the bonds were refundable to the residents upon the resident moving out or to a resident's estate upon the resident's death and \$29.4 million of the bonds were refundable after the unit has been successfully remarketed to a new resident. Excess bond redemptions over bond issuance, if any, will be funded from prior net bonds issuance reserves, to the extent available or from available operating cash flow.

Contractual Obligations and Commitments as of December 31, 2005

The following table presents CRP's contractual cash obligations and related payment periods as of December 31, 2005 (in thousands):

	_	ess than 1 Year	2 - 3 Years		4 - 5 Years	Thereafter	Total
Mortgages payable	\$	55,776	\$ 128,46	8	\$ 517,886	\$ 517,076	\$ 1,219,206
Construction loans payable		75,499	44,69	6	12,155	11,210	143,560
Ground leases		372	98	1	988	23,750	26,091
DASCO office lease		204	42	20	437	243	1,304
Revolving line of credit			75,00	00			75,000
Bonds payable(1)						98,016	98,016
Security deposits and rent support						23,954	23,954
	\$	131,851	\$ 249,56	55	\$ 531,466	\$ 674,249	\$ 1,587,131

Of this amount, \$68.7 million was due upon the resident moving out or the resident's death and \$29.4 million was due upon the unit being successfully remarketed to a new resident. It is expected that the proceeds from the issuance of new refundable life care

bonds will be used to retire the existing bonds; therefore, bond redemptions are not expected to create a current net cash obligation.

The following table presents CRP's commitments, contingencies and guarantees, and related expiration periods as of December 31, 2005 (in thousands):

2 -Less than 3 4-5 1 Year YearsYeafhereafteifotal

Pending			
investments(1)	\$ 157,100 \$	\$	\$ \$157,100
Unfunded senior			
secured term			
loan(2)	69,000		69,000
Capital			
improvements			
to properties(3)	62,620		62,620
Earnout			
provisions(4)	25,979		25,979
Guarantee of			
uncollateralized			
promissory note			
of CNL Plaza,			
Ltd.		2,313	2,313

\$314,699 \$ \$2,313 \$ \$317,012

(1) As of December 31, 2005, CRP had initial commitments to acquire 12 medical facilities for which CRP had posted a non-refundable \$10.6 million deposit. In January 2006, CRP completed the acquisition of seven medical facilities for \$84.5 million, including the application of \$6.0 million of the non-refundable deposit that CRP had posted as of December 31, 2005. The remaining properties are expected to be acquired in the first quarter of 2006. The acquisition of each of these investments is subject to the fulfillment of certain conditions. There can be no assurance that any or all of the conditions will be satisfied or, if satisfied, that CRP will acquire one or more of these investments.

(2)

Represents the unfunded portion under the \$85.0 million senior secured term loan

Commitments for the funding of properties under construction of \$38.9 million are expected to be funded with draws from construction loan facilities and the remaining balance relates to property expansion and renovation plans which CRP expects to fund from borrowings under CRP's revolving line of credit.

(4) In connection with the acquisition of 41 properties, CRP may be required to make additional payments to the seller if earnout provisions are achieved by the earnout date. The calculation generally considers the net operating income for the property, CRP's initial investment and the fair value of the property at the earnout date. In the event an amount is due, the applicable lease would be amended and annual minimum rent will increase accordingly. Amounts presented represent maximum exposure to additional payments. Earnout amounts related to six additional properties are subject to future values and events which are not quantifiable at December 31, 2005, and are not included in the table above.

Market Risk

See "Quantitative and Qualitative Disclosures About Market Risk" below.

Cash and Cash Equivalents as of December 31, 2005

Until properties are acquired or other permitted investments are entered into, cash from offering proceeds or permanent financings is held in short-term (defined as investments with an original maturity of three months or less), highly liquid investments which CRP believes to have appropriate safety of principal. This investment strategy provides high liquidity in order to facilitate CRP's use of these funds to acquire properties at such time as properties suitable for acquisition are identified or to fund other permitted investments and take advantage of favorable capital market conditions. At December 31, 2005, CRP had approximately \$94.9 million invested in

short-term investments as compared to \$51.8 million at December 31, 2004. The increase was primarily due to the fact that proceeds received from the placement of new permanent financing and offering proceeds received from the sale of shares of common stock during the year ended December 31, 2005, more than offset cash used to invest in additional properties.

Accounts and Other Receivables as of December 31, 2005

CRP's accounts and other receivables, net increased \$3.0 million to \$23.5 million at December 31, 2005, from \$20.5 million at December 31, 2004, consisting of a \$6.3 million increase in accounts and other receivables and a \$3.3 million increase in the reserve for doubtful accounts. The \$6.3 million increase in accounts and other receivables was due to (i) an increase in rental revenues receivable of \$11.3 million, (ii) an increase in other receivables of \$0.7 million, offset by (iii) a \$5.7 million reduction due to the reclassification of rental revenues receivable from current receivables to deferred receivables, as described in " Deferred Receivables" below. The \$3.3 million increase in the reserve for doubtful accounts was due to (i) a \$4.5 million increase in the reserve for balances outstanding at December 31, 2005 offset by (ii) a \$1.1 million reduction in the reserve associated with the rental revenues receivable that were reclassified to long-term receivables, as described in " Deferred Receivables" below and (iii) a \$0.1 million write-off of bad debt. Past due rent receivables were \$14.8 million and \$10.7 million at December 31, 2005 and 2004, respectively. At December 31, 2005, \$5.8 million of the \$7.2 million allowance for doubtful accounts was attributable to HRA tenants.

Effective November 2005, CRP amended and restated the leases of a five-property portfolio by (i) releasing from the pooled portfolio a non-performing property that is currently held for sale and which had been reducing the other properties' ability to pay current rent (ii) increasing the lease basis on the remaining four properties to cover the expected loss from disposing of the released property (iii) realigning the current lease rate to mirror current and forecasted portfolio operating performance

and revenue enhancements (iv) extending the lease term by six years and (v) preserving the ability to collect \$3.1 million past due receivables as of November 2005, that were reclassified to other assets. Additionally, CRP funded \$0.8 million of common area renovations, which CRP added to the lease basis, to enhance the properties' appeal to residents and improve market position. During 2005 CRP recognized a \$6.2 million impairment charge on the released property to reduce its carrying value to \$2.0 million, which is its estimated fair value less the estimated costs to dispose.

In connection with 19 properties for which CRP billed \$16.5 million in base rents for the year ended December 31, 2005, Sunrise, the operator, had guaranteed rent payments until December 31, 2005. CRP's evaluation of these properties' 2006 forecasted operating results revealed that it is likely that they will not be able to generate sufficient cash flow from operations to cover a portion of the 2006 contractual rental payments. This may result in CRP's recognition of additional provisions for doubtful accounts in 2006 and may have an adverse affect on CRP's cash available for distribution to stockholders. CRP is working with the HRA tenants (the tenant) and Sunrise to implement a plan to enhance cash flow generated from these properties. CRP is also evaluating strategic alternatives which may include lease restructure and sale of one or more of the properties.

Three additional properties had Sunrise guarantees that expired on June 30, 2005. During 2004, CRP determined to hold one of the properties for sale and recognized a \$1.9 million impairment charge to reduce the property's carrying value to \$1.6 million, which is its estimated fair value less the estimated costs to dispose. In January 2006, CRP entered into an agreement to sell this property for an expected sales price of approximately \$2.1 million. This transaction is subject to customary closing conditions and there can be no assurance that such conditions will be met, or if met, that the transaction will occur. During 2005, CRP determined to hold the two remaining properties for sale and recognized a \$1.5 million impairment charge to reduce the properties' aggregate carrying value to \$5.8 million, which is their estimated fair value less the estimated costs to dispose. In March 2006, CRP sold these two properties to an unrelated third party for \$6.0 million and CRP took back a purchase money mortgage with a three-year term secured by the properties in the amount of \$4.8 million. Interest is payable annually at a rate of 6.0% and principle is due at

maturity. CRP realized a net loss on the sale of the properties of \$0.2 million in March 2006.

Five other senior housing facilities continued to experience cash flow shortages and were unable to pay a portion of their rent obligations during 2005. Two of these properties were not supported by tenant guarantees or security deposits. During 2005, CRP worked with the operators of the properties to put initiatives in place to enhance revenue as well as certain cost containment measures and as a result, the properties began to cover a larger portion of their rent in the latter part of 2005. This trend is expected to continue during 2006. The three other properties are pooled within their portfolio but operating deficiencies of the three properties caused rent payment shortfalls. CRP continues to work with the operators of these properties to enhance revenue and reduce expenses. CRP expects that certain measures that were initiated during 2005 will enhance cash flow of the properties during 2006. However, failure of the properties to continue to enhance their cash flow from operations will result in the non-payment of a portion of their rent, and as a result, the recognition of additional provisions for doubtful accounts in 2006 which may have an adverse affect on CRP's cash available for distribution to stockholders.

CRP's analysis of estimated future cash flows to be generated by certain other properties for which CRP currently has reserves reveals that certain delinquent amounts will be collected in 2006.

CRP has been and will continue to work with the tenants and the operators of the respective properties to implement plans to increase operating efficiencies in order to enhance cash flow generated from the properties to fund current and past due rent obligations under the leases. In addition, CRP is evaluating strategic alternatives for certain facilities. The results of actual facility operations or implementation of one or more of these alternatives could result in additional reserves

for doubtful accounts or impairment losses that may impact CRP's results of operations and ability to pay distributions in future periods.

Deferred Receivables as of December 31, 2005

At December 31, 2005 and 2004, deferred receivables were \$6.6 million and \$0.9 million, respectively. Leases relating to 13 senior housing facilities provide for the amount of rental revenues receivable outstanding as of the end of the year to no longer be considered accounts receivable, but rather to be considered as a deferred receivable. Such amounts are added to the lease basis of these facilities and CRP receives a return based on the then current lease rate of the portfolio, including annual increases. Accordingly, CRP reclassifies such amounts from accounts receivable to other assets in the accompanying consolidated balance sheets. In 2006, CRP expects to reclassify an additional \$1.6 million for the outstanding rental revenues receivable at December 31, 2005, related to the 13 properties. In addition, as described under "Accounts Receivable" above, \$3.1 million of rental revenues receivable related to the leases of a five-property portfolio was reclassified from accounts receivable to deferred receivables in accordance with the lease amendment and restructure. The deferred receivables are payable from the tenant's excess cash flow after the payment of current rent and certain other fees, as permitted under the respective lease agreements. Based on CRP's evaluation of the properties' projected cash flows over the remaining term of the leases, CRP expects that the deferred receivables at December 31, 2005, and their respective rents will be collected and as such, CRP does not provide for a reserve for doubtful accounts.

Operator Rent Guarantees as of December 31, 2005

During the years ended December 31, 2005 and 2004, rental income included draws on operator rent guarantees of \$14.9 million and \$21.6 million, respectively. To mitigate credit risk, certain senior housing leases are combined into portfolios that contain cross-default and pooling terms. In addition, as of December 31, 2005, CRP held \$24.0 million in security deposits and rent support related to certain properties.

CRP had the following remaining rental support and limited guarantees from certain tenants and operators at December 31, 2005 (dollars in thousands):

Guarantor	Number of Properties	Maximum		Remaining I Balance
Horizon	21	\$ 17,500	\$ 14,391	\$ 3,109
Bay				
Aureus	11	10,000	2,255	7,745
ARC	8	(1)	9,416	\tilde{b} (1)
Eby	6	(1)	329	(1)
Encore	17	(1)	791	(1)
Greenwalt	5	(1)	2,493	(1)
Sunrise	2	(1))	(1)
Sunrise	17	(2)	6,281	(2)
Sunrise	3	(3)		

Guarantee

- (1) Unconditional guarantees
- Sunrise guaranteed the tenants' obligations to pay minimum rent and the FF&E reserve funds under the 17 leases until the later of (i) March 2006 or (ii) 18 months after the final development date of certain properties, as defined in the lease agreement. The final development property commenced operations in January 2006; accordingly, the Sunrise guarantee will terminate in July 2007.
- Sunrise guaranteed the tenants' rent obligations for these senior housing facilities that were acquired in 2004 and which commenced operations in 2004, until the later of (i) September 2006 or (ii) the properties achieving predetermined rent coverage thresholds, which are not determinable at this time.

Although CRP acquires properties located in various states and regions and screens its tenants in order to reduce risks of default, failure of Horizon Bay, the HRA tenants, or CRP's other tenants, their guarantors, or the Sunrise or Horizon Bay brands would significantly impact the results of CRP's operations.

Distributions as of December 31, 2005

During the years ended December 31, 2005, 2004 and 2003, CRP generated cash from operations of \$188.3 million, \$139.6 million and \$60.8 million, respectively, which included unrestricted security deposits received from tenants of \$4.1 million, \$8.7 million and \$3.1 million, respectively and draws on tenant and operator rent guarantees of \$14.9 million, \$21.6 million and \$5.6 million, respectively. CRP's board of directors declared distributions to CRP's stockholders of \$176.0 million, \$147.2 million and \$59.8 million during 2005, 2004 and 2003, respectively. In addition, on January 1, February 1 and March 1, 2006, CRP's board of directors declared distributions to stockholders of record on those dates, of \$0.0592 per share of common stock which are payable by March 31, 2006. Effective July 1, 2005, the board of directors amended CRP's distribution policy to discontinue the monthly payment of stockholder distributions, such that all distributions will be paid solely on a quarterly basis and will continue to be declared monthly during the offering period and quarterly thereafter.

CRP's distribution policy is based on a balanced analysis of value creation reflective of both current and long-term stabilized cash flows of CRP's properties, CRP's objective of continuing to qualify as a REIT for federal income tax purposes, the actual operating results of each quarter and anticipated operating results for the coming year, economic conditions, other operating trends, CRP's financial condition, loan restrictions, capital requirements and avoidance of volatility of distributions. During 2006 and prior to closing of the merger, CRP intends to maintain its quarterly distribution payment rate to stockholders of \$0.1776 per share. CRP expects that cash flow generated from operations will be sufficient to fund distribution payments; however, if cash flow generated from operations is not sufficient, CRP may use borrowings under its revolving line of credit to cover such shortage.

CRP's acquisition strategy is focused on opportunistically investing in larger portfolios, which allows CRP to obtain increased efficiencies. As a result, larger cash outlays are required at the time of purchase which may cause cash to accumulate for longer periods of time in short-term investments at lower returns prior to making these

purchases. Therefore, distributions paid to stockholders may periodically be greater than cash flows generated from operations. CRP expects to continue a large portfolio investment strategy during 2006, and may borrow funds from CRP's revolving line of credit to make distributions to stockholders in future quarters. During the years ended December 31, 2005, 2004 and 2003, distributions paid to stockholders were supported by borrowing under CRP's revolving line of credit of \$0, \$7.6 million and \$0 million, respectively.

For the years ended December 31, 2005, 2004 and 2003, approximately 67%, 60% and 71%, respectively, of the distributions received by stockholders were considered to be ordinary income and approximately 33%, 40% and 29%, respectively, were considered a return of capital for federal income tax purposes. No amounts distributed to stockholders for the years ended December 31, 2005, 2004 and 2003, were required to be or have been treated by us as a return of capital for purposes of calculating the stockholders' 8% return, which is equal to an 8% cumulative, non-compounded annual return on the amount calculated by multiplying the total number of shares of common stock purchased by stockholders by the issue price, without deduction for volume or other discounts, reduced by the portion of any distribution that is attributable to net sales proceeds and by any amount CRP has paid to repurchase shares under CRP's redemption plan.

Liquidity Requirements as of December 31, 2005

CRP believes that cash flow provided by operating activities will be sufficient to fund normal recurring operating expenses, regular debt service requirements and a significant portion of the distributions to stockholders. To the extent that cash flow provided by operating activities is not sufficient to meet such short-term liquidity requirements as a result, for example, of CRP's portfolio investment strategy or expenses due to the tenants defaulting under the terms of their lease agreements, CRP will use borrowings under its revolving line of credit. CRP expects to meet its other short-term liquidity requirements, including the acquisition and development of properties, the investment in other permitted investments, and the scheduled maturities of permanent financings with proceeds remaining from CRP's offerings (to the extent available), advances under the revolving line of credit, new permanent financing and the placement of permanent debt to replace maturing construction loans. CRP expects to meet its long-term liquidity requirements through short- or long-term, collateralized and uncollateralized financing or equity financing.

Senior housing facilities are generally leased on a long-term, triple-net basis, meaning the tenants are required to pay repairs and maintenance, property taxes, insurance and utilities. Generally, the tenants are also required to maintain an FF&E reserve account which is used to fund expenditures to refurbish buildings, premises and equipment to maintain the leasehold in a manner that allows operation for its intended purpose. In the event that the FF&E reserve is not sufficient, CRP may make fixed asset expenditures, in which case the annual minimum rent will be increased. CRP believes that current tenant reserves are sufficient to meet foreseen material FF&E repairs. The medical facilities are leased on either a triple-net or gross basis. CRP is responsible to fund capital improvements to medical facilities. With respect to medical facility gross leases, CRP generally recovers increases in building operating expenses (including real estate taxes, insurance, repairs, maintenance and utilities) over a specified base amount from the tenants, as specified in the lease agreement.

Results of Operations

Comparison of the year ended December 31, 2005 to the year ended December 31, 2004

Net income for the year ended December 31, 2005, totaled \$135.6 million or \$0.55 per share of common stock (\$142.4 million or \$0.57 per share of common stock from continuing operations), as compared to net income of \$117.9 million or \$0.56 per share of common stock (\$118.4 million or \$0.56 per share of common stock from continuing operations) for the year ended December 31, 2004. The increase in net income was primarily due to the rental income from the properties that CRP acquired during 2005 and the latter part of 2004, offset by increases in operating expenses related to the acquired properties and the recognition of impairment charges on three properties in 2005 as compared to one property in 2004. These changes are discussed in further detail below.

Revenues

Rental and earned income from leases. At December 31, 2005, CRP owned 262 properties, including 40 properties acquired in 2005, compared to 222 properties owned at December 31, 2004, of which 103 properties were acquired during 2004. As a result of the increase in the number of properties, rental and earned income from leases from properties from continuing operations increased \$104.8 million to \$358.1 million, including \$46.7 million as a result of straight-lining rent escalations throughout the lease terms, for the year ended December 31, 2005, compared to \$253.3 million, including \$40.4 million of straight-line rent revenue, for the vear ended December 31, 2004. The \$104.8 million increase in rental and earned income from leases was comprised of \$79.5 million from a full year of operations of the properties acquired during 2004, \$24.6 million from the new operating properties that were acquired or construction properties that commenced operations during 2005 and \$0.7 million from properties owned as of January 1, 2004. (See " Operator Rent Guarantees" above).

FF&E reserve income. FF&E reserve income from continuing operations increased \$2.9 million to \$7.5 million for the year ended December 31, 2005, from \$4.6 million for the year ended December 31, 2004. The increase was comprised of \$0.2 million additional revenues from a full year of operations of the properties acquired during 2004, \$0.1 million from the new operating properties that were acquired or construction properties that commenced operations during 2005 and \$2.6 million from properties owned as of January 1, 2004.

Tenant expense reimbursement revenue. Tenant expense reimbursement revenue from continuing operations increased \$8.6 million to \$13.3 million for the year ended December 31, 2005, from \$4.7 million for the year ended December 31, 2004. These revenue increases in 2005 reflect an increase of \$7.5 million in additional revenues from a full year of results from medical facilities acquisitions in the second and third quarters of 2004 and \$1.1 million from properties that were acquired or construction properties that commenced operations during 2005. Contractual recoveries from tenants represented 52% and 42% of CRP's medical facilities operating expenses for the years ended

Property management and development fees and loan interest income. Property management and development fees from DASCO increased revenue by \$0.7 million and loan interest income from the senior secured term loan increased revenue by \$0.5 million for the year ended December 31, 2005.

December 31, 2005 and 2004, respectively.

Expenses

Senior housing property expenses. Senior housing property expenses from continuing operations decreased \$0.5 million, to \$1.1 million for the year ended December 31, 2005, from \$1.6 million for the year ended December 31, 2004, as a result of decreased repairs resulting from increased capital spending.

Medical facilities operating expenses. Medical facilities operating expenses from continuing operations increased \$14.2 million to \$25.4 million for the year ended December 31, 2005, from \$11.2 million for the year ended December 31, 2004. The increase was comprised of \$11.9 million in additional expenses from a full year of operations of the properties acquired during 2004 and \$2.3 million from the new operating properties that were acquired or that commenced operations during 2005. CRP is generally responsible for the medical facilities' property operating expenses; however, under the terms of

the leases, CRP recovers a portion of the expenses from the tenants.

General and administrative. General and administrative expenses from continuing operations increased \$6.7 million to \$21.4 million from \$14.7 million for the years ended December 31, 2005 and 2004, respectively. The increase was due to general and administrative expenses related to a full year of DASCO operations in 2005 compared to four months of operations in 2004, the increased number of properties owned during 2005 and increased legal and consulting fees.

Asset management fees to related party. Asset management fees from continuing operations increased \$6.1 million to \$18.6 million for the year ended December 31, 2005, from \$12.5 million for the year ended December 31, 2004. The increase in expenses was comprised of \$4.9 million additional expense from a full year of operations of properties that were acquired during 2004 and \$1.2 million from new operating properties acquired or construction properties that commenced operations during 2005.

Provision for doubtful accounts. CRP recognized a provision for doubtful accounts from continuing operations for the years ended December 31, 2005 and 2004, of \$3.1 million and \$3.9 million, respectively, as discussed in the "Accounts and Other Receivables" section above. At December 31, 2005, \$5.8 million of the \$7.2 million allowance for doubtful accounts was attributable to HRA tenants.

Depreciation and

amortization. Depreciation and amortization expense increased to \$98.4 million for the year ended December 31, 2005, from \$62.5 million for the year ended December 31, 2004, as a result of the increase in properties subject to operating leases during the year ended December 31, 2005. The \$35.9 million depreciation and amortization expense increase was comprised of \$26.5 million in additional expense from a full year of operations of the properties acquired during 2004, \$8.9 million from the new operating properties that were acquired or construction properties that commenced operations during 2005 and \$0.5 million from properties owned as of January 1, 2004.

Interest and other income

During the years ended December 31, 2005 and 2004, CRP earned \$2.7 million and \$3.0 million, respectively, in interest income from investments in money market accounts and other short-term, highly liquid investments. The decrease was primarily due to reduced bank interest income as a result of lower invested cash balances. Also affecting year-over-year comparisons was the 2005 reclassification to reflect certain amounts related to CRP's medical facilities as property management and development fees revenue; for the year ended December 31, 2004, interest and other income included \$1.6 million related to these development, marketing and property management fees.

Interest and loan cost amortization expense

Interest and loan cost amortization expense was \$76.2 million and \$42.8 million for the years ended December 31, 2005 and 2004, respectively. The increase was a result of an increase in the average amount of debt outstanding to \$1.2 billion for the year ended December 31, 2005, from \$722.2 million for the year ended December 31, 2004. The weighted-average interest rate was approximately 5.8% for the year ended December 31, 2005 and 5.1% for the year ended December 31, 2004.

Discontinued operations

Loss from discontinued operations of \$6.8 million and \$0.4 million for the years ended December 31, 2005 and 2004, respectively, for four properties that CRP determined to hold for sale included operating revenues of \$1.7 million and \$2.2 million for the years ended December 31,

2005 and 2004, respectively, and expenses of \$0.8 million and \$0.7 million for the years ended December 31, 2005 and 2004, respectively. Expenses for the years ended December 31, 2005 and 2004, included a provision for doubtful accounts of \$0.3 million and \$0, respectively. Loss from discontinued operations also included impairment charges of \$7.7 million and \$1.9 million for the years ended December 31, 2005 and 2004, respectively, related to these properties as discussed in the "Accounts Receivable" section above.

Comparison of the year ended December 31, 2004 to the year ended December 31, 2003

Net income for the year ended December 31, 2004 totaled \$117.9 million or \$0.56 per share of common stock (\$118.4 million or \$0.56 per share of common stock from continuing operations), as compared to net income of \$58.5 million or \$0.66 per share of common stock (\$57.9 million or \$0.65 per share of common stock from continuing operations) for 2003. The increase in net income is primarily due to an increase in rental income from the properties that CRP acquired during the latter part of 2003 and in 2004 offset by increases in interest expense and loan cost amortization as a result of an increase in CRP's average outstanding debt, provisions for accounts receivable reserves and an impairment charge related to the proposed sale of a senior housing facility. These changes are discussed in further detail below. Although net income increased significantly for the year ended December 31, 2004, it decreased on a per share basis primarily due to the increased number of weighted-average number of common shares outstanding in 2004.

Revenues

Rental and earned income from leases. At December 31, 2004, CRP owned 218 operating properties, including 103 properties that were acquired in 2004, compared to owning 115 operating properties at December 31, 2003. As a result of the increase in the number and value of owned properties, CRP's rental and earned income from CRP's leases from continuing operations increased to \$253.3 million, including \$40.4 million as a result of straight-lining rent increases throughout the lease terms, for the year ended December 31, 2004, compared to \$90.4 million, including \$13.2 million of straight-line rent revenue, for the year ended December 31, 2003. The \$162.9 million rental and earned income from leases increase was comprised of \$71.2 million additional revenues from a full year of operations of the properties acquired or that commenced operations during 2003, \$91.0 million from the new operating properties acquired during 2004 and \$0.7 million from properties owned as of January 1, 2003. (See " Operator Rent Guarantees" above).

Rental income also included draws on operator rent guarantees of \$21.6 million and \$5.6 million during the years ended December 31, 2004 and 2003, respectively. As of December 31, 2004, CRP held \$26.3 million in security deposits and rent support related to certain properties.

FF&E reserve income. During the years ended December 31, 2004 and 2003, CRP earned \$4.6 million and \$2.6 million, respectively, in FF&E reserve income from properties from continuing operations during the years ended December 31, 2004 and 2003, respectively. The \$2.0 million FF&E reserve income increase was comprised of \$2.2 million additional revenues from a full year of operations of the properties acquired or that commenced operations during 2003 and \$0.1 million from the new operating properties that were acquired during 2004 offset by a reduction of \$0.3 million from properties owned as of January 1, 2003.

Tenant expense reimbursement revenue. During the year ended December 31, 2004, CRP recorded \$4.7 million in tenant expense reimbursement revenue, representing contractual recoveries from tenants of 42% of CRP's medical facilities operating expenses.

Expenses

Senior housing and medical facilities property expenses. Total property-related operating expenses for the years ended December 31, 2004 and 2003, were \$12.9 million

and \$136,000, respectively. The increase was primarily due to \$11.2 million from the acquisition of the medical facilities in the second and third quarters of 2004, where CRP is generally responsible for property operating expenses; however, under the terms of the leases, CRP recovers a portion of the expenses from the tenants. Property expenses related to senior housing facilities increased to \$1.6 from \$136,000 due to the increase in the number of senior housing facilities owned during the year ended December 31, 2004.

General and administrative. General and administrative expenses from continuing operations increased \$9.2 million to \$14.7 million from \$5.5 million for the years ended December 31, 2004 and 2003, respectively. The increase in expenses was related to general and administrative expenses related to the acquisition of DASCO operations in 2004 and the increased number of properties owned during 2004.

Asset management fees to related party. Asset management fees from continuing operations increased \$8.2 million to \$12.5 million for the year ended December 31, 2004, from \$4.3 million for the year ended December 31, 2003. The increase in expenses was comprised of \$3.6 million additional expense from a full year of operations of properties acquired during 2003 and \$4.6 million from new operating properties that were acquired or that commenced operations during 2004.

Provision for doubtful accounts. During the year ended December 31, 2004, CRP recognized a provision of \$3.9 million related to doubtful accounts receivable balances due to delays in receiving

current rent from certain senior housing facilities that were experiencing higher than expected property operating expenses.

Depreciation and

amortization. Depreciation and amortization expense increased to \$62.5 million for the year ended December 31, 2004, from \$17.3 million for the year ended December 31, 2003, as a result of the increase in properties subject to operating leases during the year ended December 31, 2004. The \$45.2 million depreciation and amortization expense increase was comprised of \$13.1 million additional expense from a full year of operations of the properties acquired during 2003 and \$32.1 million from the new operating properties acquired or that commenced operations during 2004.

Interest and other income

During the years ended December 31, 2004 and 2003, CRP earned \$3.0 million and \$1.6 million, respectively, in interest income from investments in money market accounts and other short-term, highly liquid investments. The increase in interest income is due to an increase in the average amount invested in short-term investments during the year ended December 31, 2004, as compared to the year ended December 31, 2003. For the year ended December 31, 2004, interest and other income also included \$1.6 million in development, marketing and property management fees related to CRP's medical facilities.

Interest and loan cost amortization expense

Interest and loan cost amortization expense was \$42.8 million and \$9.6 million for the years ended December 31, 2004 and 2003, respectively. The increase was a result of CRP's increasing the average amount of debt outstanding from \$151.4 million for the year ended December 31, 2003, to \$722.2 million for the year ended December 31, 2004. The weighted-average interest rate was approximately 5.1% for the year ended December 31, 2004 and 5.8% for the year ended December 31, 2003. In addition, CRP wrote off \$1.1 million in loan costs during the year ended December 31, 2004, as a result of the early extinguishment of debt and capitalized \$0.7 million in interest to properties under construction during 2004.

Discontinued operations

For the year ended December 31, 2004, loss from discontinued operations for four properties that CRP determined to hold for sale included operating revenues of approximately \$2.2 million offset by operating expenses of \$0.7 million and an impairment loss of \$1.9 million for the write-down of one of the properties to its estimated fair value less selling costs. For the year ended December 31, 2003, income from these properties included operating revenues of approximately \$1.0 million offset by operating expenses of \$0.4 million.

Other

Inflation and Trends as of December 31, 2005

CRP's senior housing leases are triple-net leases and contain provisions that CRP believes will mitigate the effect of inflation. These provisions include clauses requiring automatic increases in base rent at specified times during the term of the lease (generally on an annual basis) and the payment of contingent rent if properties achieve specified operating thresholds (based on factors such as a percentage of gross revenue above a specified level). CRP has also invested in medical facilities, which include both triple-net and gross basis leases. These leases also contain provisions that mitigate the effect of inflation, such as scheduled base rent increases during the lease terms and with respect to gross leases, the reimbursement of future increases in operating expenses (including real estate taxes, insurance, repairs, maintenance and utilities) over a specified base amount. Inflation and changing

prices may have an adverse impact on the potential disposition of the properties and on appreciation of the properties.

CRP believes that changes and trends in the healthcare industry will continue to create opportunities for growth of senior housing and other healthcare facilities, including (i) the growth of operators serving specific healthcare niches, (ii) the consolidation of providers and facilities through mergers, integration of physician practices, and elimination of duplicative services, (iii) the pressures to reduce the cost of providing quality healthcare, (iv) more dual-income and single-parent households leaving fewer family members available for in-home care of aging parents and necessitating more senior care facilities, and (v) an anticipated increase in the number of insurance companies and healthcare networks offering privately funded long-term care insurance. Additionally, CRP believes that demographic trends are significant when looking at the potential for future growth in the healthcare industry. Today's baby boomers (those born between 1946 and 1964) will begin reaching age 65 as early as 2011. According to the U.S. Census Bureau, the age 65 plus population is projected to more than double between now and the year 2050, to 82 million. Most of this growth is expected to occur between 2010 and 2030 when the number of older adults is projected to grow by an average of 2.8% annually.

CRP believes that during 2005, the senior housing industry experienced increased occupancies and average daily rates, and generally the facilities operated at a higher level of efficiency. The success of the future operations of CRP's properties will depend largely on each tenant's and operator's ability to adapt to dominant trends in the industry in each specific region, including, among others, greater competitive pressures, increased consolidation and changing demographics.

CRP is not aware of any material trends, favorable or unfavorable, in either capital resources or the outlook for long-term cash generation, nor does CRP expect any material changes in the availability and relative cost of such capital resources. Assuming the inflation rate remains low and long-term interest rates do not increase significantly, CRP believes that inflation will not impact the availability of equity and debt financings.

Related Party Transactions as of December 31, 2005

Certain of CRP's directors and officers hold similar positions with the Advisor, the parent

company of the Advisor and the managing dealer of CRP's public offerings, CNL Securities Corp. CRP's chairman of the board indirectly owns a controlling interest in the parent company of the Advisor. These affiliates receive fees and compensation for services provided in connection with the common stock offerings, permanent financing and the acquisition, management and sale of CRP's assets.

Pursuant to the advisory agreement, as amended and renewed, the Advisor and its affiliates earn certain fees and are entitled to receive reimbursement of certain expenses. During the years ended

December 31, 2005, 2004 and 2003, the Advisor and its affiliates earned fees and incurred reimbursable expenses as follows (in thousands):

	Years ended December 31,			
	2005	2004	2003	
Acquisition fees(1):				
From offering proceeds	\$ 5,874	\$ 38,286	\$ 47,644	
From debt proceeds	13,789	29,952	11,277	
	19,663	68,238	58,921	
	17,003	00,230	30,721	
Asset management fees(2)	19,217	13,047	4,372	
Reimbursable expenses(3):				
Acquisition expenses	210	331	403	
General and administrative expenses	5,989	4,313	2,255	
	6,199	4,644	2,658	
	\$ 45,079	\$ 85,929	\$ 65,951	

(1) For the period from May 3, 2005 through December 31, 2005, acquisition fees for, among other things, identifying properties and structuring the terms of the leases were equal to 3.0% of gross offering proceeds and loan proceeds from permanent financing under the 2004 Offering (4.0% of gross offering and loan proceeds for the period from May 14, 2004 through May 2, 2005 and 4.5% of gross offering and loan proceeds under the prior offerings). These fees are included in other assets in the accompanying consolidated balance sheets prior to being allocated to

individual properties or intangible lease

costs.

If CRP lists, the Advisor will receive an acquisition fee equal to 3.0% of amounts outstanding on the line of credit, if any, at the time of listing. Certain fees payable to the Advisor upon listing, the orderly liquidation or other sales of properties are subordinate to the return of 100% of the stockholders' invested capital plus the achievement of a cumulative, noncompounded annual 8% return on stockholders' invested capital.

- Monthly asset management fee of 0.05% of CRP's real estate asset value, as defined in the advisory agreement, and the outstanding principal balance of any mortgage loans as of the end of the preceding month.
- (3)

 Reimbursement for administrative services, including, but not limited to, accounting, financial, tax, insurance administration and regulatory compliance reporting, stockholder distributions and reporting, due diligence and marketing; and investor relations.

Pursuant to the advisory agreement, the Advisor is required to reimburse CRP the amount by which the total operating expenses CRP pays or incurs exceeds in any four consecutive fiscal quarters (the "Expense Year") the greater of 2% of average invested assets or 25% of net income (the "Expense Cap"). Operating expenses for the Expense Years ended December 31, 2005, 2004 and 2003, did not exceed the Expense Cap.

Of these amounts, approximately \$1.1 million and \$1.4 million were included in due to related parties in the accompanying consolidated balance sheets at December 31, 2005 and 2004, respectively.

CNL Securities Corp. received fees based on the amounts raised from CRP's offerings equal to: (i) selling commissions of 6.5% of gross proceeds under the 2004 offering and 7.5% under the prior offerings, (ii) a marketing support fee of 2.0% of gross proceeds under the 2004 offering and 0.5% under the prior offerings and (iii) beginning on December 31, 2003, an annual soliciting dealer servicing fee equal to 0.2% of the aggregate proceeds raised in a prior offering. Affiliates of the

Advisor are reimbursed for certain offering expenses incurred on CRP's behalf. Offering expenses incurred by the Advisor and its affiliates on CRP's behalf, together with selling commissions, the marketing support fee and due diligence expense reimbursements will not exceed 13% of the proceeds raised in connection with the offerings.

During the years ended December 31, 2005 and 2004, CRP incurred the following fees and costs (in thousands):

	Years ended December 31,		
	2005	2004	
Selling commissions	\$ 10,801	\$ 61,830	
Marketing support fee	3,313	6,648	
Offering and due diligence costs	4,250	18,328	
Soliciting dealer servicing fee		310	
	\$ 18,364	\$ 87,116	

Of these amounts, approximately \$1.3 million and \$0.2 million were included in due to related parties in the accompanying consolidated balance sheets at December 31, 2005 and 2004, respectively.

CRP owns a 9.90% interest in CNL Plaza, Ltd., or the Owner, a limited partnership that owns an office building located in Orlando, Florida, in which the Advisor and certain affiliates of CNL Financial Group, or CFG, lease office space. CFG owns a controlling interest in the parent company of the Advisor and is indirectly wholly owned by James M. Seneff, Jr., CRP's chairman of the board, and his wife. Robert A. Bourne, CRP's vice-chairman of the board and treasurer, is an officer of CFG. The remaining interests in the Owner are held by several entities with present or former affiliations with CFG, including: CNL Plaza Venture, Ltd., which has a 1% interest as general partner of the Owner and whose general partner is indirectly wholly owned by Mr. Seneff and his wife; CNL Corporate Investors, Ltd., which is indirectly wholly owned by Messrs. Seneff and Bourne, and which has a 49.50% interest, as a limited partner, in the Owner; CNL Hotels & Resorts, Inc. which has a 9.90% interest, as a limited partner, in the Owner;

Commercial Net Lease Realty, Inc., which has a 24.75% interest, as a limited partner, in the Owner; and CNL APF Partners, LP, which has a 4.95% interest, as a limited partner, in the Owner. CRP also owns a 9.90% interest in CNL Plaza Venture, Ltd., or the Borrower, a Florida limited partnership, which is the general partner of the Owner. The remaining interests in the Borrower are held by the same entities in the same proportion described above with respect to the Owner.

In 2004, the Owner conveyed a small portion of the premises underlying the parking structure adjacent to its office building, valued by the parties at approximately \$0.6 million, to CNL Plaza II, Ltd., a limited partnership in which Messrs. Seneff and Bourne own a 60% interest and 40% interest, respectively, as part of the development of the premises surrounding the building. The purpose of the conveyance was to adjust the percentage fee simple ownership under the parking structure so as to allow joint parking privileges for a new office building that was developed in 2005 and is owned by CNL Plaza II, Ltd. In connection with this transaction, the Owner received an ownership interest in a cross-bridge that was constructed and an anticipated benefit from a reduction in the allocation of its operating expenses for the parking structure. In addition, the Owner may be entitled to additional consideration pursuant to a purchase price adjustment.

On September 30, 2005, CRP executed a pro rata, several guarantee limited to 16.67%, or \$2.3 million, of a \$14.0 million uncollateralized promissory note of the Borrower that matures December 31, 2010. During each of the years ended December 31, 2005 and 2004, CRP received approximately \$0.2 million, respectively, in distributions from the Owner.

CRP maintains bank accounts in a bank in which certain of CRP's officers and directors serve as directors and are principal stockholders. The amounts deposited with this bank were \$3.1 million and \$22.9 million at December 31, 2005 and 2004, respectively.

On September 1, 2004, a company which is owned by CRP's chairman of the board sold its 30% voting membership interest in the HRA tenants to the remaining members of the limited liability company. The HRA tenants contributed 30% and 35% of CRP's total revenues for the years ended December 31, 2004 and 2003, respectively.

Century Capital Markets, LLC or CCM, an entity in which an affiliate of the Advisor was formerly a non-voting Class C member, made the arrangements for two commercial paper loans totaling \$43.9 million. The monthly interest payments due under these commercial paper loans include an annual margin of either 30 or 40 basis points, payable to CCM for the monthly services it provides related to the administration of the commercial paper loans. Effective September 30, 2005, a non-affiliated third party assumed the administration of these commercial paper loans. Therefore, CRP now pays the monthly services fee directly to the non-affiliated third party. During the years ended December 31, 2005, 2004 and 2003, \$0.1 million, \$0.1 million and \$0.2 million, respectively, was paid to CCM related to these services. During the year ended December 31, 2003, CRP also paid CCM a \$0.2 million finder's fee related to the acquisition of two properties.

CRP's chairman of the board is a director in a hospital that leases office space in seven of the medical facilities that CRP acquired in August 2004. Additionally, one of CRP's independent directors is a director in a health system that leases office space in one of the medical facilities that CRP acquired in April 2004. During the years ended December 31, 2005 and 2004, these hospitals contributed less than 1% of CRP's total revenues.

Critical Accounting Policies as of December 31, 2005

Allocation of Purchase Price for Acquisition of Properties. CRP allocates the purchase costs of properties to the tangible and intangible assets acquired and the liabilities assumed as provided by SFAS 141, "Business Combinations." For each acquisition, CRP assesses the value of the land, the as-if vacant building, equipment and intangible assets, including in-place lease origination costs, the above- or below-market lease values and the

value of customer relationships based on their estimated fair values. The values determined are based on independent appraisals, discounted cash flow models and CRP's estimates reflecting the facts and circumstances of each acquisition.

Acquisition Fees and Costs. Acquisition fees and miscellaneous acquisition costs that are directly identifiable with properties that are probable of being acquired are capitalized and included in other assets. Upon the purchase of a property, the fees and costs directly identifiable with that property are reclassified to land, building, equipment and lease intangibles or to investment in direct financing leases. In the event a property is not acquired or no longer is expected to be acquired, costs directly related to the property are charged to expense.

Leases. CRP's leases are accounted for under the provisions of Statement of Accounting Standards No. 13, "Accounting for Leases," and have been accounted for as either operating leases or direct financing leases. This statement requires CRP's management to estimate the economic life of the leased property, the residual value of the leased property and the present value of minimum lease payments to be received from the tenant. In addition, CRP assumes that all payments to be received under its leases are collectible. Changes in its estimates or assumptions regarding collectibility of lease payments could result in a change in accounting for the lease.

Impairments. CRP evaluates its properties and other long-lived assets on a quarterly basis, or upon the occurrence of significant changes in operations, to assess whether any impairment indications are present that affect the recovery of the carrying amount of an individual asset by comparing the sum

of expected undiscounted cash flows from the asset over its anticipated holding period, including the asset's estimated residual value, to the carrying value. If impairment is indicated, a loss is provided to reduce the carrying value of the property to its estimated fair value.

Allowance for Doubtful Accounts. CRP maintains an allowance for doubtful accounts for estimated losses resulting from the inability of CRP's tenants to make required rent payments. CRP bases its estimates on historical experience, projected cash flows generated from the tenants' operations of the properties and various other assumptions that CRP believes to be reasonable under the circumstances of a specific property or portfolio of properties. If the financial condition of any of CRP's tenants deteriorates, resulting in the impairment of their ability to make required rent payments, additional allowances may be required.

Goodwill. CRP allocates the excess of the aggregate purchase price paid over the fair market value of the tangible and identifiable intangible assets acquired in a business combination accounted for as a purchase to goodwill. Goodwill is not subject to amortization but is subject to quarterly impairment analysis. If quoted market prices are not available for CRP's impairment analysis, CRP uses other valuation techniques that involve measurement based on projected net earnings of the underlying reporting unit.

Derivative Instruments. The valuation of derivatives under Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities", as amended and interpreted, requires CRP to make estimates and judgments that affect the fair value of those instruments. CRP uses standard market conventions to determine the fair values of derivative instruments; and techniques such as discounted cash flow analysis, option pricing models, and termination cost are used to determine fair value at each balance sheet date. All methods of assessing fair value result in a general approximation of value and such value may never actually be realized.

REIT Qualification as of December 31, 2005

CRP made an election under Internal Revenue Code Section 856(c)(1), to be taxed as a REIT under Sections 856 through 860 of the Internal Revenue Code of 1986, as amended, and related regulations. As a REIT, for federal income tax purposes, CRP generally will not be subject to federal income tax on income that CRP distributes

to its stockholders. If CRP fails to qualify as a REIT in any taxable year, CRP will be subject to federal income tax on its taxable income at regular corporate rates and will not be permitted to qualify for treatment as a REIT for federal income tax purposes for four years following the year during which qualification is lost. Such an event could materially affect CRP's net income. However, CRP believes that it is organized and has operated in such a manner as to qualify for treatment as a REIT since CRP's formation in 1997 and specifically for the years ended December 31, 2005, 2004 and 2003. In addition, CRP intends to continue to operate so as to remain qualified as a REIT for federal income tax purposes.

CRP's Quantitative and Qualitative Disclosures About Market Risk as of March 31, 2006

At March 31, 2006, 29% of CRP's mortgages payable, construction loans payable and amount outstanding under its revolving line of credit were subject to variable interest rates; therefore, CRP is exposed to market changes in interest rates. For the three months ended March 31, 2006, a hypothetical 100 basis point increase in the LIBOR rates would have resulted in additional interest costs of \$1.0 million. This sensitivity analysis contains certain simplifying assumptions (for example, it does not consider the impact of changes in prepayment risk or credit spread risk). Therefore, although it gives an indication of CRP's exposure to interest rate change, it is not intended to predict future results and the actual results will likely vary.

To mitigate interest rate risk, CRP may pay down the mortgages or the revolving line of credit prior to their maturity dates through debt refinancing should interest rates rise substantially. In May 2005, CRP implemented a policy to further mitigate interest rate risk. CRP's primary strategy is to protect against this risk by using derivative transactions as appropriate to minimize the effect that variable interest rate fluctuations could have on cash flow. In May 2005, CRP entered into two interest rate swap agreements effective June 1, 2005, and one interest rate swap agreement effective July 1, 2005, for an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in interest rates on CRP's variable interest rate mortgage notes payable. The hedges have a 4.19% weighted-average interest rate plus a 1.26% weighted-average spread resulting in an all-in fixed interest rate of 5.45% until 2010. At March 31, 2006, these interest rate swaps had a fair value of \$7.7 million. A hypothetical 100 basis point increase or decrease in LIBOR rates would cause the fair value of these swaps to be \$15.1 million or \$(0.03) million, respectively.

Certain fixed-rate loans contain substantial prepayment penalties and/or defeasance provisions that may make it economically unfavorable to repay the loans prior to their maturity dates.

Following is a summary of CRP's mortgages payable, construction loans payable and revolving line of credit obligations at March 31, 2006 (in thousands):

Expected Maturities

	_												
	_	2006	_	2007	_	2008	_	2009	_	2010	T	hereafter	
<u>Debt</u> Obligations:													
Fixed-rate debt:	\$		\$	10,392	\$	50,211	\$	140,851	\$	138,735	\$	530,821	\$ 8
Average interest rate				7.42%	,	6.249	%	6.02%	ó	6.869	6	5.72%	, D
Variable-rate debt:	\$	124,445	\$	148,920	\$		\$	15,952	\$	240,157	\$	165,817	\$ 6
Average interest rate		7.10%)	6.62%	,			6.52%	ó	5.80%	6	5.44%	b
Interest Rate Derivatives:													
Variable to fixed swaps:	\$		\$		\$		\$		\$	233,750	\$		\$ 2
Average pay rate										4.199	6		
Average receive rate										4.55%	6		

CRP's Quantitative and Qualitative Disclosures About Market Risk as of

December 31, 2005

At December 31, 2005, 30% of CRP's mortgages payable, construction loans payable and amount outstanding under its revolving line of credit were subject to variable interest rates; therefore, CRP is exposed to market changes in interest rates. For the year ended December 31, 2005, a hypothetical 100 basis point increase in the LIBOR rates would have resulted in additional interest costs of approximately \$5.6 million (\$0.02 per share of common stock). This sensitivity analysis contains certain simplifying assumptions (for example, it does not consider the impact of changes in prepayment risk or credit spread risk). Therefore, although it gives an indication of CRP's exposure to interest rate change, it is not intended to predict future results and the actual results will likely vary.

To mitigate interest rate risk, CRP may pay down the mortgages or the revolving line of credit prior to their maturity dates with offering proceeds (to the extent available) should interest rates rise substantially. In May 2005, CRP implemented a policy to further mitigate interest rate risk. CRP's primary strategy is to protect against this risk by using derivative transactions as appropriate to minimize the effect that variable interest rate fluctuations could have on cash flow. In May 2005, CRP entered into two interest rate swap agreements effective June 1, 2005, and one interest rate swap agreement effective July 1, 2005, for an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in interest rates on CRP's variable interest rate mortgage notes payable. The hedges have a 4.19% weighted-average plus a 1.26% weighted-average spread resulting in an all-in fixed interest rate of 5.45% until 2010. At December 31, 2005, these interest rate swaps had a fair value of \$4.8 million. A hypothetical 10% increase or decrease in LIBOR rates would cause the fair value of these swaps to be \$8.7 million or \$1.0 million, respectively.

Certain fixed-rate loans contain substantial prepayment penalties and/or defeasance provisions that may make it economically unfavorable to repay the loans prior to their maturity dates.

Following is a summary of our mortgages payable, construction loans payable and revolving line of credit obligations at December 31, 2005 (in thousands):

Expected Maturities

	2006	2007	2008	2009	2010	Thereafter	7
Debt Obligations: Fixed-rate	\$	\$ 10,425	\$ 50,460	\$ 141.513	\$ 139,370	\$ 428,412	\$ 7
debt:	·	, ,, ,	, ,	. ,-	,,	/	
Average interest rate		7.429	% 6.249	6.029	% 6.749	% 5.73%	6
Variable-rate debt:	\$ 119,604	\$ 163,616	\$	\$ 12,155	\$ 240,000	\$ 132,211	\$ 6
Average interest rate	6.74%	6.339	%	6.13	% 5.47%	% 5.23%	6
Interest Rate Derivatives:							
Variable to fixed swaps:	\$	\$	\$	\$	\$ 233,750	\$	\$ 2
Average pay rate					4.19%	6	
Average receive rate					4.229	6	
		161					

DESCRIPTION OF HCP CAPITAL STOCK

HCP's authorized capital stock consists of 750,000,000 shares of common stock, par value \$1.00 per share, and 50,000,000 shares of preferred stock, par value \$1.00 per share. The following description does not contain all the information that might be important to you. Therefore you should read carefully the more detailed provisions of HCP's articles of restatement and HCP's third amended and restated bylaws. See "Where You Can Find More Information" beginning on page 181.

Common Stock

As of July 31, 2006, there were 137,301,582 shares of HCP common stock outstanding. All shares of HCP common stock participate equally in dividends payable to holders of common stock, when and as authorized by the HCP board and declared by HCP, and in net assets available for distribution to holders of HCP common stock on liquidation, dissolution, or winding up of HCP. Each outstanding share of HCP common stock entitles the holder to one vote on all matters submitted to a vote of the HCP stockholders. Holders of HCP common stock do not have cumulative voting rights in the election of directors.

All issued and outstanding shares of HCP common stock are, and the common stock offered by this proxy statement/prospectus will be upon issuance, validly issued, fully paid and nonassessable. Holders of HCP common stock do not have preference, conversion, exchange or preemptive rights. The HCP common stock is listed on The New York Stock Exchange (NYSE Symbol: HCP).

The Transfer Agent and Registrar for our common stock is Bank of New York.

Preferred Stock

Under the HCP charter, the HCP board is authorized without further stockholder action to establish and issue, from time to time, up to 50,000,000 shares of preferred stock of HCP, in one or more series, with such designations, preferences, powers and relative participating, optional or other special rights, and the qualifications, limitations or restrictions thereon, including, but not limited to, dividend rights, dividend rate or rates, conversion rights, voting rights, rights and terms of redemption (including sinking fund provisions), the redemption price or prices, and the liquidation preferences as shall be

stated in the resolution providing for the issue of a series of such stock, adopted, at any time or from time to time, by the HCP board. As of December 31, 2005, HCP had outstanding 4,000,000 shares of 7.25% Series E Cumulative Redeemable Preferred Stock ("Series E Preferred Stock") with a liquidation preference of \$100,000,000 and 7,820,000 shares of 7.10% Series F Cumulative Redeemable Preferred Stock ("Series F Preferred Stock") with a liquidation preference of \$195,500,000. Certain terms of the Series E Preferred Stock and Series F Preferred Stock are described below.

Series E Preferred Stock

Voting Rights

Holders of Series E Preferred Stock generally do not have any voting rights, except in limited circumstances.

If dividends on any shares of Series E Preferred Stock are in arrears for six or more quarterly periods, whether or not consecutive, the holders of Series E Preferred Stock (voting separately as a class with all other classes or series of preferred stock upon which like voting rights have been conferred and are exercisable) are entitled to vote for the election of a total of two additional directors of HCP at a special meeting called by the holders of record of at least 25% of the Series E Preferred Stock or the holders of any other class or series of preferred stock so in arrears or at the next annual meeting of stockholders. These voting rights continue at each subsequent annual meeting until all dividends accumulated on such shares of Series E Preferred Stock for the past dividend periods and the

dividend for the then current dividend period shall have been fully paid or declared and set aside for payment. In such case, the entire HCP board is increased by two directors.

So long as any shares of Series E Preferred Stock remain outstanding, HCP shall not, without the consent or the affirmative vote of the holders of at least two-thirds of the shares of Series E Preferred Stock outstanding at the time, given in person or by proxy, either in writing or at a meeting, with the Series E Preferred Stock voting separately as a class:

authorize, create or issue, or increase the authorized or issued amount of, any class or series of stock ranking prior to the Series E Preferred Stock with respect to the payment of dividends, or the distribution of assets on liquidation, dissolution or winding up;

reclassify any authorized stock of HCP into any such shares, or create, authorize or issue any obligation or security convertible into or evidencing the right to purchase any class or series of stock ranking prior to the Series E Preferred Stock with respect to the payment of dividends, or the distribution of assets on liquidation, dissolution or winding up;

repeal, amend or otherwise change any of the provisions applicable to the Series E Preferred Stock in any manner which materially and adversely affects the powers, preferences, voting power or other rights or privileges of the Series E Preferred Stock. However, an increase in the amount of authorized preferred stock, the creation or issuance of other classes or series of preferred stock or any increase in the amount of authorized shares of Series E Preferred Stock or of any other class or series of preferred stock, in each case ranking on a parity

with, or junior to the Series E Preferred Stock, will not be deemed to materially and adversely affect such rights, preferences, privileges or voting powers.

The consent of the holders of Series E Preferred Stock is not required for the taking of any corporate action, including any merger or consolidation involving HCP or a sale of all or substantially all of the assets of HCP, regardless of the effect that such merger, consolidation or sale may have upon the rights, preferences or voting power of the holders of the Series E Preferred Stock, except as expressly set forth in the articles supplementary relating to the Series E Preferred Stock.

Rank

With respect to dividend rights and rights upon liquidation, dissolution or winding up of HCP, the Series E Preferred Stock ranks:

senior to all classes or series of common stock of HCP, and to all equity securities issued by HCP ranking junior to the Series E Preferred Stock with respect to dividend rights or rights upon liquidation, dissolution or winding up of HCP;

on a parity with all equity securities issued by HCP the terms of which specifically provide that such equity securities rank on a parity with the Series E Preferred Stock with respect to dividend rights or rights upon liquidation, dissolution or winding up of HCP; and

junior to all equity securities issued by HCP the terms of which specifically provide that such equity securities rank senior to the Series E Preferred Stock with respect to dividend rights or rights upon liquidation, dissolution or winding up of HCP.

The term "equity securities" does not include convertible debt securities, which rank senior to

the Series E Preferred Stock prior to conversion.

Dividends

Holders of shares of the Series E Preferred Stock are entitled to receive, when, as, and if declared by the HCP board out of funds of HCP legally available for the payment of dividends, cumulative

preferential annual cash dividends at the rate of 7.25% of the liquidation preference (equivalent to \$1.8125 per annum per share).

Dividends on the Series E Preferred Stock are cumulative from the date of original issue and payable quarterly in arrears on or about the last day of each March, June, September and December or, if not a business day, the next succeeding business day.

No dividends may be declared by the HCP board or paid or set apart for payment on the Series E Preferred Stock if the terms of any agreement of HCP, including any agreement relating to its indebtedness, prohibits such a declaration, payment or setting apart for payment or provides that such declaration, payment or setting apart for payment would constitute a breach of or default under such an agreement. Likewise, no dividends may be declared by the HCP board or paid or set apart for payment if such declaration or payment is restricted or prohibited by law.

Dividends on the Series E Preferred Stock accrue, however, whether or not HCP has earnings, whether or not there are funds legally available for the payment of such dividends and whether or not such dividends are declared. Accrued but unpaid dividends on the Series E Preferred Stock do not bear interest and holders of the Series E Preferred Stock are not entitled to any dividends in excess of full cumulative dividends described above. Any dividend payment made on the Series E Preferred Stock is first credited against the earliest accrued but unpaid dividend due that remains payable.

No full dividends may be declared or paid or set apart for payment on any class or series of preferred stock ranking, as to dividends, on a parity with or junior to the Series E Preferred Stock, other than a dividend in shares of any class of stock ranking junior to the Series E Preferred Stock as to dividends and upon liquidation, for any period unless full cumulative dividends have been or contemporaneously are declared and paid or declared and set apart for such payment on the Series E Preferred Stock for all past dividend periods and the then current dividend period. When dividends are not paid in full, or full payment is not so set apart, upon the Series E Preferred Stock and the shares of any other class or series of preferred stock ranking on a parity as to dividends with the Series E Preferred Stock all dividends declared upon the Series E Preferred Stock and any other class or series of preferred stock ranking on a parity as to dividends with the Series E Preferred Stock are declared pro rata so that the amount of dividends declared per share of Series E Preferred Stock and such other class or series of preferred stock shall in all cases bear to

each other the same ratio that accrued dividends per share on the Series E Preferred Stock and such other class or series of preferred stock, which cannot include any accrual in respect of unpaid dividends for prior dividend periods if such preferred stock does not have a cumulative dividend, bear to each other.

Except as provided in the preceding paragraph, unless full cumulative dividends on the Series E Preferred Stock have been or contemporaneously are declared and paid or declared and set apart for payment for all past dividend periods and the then current dividend period, then, other than the payment of dividends in shares of HCP common stock or other shares of capital stock ranking junior to the Series E Preferred Stock as to dividends and upon liquidation:

no dividends may be declared or paid or set aside for payment upon the HCP common stock, or any other capital stock of HCP ranking junior to or on a parity with the Series E Preferred Stock as to dividends or upon liquidation;

no other distribution may be declared or made upon the HCP common stock, or any other capital stock of HCP ranking junior to or on a parity with the Series E Preferred Stock as to dividends or upon liquidation;

no shares of HCP common stock, or any other shares of capital stock of HCP ranking junior to or on a parity with the Series E Preferred Stock as to dividends or upon liquidation may be redeemed, purchased or otherwise acquired for any consideration by HCP, except by conversion into or exchange for other capital stock of HCP ranking junior to the Series E Preferred Stock

as to dividends and upon liquidation or for the purpose of preserving HCP's qualification as a real estate investment trust.

Liquidation Preferences

Upon any liquidation, dissolution or winding up of the affairs of HCP the holders of Series E Preferred Stock are entitled to be paid out of the assets of HCP legally available for distribution to its stockholders a liquidation preference of \$25 per share, plus an amount equal to any accrued and unpaid dividends to the date of payment, before any distribution of assets is made to holders of HCP common stock or any other class or series of capital stock of HCP that ranks junior to the Series E Preferred Stock as to liquidation rights.

In determining whether a distribution (other than upon voluntary or involuntary liquidation) by dividend, redemption or other acquisition of shares of stock of HCP or otherwise is permitted under the MGCL, no effect is given to amounts that would be needed if HCP would be dissolved at the time of the distribution, to satisfy the preferential rights upon distribution of holders of shares of stock of HCP whose preferential rights upon distribution are superior to those receiving the distribution.

Maturity; Redemption

The Series E Preferred Stock has no stated maturity, is not subject to any sinking fund or mandatory redemption. The Series E Preferred Stock is not redeemable prior to September 15, 2008. HCP is entitled, however, to purchase shares of the Series E Preferred Stock in order to preserve its status as a real estate investment trust for federal or state income tax purposes at any time. Following September 15, 2008, HCP may, at its option, redeem the Series E Preferred Stock at \$25 per share (\$100,000,000 in the aggregate), plus accrued and unpaid dividends.

Transfer and Ownership Restrictions

See " Transfer and Ownership Restrictions Relating to HCP Preferred Stock."

Series F Preferred Stock

Voting Rights

Holders of the Series F Preferred Stock generally do not have any voting rights, except in limited circumstances.

If dividends on any shares of Series F Preferred Stock are in arrears for six or more quarterly periods, whether or not consecutive, the holders of Series F Preferred Stock (voting separately as a class with all other classes or series of preferred stock upon which like voting rights have been conferred and are exercisable, including the Series E Preferred Stock) are entitled to vote for the election of a total of two additional directors of HCP at a special meeting called by the holders of record of at least 25% of the Series F Preferred Stock or the holders of any other class or series of preferred stock so in arrears or at the next annual meeting of stockholders. These voting rights continue at each subsequent annual meeting until all dividends accumulated on such shares of Series F Preferred Stock for the past dividend periods and the dividend for the then current dividend period shall have been fully paid or declared and set aside for payment. In such case, the entire HCP board is increased by two directors.

So long as any shares of Series F Preferred Stock remain outstanding, HCP shall not, without the consent or the affirmative vote of the holders of at least two-thirds of the shares of Series F Preferred

Stock outstanding at the time, given in person or by proxy, either in writing or at a meeting, with the Series F Preferred Stock voting separately as a class:

authorize, create or issue, or increase the authorized or issued amount of, any class or series of stock ranking prior to the Series F Preferred Stock with respect to the payment of dividends, or the distribution of assets on liquidation, dissolution or winding up;

reclassify any authorized stock of HCP into any such shares, or create, authorize or issue any obligation or security convertible into or evidencing the right to purchase any class or series of stock ranking prior to the Series F Preferred Stock with respect to the payment of dividends, or the distribution of assets on liquidation, dissolution or winding up;

repeal, amend or otherwise change any of the provisions applicable to the Series F Preferred Stock in any manner which materially and adversely affects the powers, preferences, voting power or other rights or privileges of the Series F Preferred Stock. However, an increase in the amount of authorized preferred stock, the creation or issuance of other classes or series of preferred stock or any increase in the amount of authorized shares of Series F Preferred Stock or of any other class or series of preferred stock, in each case ranking on a parity with, or junior to the Series F Preferred Stock, will not be deemed to materially and adversely affect such rights, preferences, privileges or voting powers.

The consent of the holders of Series F
Preferred Stock is not required for the taking of

any corporate action, including any merger or consolidation involving HCP or a sale of all or substantially all of the assets of HCP, regardless of the effect that such merger, consolidation or sale may have upon the rights, preferences or voting power of the holders of the Series F Preferred Stock, except as expressly set forth in the articles supplementary relating to the Series F Preferred Stock.

Rank

With respect to dividend rights and rights upon liquidation, dissolution or winding up of HCP, the Series F Preferred Stock ranks:

senior to all classes or series of common stock of HCP, and to all equity securities issued by HCP ranking junior to the Series F Preferred Stock with respect to dividend rights or rights upon liquidation, dissolution or winding up of HCP;

on a parity with the Series E
Preferred Stock and with all
equity securities issued by
HCP the terms of which
specifically provide that such
equity securities rank on a
parity with the Series F
Preferred Stock with respect to
dividend rights or rights upon
liquidation, dissolution or
winding up of HCP; and

junior to all equity securities issued by HCP the terms of which specifically provide that such equity securities rank senior to the Series F Preferred Stock with respect to dividend rights or rights upon liquidation, dissolution or winding up of HCP.

The term "equity securities" does not include convertible debt securities, which rank senior to the Series F Preferred Stock prior to conversion.

Dividends

Holders of the Series F Preferred Stock are entitled to receive, when, as, and if declared by the HCP board, out of funds legally available for the payment of dividends, cumulative preferential

annual cash dividends at the rate of 7.10% of the liquidation preference (equivalent to \$1.775 per annum per share).

Dividends on the Series F Preferred Stock are cumulative from the date of original issue and payable quarterly in arrears on or about the last day of each March, June, September and December or, if not a business day, the next succeeding business day. Any dividend payable on the Series F Preferred Stock, including dividends payable for any partial dividend period, are computed on the basis of a 360-day year consisting of twelve 30-day months.

No dividends may be declared by the HCP board or paid or set apart for payment on the Series F Preferred Stock if the terms of any agreement of HCP, including any agreement relating to its indebtedness, prohibits such a declaration, payment or setting apart for payment or provides that such declaration, payment or setting apart for payment would constitute a breach of or default under such an agreement. Likewise, no dividends may be declared by the HCP board or paid or set apart for payment if such declaration or payment is restricted or prohibited by law.

Dividends on the Series F Preferred Stock accrue, however, whether or not HCP has earnings, whether or not there are funds legally available for the payment of such dividends and whether or not such dividends are declared. Accrued but unpaid dividends on the Series F Preferred Stock do not bear interest and holders of the Series F Preferred Stock are not entitled to any dividends in excess of full cumulative dividends described above. Any dividend payment made on the Series F Preferred Stock is first credited against the earliest accrued but unpaid dividend due that remains payable.

No full dividends may be declared or paid or set apart for payment on any class or series of preferred stock ranking, as to dividends, on a parity with or junior to the Series F Preferred Stock, other than a dividend in shares of any class of stock ranking junior to the Series F Preferred Stock as to dividends and upon liquidation, for any period unless full cumulative dividends have been or contemporaneously are declared and paid or declared and set apart for such payment on the Series F Preferred Stock for all past dividend periods and the then current dividend period. When dividends are not paid in full (or full payment is not so set apart) upon the Series F Preferred Stock and the shares of any other class or series of preferred stock ranking on a parity as to dividends with the Series F Preferred Stock, including the Series E Preferred Stock, all dividends declared upon the Series F Preferred Stock and any other class or series of preferred stock ranking on a parity as to dividends with the Series F Preferred Stock are declared pro rata so that the amount of dividends declared per share of Series F Preferred Stock and such other class or

series of preferred stock shall in all cases bear to each other the same ratio that accrued dividends per share on the Series F Preferred Stock and such other class or series of preferred stock, which shall not include any accrual in respect of unpaid dividends for prior dividend periods if such preferred stock does not have a cumulative dividend, bear to each other.

Except as provided in the preceding paragraph, unless full cumulative dividends on the Series F Preferred Stock have been or contemporaneously are declared and paid or declared and set apart for payment for all past dividend periods and the then current dividend period, then, other than the payment of dividends in shares of HCP common stock or other shares of capital stock ranking junior to the Series F Preferred Stock as to dividends and upon liquidation:

no dividends may be declared or paid or set aside for payment upon the HCP common stock, or any other capital stock of HCP ranking junior to or on a parity with the Series F Preferred Stock as to dividends or upon liquidation;

no other distribution may be declared or made upon the HCP common stock, or any other capital stock of HCP ranking junior to or on a parity with the Series F Preferred Stock as to dividends or upon liquidation;

no shares of HCP common stock, or any other shares of capital stock of HCP ranking junior to or on a parity with the Series F Preferred Stock as to dividends or upon liquidation may be redeemed, purchased or otherwise acquired for any consideration by HCP, except by conversion

into or exchange for other capital stock of HCP ranking junior to the Series F Preferred Stock as to dividends and upon liquidation or for the purpose of preserving HCP's qualification as a real estate investment trust.

Liquidation Preferences

Upon any liquidation, dissolution or winding up of the affairs of HCP the holders of Series F Preferred Stock are entitled to be paid out of the assets of HCP legally available for distribution to its stockholders a liquidation preference of \$25 per share, plus an amount equal to any accrued and unpaid dividends to the date of payment, before any distribution of assets is made to holders of HCP common stock or any other class or series of capital stock of HCP that ranks junior to the Series F Preferred Stock as to liquidation rights.

In determining whether a distribution, other than upon voluntary or involuntary liquidation, by dividend, redemption or other acquisition of shares of stock of HCP or otherwise is permitted under the MGCL, no effect is given to amounts that would be needed if HCP would be dissolved at the time of the distribution, to satisfy the preferential rights upon distribution of holders of shares of stock of HCP whose preferential rights upon distribution are superior to those receiving the distribution.

Maturity; Redemption

The Series F Preferred Stock has no stated maturity, is not subject to any sinking fund or mandatory redemption. The Series F Preferred Stock is not redeemable prior to December 8, 2008. HCP is entitled, however, to purchase shares of the Series F Preferred Stock in order to preserve its status as a real estate investment trust for federal or state income tax purposes at any time. Following December 8, 2008, HCP may, at its option, redeem the Series F Preferred Stock at \$25 per share (\$195,500,000 in the aggregate), plus accrued and unpaid dividends.

Transfer and Ownership Restrictions

See " Transfer and Ownership Restrictions Relating to HCP Preferred Stock."

Transfer and Ownership Restrictions Relating to HCP Common Stock

The HCP charter contains restrictions on the ownership and transfer of voting stock of HCP

which are intended to assist HCP in complying with the requirements to continue to qualify as a real estate investment trust. Subject to limited exceptions, no person or entity may own, or be deemed to own by virtue of the applicable constructive ownership provisions of the Internal Revenue Code, more than 9.8% (by number or value, whichever is more restrictive) of the outstanding shares of HCP common stock. The HCP board may, but is in no event required to, waive the applicable ownership limit with respect to a particular stockholder if it determines that such ownership will not jeopardize HCP's status as a real estate investment trust and the HCP board otherwise decides such action would be in the best interest of HCP.

The HCP charter further prohibits:

any person from actually or constructively owning shares of stock of HCP that would result in HCP being "closely held" under Section 856(h) of the Internal Revenue Code or otherwise cause HCP to fail to qualify as a real estate investment trust (including but not limited to ownership that would result in HCP owning, actually or constructively, an interest in a tenant as described in Section 856(d)(2)(B) of the Internal Revenue Code if the income derived by HCP, either directly or indirectly, from such tenant would cause HCP to fail to satisfy any of the gross income requirements of Section 856(c) of the Internal Revenue Code); and

any person from transferring shares of HCP capital stock if such transfer would result in shares of stock of HCP being beneficially owned by fewer than 100 persons (determined without reference to any rules of attribution).

Any person who acquires or attempts or intends to acquire actual or constructive ownership of shares of HCP stock that will or may violate any of these restrictions on ownership and transfer is required to give notice immediately to HCP and provide HCP with such other information as HCP may request in order to determine the effect of the transfer on HCP's status as a real estate investment trust. Under HCP's charter, if any purported transfer of HCP stock or any other event would otherwise result in any person violating the applicable ownership limit or such other limit as permitted by the HCP board, then any such purported transfer is void and of no force or effect with respect to the purported transferee as to that number of shares of HCP stock in excess of the ownership limit or such other limit, and the transferee will acquire no right or interest in such excess shares. Any excess shares described above are transferred automatically, by operation of law, to a trust, the beneficiary of which is a qualified charitable organization selected by HCP. Such automatic transfer will be deemed to be effective as of the close of business on the business day prior to the date of such violative transfer. Within 20 days of receiving notice from HCP of the transfer of shares to the trust, the trustee of the trust is required to sell the excess shares to a person or entity who could own the shares without violating the applicable ownership limit, or such other limit as permitted by the HCP board, and distribute to the prohibited transferee an amount equal to the lesser of the price paid by the prohibited transferee for the excess shares or the sales proceeds received by the trust for the excess shares. Any proceeds in excess of the amount distributable to the prohibited transferee are distributed to the beneficiary of the trust. Prior to a sale of any such excess shares by the trust, the trustee is entitled to receive, in trust for the beneficiary, all dividends and other distributions paid by HCP with respect to such excess shares, and also is entitled to exercise all voting rights with respect to such excess shares.

Subject to Maryland law, effective as of the date that such shares have been transferred to the trust, the trustee will have the authority, at the trustee's sole discretion:

to rescind as void any vote cast by a prohibited transferee prior to the discovery by HCP that the shares have been transferred to the trust; or

to recast such vote in accordance with the desires of the trustee acting for the benefit of the beneficiary of the trust.

However, if HCP has already taken irreversible corporate action, then the trustee will not have the authority to rescind and recast such vote. Any dividend or other distribution paid to the prohibited transferee, prior to the discovery by HCP that such shares had been automatically transferred to a trust as described above, are required to be repaid to the trustee upon demand for distribution to the beneficiary of the trust. In the event that the transfer to the trust as described above is not automatically effective, for any reason, to prevent violation of the ownership limit or such other limit as permitted by the HCP board, then HCP's charter provides that the transfer of the excess shares is void ab initio.

In addition, shares of HCP common stock held in the trust shall be deemed to have been offered for sale to HCP, or its designee, at a price per share equal to the lesser of:

> the price per share in the transaction that resulted in such transfer to the trust or, in the case of a devise or gift, the market price at the time of such devise or gift, and

the market price on the date HCP, or its designee, accepted the offer. HCP will have the right to accept the offer until the trustee has sold the shares of stock held in the trust. Upon a sale to HCP, the interest of the beneficiary in the shares sold will terminate and the trustee will distribute the net proceeds of the sale to the prohibited transferee.

If any purported transfer of shares of HCP common stock would cause HCP to be beneficially owned by fewer than 100 persons, such transfer will be null and void ab initio in its entirety and the intended transferee will acquire no rights to the stock.

All certificates representing shares of HCP common stock bear a legend referring to the restrictions described above. The foregoing ownership limitations could delay, defer or prevent a transaction or a change in control of HCP that might involve a premium price for the HCP common stock or otherwise be in the best interest of HCP stockholders.

In addition, if HCP's board of directors shall, at any time and in good faith, be of the opinion that direct or indirect ownership of at least 9.9% of the voting shares of capital stock has or may become concentrated in the hands of one beneficial owner, it shall have the power:

by lot or other means deemed equitable by it to call for the purchase from any stockholder of a number of voting shares sufficient, in the opinion of HCP's board of directors, to maintain or bring the direct or indirect ownership of voting shares of capital stock of the beneficial owner to a level of no more than 9.9% of HCP's outstanding voting shares; and

to refuse to transfer or issue voting shares of capital stock to any person whose acquisition of such voting shares would, in the opinion of the board of directors, result in the direct or indirect ownership by that person of more than 9.9% of the outstanding voting shares of our capital stock.

If HCP's board of directors fails to grant an exemption from this 9.9% ownership limitation, then the transfer of shares, options, warrants, or other securities convertible into voting shares that would create a beneficial owner of more than 9.9% of the outstanding voting shares shall be deemed void ab initio, and the intended transferee shall be deemed never to have had an interest in the transferred securities. The purchase price for any voting shares of capital stock so redeemed shall be equal to the fair market value of the shares reflected in the closing sales price for the shares, if

then listed on a national securities exchange, or the average of the closing sales prices for the shares if then listed on more than one national securities exchange, or if the shares are not then listed on a national securities exchange, the latest bid quotation for the shares if then traded over-the-counter, on the last business day immediately preceding the day on which HCP sends notices of such acquisitions, or, if no such closing sales prices or quotations are available, then the purchase price shall be equal to the net asset value of such stock as determined by the board of directors in accordance with the provisions of applicable law. From and after the date fixed for purchase by the board of directors, the holder of any shares so called for purchase shall cease to be entitled to distributions, voting rights and other benefits with respect to such shares, except the right to payment of the purchase price for the shares.

Business Combination Provisions

The HCP charter requires that, except in some circumstances, "business combinations" between HCP and a beneficial holder of 10% or more of HCP's outstanding voting stock (a "Related Person") be approved by the affirmative vote of at least 90% of the outstanding voting shares of HCP. A "business combination" is defined in the HCP charter as:

any merger or consolidation of HCP with or into a Related Person;

any sale, lease, exchange, transfer or other disposition, including without limitation a mortgage or any other security device, of all or any "Substantial Part" (as defined below) of the assets of HCP, including any voting securities of a subsidiary, to a Related Person;

any merger or consolidation of a Related Person with or into HCP;

any sale, lease, exchange, transfer or other disposition of all or any Substantial Part of the assets of a Related Person to HCP;

the issuance of any securities, other than by way of pro rata distribution to all stockholders, of HCP to a Related Person; and

any agreement, contract or other arrangement providing for any of the transactions described above.

The term "Substantial Part" means more than 10% of the book value of the total assets of HCP as of the end of its most recent fiscal year ending prior to the time the determination is being made.

In addition to the restrictions on business combinations contained in HCP's charter, Maryland law also contains restrictions on business combinations. See "Comparison of Stockholder Rights Mergers, Sales of Assets and Other Transactions".

The foregoing provisions may have the effect of discouraging unilateral tender offers or other takeover proposals which stockholders might deem to be in their interests or in which they might receive a substantial premium. The HCP board's authority to issue and establish the terms of currently authorized preferred stock, without stockholder approval, may also have the effect of discouraging takeover attempts. See " Preferred Stock."

The foregoing provisions could also have the effect of insulating current management against the possibility of removal and could, by possibly reducing temporary fluctuations in market price caused by accumulations of shares of HCP common stock, deprive stockholders of opportunities to sell at a temporarily higher market price. The HCP board believes, however, that inclusion of the business combination provisions in the HCP charter may help assure fair treatment of HCP stockholders and preserve HCP's assets.

Transfer and Ownership Restrictions Relating to HCP Preferred Stock

HCP's charter contains restrictions on the ownership and transfer of preferred stock which are intended to assist HCP in complying with the requirements to maintain its status as a REIT. Subject to limited exceptions, no person or entity may own, or be deemed to own by virtue of the applicable constructive ownership provisions of the Internal Revenue Code, more than 9.8% (by number or value, whichever is more restrictive) of the outstanding shares of series E preferred stock

or series F preferred stock. HCP's board of directors may, but in no event is required to, waive the applicable ownership limit with respect to a particular stockholder if it determines that such ownership will not jeopardize HCP's status as a REIT and our board of directors otherwise decides such action would be in the best interest of HCP. The mechanics for the ownership limits on our preferred stock are similar to the mechanics related to our common stock, as described in "Transfer and Ownership Restrictions Relating to HCP Common Stock" above.

ANTI-TAKEOVER CONSIDERATIONS

Maryland law and HCP's articles of amendment and restatement, or charter, and third amended and restated bylaws contain a number of provisions which may have the effect of discouraging transactions that involve an actual or threatened change of control of HCP. These provisions of HCP's charter and bylaws include, among others, the restrictions on ownership described above. See "Comparison of Stockholder Rights".

COMPARISON OF STOCKHOLDER RIGHTS

HCP and CRP are each incorporated under the laws of the State of Maryland. Upon completion of the merger, each outstanding share of CRP common stock will be converted in part into 0.0865 of a share of HCP common stock. Consequently, the rights of CRP stockholders who receive shares of HCP common stock as a result of the merger will continue to be governed by Maryland General Corporation Law, but will be governed by HCP's charter and HCP's restated bylaws as opposed to CRP's charter and bylaws. The following discussion summarizes certain material differences between the rights of holders of CRP common stock and HCP common stock resulting from the differences in their governing documents.

This discussion does not purport to be a complete statement of the rights of holders of HCP common stock under applicable Maryland law, HCP's amended and restated articles of incorporation and HCP's restated bylaws or the rights of holders of CRP common stock under applicable Maryland law, CRP's articles of amendment and restatement and CRP's amended and restated bylaws and is qualified in its entirety by reference to the governing corporate documents of HCP and CRP and applicable law. See "Where You Can Find More Information" beginning on page 181.

Number of Directors

HCP. Under HCP's bylaws, HCP has ten directors unless increased or decreased by HCP's board of directors or its stockholders. However, this number may not be fewer than three. All of HCP's directors are elected on an annual basis. In addition, HCP has adopted a policy whereby any nominee for director who receives a number of "withhold" votes equal to greater than 50% of the shares of HCP common stock entitled to vote at the annual meeting will tender his or her resignation for consideration by HCP's Nominating and Corporate Governance Committee. The Nominating and Corporate Governance Committee will recommend to the HCP Board of Directors the action to be taken with respect to such offer of resignation.

CRP. CRP's charter provides that the number of directors cannot be less than three nor more than 15. CRP currently has 5 directors. All of CRP's directors are elected on an annual basis.

Limitation On Personal Liability and Indemnification of Directors

HCP. Maryland law permits a Maryland corporation to include in its charter a provision limiting the liability of its directors and officers to the corporation and its stockholders for money damages. However, a Maryland corporation may not limit liability resulting from actual receipt of an improper benefit or profit in money, property or services. Also, liability resulting from active and deliberate dishonesty may not be eliminated if a final judgment establishes that the dishonesty is material to the cause of action. HCP's charter contains a provision which limits the liability of directors and officers for money damages to the maximum extent permitted by Maryland law. This provision does not limit HCP's right or that of HCP's stockholders to obtain equitable relief, such as an injunction or rescission.

HCP's bylaws obligate HCP, to the maximum extent permitted by Maryland law, to indemnify and, without requiring a preliminary determination as to the ultimate entitlement to indemnification, to pay or reimburse reasonable expenses before final disposition of a proceeding to:

any present or former director or officer who is made a party to the proceeding by reason of his service in that capacity; or

any individual who, while one of HCP's directors or officers and at HCP's request, serves or has served another corporation, partnership, joint venture, trust, employee benefit plan or any other enterprise as a director, officer, partner or trustee of such corporation, partnership, joint venture, trust, employee benefit plan, or other enterprise and who is made a party to the proceeding by reason of his service in that capacity.

The bylaws authorize HCP, with the approval of HCP's board of directors, to provide indemnification and advancement of expenses to HCP's agents and employees.

Unless limited by a corporation's charter, Maryland law requires a corporation to indemnify a director or officer who has been successful, on the merits or otherwise, in the defense of any proceeding to which he is made a party by reason of his service in that capacity, or in the defense of any claim, issue or matter in the proceeding. HCP's charter does not alter this requirement.

Maryland law permits a corporation to indemnify its present and former directors and officers, among others, against: judgments; penalties; fines; settlements; and reasonable expenses actually incurred by them in connection with any proceeding to which they may be made a party by reason of their service in those or other capacities.

Maryland law does not permit a corporation to indemnify its present and former directors and officers if it is established that: the act or omission of the director or officer was material to the matter giving rise to the proceeding and was committed in bad faith or was the result of active and deliberate dishonesty; the director or officer actually received an improper personal benefit in money, property or services; or in the case of any criminal proceeding, the director or officer had reasonable cause to believe that the act or omission was unlawful.

Under Maryland law, a Maryland corporation generally may not indemnify for an adverse judgment in a suit by or in the right of the corporation. Also, a Maryland corporation generally may not indemnify for a judgment of liability on the basis that personal benefit was improperly received. In either of these cases, a court may nevertheless determine that the director or officer is fairly and reasonably entitled to indemnification under all of the relevant circumstances and may order such indemnification as it deems proper, but such indemnification is limited to expenses.

Maryland law permits a corporation to advance reasonable expenses to a director or officer. First, however, the corporation must receive a written affirmation by the director or officer of his good faith belief that he has met the standard of conduct necessary for indemnification by the corporation. The corporation must also receive a written undertaking, either by the director or officer or on his behalf, to repay the amount paid or reimbursed by the corporation if it shall ultimately be determined that the standard of conduct was not met. The termination of any

proceeding by conviction, or upon a plea of nolo contendere or its equivalent, or an entry of any order of probation prior to judgment, creates a rebuttable presumption that the director or officer did not meet the requisite standard of conduct required for indemnification to be permitted.

It is the position of the Commission that indemnification of directors and officers for liabilities arising under the Securities Act is against public policy and is unenforceable pursuant to Section 14 of the Securities Act.

CRP. Pursuant to CRP's charter, CRP is required to indemnify and hold harmless a present or former director, advisor, or affiliate of CRP against any or all losses or liabilities reasonably incurred by the indemnitee in connection with or by reason of any act or omission performed or omitted to be performed on behalf of CRP while a director, advisor or affiliate, and in such capacity, provided, that the indemnitee has determined, in good faith, that the act or omission which caused the loss or liability was in the best interests of CRP. CRP will not indemnify or hold harmless the indemnitee if: (i) the loss or liability was the result of negligence or misconduct, or if the indemnitee is an independent director, the loss or liability was the result of gross negligence or willful misconduct, (ii) the act or omission was material to the loss or liability and was committed in bad faith or was the result of active or deliberate dishonesty, (iii) the indemnitee actually received an improper personal benefit in money, property, or services, (iv) in the case of any criminal proceeding, the indemnitee had reasonable cause to believe that the act or omission was unlawful, or (v) in a proceeding by or in the right of CRP, the indemnitee is adjudged to be liable to CRP.

In addition, CRP will not provide indemnification for any loss or liability arising from an alleged violation of federal or state securities laws unless one or more of the following conditions are met: (i) there has been a successful adjudication on the merits of each count involving alleged securities law violations as to the particular indemnitee; (ii) such claims have been dismissed with prejudice on the merits by a court of competent jurisdiction as to the particular indemnitee; or (iii) a court of competent jurisdiction approves a settlement of the claims against a particular indemnitee and finds that indemnification of the settlement and the related costs should be made, and the court considering the request for indemnification has been advised of the position of the SEC and of the published position of any state securities regulatory authority in which our securities were offered or sold as to indemnification for violations of securities laws.

Pursuant to CRP's charter, CRP is required to pay or reimburse reasonable expenses incurred by a present or former director, advisor or affiliate in advance of final disposition of a proceeding if the following are satisfied: (i) the indemnitee was made a party to the proceeding by reasons of his or her service as a director, advisor or affiliate of CRP, (ii) the indemnitee provides us with written affirmation of his or her good faith belief that he or she has met the standard of conduct necessary for indemnification by CRP as authorized by CRP's charter, (iii) the indemnitee provides CRP with a written agreement to repay the amount paid or reimbursed by us, together with the applicable legal rate of interest thereon, if it is ultimately determined that the indemnitee did not comply with the requisite standard of conduct, and (iv) the legal proceeding was initiated by a third party who is not a stockholder or, if by a stockholder of the Company acting in his or her capacity as such, a court of competent jurisdiction approves such advancement. CRP's charter further provides that any indemnification, payment, or reimbursement of the expenses permitted by such charter will be furnished in accordance with the procedures in Section 2-418 of the MGCL.

Any indemnification may be paid only out of CRP's net assets, and no portion may be recoverable from CRP's stockholders.

CRP has entered into indemnification agreements with certain of its executive officers and directors. The indemnification agreements require, among other things, that CRP indemnify such executive officers and directors to the fullest extent permitted by law, and advance to such executive officers and directors all related

expenses, subject to reimbursement if it is subsequently determined that indemnification is not permitted. In accordance with these agreements, CRP must indemnify and advance all expenses reasonably incurred by such executive officers and directors seeking to enforce their rights under the indemnification agreements. CRP also must cover such executive officers and directors under its directors' and officers' liability insurance. Although, with respect to directors, these indemnification agreements offer substantially the same scope of coverage afforded by the indemnification provisions in CRP's charter and bylaws, it provides greater assurance to such directors and executive officers that indemnification will be available because these contracts cannot be modified unilaterally by CRP's board of directors or by its stockholders.

Anti-Takeover Effect of Provisions of Maryland Law and of the Charter and Bylaws

HCP. The provisions in the charter on removal of directors and business combinations, the business combinations and control share acquisition provisions of Maryland law, the advance notice provisions of HCP's bylaws and the provisions of HCP's bylaws relating to stockholder-requested special meetings may delay, defer or prevent a change of control or other transaction in which holders of some, or a majority, of the common stock might receive a premium for their common stock over the then prevailing market price or which such holders might believe to be otherwise in their best interests.

CRP. The provisions in CRP's charter on (i) removal of directors, (ii) limitations on ownership and transfer of CRP's common stock and (iii) the power to issue additional capital stock, and the advance notice provisions in CRP's bylaws, could delay, defer or prevent a transaction or a change in

control of CRP that might involve a premium price for holders of CRP's common stock or otherwise be in their best interest.

Transfer and Ownership Restrictions

To qualify as a REIT under the Code, among other things, (i) not more than 50% of the value of the REIT's outstanding stock may be owned, directly or indirectly (applying certain attribution rules), by five or fewer individuals (as defined in the Code to include certain entities) during the last half of a taxable year, (ii) the REIT's stock must be beneficially owned (without reference to any attribution rules) by 100 or more persons during at least 335 days of a taxable year of 12 months or during a proportionate part of a shorter taxable year; and (iii) certain other requirements must be satisfied.

HCP. The HCP charter contains restrictions on the ownership and transfer of voting stock of HCP which are intended to assist HCP in complying with the requirements to continue to qualify as a real estate investment trust. For a description of these restrictions, see "Description of HCP Capital Stock Transfer and Ownership Restrictions Relating to HCP Common Stock".

CRP. CRP's charter restricts the direct or indirect ownership (applying certain attribution rules) of shares of its common stock and preferred stock by any "person" (as defined in the charter) to no more than 9.8% (by value) of the outstanding shares of such common stock or 9.8% of any series of preferred stock. It is the responsibility of each person owning (or deemed to own) more than 5% of the outstanding shares of CRP's common stock or any series of outstanding preferred stock to give CRP written notice of such ownership. In addition, to the extent deemed necessary by CRP's board of directors, CRP may demand that each of its stockholders disclose to CRP in writing all information regarding the beneficial and constructive ownership (as such terms are defined in the charter) of CRP's common stock and preferred stock. However, the charter generally provides that CRP's board of directors, upon a receipt of a ruling from the Internal Revenue Service or an opinion of counsel or other evidence, representations or undertakings acceptable to such board of directors, may, in its sole discretion, waive the application of certain transfer restrictions or the ownership limit to a person if CRP's board of directors determines that such person's ownership of CRP's common stock and/or preferred stock will not jeopardize its status as a REIT under the Code. In addition, the foregoing restrictions on transfer, ownership limitations and

information requirements will not apply if CRP's board of directors determines that it is no longer in the best interests of CRP to attempt to qualify, or to continue to qualify, as a REIT under the Code.

Subject to CRP's board of director's ability to waive certain of the following restrictions in certain circumstances (as described below), transfers of shares of CRP's common stock or preferred stock or other events that would create a direct or indirect ownership of such stock that would: (i) violate the ownership limit; (ii) result in CRP's disqualification as a REIT under the Code, including any transfer that results in: (a) CRP's common stock and/or preferred stock being owned by fewer than 100 persons, or (b) CRP's being "closely held" within the meaning of Section 856(h) of the Code; or (iii) potentially jeopardize CRP's status as a REIT under the Code, will be null and void and of no effect with respect to the shares in excess of the applicable limit and, accordingly, the intended transferee, or "prohibited owner", will acquire no right or interest in such shares. Any shares in excess of an applicable limitation will be converted automatically into an equal number of shares of CRP's excess stock that will be transferred by operation of law to an unaffiliated trust for the exclusive benefit of one or more qualified charitable organizations selected by CRP. As soon as practicable after the transfer of shares to the trust, the trustee of the trust will be required to sell the shares of excess stock to a person or entity who could own the shares without violating the applicable limit and distribute to the prohibited owner an amount equal to the lesser of: (i) the proceeds of the sale; (ii) the price paid for the stock in excess of the applicable limit by the prohibited owner or, in the event that the original violative transfer was a gift or an event other than a transfer, the market price of the shares on the date of the transfer or

other event; or (iii) the pro rata amount of the prohibited owner's initial capital investment in CRP properly allocated to such shares of excess stock.

Removal of Directors

HCP. Under the terms of HCP's charter, a director of HCP may be removed by the affirmative vote of the holders of two-thirds of the outstanding shares of our voting stock or by a unanimous vote of all other directors. HCP's stockholders may elect a successor to fill any vacancy which results from the removal of a director.

CRP. Under CRP's charter, a director may be removed with or without cause by the affirmative vote of a majority of CRP's outstanding capital stock entitled to vote.

Filling Vacancies On the Board of Directors

HCP. At each annual meeting of stockholders, the election of directors shall be by a plurality of the votes cast. Holders of common stock have no right to cumulative voting for the election of directors. Consequently, at each annual meeting of stockholders, the holders of a majority of the outstanding shares of our common stock can elect all of our directors. Additionally, HCP has adopted a policy whereby any nominee for director who receives a number of "withhold" votes equal to or greater than 50% of the shares of HCP common stock entitled to vote at the annual meeting must tender his or her resignation for consideration by the Nominating and Corporate Governance Committee. A vacancy resulting from an increase in the number of directors may be filled by a majority vote of the entire board of directors or by the affirmative vote of the holders of a majority of HCP's shares then entitled to vote at an election of directors. Other vacancies may be filled by the vote of a majority of the remaining directors.

CRP. Under CRP's charter and Maryland law, each director, other than a director elected to fill the unexpired term of another director, will be elected at each annual meeting or at any special meeting of CRP's stockholders called for that purpose, by holders of a majority of CRP's shares of common stock present in person or by proxy and entitled to vote. Independent directors will nominate replacements for vacancies among CRP's independent directors.

Under CRP's charter, the term of office for each director is one year, expiring at each annual meeting of stockholders; however, nothing in CRP's charter prohibits a director from being reelected by its stockholders.

Amendment of Charter

HCP. Provisions of HCP's charter on business combinations, the number of directors and certain ownership restrictions may be amended only if approved by HCP's board of directors and by HCP's stockholders by the affirmative vote of two-thirds of all of the votes entitled to be cast by HCP's stockholders on the matter. Other amendments to HCP's charter require approval by HCP's board of directors and approval by HCP's stockholders by the affirmative vote of a majority of all the votes entitled to be cast by HCP's stockholders on the matter.

CRP. CRP's charter provides that CRP's board of directors can amend CRP's charter by a two-thirds majority from time to time if necessary in order to qualify initially or in order to continue to qualify as a REIT. Although CRP's charter so provides, Maryland law does not permit such an amendment without stockholder approval. Otherwise, the charter may be amended only by action of CRP's board of directors and the affirmative vote of a majority, and, in some cases a two-thirds majority, of holders of outstanding shares of CRP's capital stock entitled to vote. CRP's charter provides that CRP's stockholders may vote to amend the charter, terminate or dissolve CRP or remove one or more directors without the necessity for concurrence by CRP's board of directors. However,

Maryland law does not permit the amendment of corporate charters or the dissolution of a Maryland corporation without action by the board of directors.

Amendment of Bylaws

HCP. Provisions of HCP's bylaws on the number of directors and the vote required to amend the bylaws may be amended only by unanimous vote of the board of directors or by the affirmative vote of not less than 90% of all of the votes entitled to be cast by our stockholders on the matter. Other amendments to HCP's bylaws require the affirmative vote of a majority of the entire board of directors or the affirmative vote of two-thirds of all of the votes entitled to be cast by HCP's stockholders on the matter.

CRP. CRP's bylaws may be amended by the affirmative vote of either a majority of holders of all outstanding shares of CRP's capital stock entitled to vote generally in the election of directors or by the affirmative vote of a majority of CRP's directors (including a majority of the independent directors), provided that such amendments are not inconsistent with CRP's charter, and provided further that CRP's board of directors may not amend the bylaws without the affirmative vote of holders of a majority of the outstanding shares of CRP's capital stock, to the extent such amendments adversely affect the rights, preferences and privileges of CRP's stockholders.

Right to Call Special Meeting of Stockholders

HCP. HCP's bylaws provide that nominations of persons for election to the board of directors and the proposal of business to be considered by stockholders at the annual or special meeting of stockholders may be made only:

pursuant to our notice of the meeting;

by or at the direction of the board of directors; or

by a stockholder who was a stockholder at the time the notice of meeting was given and is entitled to vote at the meeting and who has complied with the advance notice

procedures, including the minimum time period, described in the bylaws.

HCP's bylaws also provide that only the business specified in HCP's notice of meeting may be brought before a special meeting of stockholders. HCP's bylaws provide that HCP's stockholders have the right to call a special meeting only upon the written request of the stockholders holding in the aggregate not less than 50% of the outstanding shares entitled to vote on the business proposed to be transacted at such meeting.

CRP. Under CRP's bylaws, special meetings of the stockholders may be called upon the written request of stockholders holding in the aggregate not less than 10% of the outstanding common stock entitled to vote at such meeting. The MGCL requires that a special meeting be called by the secretary upon the written request of stockholders entitled to cast at least 25% of the votes entitled to be cast, or such greater or less percentage (but not more than a majority) as may be provided in the charter or bylaws. Upon receipt of such a written request, either in person or by mail, stating the purpose or purposes of the meeting, CRP must provide all stockholders, within 10 days of receipt of the written request, written notice, either in person or by mail, of the meeting and its purpose. Such meeting must be held not less than 15 nor more than 60 days after such request is received, at a time and place specified in the request that must be at a time and place convenient to stockholders.

Mergers, Sales of Assets and Other Transactions

General. Under Maryland law, "business combinations" between a Maryland corporation and an interested stockholder or an affiliate of an interested stockholder are prohibited for five years after the

most recent date on which the interested stockholder becomes an interested stockholder. These business combinations include a merger, consolidation, share exchange, or, in circumstances specified in the statute, an asset transfer or issuance or reclassification of equity securities. An interested stockholder is defined as:

any person who beneficially owns ten percent or more of the voting power of the corporation's shares; or

an affiliate of the corporation who, at any time within the two-year period prior to the date in question, was the beneficial owner of ten percent or more of the voting power of the then outstanding voting stock of the corporation.

A person is not an interested stockholder under the statute if the board of directors approved in advance the transaction by which he or she would have become an interested stockholder. However, in approving the transaction, the board of directors may provide that its approval is subject to compliance, at or after the time of approval, with any terms and conditions determined by the board of directors.

After the five-year prohibition, any business combination between the Maryland corporation and an interested stockholder generally must be recommended by the board of directors of the corporation and approved by the affirmative vote of at least:

80% of the votes entitled to be cast by holders of outstanding shares of voting stock of the corporation; and

two-thirds of the votes entitled to be cast by holders of voting stock of the corporation other than shares held by the interested stockholder with whom or with whose affiliate the business combination is to be effected or which are held by an affiliate or associate of the interested stockholder.

These super-majority vote requirements do not apply if the corporation's common stockholders

receive a minimum price, as defined under Maryland law, for their shares in the form of cash or other consideration in the same form as previously paid by the interested stockholder for its shares. None of these provisions of the Maryland law will apply, however, to business combinations that are approved or exempted by the board of directors of the corporation prior to the time that the interested stockholder becomes an interested stockholder.

HCP. In addition to the restrictions on business combinations provided under Maryland law, HCP's charter also contains restrictions on business combinations. See "Description of HCP Capital Stock Business Combination Provisions."

CRP. A merger in which CRP is not the surviving entity, a combination, sale, or other disposition of all or substantially all of its assets other than in the ordinary course of business must be approved by CRP's board of directors and a majority of the holders of outstanding shares of capital stock entitled to vote. In addition, any such transaction involving an affiliate of CRP or the Advisor also must be approved by a majority of CRP's directors (including a majority of the independent directors) not otherwise interested in such transaction as fair and reasonable to us and on terms and conditions not less favorable to CRP than those available from unaffiliated third parties.

CRP's charter provides that the prohibitions and restrictions set forth in the Maryland Business Combinations Statute are inapplicable to any business combination between CRP and any person. Consequently, business combinations between CRP and interested stockholders can be effected upon the affirmative vote of the holders of a majority of the outstanding shares of capital stock entitled to vote thereon and do not require the approval of a supermajority of the outstanding shares of capital stock held by disinterested stockholders.

Dissolution

HCP. HCP's dissolution must be approved by HCP's board of directors by a majority vote of the entire board and by HCP's stockholders by the affirmative vote of a majority of all the votes entitled to be cast by HCP's stockholders on the matter.

CRP. CRP's charter provides for the voluntary termination and dissolution of CRP by the affirmative vote of the holders of a majority of the outstanding shares of common stock outstanding and entitled to vote at a meeting called for that purpose. In addition, CRP's charter permits the stockholders to terminate the status of CRP as a REIT under the Code only by the affirmative vote of the holders of a majority of the outstanding shares of common stock entitled to vote. Under CRP's charter, CRP will automatically terminate and dissolve on December 31, 2008, unless a "listing" occurs, in which event CRP will automatically become a perpetual life entity.

Control Share Acquisitions

General. Maryland law provides that "control shares" of a Maryland corporation acquired in a "control share acquisition" have no voting rights except to the extent approved by a vote of two-thirds of the votes entitled to be cast on the matter. Shares of stock owned by the acquiror, by officers or by directors who are employees of the corporation are excluded from shares entitled to vote on the matter.

"Control shares" are voting shares of stock which, if aggregated with all other shares of stock owned by the acquiror or shares of stock for which the acquiror is able to exercise or direct the exercise of voting power except solely by virtue of a revocable proxy, would entitle the acquiror to exercise voting power in electing directors within one of the following ranges of voting power:

one-tenth or more but less than one-third;

one-third or more but less than a majority; or

a majority or more of all voting power.

Control shares do not include shares the acquiring person is then entitled to vote as a result of having previously obtained stockholder approval. Except as otherwise specified in the statute, a "control share acquisition" means the acquisition of control shares.

Once a person who has made or proposes to make a control share acquisition has undertaken to pay expenses and satisfied other conditions, the person may compel the board of directors to call a special meeting of stockholders to be held within 50 days of demand to consider the voting rights of the shares. If no request for a meeting is made, the corporation may itself present the question at any stockholders meeting.

If voting rights are not approved at the meeting or if the acquiring person does not deliver an acquiring person statement as required by the statute, then the corporation may be able to redeem any or all of the control shares for fair value, except for control shares for which voting rights previously have been approved. The right of the corporation to redeem control shares is subject to certain conditions and limitations. Fair value is determined without regard to the absence of voting rights for control shares, as of the date of the last control share acquisition by the acquiror or of any meeting of stockholders at which the voting rights of control shares are considered and not approved. If voting rights for control shares are approved at a stockholders meeting and the acquiror becomes entitled to vote a majority of the shares entitled to vote, all other stockholders may exercise appraisal rights. The fair value of the shares as determined for purposes of these appraisal rights may not be less than the highest price per share paid by the acquiror in the control share acquisition. Some of the limitations and restrictions otherwise applicable to the exercise of dissenters' rights do not apply in the context of a control share acquisition.

The control share acquisition statute does not apply to shares acquired in a merger, consolidation or share exchange if the corporation is a party to the transaction or to acquisitions approved or exempted by the charter or bylaws of the corporation. HCP's charter and bylaws do not provide for any such exemption.

Duties of Directors with Respect to Unsolicited Takeovers

General. Maryland law provides protection for Maryland corporations against unsolicited takeovers by limiting, among other things, the duties of the directors in unsolicited takeover situations. The duties of directors of Maryland corporations do not require them to (a) accept, recommend or respond to any proposal by a person seeking to acquire control of the corporation, (b) make a determination under the Maryland business combination or control share acquisition statutes described above, or (c) act or fail to act solely because of the effect the act or failure to act may have on an acquisition or potential acquisition of control of the corporation or the amount or type of consideration that may be offered or paid to the stockholders in an acquisition. Moreover, under Maryland law the act of a director of a Maryland corporation relating to or affecting an acquisition or potential acquisition of control is not subject to any higher duty or greater scrutiny than is applied to any other act of a director. Maryland law also contains a statutory presumption that an act of a director of a Maryland corporation satisfies the applicable standards of conduct for directors under Maryland law.

Unsolicited Takeovers

Under Maryland law, a Maryland corporation with a class of equity securities registered under the Securities Exchange Act of 1934, as amended, and at least three independent directors may elect to be subject to certain statutory provisions relating to unsolicited takeovers which, among other things, would classify the board of directors into three classes with staggered terms of three years each and vest in the board of directors the exclusive right to determine the number of directors and the exclusive right, by the affirmative vote of a majority of the remaining directors, to fill vacancies on the board of directors, even if the remaining directors do not constitute a quorum. These statutory provisions relating to unsolicited takeovers also provide that any director elected to fill a vacancy shall hold office for the remainder of the full term of the class of directors in which the vacancy occurred, rather than the next annual meeting of directors as would otherwise be the

case, and until his successor is elected and qualified.

An election to be subject to any or all of the foregoing statutory provisions may be made in the charter or bylaws of HCP or CRP, or by resolution of the board of directors, without stockholder approval. Any such statutory provision to which either company elects to be subject will apply even if other provisions of Maryland law or the charter or bylaws provides to the contrary. Neither the charter nor the bylaws of either company provides that the company is subject to any of the foregoing statutory provisions relating to unsolicited takeovers. However, the board of directors could adopt a resolution, without stockholder approval, to elect to become subject to some or all of these statutory provisions.

If either company made an election to be subject to such statutory provisions and the board of directors was divided into three classes with staggered terms of office of three years each, the classification and staggered terms of office of the directors would make it more difficult for a third party to gain control of the board of directors since at least two annual meetings of stockholders, instead of one, generally would be required to effect a change in the majority of the board of directors.

VALIDITY OF COMMON STOCK

The validity of the shares of HCP common stock to be issued in the merger will be passed upon for HCP by Ballard Spahr Andrews & Ingersoll, LLP, Baltimore, Maryland. In addition, each of Latham & Watkins, LLP and Sullivan & Cromwell LLP will provide opinions to HCP with respect to certain tax matters.

EXPERTS

The consolidated financials statements and schedule of Health Care Property Investors, Inc. for the year ended December 31, 2005 appearing in Health Care Property Investors, Inc.'s Current Report (Form 8-K) dated June 30, 2006 and Health Care Property Investors, Inc. management's assessment of the effectiveness of internal control over financial reporting as of December 31, 2005 included in Health Care Property Investors, Inc.'s Annual Report (Form 10-K) for the year ended December 31, 2005, have been audited by Ernst & Young, independent registered public accounting firm, as set forth in their reports thereon, included therein, and incorporated herein by reference. Such consolidated financial statements and schedule and management's assessment are incorporated herein by reference in reliance upon such reports given on the authority of such firm as experts in accounting and auditing.

The financial statements of CRP as of December 31, 2005 and 2004 and for each of the three years in the period ended December 31, 2005 and management's assessment of the effectiveness of internal control over financial reporting (which is included in Management's Report on Internal Control over Financial Reporting) as of December 31, 2005 and the financial statements of the Advisor and its subsidiaries as of December 31, 2005 and 2004 and for the two years then ended included in this registration statement, have been so included in reliance on the reports of PricewaterhouseCoopers LLP, an independent registered certified public accounting firm, given on the authority of said firm as experts in auditing and accounting. The report on the financial statements of the Advisor contains an emphasis of a matter paragraph relating to the advisory services provided to CRP as described in Note 1 to the financial statements.

STOCKHOLDER PROPOSALS

CRP will hold an annual meeting of its stockholders in 2006 only if the merger is not

completed. If CRP's 2006 annual meeting is held, in order to be considered for inclusion in CRP's proxy statement for the meeting, stockholder proposals must be in writing and must be timely received by CRP at its executive offices and must have otherwise complied with Rule 14a-8 under the Exchange Act and CRP's bylaws to be eligible for inclusion in the proxy statement and form of proxy to be distributed in connection with that meeting. Any such proposal should be mailed to: CNL Retirement Properties, Inc., CNL Center II at City Commons, 420 South Orange Avenue, Suite 500, Orlando, Florida 32801, Attention: Corporate Secretary. If CRP does not complete the merger when currently anticipated, CRP intends to hold its next annual meeting in December, 2006.

WHERE YOU CAN FIND MORE INFORMATION

HCP and CRP file annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any reports, statements or other information on file with the SEC at the SEC's public reference room located at 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the public reference room. SEC filings are also available for free to the public at the SEC's website at http://www.sec.gov. HCP has filed a registration statement on Form S-4 to register with the SEC the shares of HCP common stock that CRP's stockholders will receive in the merger. This document is part of the registration statement of HCP on Form S-4 and is a prospectus of HCP and a proxy statement of CRP for the CRP special meeting.

The SEC permits HCP to "incorporate by reference" information into this proxy statement/prospectus. This means that HCP can disclose important information to you by referring to another document filed separately with the SEC. The information incorporated by reference is considered a part of this proxy statement/prospectus, except for any information superseded by information contained directly in this proxy statement/prospectus or by information contained in documents filed with or furnished to the SEC after the date of this proxy statement/prospectus that is incorporated by reference in this proxy statement/prospectus.

This proxy statement/prospectus incorporates by reference the documents set forth below that have been previously filed with the SEC. These documents contain important information about HCP and its financial condition.

HCP Securities and Exchange Commission Filings:	Period or Date Filed
Annual Report on Form 10-K	Year ended December 31, 2005
Quarterly Report on Form 10-Q	Quarter ended March 31, 2006
Current Reports on Form 8-K and 8-K/A	Filed on February 9, 2006, February 17, 2006, February 21, 2006, May 2, 2006, May 4, 2006, May 17, 2006, June 30, 2006, and August 2, 2006 (other than the portions of those documents not deemed to be
addition. HCP als	filed) so incorporates by

In addition, HCP also incorporates by reference additional documents that it may file with the SEC under Sections 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934, as amended, between the date of this proxy statement/prospectus and the date of the CRP special meeting. These documents include Annual Reports on Form 10-K, Quarterly Reports on

Form 10-Q and Current Reports on Form 8-K as well as proxy statements.

CRP is not incorporating by reference any documents or filings into this proxy statement/prospectus; however, for your information, CRP has filed the following documents:

CRP Securities and Exchange Commission Filings:	Period or Date Filed
Annual Report on Form 10-K	Year ended December 31, 2005
Quarterly Report on Form 10-Q	Quarter ended March 31, 2006
Current Reports on Form 8-K and 8-K/A	Filed on March 14, 2006, May 2, 2006, May 4, 2006 and May 4, 2006 (other than the portions of those documents not deemed to be filed)

Documents included or incorporated by reference are available from HCP and CRP, without charge, excluding all exhibits unless an exhibit has been specifically incorporated by reference into this proxy statement/ prospectus. You can obtain documents incorporated by reference in this proxy

statement/prospectus by requesting them in writing or by telephone from the appropriate company at the following addresses:

By CNL By Health Care Mail: Retirement Mail: Property Properties, Inc. Investors, Inc. 3760 Kilroy CNL Center II at City Airport Way, Commons Suite 300 420 South Long Beach, California Orange Avenue 90806 Suite 500 Attention: Edward J. Henning, Esq.

Orlando, Florida 32801

Attention: CNL By Telephone: Client Services (562) 733-5100

By Telephone: 1-866-650-0650

This proxy statement/prospectus does not constitute an offer to sell, or a solicitation of an offer to purchase, the securities offered by this proxy statement/prospectus, or the solicitation of a proxy, in any jurisdiction to or from any person to whom or from whom it is unlawful to make such offer, solicitation of an offer or proxy solicitation in such jurisdiction. Neither the delivery of this proxy statement/prospectus nor any distribution of securities pursuant to this proxy statement/prospectus shall, under any circumstances, create any implication that there has been no change in the information set forth or incorporated into this proxy statement/prospectus by reference or in the affairs of HCP or CRP since the date of this proxy statement/ prospectus. The information contained in this proxy statement/prospectus with respect to HCP and Merger Sub was provided by HCP, and the information contained in this proxy statement/prospectus with respect to CRP was provided by CRP.

INDEX TO FINANCIAL STATEMENTS

	Page
HCP Unaudited Pro Forma Condensed Consolidated	
Financial Statements	
Unaudited Pro Forma	
Condensed Consolidated	
Balance Sheet as of March 31,	
2006	F-3
Unaudited Pro Forma	
Condensed Consolidated	
Statement of Income for the	
year ended December 31, 2005	F-4
Unaudited Pro Forma	
Condensed Consolidated	
Statement of Income for the	
three months ended March 31,	
2006	F-5
Notes to Unaudited Pro Forma	
Condensed Consolidated	/
Financial Statements	F-6
CRP Financial Statements for	
the years ended December 31,	
2003, 2004 and 2005:	
Report of Independent	
Registered Certified Public	E 10
Accounting Firm	F-13
Management's Report on	
Internal Control Over Financial	D 15
Reporting Consolidated Balance Sheets	F-15 F-16
Consolidated Statements of	F-10
Income	F-17
Consolidated Statements of	1'-1/
Stockholders' Equity	F-18
Consolidated Statements of	1-10
Cash Flows	F-19
Notes to Consolidated Financial	1 1/
Statements	F-21
Schedule II Valuation and	
Qualifying Accounts	F-50
Schedule III Real Estate and	
Accumulated Depreciation	F-51
·	
CRP Financial Statements for	
the three month period ended	
March 31, 2006:	
Unaudited Condensed	
Consolidated Balance Sheets	F-62
Unaudited Condensed	
Consolidated Statements of	
Income	F-63
Unaudited Condensed	
Consolidated Statements of	
Stockholders' Equity	F-64
Unaudited Condensed	F-65
Consolidated Statements of	

	Page
Cash Flows	
Notes to Condensed	
Consolidated Financial	
Statements	F-66
Advisor Financial Statements	
for the years ended	
December 31, 2004 and 2005:	
Report of Independent Certified	
Public Accountants	F-73
Consolidated Balance Sheets	F-74
Consolidated Statements of	
Income	F-75
Consolidated Statements of	
Stockholders' Equity	F-76
Consolidated Statements of	
Cash Flows	F-77
Notes to Consolidated Financial	
Statements	F-78
Advisor Financial Statements	
for the three month period	
ended March 31, 2006:	
Unaudited Condensed	
Consolidated Balance Sheets	F-84
Unaudited Condensed	
Consolidated Statements of	
Earnings	F-85
Unaudited Condensed	
Consolidated Statements of	
Stockholders' Equity	F-86
Unaudited Condensed	
Consolidated Statements of	
Cash Flows	F-87
Notes to Condensed	
Consolidated Financial	T 05
Statements	F-88
F-1	

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The unaudited pro forma condensed consolidated financial statements presented below have been prepared based on certain pro forma adjustments to the historical consolidated financial statements of HCP, CRP and the Advisor as of and for the three months ended March 31, 2006 and for the year ended December 31, 2005. The historical consolidated financial statements of HCP are contained in its Annual Report on Form 10-K, Quarterly Reports on Form 10-Q and Current Report on Form 8-K on file with the SEC and are incorporated by reference. The historical consolidated financial statements of CRP and the Advisor are included in this document beginning on page F-13. The historical financial information with respect to HCP for the year ended December 31, 2005 has been restated to reflect as discontinued operations the results of operations of certain properties that were initially classified as discontinued operations during the first quarter of 2006. The accompanying unaudited pro forma condensed consolidated balance sheet as of March 31, 2006 has been prepared as if the merger of HCP and CRP and the Advisor and the incurrence of debt by HCP to finance the acquisitions of CRP and the Advisor had occurred as of that date.

The accompanying unaudited pro forma condensed consolidated statements of income for the three months ended March 31, 2006 and for the year ended December 31, 2005 have been prepared as if the merger and the Advisor merger had occurred as of January 1, 2005 and reflects the incurrence of debt by HCP in order to finance the acquisition of CRP, including the cash consideration needed for the merger. The allocation of the purchase price of CRP and the Advisor as reflected in these unaudited pro forma condensed consolidated financial statements has, with the assistance of independent valuation specialists, been based upon preliminary estimates of the fair value of assets acquired and liabilities assumed. In the opinion of HCP's management, all significant adjustments necessary to reflect the effects of the merger and the Advisor merger that can be factually supported within the SEC regulations covering the preparation of pro forma financial statements have been made.

A final determination of the fair values of CRP's and the Advisor's assets and liabilities, which cannot be made prior to the completion of the transactions, will be based on the actual net

tangible and intangible assets of CRP and the Advisor that exist as of the date of completion of the transactions. Consequently, amounts preliminarily allocated to goodwill and identifiable intangibles could change significantly from those used in the pro forma condensed consolidated financial statements presented below and could result in a material change in amortization of tangible and intangible assets and liabilities.

The unaudited pro forma condensed consolidated financial statements are provided for informational purposes only. The unaudited pro forma condensed consolidated financial statements are not necessarily and should not be assumed to be an indication of the results that would have been achieved had the transactions been completed as of the dates indicated or that may be achieved in the future. The unaudited pro forma condensed consolidated balance sheet does not include restructuring charges and other related liabilities expected to result from HCP's integration of CRP and the Advisor as these are not presently estimable. In addition to the completion of the valuation, the impact of ongoing integration activities, the timing of completion of the transactions and other changes in CRP's and the Advisor's net tangible and intangible assets that occur prior to completion of the transactions could cause material differences in the information presented. Furthermore, following consummation of the transaction, HCP expects to apply its own methodologies and judgments in accounting for the assets and liabilities acquired in the transaction, which may differ from those reflected in CRP's historical financial statements and the pro forma financial statements.

The unaudited pro forma condensed consolidated financial statements should be read in conjunction with the respective historical financial statements and the notes thereto of HCP, CRP and the Advisor, which are incorporated by reference or included in this document. See the section entitled "Where You Can Find More Information" for more information on where you can obtain copies of these documents.

F-2

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET March 31, 2006 (In thousands)

His	HCP torical(A)	CRP Histor Rø tlas	CRP ssification	CRP R(B) assifie A	CRP Pro Forma A djustments(C) H	Adv P Advisor For istor ialji ustn
ASSETS						
Real						
estate:						
Real	1 022 556	¢ 2 272 206 0	r	¢ 2 272 206	¢ 1 202 009 (D)	\$ \$
Less 2	+,032,330	\$ 3,273,360	Þ	\$ 3,273,380	\$ 1,203,008 (D)S	р Ф
accumulated depreciation and	n	180,953		180,953	(180,953)(D)	
_						
Net						
real	2 200 701	2 002 422		2 002 422	1 202 061	
estate 3	5,396,791	3,092,433		3,092,433	1,383,961	
D:						
Direct financing						
leases		491,239		491,239	3,318 (E)	
Loans		,>		,/	1,210 (2)	
receivable,						
net:						
Joint venture						
partners						
and						
affiliates	7,006		34,000	34,000		
Others	145,638					
Investments in and advances to						
unconsolidat	ted					
joint	40.050					
ventures Accounts	49,058					
receivable,						
net of						
allowance	13,696	19,848		19,848		
Cash and cash						
equivalents	55,957	102,204		102,204		1,427
Restricted		, •				
cash	2,694	22,767		22,767		
Intangibles,	£1.55/	110 500		110 500	127.240 (5)	~
net	51,556	110,580		110,580	137,349 (F) 113,336 (F)	54
					(110,580)(F)	2
Goodwill		5,791		5,791	(5,791)(G)	4:
Other assets,		·				
net	64,420	204,577	(34,000)	170,577	16,375 (H)	4,314
					(110,330)(H)	

	HCP Historical(A)	CRP Histor Rød l		CRP Pro Forma djustments(C)	Advisor Histor ic	
				(15,900)(H		
				(7,383)(H)	
Total						
assets	\$ 3,788,816	\$ 4,049,439	\$ \$ 4,049,439	\$ 1,404,355	\$ 5,741	\$ 120
LIABILITI AND FOCKHOLI EQUITY	DERS'					
Bank	L					
lines of credit	\$ 375,000	\$ 105,000	\$ \$ 105,000	\$ 154,525 (I)	\$	\$:
Bridge	, , , , , , , , , , , , , , , , , , , ,		,,	(105,000)(I)		
and term						
financin Senior	g			2,940,281 (I)		
unsecure	ed 1,476,215					
notes Mortgag	ge	1 252 712	1 252 712	(21,000)(1)		
debt Constru	268,654 ction	1,353,713	1,353,713	(21,990)(J)		
loans Entrance fee	e	108,472	108,472			
bonds payable		101,188	101,188			
Account payable, other liabilitie and deferred	s					
revenue	104,648	73,234	73,234	54,400 (K 86,779 (K	.)	
				(4,744)(K		
				(5,666)(K		
Total liabiliti	es 2,224,517	1,741,607	1,741,607	3,098,585	3,795	4
Minority interests		8,270	8,270			
Stockho equity:	lders'					
Preferre stock	ed 285,173					
Commo		2,642	2,642	22,853 (L) 2	,
Additio	·	2,042	2,042	(2,642)(L)		
paid-in capital	1,457,108	2,373,850	2,373,850	583,429 (L) 1,944	111
Jupitul	1,107,100	2,373,030	_,575,650	(950)(L))	
Cumula	ative			(2,373,850)(L)		(:
net income						
	/- · · · • · ·					

	HCP Historical(A)	CRP Histor Rad la	CRP ssification	CRP n R@) assifie 4	CRP Pro Forma djustments(C)	Advisor Histor ic ij	
	nulative						
	dends (2,052,009)						
	nulative ributions						
in	Toutions						
exce	ess						
of n							
inco	me	(84,641)		(84,641)	84,641 (L)	
	umulated						
othe	-						
	prehensive	7.711		7.711	(7.711)(1.)		
inco	me 1,357	7,711		7,711	(7,711)(L)		
Total							
	tholders'	2 200 562		2 200 562	(1.604.220)	1.946	11
equit	y 1,407,505	2,299,302		2,299,302	(1,694,230)	1,940	11
Tota							
and	ilities						
	kholders'						
	ty \$ 3,788,816	\$ 4,049,439	\$	\$ 4,049,439	\$ 1,404,355	\$ 5,741	\$ 12
1					. ,		

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

F-3

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF INCOME

For the year ended December 31, 2005 (In thousands, except per share data)

77.	HCP	CRP	CRP Reclass-	CRP	CRP Pro Forma Idjustments	Advisor l
п	istoricai(A)	nistorical	incations(B)	eciassinee	tajustments	Historicadj
Revenues and						
other income:						
Rental and						
other revenues	\$ 450,477	\$	\$ 321,649	\$ 321,649	\$ 44,554 (T)	\$ \$
					(1,422)(T)	
					(46,672)(T) 523 (T)	
Seniors'					323 (1)	
housing rental						
income		237,892	(237,892)			
Earned income						
from direct						
financing leases		61,202		61,202	(138)(U)	
FF&E reserve						
income		7,500	(7,500)			
Contingent rent		3,955	(3,955)			
Medical						
facilities rental						
income and other revenues		72,302	(72.202)			
Equity income		72,302	(72,302)			
(loss) from						
unconsolidated						
joint ventures	(1,123)	227		227		
Acquisition	(1,111)					
fees						6,349
Debt						
acquisition fees						13,789
Management						
fees						19,144
Interest and						
other income	26,154	4,202		4,202		3,035
	475,508	387,280		387,280	(3,155)	42,317
ļ						
Costs and						
expenses:	107 201	76 171		76 171	172 477 (1)	
Interest	107,201	76,171		76,171	173,477 (V)	
					3,418 (V)	
					10,667 (V) (5,576)(V)	
Depreciation and					(3,370)(*)	
amortization	106,342	98,446		98,446	27,421 (W))
					7,445 (W)	
Operating	58,983		26,443	26,443	484 (X)	
Seniors'						
housing						
property expenses		1,075	(1,075)			
Medical		25,368				
facilities operating		23,300	(23,300)			

1	HCP Historical(A)	CRP Historicali	CRP Reclass- fications(B)	CRP Reclassified	CRP Pro Forma Idjustment	Advisor l s Historicadj
expenses General and administrative	32,767	21,376	2,706	24,082		22,779
Asset management fees paid to related party Provision for doubtful		18,641		18,641		
accounts		3,082	(3,082)			
	305,293	244,159	(376)	243,783	217,336	22,779
Income before minority		142 121	376	142 407	(220,401)	10.529
interests Minority	170,215	ĺ		·	(220,491)	19,538
interests	(12,950)	(706)		(706)		
Earnings before income taxes	157,265	142,415	376	142,791	(220,491)	19,538
Income tax expense (benefit)	(700)		376	376		7,473
Income from continuing operations Less: preferred stock dividends		142,415		142,415	(220,491)	12,065
Income from continuing operations applicable to common share. Income from continuing operations per common share basic(AA)		\$ 142,415		\$ 142,415 \$ 0.57	\$ (220,491)	\$ 12,065 \$
Income from continuing operations per common share diluted(A	A.S i) 1.01			\$ 0.57		
Weighted-aver shares used to calculate income per common share	»:					
Basic(AA)	134,673			248,298	22,853	(BB)
Diluted(AA)	135,560			248,298	22,853	(BB)

				CRP	A
		CRP		Pro	
HCP	CRP	Reclass-	CRP	Forma	Advisor I
Historical(A	Historica	lifications(BB	Reclassifie	Adjustments	Historicadj

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

F-4

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF INCOME

For the three months ended March 31, 2006 (In thousands, except per share data)

Hi	HCP istorical(A	CRP Histo Neal a	-	CRP	CRP Pro Forma Idjustments	Advisor Historica	
Revenues and							
other income:							
Rental and							
other revenues	\$ 122,126	\$	\$ 87,998 \$	87,998	\$ 9,892 (T)	\$	\$
					(356)(T)		
					(11,115)(T)		
					131 (T)		
Seniors'							
housing rental income		62.265	(62.265)				
Earned income		62,265	(62,265)				
from direct							
financing leases		15,969		15,969	(35)(U)		
FF&E reserve		10,707		10,707	(33)(3)		
income		1,992	(1,992)				
Contingent rent		101	(101)				
Medical							
facilities rental							
income and							
other revenues		23,640	(23,640)				
Equity income							
(loss) from unconsolidated							
joint ventures	3,822	2		2			
Acquisition	3,022	2					
fees						2,581	
Debt						_,-,-	
acquisition fees						4,328	
Management							
fees						5,127	
Interest and							
other income	15,747	1,609		1,609		626	
_	141,695	105,578		105,578	(1,483)	12,662	
Costs and							
expenses:							
Interest	32,093	23,187		23,187	43,369 (V)		
					995 (V)		
					1,010 (V)		
D 1.1					(1,284)(V)		
Depreciation							
and amortization	20.670	26.052		26.052	6 055 (W)		
amoruzation	30,679	26,952		26,952	6,855 (W) 1,861 (W)		1
Operating	17,564		7,767	7,767	1,801 (W)		1
Seniors'	17,501		.,,,,,,	.,,,,,,	.21 (A)		
housing							
property							
expenses		283	(283)				
Medical facilities operating		7,484	(7,484)				

I	HCP Historical(Al	CRP Histo Neal a	CRP ssificatio R s	CRP	CRP Pro Forma Hjustments	Advisor Historic	
expenses General and							
administrative Asset	8,742	4,744	1,328	6,072		5,553	1
management fees paid to related party		5,098		5,098			
Provision for doubtful		1.500	(1.502)				
accounts		1,523	(1,523)				
	89,078	69,271	(195)	69,076	52,927	5,553	1
Income before minority interests	52,617	36,307	195	36,502	(54,410)	7,109	(1
Minority	·	·	1)3	·		7,107	(1
interests	(3,777)	(86)		(86)			_
Earnings before income taxes Income tax	48,840	36,221	195	36,416	(54,410)	7,109	(1
expense (benefit)			195	195		2,719	(2
Income from continuing							
operations Less: preferred	48,840	36,221		36,221	(54,410)	4,390) 1
stock dividends	(5,283)						
Income from continuing operations applicable to common shares	s\$ 43,557	\$ 36,221	5 \$	36,221	\$ (54,410)	\$ 4,390	\$ 1
Income from continuing operations per common share basic(AA	0.32		\$	6 0.14			
Income from continuing operations per common share diluted(A	(A) 0.32		\$	6 0.14			
Weighted-aver shares used to calculate income per common share			•				
Basic(AA)	136,040		_	257,507	22,853 (BB)	4
Diluted(AA)	136,856		•	257,507	22,853 (BB)	4

CRP Adv
Pro P
HCP CRP CRP CRP Forma Advisor For
Historical(Allisto Real assifications (R) assified djustments Historical ljustments

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

F-5

Health Care Property Investors, Inc. Notes to Unaudited Pro Forma Condensed Consolidated Financial Statements

(A)

Includes reclassification of certain HCP amounts as follows:

Balance Sheet:

Real estate assets held for sale were reclassified to "Other assets, net" from "Real estate" by HCP.

Statement of Income:

Income taxes have been reclassified from "General and administrative" to a separate line item.

(B)

Includes the following reclassifications to conform certain CRP amounts with HCP's presentation:

Balance Sheet:

Loans receivable by CRP have been reclassified to "Loans receivable, net joint venture partners and affiliates" from "Other assets, net".

Statement of Income:

"Senior housing rental income," "FF&E reserve income," "Contingent rent," and "Medical facilities rental income and other revenues" have been reclassified to "Rental and other revenues".

"Senior Housing property expenses" and "Medical facilities operating expenses" have been reclassified to "Operating".

"Provision for doubtful accounts" has been reclassified to "General and administrative".

Income taxes have been reclassified from "General and administrative" to a separate line item.

(C)

In the CRP merger, each CRP stockholder will receive 0.0865 of a share of HCP common stock and \$11.1293 in cash, without interest, for each share of CRP common stock that the stockholder owns immediately prior to the effective date of the merger.

For purposes of the unaudited pro forma condensed consolidated balance sheet presentation, the total purchase price is based on the number of shares of CRP common stock outstanding at March 31, 2006 and \$26.53, which represents the average of closing trading prices for each of the two trading days before, the day, and the two trading days after the merger was announced (April 29 and 30, and May 1, 2 and 3, 2006). The calculation of the merger consideration and total purchase price follows (dollar amounts in thousands):

Calculation of CRP purchase price Issuance of

22,852,680 shares
of HCP common
stock (based on a
conversion ratio of
0.0865) exchanged
for 264,192,829
shares of CRP
common stock
Payment of
aggregate cash
consideration
2,940,281

Total merger	
consideration	3,546,563
CRP secured debt	
and bonds	
outstanding at book	
value	1,668,373
Adjustment to	(21,990)
record CRP secured	
debt and bonds at	
fair value under	

purchase accounting	
All other CRP	
liabilities at book	
value	73,234
Adjustment to	
record CRP	
liabilities at fair	
value under	
purchase accounting	130,769
CRP minority	
interest at book	
value	8,270
Estimated fees and	
other expenses	
related to the	
merger	33,150
Total purchase	
price	\$ 5,438,369
F-6	

The calculation of the estimated fees and other expenses related to the merger is as follows (in thousands):

Advisory fees	\$ 17,850
Legal, accounting and	6,850
other fees and costs	
Share registration and	950
issuance costs	
Debt assumption fees	7,500
Total	\$ 33,150

(D)

CRP's real estate assets have been adjusted to their estimated fair values as of March 31, 2006 and CRP's historical accumulated depreciation balance is eliminated when real estate assets are recorded at fair value.

- (E)

 Adjustment reflects CRP's existing fixed rate direct financing leases at their estimated fair value based on HCP management's estimates of current market rates for direct financing leases.

 The payments on the assumed leases are
- (F)
 Adjustments to CRP's historical balance of intangible assets are as follows (in thousands):

estimated on average to be above market.

\$	137,349
	113,336
	(110,580)
_	
\$	140,105

- (G)
 Adjustment reflects the elimination of CRP's historical goodwill.
- (H)

Adjustments to CRP's historical balance of other assets are as follows (in thousands):

Deferral of issuance costs associated with		
costs associated with		
debt issued in the		
merger	\$	16,375
Elimination of		
historical straight-line		
rent balance	(110,330)
Elimination of		
historical deferred		
debt issuance costs		(15,900)
Elimination of		
historical deferred		
leasing costs		(7,383)
	\$ (117,238)

(I)

Borrowings under lines of credit and short-term borrowings are assumed to fund the cash consideration and other associated costs of the merger aggregating \$3.0 billion. HCP expects to: (i) borrow \$160.425 million on its existing lines of credit to pay the combined estimated transaction costs of the merger and the Advisor merger of \$39.05 million, the costs associated with debt issued in the merger of \$16.375 million, and the outstanding balance of CRP's line of credit of \$105 million; and (ii) obtain a 365-day bridge loan financing of \$1.3 billion and a 2-year term loan of \$1.5 billion. The portion of the bridge financing is expected to be repaid after the merger with borrowings under new credit facilities.

(J)

Adjustment reflects CRP's existing fixed rate debt at its estimated fair value based on HCP management's estimates of the interest rates that would be available to HCP for the issuance of debt with similar terms and remaining maturities. The fixed rate debt of CRP will be assumed by HCP in the merger. The interest rates on the assumed debt are considered to be below market.

F-7

Estimated market interest rates assumed to compute the fair value adjustments of CRP's existing fixed rate debt ranged from 5.95% to 6.45%.

(K)

Adjustments to CRP's historical balance of other liabilities are as follows (in thousands):

Recognition of		
liability associated		
with the acquired		
advisory agreement		
between the Advisor		
and CRP	\$	54,400
Recognition of		
liabilities associated		
with the acquired		
in-place leases that		
have below-market		
rental rates		86,779
Elimination of		
historical intangible		
liabilities		(4,744)
Elimination of		
historical deferred		
revenues		(5,666)
	_	
	\$	130,769

(L)

Adjustments represent the elimination of historical CRP balances and the issuance of shares of HCP common stock in the merger. The shares of HCP common stock issued are valued as follows (dollar amounts in thousands, except per share data):

Number of shares		
issued	2	2,852,680
Assumed price of		
shares of HCP		
common stock	\$	26.53
	_	
Value of shares		
issued	\$	606,282
Less: share		
registration and		
issuance costs		(950)
Total value of		
shares issued	\$	605,332

The total value of the shares of HCP common stock issued is presented as follows:

Par value, \$1.00		
par value per share	\$	22,853
Additional paid-in		
capital		583,429
Less: share		
registration and		
issuance costs		(950)
	_	

605,332

(M)

As discussed in this proxy statement/prospectus, HCP has also agreed to acquire the Advisor for 4,378,923 shares of HCP common stock. The merger and the Advisor merger are each conditioned upon the consummation of the other.

For purposes of the unaudited pro forma condensed consolidated balance sheet presentation, the total purchase price is based on an average trading price of HCP's common stock of \$26.53, which represents the average of the closing prices for each of the two trading days before, the day, and the two trading days after the merger was announced (April 29, 30 and May, 1, 2 and 3, 2006).

F-8

The calculation of the merger consideration and total purchase price is as follows (dollar amounts in thousands):

Calcula	ation of
Adviso	r purchase
price	
_	

Issuance of 4,378,923
shares of HCP
common stock \$116,173
All Advisor liabilities
at book value 3,795
Estimated fees and
other expenses related
to the merger 5,900

Total purchase price \$ 125,868

The calculation of the estimated fees and other expenses related to the merger is as follows:

Advisory fees	\$	500
Legal, accounting and		
other fees and costs		5,350
Share registration and		
issuance costs		50
	_	
Total	\$	5,900
		,

- (N)
 Adjustment reflects the cash of \$270,000 that the Advisor will receive in exchange for CRP shares owned by it, with a book value of \$200,000 at March 31, 2006.
- (O)

 Represents intangible assets associated with the advisory agreement between the Advisor and CRP of \$54.4 million, employee non-compete agreements of \$2.9 million, and a non-compete agreement with CNL Financial Group, CNL Real Estate Group and two other named individuals of \$21.3 million.
- (P)

 Represents the recognition of goodwill
 for the excess of the purchase price over
 the fair value of the assets acquired and
 liabilities assumed.
- (Q)
 Fixed assets and other receivables have been adjusted to their estimated fair

values.

(R)

Adjustments reflect the elimination of historical Advisor equity balances and the issuance of shares of HCP common stock in the merger. The shares of HCP common stock issued are valued as follows (dollars in thousands, except per share data):

Number of shares		
issued	4	1,378,923
Assumed price of		
shares of HCP		
common stock	\$	26.53
	_	
Value of shares		
issued	\$	116,173
Less: share		
registration and		
issuance costs		(50)
	_	
Total value of		
shares issued	\$	116,123
	_	
The total value of the		
shares of HCP		
common stock issued		
is reported as		
follows:		
Par value, \$1.00 par		
value per share	\$	4,379
Additional paid-in	-	1,0 / 2
capital		111,794
Less: share		
registration and		
issuance costs		(50)
	_	
	\$	116,123
	Ψ	110,123
F-9		
Γ-9		

(S)

Represents the elimination of the intangible for advisory agreement between the Advisor and CRP of \$54.4 million, which is an asset to the Advisor and an obligation to CRP (See Note K and Note O), and the elimination of management fees payable of \$259,000 to the Advisor from CRP.

(T)
Adjustments to rental income and other revenues are as follows (in thousands):

		Year Ended ecember 31, 2005	M H N	Three Ionths Ended March I, 2006
Recognize the				
total				
minimum				
lease				
payments				
provided				
under the				
acquired				
leases on a				
straight-line				
basis over the				
remaining				
term from the				
assumed				
merger date of				
January 1,				
2005	\$	44,554	\$	9,892
Recognize the				
amortization				
of above-				
and-below				
market lease				
intangibles		(1,422)		(356)
Remove				
CRP's				
historical				
straight-line				
rent				
adjustment		(46,672)	(11,115)
Eliminate				
CRP's				
historical				
amortization				
of above- and				
below-market				
lease		500		101
intangibles		523		131
	_		_	
	\$	(3,017)	\$	(1,448)

Year	Three
Ended	Months
December	Ended
31,	March
2005	31, 2000

(U)

Earned income from direct financing leases is adjusted to reflect market rates for interest rates at May 1, 2006 (the date that the merger agreement was executed). (See Note E)

(V)

Adjustments to interest expense are as follows (in thousands):

Year

Three

	Ended December 31, 2005	Months Ended March 31, 2006
Incremental		
increase in		
interest		
expense		
associated		
with new debt		
issued in the		
merger and		
Advisor		
merger	\$ 173,477	\$ 43,369
Increase in		
interest		
expense		
resulting from		
the		
amortization of the		
discount		
recognized at		
the merger		
date to adjust		
the assumed		
CRP secured		
debt at fair		
value	3,418	995
Increase in	10,667	1,010
interest		
expense		
resulting from		
the		
amortization		
of debt		
issuance costs		
associated		

Three Ended Months December Ended 31, March 2005 31, 2006 with the new debt issued in the merger and Advisor merger Eliminate historical debt issuance costs and loan premium amortization (5,576) (1,284) \$ 181,986 \$ 44,090

The pro forma increase in interest expense as a result of the assumed issuance of new debt in the merger is calculated using market rates management believes would have been available to HCP for the lines of credit and short-term borrowings assumed to have been issued as of May 1, 2006 (the date that the merger agreement was executed). Each ¹/₈ of 1% increase in the annual interest assumed with respect to the debt will increase pro forma interest expense by \$3.675 million for the year ended December 31, 2005 and \$919,000 for the three months ended March 31, 2006.

(W)

Adjustments to depreciation expense are as follows (in thousands):

		Year Ended ecember 31, 2005	N I	Three Ionths Ended March 31, 2006
Represents the increase in real estate depreciation expense as a result of the recording of CRP's real estate its estimated fair value at the assumed merger date of January 1, 2005	\$	27,421	\$	6,855
Represents the incremental amortization expense related to lease-up related intangible assets associated with acquired leases	_	7,445		1,861
	\$	34,866	\$	8,716

An estimated useful life of 45 years was assumed to compute the adjustment to real estate depreciation. For assets and liabilities associated with the value of in-place leases, an average remaining lease term of 9.1 years was used to compute amortization expense.

- (X)

 Operating expenses are adjusted to include amortization of below market-ground lease intangibles.
- (Y)

 Depreciation and amortization is adjusted to include the amortization of non-compete contract intangibles. A 4 year period was used to compute amortization expense.

Management of HCP expects that the merger and Advisor merger will create operational and general and administrative cost savings, including property management costs, costs associated with corporate administrative functions and executive compensation. There can be no assurance that HCP will be successful in achieving these anticipated cost savings. No estimate of these expected future cost savings has been included in the pro forma financial statements. Such adjustments cannot be factually supported within the SEC regulations governing the preparation of pro forma financial statements until such time as the operations of the companies have been fully integrated. Additionally, no adjustment has been made for anticipated property tax increases resulting from the merger since HCP expects that such increases will not be significant.

(Z)

Income taxes of the Advisor have been eliminated as a result of the Advisor merger, which is assumed as of January 1, 2005. As a condition to closing of the merger with CRP, the Advisor merger is assumed to have been consummated; at the closing of the Advisor merger, the Advisor will be merged into a Qualifying REIT Subsidiary "QRS", which assuming the merger with the Advisor was effective as of January 1, 2005, would eliminate the Advisor's income tax obligations.

(AA)

The calculations of basic and diluted earnings from continuing operations attributable to common stock per share are as follows (in thousands, except per share data):

		Year Ended December 31, 2005						Months E ch 31, 20				
		CP 1 orical		assifiec CRP	d l	Pro Forma HCP	Н	HCP l		classified CRP	1]	Pro Forma HCP
Income from												
continuing												
operations	\$ 157	7,965	\$ 1	42,415	\$	72,736	\$	48,840	\$	36,221	\$	29,309
Less: preferred												
stock dividends	(2)	1,130))			(21,130)	(5,283))			(5,283)
Earnings from												
continuing												
operations												
attributable to												
common shares	\$ 130	5,835	\$ 1	42,415		51,606	\$	43,557	\$	36,221	\$	24,026
Weighted-average												
shares used to												
calculate earnings												
per common												
share Basic	134	4,673	2	48,298		161,905	•	136,040		257,507		163,272
Incremental												
weighted-average												
effect of												
potentially dilutive												
instruments		887				887	,	816				816
		007				007		810				810
Adjusted weighted-average												
shares used to												
calculate earnings												
per common												
share Diluted	135	5,560	2	48,298		162,792		136,856		257,507		164,088
Earnings from		.,		,_,		,		,		,,		,
continuing												
operations per												
common												
share Basic	\$	1.02	\$	0.57	\$	0.32	\$	0.32	\$	0.14	\$	0.15
Earnings from												
continuing												
operations per												
common												
share Diluted	\$	1.01	\$	0.57	\$	0.32	\$	0.32	\$	0.14	\$	0.15

(BB)

The pro forma weighted-average shares outstanding are the historical weighted-average shares of HCP for the periods presented, adjusted for the assumed issuance of 27.23 million shares of HCP common stock on a weighted-average basis for the year ended December 31, 2005, and the three months ended March 31, 2006.

(CC)

Represent the elimination of acquisition, debt acquisition, management and other fees earned by the Advisor from CRP. Because acquisition fees and debt acquisition fees paid by CRP to the Advisor are capitalized by CRP, only management fees and other fees are eliminated within costs and expenses.

Report of Independent Registered Certified Public Accounting Firm

To the Board of Directors and Stockholders of CNL Retirement Properties, Inc.

We have completed integrated audits of CNL Retirement Properties, Inc. and its subsidiaries' (the "Company") December 31, 2005 and 2004 consolidated financial statements and of its internal control over financial reporting as of December 31, 2005, and an audit of its December 31, 2003 consolidated financial statements in accordance with the standards of the Public Company Accounting Oversight Board (United States). Our opinions, based on our audits, are presented below.

Consolidated financial statements and financial statement schedules

In our opinion, the consolidated financial statements listed in the accompanying index present fairly, in all material respects, the financial position of CNL Retirement Properties, Inc. and its subsidiaries at December 31, 2005 and 2004, and the results of their operations and their cash flows for each of the three years in the period ended December 31, 2005 in conformity with accounting principles generally accepted in the United States of America. In addition, in our opinion, the financial statement schedules present fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements. These financial statements and financial statement schedules are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements and financial statement schedules based on our audits. We conducted our audits of these statements in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit of financial statements includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

Internal control over financial reporting

Also, in our opinion, management's assessment, included in the acompanying Management's Report on Internal Control Over Financial Reporting that the Company maintained

effective internal control over financial reporting as of December 31, 2005 based on criteria established in Internal Control Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO), is fairly stated, in all material respects, based on those criteria. Furthermore, in our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of December 31, 2005, based on criteria established in Internal Control Integrated Framework issued by the COSO. The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting. Our responsibility is to express opinions on management's assessment and on the effectiveness of the Company's internal control over financial reporting based on our audit. We conducted our audit of internal control over financial reporting in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. An audit of internal control over financial reporting includes obtaining an understanding of internal control over financial reporting, evaluating management's assessment, testing and evaluating the design and operating effectiveness of internal control, and performing such other procedures as we consider necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinions.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for

external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (i) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ PricewaterhouseCoopers LLP

Orlando, Florida March 24, 2006

CRP'S MANAGEMENT'S REPORT ON INTERNAL CONTROL OVER FINANCIAL REPORTING

CRP's management is responsible for establishing and maintaining adequate internal control over financial reporting. Internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with U.S. GAAP.

CRP's management has assessed the effectiveness of CRP's internal control over financial reporting as of December 31, 2005, using the criteria described in Internal Control Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (or the COSO criteria). Based on its assessment, CRP's management has concluded, as of December 31, 2005, CRP maintained effective internal control over financial reporting.

PricewaterhouseCoopers LLP, an independent registered public accounting firm, has audited CRP's management's assessment of the effectiveness of CRP's internal control over financial reporting as of December 31, 2005, as stated in their report, which appears herein.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (in thousands, except per share data)

December

December

	31, 2005	31, 2004
Assets		
Real estate		
investment		
properties:		
Accounted for		
using the		
operating method,		
net	\$ 2,914,817	\$ 2,580,948
Accounted for		
using the direct		
financing method	488,683	480,051
Intangible lease		
costs, net	99,611	98,237
	3,503,111	3,159,236
Cash and cash		
equivalents	94,902	51,781
Restricted cash	21,920	34,430
Accounts and		
other receivables,		
net	23,486	20,545
Deferred costs, net	24,705	17,469
Accrued rental		
income	99,219	51,795
Other assets	52,935	11,412
Real estate held		
for sale	12,692	17,182
Goodwill	5,791	5,791
	\$ 3,838,761	\$ 3,369,641
Liabilities and stockholders' equity		

equity

Liabilities:		
Mortgages		
payable	\$1,220,190 \$	937,589
Bonds payable	98,016	94,451
Construction		
loans payable	143,560	81,508
Line of credit	75,000	20,000
Term loan		60,000
Due to related		
parties	2,386	1,632
Accounts payable		
and other		
liabilities	31,035	33,937
Intangible lease		
liability, net	4,505	3,742

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	December 31, 2005	December 31, 2004
Deferred income Security deposits	6,607 23,954	4,811 26,253
Total liabilities	1,605,253	1,263,923
Commitments and contingencies		
Minority interests	5,701	2,361
Stockholders' equity: Preferred stock, without par value Authorized and unissued 3,000 shares		
Excess shares, \$.01 par value per share Authorized and unissued 103,000 shares		
Common stock, \$.01 par value per share Authorized one billion shares, issued 260,923 and 238,485 shares, respectively, outstanding 255,527 and		
237,547 shares, respectively Capital in excess	2,555	2,376
of par value Accumulated distributions in excess of net	2,295,307	2,135,498
income Accumulated other	(74,894)	(34,517)
comprehensive income	4,839	
Total stockholders' equity	2,227,807	2,103,357
	\$ 3,838,761	\$ 3,369,641

See accompanying notes to consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME

(in thousands, except per share data)

	Years Ended December 31,					
	2005	2005 2004				
Revenues:						
Seniors'						
Housing:						
Rental income						
from operating leases	\$ 237 892	\$ 172,245	\$ 59 262			
Earned income	Ψ 231,072	Ψ 172,243	Ψ 57,202			
from direct						
financing leases	61,202	54,873	31,107			
FF&E reserve						
income	7,500	4,601	2,592			
Contingent rent	3,955	90	47			
Medical						
Facilities: Rental income						
from operating						
leases	59,048	26,225				
Tenant expense	23,010	20,223				
reimbursements	13,254	4,735				
Property	,	·				
management and						
development fees	701					
Loan interest						
income	531					
	384,083	262,769	93,008			
Expenses:						
Seniors'						
Housing						
property						
expenses	1,075	1,639	136			
Medical						
Facilities						
operating	25,368	11 224				
expenses General and	25,506	11,234				
administrative	21,376	14,740	5,462			
Asset	_1,0 / 0	1,,, 10	2,102			
management						
fees to related						
party	18,641	12,463	4,318			
Provision for						
doubtful	2.002	2.000				
accounts	3,082	3,900				
Depreciation	00 446	62.512	17 277			
and amortization	98,446	62,512	17,277			
	1/5 000	107 400	07.100			
	167,988	106,488	27,193			

Years Ended	December 31,
-------------	--------------

Operating income	216,095	156,281	65,815
Interest and other			
income	2,970	4,768	1,626
Interest and loan			
cost amortization			
expense	(76,171)	(42,783)	(9,588)
•			
Income before equity in earnings of unconsolidated entity, minority interests in income of consolidated			
subsidiaries and			
discontinued			
operations	142,894	118,266	57,853
Equity in	142,024	110,200	31,033
earnings of			
unconsolidated			
entity	227	178	11
Minority interests		170	11
in income of			
consolidated			
subsidiaries	(706)	(93)	1
5 d o 5 d d d d d d d d d d d d d d d d	(.00)	(70)	
Income from			
continuing			
operations	142,415	118,351	57,864
Income (loss)			
from			
discontinued			
operations	(6,834)	(433)	596
Net income	\$ 135,581	\$ 117,918	\$ 58,460
Net income (loss) per share of common stock (basic and diluted)			
From continuing			
operations	\$ 0.57	\$ 0.56	\$ 0.65
From			
discontinued			
operations	(0.02)		0.01
	\$ 0.55	\$ 0.56	\$ 0.66
		, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Weighted-average number of shares of common stock outstanding (basic and diluted)	248,298	210,343	88,840
*			

Years Ended December 31,

See accompanying notes to consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

Years Ended December 31, 2005, 2004 and 2003 (in thousands, except per share data)

	Common stock				
	Number of shares	Par value		AccumulatAdo distributions in excess ofm net income	other prehensive
Balance at December 31, 2002 Net income Subscriptions received for common stock through public offerings and reinvestment	44,211 S			\$ (3,955)\$ 58,460	\$ \$ 389,795 58,460
plan Retirement of	103,998	1,060	1,058,921		1,039,981
common stock	(132)	(1)	(1,211)	(1,212
Stock issuance costs			(101,299)	(101,299
Distributions declared (\$0.7067 per share)				(59,784)	(59,784
Balance at December 31, 2003 Net income Subscriptions received for common stock through public offerings and reinvestment	150,077	1,501	1,349,719	(5,279) 117,918	1,345,941 117,918
plan	88,155	882	879,386		880,268
Retirement of common stock	(685)	(7)	(6,491)	(6,498
Stock issuance costs Distributions	(003)	(7)	(87,116		(87,116
declared (\$0.7104 per share)				(147,156)	(147,156
Balance at December 31, 2004 Net income Change in fair value of cash	237,547	2,376	2,135,498	(34,517) 135,581	2,103,357 135,581 4,839 4,839

Common stock

flow hedges						
			-			
Total comprehensive income				135,581	4,839	140,420
Subscriptions received for common stock through public offerings and reinvestment						
plan	21,884	218	215,218			215,436
Retirement of common stock	(3,904)	(39)	(37,045)			(37,084)
Stock issuance costs			(18,364)			(18,364)
Distributions declared (\$0.7104 per share)				(175,958)		(175,958)
Balance at December 31, 2005	255 527 \$	2 555 \$	2.295.307 \$	k (74 894)	\$ 4 839 (\$ 2, 227 807

See accompanying notes to consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

	Years Ended December 31,				
	2005	2004	2003		
Increase (decrease) in cash and cash equivalents Operating activities: Net income\$ Adjustme to reconcile net income to net cash provided by	135,581	\$ 117,918	\$ 58,460		
operating activities: Deprecia and amortiza Minority interests	fi04,232	68,062	18,925		
in income	706	93			
Impairme provision Net rental income from above (below) market		1,883			
leases Lease incentive cost amortiza Provision for doubtful	tion 289 1	90			
Equity in earnings	3,300	3,900	138		

Years Ended December 31,

of		
unconsolidated		
entity,		
net		
of		
cash		
distributions		
received		
Changes		
in		
operating		
assets		
and		
liabilities:		
Accounts		
and		
other		
receivab(112,239)	(10,171)	(11,031)
Accrued		
rental		
and		
direct		
financing		
lease		
income (55,519)	(49.520)	(13,426)
Deferred	(15,520)	(13,120)
lease		
incentive		
costs (893)	(2,678)	
Other	(2,076)	
	(4,528)	(1,906)
	(4,326)	(1,900)
Accounts		
payable		
and		
other	5.000	c 10c
liabilities 10,365	5,209	6,426
Due		
to		
related		
parties 28	457	195
Security		
deposits		
and		
prepaid		
rents 4,339	8,840	3,026
Net		
cash		
provided		
by approxima		
operating	120 572	(0.007
activiti l:88,309	139,573	60,807
Investing		
activities:		
Investm(371,026)	(921,698)	(661,946)
in	(==,0,0)	(===,> .0)
land,		
buildings		
Juliani		

Years Ended December 31,

and		
equipment		
Investment		
in		
direct		
financing leases (278)	(50.230)	(263,330)
Investment	(50,250)	(203,330)
in		
intangible		
lease		
costs (15,044)	(50,064)	(23,220)
DASCO		
Acquisition	(204,441)	
Investment		
in note		
receivabl (16,000))	
Proceeds	,	
from		
note		
receivable		2,000
Payment		
of		
acquisition fees		
and		
	(73,124)	(53,126)
Payment	, (12,)	(==, =,
of		
deferred		
leasing		
costs (1,039)	(864)	
Increase in		
restricted		
cash 6,082	(9,448)	(13,127)
0,002	(5,1.0)	(10,127)
Net		
cash		
used		
in		
investing		
activi(417,880)	(1,309,869)	(1,012,749)
		_
Financing		
activities:		
Proceeds		
from borrowings		
on		
mortgages		
payable 305,485	315,045	170,800
Principal		
payments		
on		
mortgages	(20.05)	(12.222
payable (66,219)		(13,832)
63,367	73,618	7,402

Years Ended December 31,

Proceeds		
from		
construction		
loans		
payable		
Repayments		
of		
construction		
loans		
payable (1,315)		
Proceeds		
from		
borrowings		
on		
line		
of		
credit 115,000		71,370
Repayments		·
on		
line		
of		
credit (60,000)		(51,370)
Proceeds		, , ,
from		
term		
loan	60,000	
Repayment	,	
of		
term		
loan (60,000)		
Proceeds		
from		
11 () 111		
issuance		
issuance		
issuance of	12,063	8,203

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Financing	1		
activities Retireme			
of	III		
bonds			
payable \$	(9,057)\$	(7,736)\$	(6,589)
Payment	γ (2,021)ψ	(1,150)ψ	(0,507)
of loan			
costs	(11,707)	(10,149)	(7,523)
Contribut		, , ,	, ,
from			
minority			
interests	3,093	997	
Distributi	ions		
to			
minority			
interest	(459)	(45)	
Subscript	ions		
received			
from	215 205	000 260	1.050.001
Stockholo	ler 215,397	880,268	1,059,981
to	OHS		
	le(1175,958)	(147,138)	(59,784)
Retireme		(147,136)	(33,704)
of	III.		
common			
stock	(40,303)	(3,933)	(1,117)
Payment	(10,000)	(3,755)	(1,117)
of			
stock			
issuance			
costs	(17,254)	(89,039)	(99,309)
-			
Net			
cash			
provide	d		
by	-		
financin	ıg		
	s 272,692	1,054,987	1,078,232
•			
Net			
increase			
(decrease)			
in cash			
and cash			
equivalent	s 43,121	(115,309)	126,290
Cash and	,	(110,000)	,
cash			
equivalent	S		
at			
beginning			
of year	51,781	167,090	40,800
•			
Cash and			
cash			
equivalents	s		
at end of			
year S	94,902 \$	51,781 \$	167,090
_			·

		•				
ç	Suppleme	ental	ı			
	lisclosure					
	of cash					
	low					
i	nformatio	on:				
,	Cash					
	paid					
	during					
	the year					
	for					
	interest,					
	net of					
	capitalize	ed				
	interest	\$	75,654	\$ 39,028	\$	7,534
		_				
	Amounts	,				
	incurred	,				
	by us					
	and paid					
	by					
	related					
	parties					
	on our					
	behalf					
	were as					
	follows:					
	Acquisit	ion				
	costs	\$	210	\$ 331	\$	403
	Stock					
	issuance	;				
	issuance costs	;	4,250	18,987		17,246
			4,250	 18,987	_	17,246
				\$ 	\$	
		\$	4,250	\$ 18,987 19,318	\$	17,246 17,649
	costs	\$	4,460	\$ 	\$	
	costs	\$	4,460	\$ 	\$	
S	costs Suppleme	\$	4,460	\$ 	\$	
s	costs Suppleme chedule of	\$	4,460	\$ 	\$	
s c	Costs Supplement chedule of the con-cash	\$	4,460	\$ 	\$	
s r i	Supplement chedule of son-cash investing	\$	4,460	\$ 	\$	
s c r i	Suppleme chedule of ion-cash nvesting	\$	4,460	\$ 	\$	
s c r i a f	Suppleme chedule of ion-cash nvesting nd inancing	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash nivesting nd inancing ctivities:	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash nvesting and inancing ctivities:	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash inancing ctivities: DASCO Acquisiti	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash nvesting and inancing ctivities: DASCO Acquisiti Purchaso	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash inancing ctivities: DASCO Acquisiti Purchase accounti	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of con-cash nvesting and inancing ctivities: DASCO Acquisiti Purchase accounti	\$ con e ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash inancing ctivities: DASCO Acquisiti Purchase accounti	\$ con e ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of ion-cash inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquire	\$ con e ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of hon-cash nvesting nd inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquire Real	\$ ental	4,460	\$ 	\$	
s r i a f	Suppleme chedule of hon-cash nvesting nd inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquired Real estate	\$ donne ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of hon-cash inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquired Real estate propertions.	\$ donne ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of hon-cash nvesting nd inancing ctivities: DASCO Acquisiti Assets accounti Assets acquired Real estate propert accounti accounting the state propert accounting the state propertion accounting the state properties accounting the state pr	\$ donne ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of foon-cash nivesting and inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquired Real estate propertiaccount for	\$ donne ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of toon-cash nivesting and inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquired Real estate propert account for using	\$ aon e ing:	4,460	\$ 	\$	
s r i a f	Suppleme chedule of non-cash nvesting nd inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquire Real estate propert account for using the operati method	\$ aon e ing: d: ties ated	4,460	\$ 		
s r i a f	Suppleme chedule of for non-cash nowesting and inancing ctivities: DASCO Acquisiti Assets accounti Assets acquire Real estate propert account for using the operati	\$ aon e ing: d: ties ated	4,460	19,318		
s r i a f	Suppleme chedule of non-cash nvesting nd inancing ctivities: DASCO Acquisiti Purchase accounti Assets acquire Real estate propert account for using the operati method	\$ aon e ing: d: ties ated	4,460	19,318		

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

•		
Cash		
and		
cash		
equivalents	470	
Restricted	170	
cash	633	
Deferred	033	
costs	124	
Other	124	
assets	1,088	
Goodwill	5,487	
Goodwiii	3,467	
Total \$	\$ 222,536	\$
Liabilities		
assumed:		
Mortgages		
payable\$	\$ 10,562	¢
Construction	\$ 10,302	φ
loans		
	487	
payable Accounts	407	
payable		
and		
other	2 270	
liabilities	3,379	
Intangible		
lease	2 204	
liability	2,304	
Security	002	
deposits	893	
Total \$	\$ 17,625	\$
NT 4		
Net		
assets	¢ 204.011	Ф
acquired\$	\$ 204,911	\$
Net		
assets		
acquired,		
net of		
cash \$	\$ 204,441	\$
,	, ,	
-		
Mortgage		
loans		
assumed		
on		
properties		
acquired \$ 43,076	\$ 365,166	\$ 72,762
Bonds		
assumed		
on		
properties		
acquired \$	\$	\$ 88,511
	Ψ	<u> </u>

See accompanying notes to consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Years Ended December 31, 2005, 2004 and 2003

1. Organizational and Basis of Presentation:

Organization CNL Retirement
Properties, Inc., a Maryland corporation, was organized in December 1997 to operate as a real estate investment trust (a "REIT") for federal income tax purposes. Throughout this document, CNL Retirement Properties, Inc. and each of its subsidiaries and several consolidated partnerships and joint ventures are referred to as "we", "us" and "our." Various other wholly owned or majority owned subsidiaries are expected to be formed in the future for the purpose of acquiring or developing additional real estate properties and holding other permitted investments.

We acquire primarily real estate properties related to seniors' housing and health care facilities (the "Properties") located primarily across the United States. The Properties may include independent living, assisted living and skilled nursing facilities, continuing care retirement communities ("CCRC") and life care communities (collectively "Seniors' Housing"), medical office buildings, specialty and walk-in clinics, free standing ambulatory surgery centers, specialty or general hospitals and other types of health care-related facilities (collectively "Medical Facilities"). Seniors' Housing facilities are generally leased on a long-term, triple-net basis and Medical Facilities are generally leased on a shorter-term, gross or triple-net basis. We may provide mortgage financing loans ("Mortgage Loans"), furniture, fixture and equipment financing ("Secured Equipment Leases") and other loans to operators or developers of Seniors' Housing. In addition, we may invest up to a maximum of 5% of total assets in equity interests in businesses, including those that provide services to or are otherwise ancillary to the retirement and health care industries. We operate in one business segment, which is the ownership, development, management and leasing of health care-related real estate. At December 31, 2005, we owned 184 Seniors' Housing facilities, 73 Medical Facilities, including a specialty hospital, 2 walk-in clinics, and 4 Seniors' Housing facilities and a parcel of land that we hold for sale.

In August 2004, we acquired a 55% controlling interest in The DASCO Companies, LLC ("DASCO"), a development and property management company that managed forty-eight of

our Medical Facilities, including two of our walk-in clinics and was developing five of our Medical Facilities at December 31, 2005. DASCO also provides development and property management services to unrelated third parties.

We retained CNL Retirement Corp. (the "Advisor") as our advisor to provide management, acquisition, advisory and administrative services relating to our Properties, Mortgage Loans, Secured Equipment Lease program, other loans and other permitted investments pursuant to an advisory agreement that was renewed pursuant to a Renewal Agreement effective May 3, 2005 for a one-year term and was amended by an amendment to the Renewal Agreement on July 13, 2005 (the "Advisory Agreement").

Basis of Presentation The accompanying consolidated financial statements include the accounts of our wholly owned subsidiaries, DASCO and other entities in which we own a majority and controlling interest. Interests of unaffiliated third parties in less than 100% owned and majority controlled entities are reflected as minority interests. All significant inter-company balances and transactions have been eliminated in consolidation.

2. Summary of Significant Accounting Policies:

Use of Estimates The preparation of financial statements in conformity with accounting principles generally accepted in the U.S. ("GAAP") requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Actual results could differ from those estimates.

Cash equivalents, accounts and other receivables, and accounts payable and other liabilities are carried at amounts which approximate their fair values because of the short-term nature of these instruments.

Investment Properties and Lease Accounting Seniors' Housing Properties are leased on a long-term (generally 15 years), triple-net basis whereby the tenants are responsible for all operating expenses relating to the Property, including property taxes, maintenance, repairs, utilities and insurance, as well as capital expenditures that may be reasonably necessary to maintain the leasehold in a manner that allows operation for its intended purpose. Seniors' Housing leases generally provide for minimum and contingent rent and contain renewal options from 5 to 20 successive years subject to the same terms and conditions as the initial term. Medical Facilities are leased on either a triple-net or gross basis, generally have initial lease terms of 5 to 15 years and are generally subject to renewal options. In addition, Medical Facilities gross leases provide for the recovery of a portion of the properties' operating expenses from the tenants. Substantially all Seniors' Housing and Medical Facilities leases require minimum annual rents to increase at predetermined intervals during the lease terms. For the years ended December 31, 2005, 2004 and 2003, our tenants paid \$32.3 million, \$21.5 million and \$8.1 million, respectively, in property taxes on our behalf. The leases are accounted for using either the operating or direct financing method.

Operating method For leases accounted for as operating leases, Properties are recorded at cost.

Minimum rent payments contractually due under the leases are recognized as revenue on a straight-line basis over the initial lease terms so as to produce constant periodic rent recognition over the lease terms. The excess of rents recognized over amounts contractually due are included in accrued rental income in the accompanying financial

statements. Buildings, land improvements and equipment are depreciated on the straight-line method over their estimated useful lives of 39 to 40 years, 15 years and 3 to 7 years, respectively. Tenant improvements are depreciated over the initial lease term. Expenditures for ordinary maintenance and repairs are charged to operations as incurred, while significant renovations and enhancements that improve and/or extend the useful life of an asset are capitalized and depreciated over the estimated useful life.

Direct financing method For leases accounted for as direct financing leases, future minimum lease payments are recorded as a receivable. The difference between the rents receivable and the estimated residual values less the cost of the Properties is recorded as unearned income. Unearned income is deferred and amortized to income over the lease terms to provide a constant rate of return. Investments in direct financing leases are presented net of unamortized unearned income. Direct financing leases have initial terms that range from 10 to 35 years and provide for minimum annual rent. Certain leases contain provisions that allow the tenants to elect to purchase the Properties during or at the end of the lease terms for our aggregate initial investment amount plus adjustments, if any, as defined in the lease agreements. Certain leases also permit us to require the tenants to purchase the Properties at the end of the lease terms for the same amount.

Impairment of Long-Lived

Assets We evaluate our Properties and other long-lived assets on a quarterly basis, or upon the occurrence of significant changes in operations, to assess whether any impairment indications are present that affect the recovery of the carrying amount of an individual asset. We compare the sum of expected undiscounted cash flows from the asset over its anticipated holding period, including the asset's estimated residual value, to the carrying value. If impairment is indicated, a loss is provided to reduce the carrying value of the property to the lower of its cost or its estimated fair value.

Real estate held for sale Based on the ongoing evaluation of our Properties, we have determined to hold certain Properties for sale (see Note 10). Statement of Financial Accounting Standards No. 144, "Accounting for Impairment or Disposal of Long-Lived Assets," ("SFAS 144") requires that long-lived assets to be disposed of be reported at the lower of their carrying amount or their fair value less costs to dispose. SFAS 144 also requires us to stop depreciating these assets at the time the assets are classified as discontinued operations. In accordance with SFAS 144 we have reclassified the assets and operating results from certain Seniors' Housing Properties as discontinued operations, restating previously reported results to reflect the reclassification on a comparable basis. These reclassifications had no effect on reported equity or net income.

When a Property is sold, the related costs and accumulated depreciation, plus any accrued rental income, are removed from the accounts and any gain or loss from sale is reflected in income.

Intangible Lease Costs In accordance with Statement of Financial Accounting Standards No. 141, "Business Combinations" ("SFAS 141"), we allocate the purchase price of acquired Properties to tangible and identified intangible assets based on their respective fair values. The allocation to tangible assets (building and land) is based upon management's determination of the value of the Property as if it were vacant using discounted cash flow models similar to those used by independent appraisers. The allocation to intangible assets is based upon factors considered

by management including an estimate of carrying costs during the expected lease-up periods considering current market conditions and costs to execute similar leases. Additionally, the purchase price is allocated to the above- or below-market value of in-place leases and the value of customer relationships. The value allocable to the above- or below-market component of the acquired in-place lease is determined based upon the present value (using a discount rate which reflects the risks associated with the acquired leases) of the difference between (i) the contractual amounts to be paid pursuant to the lease over its remaining term, and (ii) management's estimate of the amounts that would be paid using fair market rates over the remaining term of the lease. The amounts allocated to above-market leases are included in intangible lease costs and are amortized to rental income over the remaining terms of the leases acquired with each Property. The amounts allocated to below-market lease values are included in an intangible lease liability and amortized to rental income over the remaining term of the associated lease, including below-market lease extension, if any.

The total amount of other intangible assets acquired is further allocated to in-place lease origination costs and customer relationship values based on management's evaluation of the specific characteristics of each tenant's lease and our overall relationship with that respective tenant. Characteristics considered by management in allocating these values include the nature and extent of the credit quality and expectations of lease renewals, among other factors.

Cash and Cash Equivalents All highly liquid investments with an original maturity of three months or less when purchased are considered cash equivalents. Cash and cash equivalents consist of demand deposits at commercial banks and money market funds (some of which are backed by government securities). Cash equivalents are stated at cost plus accrued interest, which approximates market value.

Cash accounts maintained in demand deposits at commercial banks and money market funds may exceed federally insured levels; however, we have not experienced any losses in such accounts. We believe we are not exposed to any significant credit risk on cash and cash equivalents.

Accounts and Other Receivables Accounts and other receivables consist primarily of lease payments contractually due from tenants. On a monthly basis, we review the contractual payments versus the actual cash received. When we identify delinquencies, an estimate is made as to the amount of provision for loss related to doubtful accounts, if any, that may be needed based on our review of Property specific circumstances, including the analysis of the Property's operations and operating trends, current economic conditions and tenant payment history. At December 31, 2005 and 2004, we had reserves for doubtful accounts and other receivables of \$7.2 million and \$3.9 million, respectively. The total amount of the reserves, which represent the cumulative provisions less write-offs of uncollectible rent, if any, are recorded against accounts and other receivables in our consolidated balance sheets.

Deferred Loan Costs Loan costs are capitalized and are amortized as interest over the terms of the respective loan agreements on a basis which approximates the effective interest method. Unamortized deferred loan costs are expensed when the associated debt is retired before maturity.

Goodwill In connection with the acquisition of DASCO, we allocated \$5.8 million to goodwill, which represented the excess of the purchase price plus closing costs paid over the fair market value of the tangible assets acquired in the business acquisition (see Note 19). In accordance with SFAS 141, and Statement of Financial Accounting Standards No. 142 "Goodwill and Other Intangible Assets," goodwill is not amortized but is tested quarterly for impairment. If quoted market prices are not available for the impairment analysis, we use other valuation techniques that involve measurement based on projected net earnings of the underlying reporting unit.

Investment in Unconsolidated Entity We own a 9.90% interest in CNL Plaza, Ltd., a limited partnership that owns an office building located in Orlando, Florida, in which the Advisor and its affiliates lease office space. Our investment in the partnership is accounted for using the equity method because we have significant influence.

Development Costs Development costs, including interest, real estate taxes, insurance and other costs incurred in developing new Properties, are capitalized during construction. Upon completion of construction, development costs are depreciated on a straight-line basis over the useful lives of the respective assets.

Capitalized Interest Interest, including loan costs for borrowings used to fund development and construction, is capitalized as construction in progress and allocated to the respective assets. For the years ended December 31, 2005 and 2004, interest of \$4.8 million and \$0.7 million, respectively, was capitalized to construction in progress.

Deferred Income Rental income contractually due under leases from Properties that are under development are recorded as deferred income. Upon completion of construction, deferred income is amortized to revenue on a straight-line basis over the remaining lease term.

Bonds Payable Our two CCRCs hold non-interest bearing life care bonds payable to certain residents of the CCRCs. Generally, the bonds are refundable to the resident or to the resident's estate upon termination or cancellation of the CCRC agreement. One of our other Seniors' Housing facilities requires that certain residents of the facility post non-interest bearing occupancy fee deposits that are refundable to the resident or the resident's estate the earlier of the re-letting of the unit or after two years of vacancy. Proceeds from the issuance of new bonds are used to retire existing bonds. As the maturity of these obligations is not determinable, no interest is imputed.

Minority Interests Minority interests in consolidated real estate partnerships represents the minority partners' share of the underlying net assets of the consolidated real estate partnerships. Net income or net losses, contributions and distributions for each partnership are allocated to the minority partner in accordance with the partnership agreement.

FF&E Reserve Income A furniture, fixture and equipment ("FF&E") cash reserve has been established with substantially all of the Seniors' Housing lease agreements. In accordance with the agreements, the tenants deposit funds into restricted FF&E cash reserve accounts and periodically use these funds to cover the cost of the replacement, renewal and additions to FF&E. In the event that the FF&E reserve is not sufficient to maintain the Property in good working condition and repair, we may make fixed asset expenditures, in which case annual rent would be increased.

All funds in the FF&E reserve accounts held by us, including the interest earned on the funds and all property purchased with the funds from the FF&E reserve are our assets; therefore, we recognize the FF&E reserve payments as income. FF&E purchased with FF&E reserve funds that improve or extend the useful lives of the respective Properties are capitalized. All other FF&E costs are recorded as property operating expenses in the accompanying consolidated financial statements. For a number of our leases, FF&E reserve accounts are held by each tenant until the end of the lease term at which time all property purchased with funds from the FF&E reserve accounts become our

assets.

With respect to 13 Properties subject to direct financing leases, FF&E reserve accounts are held by each tenant and all property purchased with funds from the FF&E accounts will remain the property of the tenants. Accordingly, we do not recognize FF&E reserve income relating to these direct financing leases.

Derivative Instruments and Hedging Activities Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities" ("SFAS 133"), as amended and interpreted, establishes accounting and reporting standards for derivative instruments, including certain derivative instruments embedded in other contracts, and for hedging activities. As required by SFAS 133, we record all derivatives on the balance sheet at fair value. We do not invest in derivatives for trading purposes. Our objective in using derivatives is to limit exposure to changes in interest rates on our debt obligations. To accomplish this objective, we use interest rate swaps to hedge the variable cash flows associated with existing variable-rate debt. These interest rate swaps, designated as cash flow

hedges, involve the receipt of variable-rate amounts in exchange for fixed-rate payments over the life of the agreements without exchange of the underlying principal amount.

For derivatives designated as cash flow hedges, the effective portion of changes in the fair value of the derivative is initially reported in other comprehensive income and subsequently reclassified to earnings when the hedged transaction affects earnings, and the ineffective portion of changes in the fair value of the derivative is recognized directly in earnings. We assess the effectiveness of each hedging relationship by comparing the changes in fair value or cash flows of the derivative hedging instrument with the changes in fair value or cash flows of the designated hedged item or transaction. (See Note 12.)

Income Taxes We are taxed as a REIT under Sections 856 through 860 of the Internal Revenue Code of 1986, as amended, and related regulations. As a REIT, we generally will not be subject to federal corporate income taxes on amounts distributed to stockholders, providing we distribute at least 90% of our REIT taxable income and meet certain other requirements for qualifying as a REIT. At December 31, 2005, 2004 and 2003, we were in compliance with all REIT requirements and were not subject to federal income taxes. State and local taxing authorities apply their own rules and regulations that adjust the federal taxable income of REITs to arrive at state taxable income. For the year ended December 31, 2005, we determined that the REIT would be subject to state and local income taxes of approximately \$40,000.

We hold five wholly owned taxable REIT subsidiaries which enable us to engage in non-REIT activities. Taxable REIT subsidiaries are subject to federal, state, and local income taxes. For the year ended December 31, 2005, we determined that the taxable REIT subsidiaries collectively would be subject to federal income taxes and applicable state income taxes of approximately \$0.4 million. We recorded a tax provision on each of the taxable REIT subsidiaries for their respective share of the estimated federal, state, and local income taxes. This provision is included in general and administrative expenses in the accompanying consolidated statements of income.

Income Per Share Basic income per common share is calculated based upon net income (income available to common stockholders) divided by the weighted-average number of shares of common stock outstanding during the period. As of

December 31, 2005, 2004 and 2003, we did not have any potentially dilutive common shares.

Reclassifications Certain items in the prior periods' financial statements have been reclassified to conform to the 2005 presentation, including those related to our real estate held for sale (see Note 10). These reclassifications had no effect on reported equity or net income.

Recently Issued Accounting Pronouncements In May 2005, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards No. 154, "Accounting Changes and Error Corrections A Replacement of APB Opinion No. 20 and SFAS No. 3" ("SFAS 154"). SFAS 154 changes the requirements for the accounting and reporting of a change in accounting principle by requiring that a voluntary change in accounting principle be applied retrospectively with all prior periods' financial statements presented on the new accounting principle, unless it is impracticable to do so. SFAS 154 also requires that a change in depreciation or amortization for long-lived, non-financial assets be accounted for as a change in accounting estimate effected by a change in

accounting principle and corrections of errors in previously issued financial statements should be termed a "restatement." SFAS 154 is effective for accounting changes and corrections of errors made in fiscal years beginning after December 15, 2005. We believe that the adoption of SFAS 154 will not have a material effect on our consolidated financial statements.

In March 2005, the FASB issued Interpretation No. 47, "Accounting for Conditional Asset Retirement Obligations," ("FIN 47") an interpretation of Statement of Financial Accounting Standards No. 143, "Accounting for Asset Retirement Obligations" ("SFAS 143"). FIN 47, which is effective for fiscal years ended after December 15, 2005, clarifies that the term "conditional asset retirement obligation," as used in SFAS 143 refers to a legal obligation to perform an asset retirement activity in which the timing and/or method of settlement are conditional on a future event that may or may not be within the control of the entity. The obligation to perform the asset retirement activity is unconditional even though uncertainty exists about the timing and/or method of settlement. FIN 47 requires a company to recognize the liability for the fair value of the conditional asset retirement obligation if the fair value of the liability can be reasonably estimated. Any liability accrued is offset by an increase in the value of the asset. Adoption of FIN 47 did not have a material impact on our financial statements.

3. Public Offerings:

Upon formation in December 1997, we received an initial capital contribution of \$200,000 for 20,000 shares of common stock from the Advisor. From our inception through December 31, 2005, we have made five public offerings and received subscriptions as follows (in thousands):

		Offering		Subse	criptions
Offering	Date Completed	Shares(b)	Amount	Shares(c)	Amount
Initial Offering	September 2000	15,500	\$ 155,000	972	\$ 9,719
2000 Offering	May 2002	15,500	155,000	15,500	155,000
2002 Offering	April 2003	45,000	450,000	45,000	450,000
2003 Offering	April 2004	175,000	1,750,000	156,793	1,567,925
2004 Offering	Open(a)	400,000	4,000,000	41,548	415,485

Offering	Subscriptions
651,000 \$ 6,510,000	259,813 \$ 2,598,129

- (a) 2004 Offering will close on or before March 26, 2006.
- (b)
 Includes reinvestment plan shares of 500 in each of the Initial and 2000 Offerings, 5,000 in the 2002 Offering, 25,000 in the 2003 Offering and 15,000 in the 2004 Offering.
- Includes reinvestment plan shares of 5 in the Initial Offering, 42 in the 2000 Offering, 129 in the 2002 Offering, 1,728 in the 2003 Offering and 8,749 in the 2004 Offering.

The price per share of all of the equity offerings of our common stock has been \$10.00 per share, with the exception of (i) shares purchased pursuant to volume or other discounts and (ii) shares purchased through our reinvestment plan, which are currently priced at \$9.50 per share.

In July 2004, the stockholders approved a resolution to amend our Amended and Restated Articles of Incorporation to increase the number of authorized shares of common stock from 450 million to one billion.

We incurred offering expenses, including selling commissions, marketing support fees, due diligence expense reimbursements, filing fees, legal, accounting, printing and escrow fees, which have been deducted from the gross proceeds of the offerings. Offering expenses together with selling commissions, marketing support fees and due diligence expense reimbursements will not exceed 13% of the proceeds raised in connection with our public offerings. Under our first four public offerings ("Prior Offerings"), the Advisor and its affiliates were entitled to selling commissions of 7.5%, a marketing support fee of 0.5% and an acquisition fee of 4.5% of gross offering and debt proceeds. Under the 2004 Offering, the Advisor and its affiliates are entitled to selling commissions of 6.5%, a marketing support fee of 2.0% and acquisition fees equal to 3.0% of gross offering and loan proceeds from permanent financing for the period from May 3, 2005 through December 31, 2005 (4.0% of gross proceeds and loan proceeds for the period from May 14, 2004 through May 2, 2005).

During the years ended December 31, 2005, 2004 and 2003, we incurred \$18.4 million, \$87.1 million and \$101.3 million, respectively, in offering costs, including \$14.1 million, \$68.8 million and \$85.1 million, respectively, in selling commissions and marketing support fees. These amounts are treated as stock issuance costs and charged to stockholders' equity.

4. Investment Properties:

Accounted for Using the Operating
Method Properties subject to operating leases
consisted of the following at December 31 (dollars
in thousands):

	2005	2004
Land and land	\$ 345,936	\$ 311,198
improvements		
Buildings and	2,523,724	2,118,086
building		
improvements		
Tenant	96,084	62,641
improvements		
Equipment	75,700	65,936
	3,041,444	2,557,861
Less	(157,746	
accumulated		
depreciation		
	2,883,698	2,484,145
Construction	31,119	96,803
in progress		
	\$ 2,914,817	\$ 2,580,948
Number of		
Properties(1)(2	2):	
Seniors'		
Housing:		
Operating	150	130
Under	1	3
construction		
	151	133
Medical	131	133
Facilities:		
Operating	68	49
Under	5	3
construction		
	73	52
	224	185

(1) At December 31, 2005, excludes four Seniors' Housing facilities and a parcel

of land held for sale. At December 31, 2004, excludes four Seniors' Housing facilities held for sale.

(2)

At December 31, 2005, includes 26 Medical Facilities and one Seniors' Housing facility subject to long-term ground lease agreements. At December 31, 2004, includes 20 Medical Facilities subject to long-term ground lease agreements.

For the years ended December 31, 2005, 2004 and 2003, we recognized \$46.7 million, \$40.4 million and \$13.2 million, respectively, of revenue from the straight-lining of lease revenues over current contractually due amounts. These amounts are included in rental income from operating leases in the accompanying consolidated statements of income.

Future minimum lease payments contractually due under the noncancellable operating leases at December 31, 2005, exclusive of renewal option periods and contingent rents, were as follows (in thousands):

2006	\$ 269,257
2007	272,121
2008	274,547
2009	275,613
2010	274,695
Thereafter	2,475,678
	\$ 3,841,911

Accounted for Using the Direct Financing
Method The components of net investment in direct
financing leases consisted of the following at
December 31 (dollars in thousands):

	2005	2004
Minimum lease payments receivable	\$ 1,477,576	\$ 1,529,171
Estimated residual values	449,099	449,099
Less unearned income	(1,437,992)	(1,498,219)
Net investment in direct financing leases	\$ 488,683	\$ 480,051
Properties subject to direct financing leases	33	33

Lease payments due to us relating to six land-only direct financing leases with a carrying value of \$131.9 million are subordinate to first mortgage construction loans with third parties entered into by the tenants to fund development costs related to the Properties.

Future minimum lease payments contractually due on direct financing leases at December 31, 2005, were as follows (in thousands):

2006	\$ 54,235
2007	55,224

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

08 56,314 09 58,126 10 60,155 ereafter 1,193,522 \$ 1,477,576
10 60,155 ereafter 1,193,522 \$ 1,477,576
\$ 1,477,576
\$ 1,477,576
<u> </u>
<u> </u>
F-30

5. Intangible Lease Costs:

Intangible lease costs included the following at December 31 (in thousands):

	2005	2004
Intangible lease origination costs:		
In-place lease costs	\$ 103,736	\$ 88,740
Customer relationship values	12,152	11,698
	115,888	100,438
Less accumulated amortization	(23,643)	(9,934)
	92,245	90,504
Above-market lease values	9,744	8,475
Less accumulated amortization	(2,378)	(742)
	7,366	7,733
	\$ 99,611	\$ 98,237

Above-market lease values are amortized to rental income over the remaining terms of the leases acquired in connection with each applicable Property acquisition. Above-market lease amortization charged against rental income from operating leases in the accompanying consolidated statements of income was \$1.7 million, \$0.7 million and \$0, respectively, for the years ended December 31, 2005, 2004 and 2003.

The estimated amortization expense for in-place lease costs and customer relationship values, and the estimated rental income amortization for above-market lease values at December 31, 2005, were as follows (in thousands):

	-	Custom&bo elationship values	lease	ŧ
2006	\$ 10,712	\$ 1,902 5	\$ 1,523	
2007	8,849	1,294	1,225	
2008	7,873	1.161	1.051	

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	In-place Custom&bove-market lease relationship lease costs values values			
2009	7,147	984	934	
2010	6,389	776	655	
Thereafter	42,770	2,388	1,978	
	\$ 83,740	\$ 8,505	\$ 7,366	
	F-31			

6. Restricted Cash:

Restricted cash included the following at December 31 (in thousands):

		2005	2004
TD 6 .	ф	4.000	Ф 12 21 4
Transfer agent	\$	4,980	\$ 13,214
escrows			
Horizon Bay		3,109	9,537
tenant rent			
deposit			
FF&E reserves		4,509	4,894
Lender escrow		6,908	3,808
reserves			
Property			1,950
acquisition			
deposits			
Other		2,414	1,027
	_		
	Φ.	24 020	A. 2.4.420
	\$	21,920	\$ 34,430

7. Accounts and Other Receivables:

Accounts and other receivables included the following at December 31 (in thousands):

	2005	2004
Rental	\$ 27,301	\$ 21,790
revenues		
receivable		
Other	3,385	2,655
receivables		
	30,686	24,445
Allowance for	(7,200)	(3,900)
doubtful		
accounts		
	\$ 23,486	\$ 20,545

At December 31, 2005 and 2004, past due rents aggregated \$14.8 million and \$10.7 million, respectively. The provision for doubtful accounts for the years ended December 31, 2005, 2004 and 2003, was \$3.4 million, \$3.9 million and \$0, respectively, which included \$3.1 million, \$3.9 million and \$0, respectively, from continuing operations and \$0.3 million, \$0 and \$0, respectively, from discontinued operations. Additionally, during 2005, accounts receivable of \$0.1 million related to certain Medical Facilities tenants were written off.

8. Deferred Costs:

Deferred costs included the following at December 31 (in thousands):

	2005	2004
Financing costs	\$ 26,679	\$ 17,989
Leasing commissions	989	523
Other lease costs	767	341
	28,435	18,853
Less accumulated amortization	(8,503)	(5,408)
	19,932	13,445
Lease incentives	5,153	4,114
Less accumulated amortization	(380)	(90)
	4,773	4,024
	\$ 24,705	\$ 17,469

Lease incentive costs are amortized to rental income over the terms of the leases. Lease incentive cost amortization charged against rental income was \$0.3 million, \$0.1 million and \$0, for the years ended December 31, 2005, 2004 and 2003, respectively.

9. Other Assets:

Other assets included the following at December 31 (in thousands):

2005	2004
¢ 17 000	¢.
\$ 10,000	ф
10,601	
7,633	2,972
((20	0.42
0,038	942
4,950	6,400
4 920	
4,839	
	\$ 16,000 10,601 7,633 6,638

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	2005	2004
Other	2,274	1,098
	\$ 52,935	\$ 11,412

(1)

In August 2005, we entered into an agreement to provide an affiliate of the Cirrus Group, LLC ("Cirrus") with an interest only, five-year, senior secured term loan under which up to \$85.0 million (plus capitalized interest) may be borrowed to finance the acquisition, development, syndication and operation of new and existing surgical partnerships ("Senior Secured Term Loan"). Certain of these surgical partnerships are tenants in the Medical Facilities acquired from Cirrus. During the first 48 months of the term, interest at a rate of 14.0%, will accrue, of which 9.5% will be payable monthly and the balance of 4.5% will be capitalized; thereafter, interest at the greater of 14.0% or LIBOR plus 9.0% will be payable monthly. The loan is subject to equity contribution requirements and borrower financial covenants that will dictate the draw down availability, is

collateralized by all of the assets of the borrower (comprised primarily of interest in partnerships operating surgical facilities in premises leased from a Cirrus affiliate) and is guaranteed up to \$50.0 million through a combination of (i) a personal guarantee of up to \$13.0 million by a principal of Cirrus and (ii) a guarantee of the balance by other principals of Cirrus under arrangements for recourse limited only to their interests in certain entities owning real estate. The carrying value of the loan at December 31, 2005, approximated its fair value.

In connection with the Senior Secured Term Loan, we received stock warrants which are exercisable into a 10% to 15% ownership interest of the borrower. The stock warrants are exercisable at the earlier of an event of default or the full repayment of the Senior Secured Term Loan and expire in September 2015.

(2)

Represents rental revenue receivable reclassified from accounts receivable to other assets in accordance with certain lease provisions.

10. Real Estate Held For Sale:

As of December 31, 2005, real estate held for sale included four Seniors' Housing facilities with an aggregate net carrying value of \$9.4 million and a 10.4 acre parcel of land that was acquired in 2005 for \$3.2 million as part of a portfolio of Seniors' Housing Properties. We determined to hold these Properties for sale during late 2004 and 2005 and recognized aggregate impairment charges of approximately \$9.6 million to reduce the Properties' carrying value to their estimated fair value less the estimated costs to dispose. In July 2005, we entered into an agreement with a buyer to sell two of the Properties for an expected aggregate sales price of approximately \$6.0 million. In January 2006, we entered into an agreement to sell one additional Property for an expected sales price of approximately \$2.1 million.

In accordance with SFAS 144, we have reclassified the assets and operating results from the Seniors' Housing Properties as discontinued operations, restating previously reported results to reflect the reclassification on a comparable basis. These reclassifications had no effect on reported equity or net income.

The assets of the real estate held for sale were presented separately in the accompanying

consolidated balance sheets and consisted of the following at December 31 (in thousands):

	2005	2004
Real estate		
investment		
properties		
accounted for		
using the		
operating		
method, net	\$ 12,066	\$ 16,599
Accrued rental		
income	626	583
	\$ 12,692	\$ 17,182

The operational results associated with the Properties were presented as income (loss) from discontinued operations in the accompanying consolidated statements of income. Summarized financial information was as follows (in thousands):

		2005		2004	2	003
	_		_		_	
Rental	\$	1,716	\$	2,196	\$	960
income from						
operating						
leases						
Provision for		(350))			
doubtful						
accounts						
Impairment		(7,740))	(1,883))	
provisions						
Income (loss)		(6,834))	(433))	596
from						
discontinued						
operations						

11. Indebtedness:

Mortgage Notes Payable Mortgage notes payable and the Net Book Value ("NBV") of the associated collateral as of December 31, 2005, consisted of the following at December 31 (in thousands):

5.63% and variable rate based on the 3 to 9 month Fannie Mae Discount MBS rate plus 0.95% (5.39% combined weighted-average interest rate at December 31,			2005	2004	NBV
October 2013	with both variable-rate and fixed-rate components. Fixed rate at 5.63% and variable rate based on the 3 to 9 month Fannie Mae Discount MBS rate plus 0.95% (5.39% combined weighted-average interest rate at December 31, 2005), maturing October 2013 Various mortgages payable, interest only payments at variable rates ranging from LIBOR plus 1.0% to 3.0% (5.70%	\$			-

	2005	2004	NBV
December 31, 2005), maturing from November 2006 to March 2010			
Two mortgages payable, interest only payments at a 30-day commercial paper rate plus 1.82% or 2.15% (6.36% weighted-average interest rate at December 31, 2005), maturing March 2007 and May 2007	43,920	43,920	98,103
Various fixed-rate mortgages payable, interest only payments, bearing interest at rates ranging from 4.85% to 6.06%, (5.71% weighted-average interest rate at December 31, 2005), maturing September 2010 through	13,720	13,720	90,103
November 2015 Various fixed-rate mortgages payable, principal and interest payments, including net premiums of \$1.0 million and \$0.7 million at December 31, 2005 and 2004, respectively, bearing interest at rates ranging from 4.91% to 8.42% (6.25% weighted-average interest rate at December 31, 2005), maturing July 2007 through	263,810	167,145	517,757
November 2038	386,484(3)	339,913	629,632
	\$ 1,220,190	\$ 937,589	\$ 2,188,931

(1)

On October 3, 2005, we (i) exercised an extension option available under the \$140.4 million mortgage notes that were to mature in October 2005, (ii) negotiated the inclusion of an

\$82.2 million variable-rate mortgage loan due to mature in April 2008 and (iii) drew an additional \$19.4 million under the facility, all with a new maturity date of October 2013. The facility contains provisions that will allow us to draw an additional \$58.0 million upon providing additional collateral. Of the new \$242.0 million mortgage note payable, \$121.0 million bears fixed-rate interest at 5.63% requiring principal and interest payments through maturity and \$121.0 million bears variable-rate interest based on the 3 to 9 month Fannie Mae Discount MBS rate plus 0.95% (5.16% at December 31, 2005) requiring interest only payments through maturity. We also have the option to convert the variable-rate debt component to fixed-rate debt.

- We entered into interest rate swap agreements tied to debt with an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in LIBOR rates (see Note 12).
- (3)

 Certain fixed-rate loans contain substantial prepayment penalties and/or defeasance provisions that could preclude the repayment of the loans prior to their maturity dates.

Maturities for all mortgage notes payable, excluding loan premiums of \$1.0 million, at December 31, 2005 were as follows (in thousands):

2006	\$ 55,776
2007	66,989
2008	61,479
2009	143,453
2010	374,433
Thereafter	517,076
	\$ 1,219,206

Bonds Payable At December 31, 2005 and 2004, we had \$98.0 million and \$94.5 million, respectively, of non-interest bearing life care bonds at our two CCRCs and non-interest bearing occupancy fee deposits at a Seniors' Housing facility, all of which were payable to certain residents of the facilities (collectively "Bonds Payable"). During 2005, the tenants of the facilities issued new Bonds Payable to new residents of the facilities totaling \$12.6 million and used the proceeds from the Bonds issued in the current

period and prior periods to retire \$9.1 million of Bonds on our behalf. At December 31, 2005, \$68.7 million of the Bonds were refundable to the residents upon the resident moving out or to a resident's estate upon the resident's death and \$29.4 million of the Bonds were refundable after the unit has been successfully remarketed to a new resident.

Construction Loans Payable Construction loans payable consisted of the following at December 31 (in thousands):

	Total Facility	2005	2004
Five			
construction			
loans payable,			
each bearing			
interest at			
30-day			
LIBOR plus			
2.25% (6.62%			
at			
December 31,			
2005), with			
monthly			
interest only			
payments,			
maturing			
November			
2006	\$ 83,100	\$ 75,499	\$ 47 148
Construction	Ψ 03,100	Ψ 13,477	Ψ 77,170
loan payable			
bearing			
interest at the			
lender's base			
rate, as			
defined, less			
0.75% with a			
minimum rate			
of 6.50%			
(6.50% at			
December 31,			
2005), with			
monthly			
interest only			
payments,			
maturing			
December			
2007	48,000	44,696	32,339
Construction	ĺ	,	
loan payable			
bearing			
interest at			
30-day			
LIBOR plus			
1.75% (6.12%			
at			
December 31,			
2005), with			
monthly			
interest only			
payments,			
maturing July			
2009	14,287	11,750	2,021
	19,148	405	

	Total Facility	2005	2004
Two			
construction			
loans payable			
bearing			
interest at			
30-day			
LIBOR plus			
1.60% (6.31%			
at December 31,			
2005), with			
monthly			
interest only			
payments,			
maturing			
December			
2009			
Construction			
loan payable			
bearing			
interest at 30-day			
LIBOR plus			
1.70% (5.99%			
at			
December 31,			
2005), with			
monthly			
interest only			
payments,			
maturing	11 200	(00(
April 2012	11,280	6,096	
Construction			
loan payable bearing			
interest at			
30-day			
LIBOR plus			
1.80% (6.09%			
at			
December 31,			
2005), with			
monthly			
interest only			
payments, maturing			
December			
2013	6,600	5,114	
	-,		
	\$ 182,415	\$ 143,560	\$ 81,508

Line of Credit On August 23, 2005, we amended and restated our \$85.0 million credit agreement and closed on a \$320.0 million amended and restated senior secured revolving line of credit, which permits us to expand the borrowing capacity up to \$400.0 million and extended the initial maturity date to August 23,

2007 (the "Revolving LOC"). The amount available for use under the Revolving LOC is subject to certain limitations based on the pledged collateral. The Revolving LOC is collateralized by 36 Properties with a carrying value of approximately \$390.4 million at December 31, 2005, that in the aggregate, currently allows us to draw up to \$283.0 million. The Revolving LOC contains two one-year extension options and may be used to fund the acquisition and development of Properties, purchase other permitted investments and for general corporate purposes. The Revolving LOC requires interest only payments at LIBOR plus a percentage that fluctuates depending on our aggregate amount of debt outstanding in relation to our total assets (6.20% all-in rate at December 31, 2005, which represents a pricing of LIBOR plus 170 basis points). At December 31, 2005, \$75.0 million was outstanding under the Revolving LOC.

Term Loan. On January 13, 2005, we repaid and terminated a \$60.0 million 14-day term loan used for the acquisition of Properties for which permanent financing was obtained in January 2005.

Interest and loan cost amortization expense was \$76.2 million, \$42.8 million and \$9.6 million for the years ended December 31, 2005, 2004 and 2003, respectively, including \$0.4 million, \$1.1 million and \$0 of loan costs written off related to the early termination of debt for the years ended December 31, 2005, 2004 and 2003, respectively. For the years ended December 31, 2005, 2004 and 2003, interest of \$4.8 million, \$0.7 million and \$0, respectively, was capitalized to construction in progress.

The fair market value of our outstanding mortgage notes and construction loans payable was \$1.4 billion at December 31, 2005.

We were in compliance with all of our financial covenants as of December 31, 2005.

12. Financial Instruments: Derivatives and Hedging:

In May 2005, we entered into two interest rate swap agreements effective June 1, 2005, and one interest rate swap agreement effective July 1, 2005, for an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in interest rates on our variable interest rate mortgage notes payable. At December 31, 2005, derivatives with a fair value of \$4.8 million were included in other assets in the accompanying consolidated balance sheets. The change in net unrealized gain of \$4.8 million as of December 31, 2005, for derivatives designated as cash flow hedges is disclosed separately in the accompanying consolidated statements of stockholders' equity as the change in fair value of cash flow hedges.

Amounts reported in accumulated other comprehensive income related to derivatives will be reclassified to interest expense as interest payments are made on our variable-rate debt. The change in net unrealized income on cash flow hedges reflects a reclassification of \$0.6 million of net unrealized gain from accumulated other comprehensive income to interest expense during the year ended December 31, 2005. During the next twelve months, we expect to reclassify approximately \$1.2 million of the current balance held in accumulated other comprehensive income related to the interest rate swaps to earnings as a reduction of interest expense.

Cash flow hedges at December 31, 2005 consisted of the following:

Hedge Notional Rate Maturity Fair Type Amount _____ Value

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	(in thousands)	ſ	(in chousands)
Swap, Cash Flow	\$ 100,000	4.1840% January 12, 2010	\$ 2,062
Swap, Cash Flow	83,750	4.1764% January 1, 2010	1,738
Swap, Cash Flow	50,000	4.2085% March 31, 2010	1,039
	\$ 233,750		\$ 4,839
		F-38	

13. Intangible Lease Liability:

Intangible lease liability at December 31, 2005 and 2004, was \$4.5 million and \$3.7 million, respectively, consisting of the unamortized carrying value of below-market-rate leases associated with Properties acquired. Intangible lease liability is amortized over the remaining term of the associated lease, including below-market lease extension, if any. Intangible lease liability accreted to rental income from operating leases in the accompanying consolidated statements of income was \$1.1 million, \$0.7 million and \$0, for the years ended December 31, 2005, 2004 and 2003, respectively.

14. Commitments and Contingencies:

Commitments Commitments, contingencies and guarantees by expiration period as of December 31, 2005 (in thousands):

Less than 3 4-5
1 Year YearsYeaFhereafteifotal

Pending			
investments(1)	\$ 157,100 \$ 5	\$	\$ 157,100
Unfunded			
Senior Secured			
Term Loan(2)	69,000		69,000
Capital			
improvements			
to Properties	62,620		62,620
Earnout			
provisions(3)	25,979		25,979
Guarantee of			
uncollateralized			
promissory note			
of CNL Plaza,			
Ltd.(4)		2,313	2,313

\$ 314,699 \$ \$ 2,313 \$ \$ 317,012

(1)

As of December 31, 2005, we had initial commitments to acquire 12 Medical Facilities for which we had posted a non-refundable \$10.6 million deposit. In January 2006, we completed the acquisition of seven Medical Facilities for \$84.5 million, including the application of \$6.0 million of the non-refundable deposit that we had posted as of December 31, 2005. The remaining Properties are expected to be acquired in the first quarter of 2006. The

acquisition of each of these Properties is subject to the fulfillment of certain conditions. There can be no assurance that any or all of the conditions will be satisfied or, if satisfied, that we will acquire one or more of these investments.

- (2)

 Represents the unfunded portion under the \$85.0 million Senior Secured Term Loan.
- (3) In connection with the acquisition of 41 Properties, we may be required to make additional payments to the seller if certain earnout provisions are achieved by the earnout date for each Property. The calculation generally considers the net operating income for the Property, our initial investment in the Property and the fair value of the Property. In the event an amount is due, the applicable lease will be amended and annual minimum rent will increase accordingly. Amounts presented represent maximum exposure to additional payments. Earnout amounts related to six additional Properties are subject to future values and events which are not quantifiable at December 31, 2005, and are not included in the table above.
- (4)
 In connection with the ownership of a 9.90% limited partnership interest in CNL Plaza, Ltd., we severally guaranteed 16.67%, or \$2.3 million, of a \$14.0 million uncollateralized promissory note of the general partner of the limited partnership that matures December 31, 2010. As of

December 31, 2005, the uncollateralized promissory note had an outstanding balance of \$13.9 million.

Ground Leases Twenty-seven of our Properties are subject to ground leases. These ground leases have predetermined rent increases based on the CPI index or a defined percentage and termination dates ranging from 2038 to 2084. Twenty-one of the ground leases contain renewal options for terms of 30 to 50 years. During the years ended December 31, 2005, 2004 and 2003, we recognized ground lease expense of \$0.5 million, \$0.2 million and \$0, respectively, including \$0.2 million, \$13,000 and \$0, respectively, from the straight-lining of ground lease expense, which is included in Seniors' Housing property expenses and Medical Facilities operating expenses in the accompanying consolidated statements of income.

Future minimum lease payments due under ground leases at December 31, 2005, exclusive of renewal option periods, were as follows (in thousands):

2006	9	\$ 372
2007		490
2008		491
2009		493
2010		495
Thereafter		23,750
		\$ 26,091
		\$ 26,091

Operating Leases At December 31, 2005, future minimum lease payments due under an operating lease for DASCO's administrative offices which terminates in 2011 were \$0.2 million for each of the next six years.

Legal Matters From time to time, we are exposed to litigation arising from an unidentified pre-acquisition contingency or from the operation of our business. We do not believe that resolution of these matters will have a material adverse effect on our financial condition or results of operations.

15. Redemption of Shares:

We have a redemption plan under which we may elect to redeem shares, subject to certain conditions and limitations. Under the redemption plan, prior to such time, if any, as listing of our common stock on a national securities exchange or over-the-counter market occurs, any stockholder who has held shares for at least one year may present to us all or any portion equal to at least 25% of their shares for redemption in accordance with the procedures outlined in the redemption

plan. Upon presentation, we may, at our option, redeem the shares, subject to certain conditions and limitations. However, at no time during a 12-month period may the number of shares we redeem exceed 5% of the number of shares of our outstanding common stock at the beginning of the 12-month period. During the years ended December 31, 2005, 2004 and 2003, 3,904,039 shares, 685,396 shares and 131,781 shares of common stock were redeemed and retired for \$37.1 million, \$6.5 million and \$1.2 million, respectively. In the second quarter of 2004, we amended our redemption plan to change the redemption price from \$9.20 per share to \$9.50 per share.

16. Distributions:

For the years ended December 31, 2005, 2004 and 2003, approximately 67%, 60% and 71%, respectively, of the distributions paid to stockholders were considered ordinary income and approximately 33%, 40% and 29%, respectively, were considered a return of capital to stockholders for federal income tax purposes. For the years ended December 31, 2005, 2004 and 2003, no amounts distributed to the stockholders are required to be or have been treated by us as a return of capital for purposes of calculating the stockholders' 8% return, which is equal to an 8% cumulative, non-compounded annual return on the amount calculated by multiplying the total number of shares of common stock purchased by stockholders by the issue price, without deduction for volume or other discounts, reduced by the portion of any distribution that is attributable to net sales proceeds and by any amount we have paid to repurchase shares under our redemption plan.

17. Related Party Arrangements:

Certain of our directors and officers hold similar positions with the Advisor, the parent company of the Advisor and the managing dealer of our public offerings, CNL Securities Corp. Our chairman of the board indirectly owns a controlling interest in the parent company of the Advisor. These affiliates receive fees and compensation for services provided in connection with the common stock offerings, permanent financing and the acquisition, management and sale of our assets.

Pursuant to the Advisory Agreement, as amended and renewed, the Advisor and its affiliates earn certain fees and are entitled to receive reimbursement of certain expenses. During the years ended December 31, 2005, 2004 and 2003, the Advisor and its affiliates earned fees and incurred reimbursable expenses as follows (in thousands):

	Years ended December 31,			
	2005	2004	2003	
Acquisition fees(1):				
From offering proceeds	\$ 5,874	\$ 38,286	\$ 47,644	
From debt proceeds	13,789	29,952	11,277	

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	Years ended December 31,		
	19,663	68,238	58,921
Asset management fees(2)	19,217	13,047	4,372
Reimbursable expenses(3):			
Acquisition expenses	210	331	403
General and administrative expenses	5,989	4,313	2,255
	6,199	4,644	2,658
	\$ 45,079	\$ 85,929	\$ 65,951

For the period from May 3, 2005 through December 31, 2005, acquisition fees for, among other things, identifying Properties and structuring the terms of the leases were equal to 3.0% of gross offering proceeds and loan proceeds from permanent financing under the 2004 Offering (4.0% of gross offering and loan proceeds for the period from May 14, 2004 through May 2, 2005 and 4.5% of gross offering and loan proceeds under the Prior Offerings). These fees

F-41

are included in other

assets in the accompanying consolidated balance sheets prior to being allocated to individual Properties or intangible lease costs.

If we list our common stock on a national securities exchange or over-the-counter market ("List" or "Listing"), the Advisor will receive an acquisition fee equal to 3.0% of amounts outstanding on the line of credit, if any, at the time of Listing. Certain fees payable to the Advisor upon Listing, the orderly liquidation or other sales of Properties are subordinate to the return of 100% of the stockholders' invested capital plus the achievement of a cumulative, noncompounded annual 8% return on stockholders' invested capital.

- Monthly asset management fee of 0.05% of our real estate asset value, as defined in the Advisory Agreement, and the outstanding principal balance of any Mortgage Loans as of the end of the preceding month.
- Reimbursement for administrative services, including, but not limited to, accounting; financial, tax, insurance administration and regulatory compliance reporting; stockholder distributions and reporting; due diligence and marketing; and investor relations.

Pursuant to the advisory agreement, the Advisor is required to reimburse us the amount by which the total operating expenses we pay or incur exceeds in any four consecutive fiscal quarters (the "Expense Year") the greater of 2% of average invested assets or 25% of net income (the "Expense Cap"). Operating expenses for the Expense Years ended December 31, 2005, 2004 and 2003, did not exceed the Expense Cap.

Of these amounts, approximately \$1.1 million and \$1.4 million were included in due to related parties in the accompanying consolidated balance sheets at December 31, 2005 and 2004, respectively.

CNL Securities Corp. received fees based on the amounts raised from our offerings equal to: (i) selling commissions of 6.5% of gross proceeds under the 2004 Offering and 7.5% under the Prior Offerings, (ii) a marketing support fee of 2.0% of gross proceeds under the 2004 Offering and 0.5%

under the Prior Offerings and (iii) beginning on December 31, 2003, an annual soliciting dealer servicing fee equal to 0.2% of the aggregate proceeds raised in a prior offering. Affiliates of the Advisor are reimbursed for certain offering expenses incurred on our behalf. Offering expenses incurred by the Advisor and its affiliates on our behalf, together with selling commissions, the marketing support fee and due diligence expense reimbursements will not exceed 13% of the proceeds raised in connection with the offerings.

During the years ended December 31, 2005 and 2004, we incurred the following fees and costs (in thousands):

	Years ended December 31,					
	2005	2004				
Selling commissions	\$ 10,801	\$ 61,830				
Marketing support fee	3,313	6,648				
Offering and due diligence costs	4,250	18,328				
Soliciting dealer servicing fee		310				
	\$ 18,364	\$ 87,116				

Of these amounts, approximately \$1.3 million and \$0.2 million were included in due to related parties in the accompanying consolidated balance sheets at December 31, 2005 and 2004, respectively.

We own a 9.90% interest in CNL Plaza, Ltd. (the "Owner"), a limited partnership that owns an office building located in Orlando, Florida, in which the Advisor and certain affiliates of CNL Financial Group ("CFG") lease office space. CFG owns a controlling interest in the parent company of the Advisor and is indirectly wholly owned by James M. Seneff, Jr., our chairman of the board, and his wife. Robert A. Bourne, our vice-chairman of the board and treasurer, is an officer of CFG. The remaining interests in the Owner are held by several entities with present or former affiliations with CFG, including: CNL Plaza Venture, Ltd., which has a 1% interest as general partner of the Owner and whose general partner is indirectly wholly owned by Mr. Seneff and his wife; CNL Corporate Investors, Ltd., which is indirectly wholly owned by Messrs. Seneff and Bourne, and which has a 49.50% interest, as a limited partner, in the Owner; CNL Hotels & Resorts, Inc. which has a 9.90% interest, as a limited partner, in the Owner; Commercial Net Lease Realty, Inc., which has a 24.75% interest, as a limited partner, in the Owner; and CNL APF Partners, LP, which has a 4.95% interest, as a limited partner, in the Owner. We also own a 9.90% interest in CNL Plaza Venture, Ltd. (the "Borrower"), a Florida limited partnership, which is the general partner of the Owner. The remaining interests in the Borrower are held by the same entities in the same

proportion described above with respect to the Owner.

In 2004, the Owner conveyed a small portion of the premises underlying the parking structure adjacent to its office building, valued by the parties at approximately \$0.6 million, to CNL Plaza II, Ltd., a limited partnership in which Messrs. Seneff and Bourne own a 60% interest and 40% interest, respectively, as part of the development of the premises surrounding the building. The purpose of the conveyance was to adjust the percentage fee simple ownership under the parking structure so as to allow joint parking privileges for a new office building that was developed in 2005 and is owned by CNL Plaza II, Ltd. In connection with this transaction, the Owner received an ownership interest in a cross-bridge that was constructed and an anticipated benefit from a reduction in the allocation of its operating expenses for the garage. In addition, the Owner may be entitled to additional consideration pursuant to a purchase price adjustment.

On September 30, 2005, we executed a pro rata, several guarantee limited to 16.67%, or \$2.3 million, of a \$14.0 million uncollateralized promissory note of the Borrower that matures

December 31, 2010. During each of the years ended December 31, 2005 and 2004, we received approximately \$0.2 million, respectively, in distributions from the Owner.

We maintain bank accounts in a bank in which certain of our officers and directors serve as directors and are principal stockholders. The amounts deposited with this bank were \$3.1 million and \$22.9 million at December 31, 2005 and 2004, respectively.

On September 1, 2004, a company which is owned by our chairman of the board sold its 30% voting membership interest in a limited liability company which is affiliated with ten of our tenants (the "HRA Tenants") to the remaining members of the limited liability company. The HRA Tenants contributed 30% and 35% of our total revenues for the years ended December 31, 2004 and 2003, respectively.

Century Capital Markets, LLC ("CCM"), an entity in which an affiliate of the Advisor was formerly a non-voting Class C member, made the arrangements for two commercial paper loans totaling \$43.9 million. The monthly interest payments due under these commercial paper loans include an annual margin of either 30 or 40 basis points, payable to CCM for the monthly services it provides related to the administration of the commercial paper loans. Effective September 30, 2005, a non-affiliated third party assumed the administration of these commercial paper loans. Therefore, we now pay the monthly services fee directly to the non-affiliated third party. During the years ended December 31, 2005, 2004 and 2003, \$0.1 million, \$0.1 million and \$0.2 million, respectively, was paid to CCM related to these services. During the year ended December 31, 2003, we also paid CCM a \$0.2 million finder's fee related to the acquisition of two Properties.

Our chairman of the board is a director in a hospital that leases office space in seven of the Medical Facilities that we acquired in August 2004. Additionally, one of our independent directors is a director in a health system that leases office space in one of the Medical Facilities that we acquired in April 2004. During the years ended December 31, 2005 and 2004, these hospitals contributed less than 1% of our total revenues.

18. Concentration of Credit Risk:

At December 31, 2005, we leased our Seniors' Housing facilities to 22 tenants. Two tenants affiliated with Horizon Bay Management, LLC

("Horizon Bay") contributed 21% of total revenues for each of the years ended December 31, 2005 and 2004. The HRA Tenants contributed 22%, 30% and 35% of total revenues for the years ended December 31, 2005, 2004 and 2003, respectively. No other Seniors' Housing tenant contributed more than 10% of total revenues for the three years ended December 31, 2005. Several of our tenants, including the HRA Tenants, are thinly capitalized corporations that rely on the net operating income generated from the Seniors' Housing facilities to fund rent obligations under their leases. At December 31, 2005, \$5.8 million of the \$7.2 million allowance for doubtful accounts pertained to HRA Tenants. At December 31, 2005, our Medical Facilities were leased to more than 700 tenants.

At December 31, 2005, 107 of the 188 Seniors' Housing facilities were operated by Sunrise Senior Living Services, Inc. ("Sunrise"), a wholly owned subsidiary of Sunrise Senior Living, Inc. Additionally, as of December 31, 2005, a Seniors' Housing Property was being developed by Sunrise Development, Inc., a wholly owned subsidiary of Sunrise Senior Living, Inc. Upon completion of the development, the facility will be operated by Sunrise. Horizon Bay operates 27 Seniors' Housing facilities and six additional operators manage the remaining 53 Seniors' Housing facilities. At December 31, 2005, DASCO managed or was developing 53 of our 73 Medical Facilities, Cirrus managed 10 of our Medical Facilities and the remaining 10 Medical Facilities were managed by four third-party property managers. Sunrise, Horizon Bay, DASCO and ARC operated Property portfolios that, for each in the aggregate as operator, contributed 10% or more of total rental and earned income from leases. Sunrise contributed 42%, 45% and 76%, for the years ended December 31, 2005, 2004 and 2003, respectively; Horizon Bay contributed 22%, 24%, for the years ended December 31, 2005 and 2004, respectively; DASCO contributed 10% for the year ended December 31, 2005 and ARC contributed 13% for the year ended December 31, 2003. No other operator contributed more than 10% of total rental and earned income from leases.

To mitigate credit risk, certain Seniors' Housing leases are combined into portfolios that contain cross-default terms, so that if a tenant of any of the Properties in a portfolio defaults on its obligations under its lease, we may pursue remedies under the lease with respect to any of the Properties in the portfolio ("Cross-Default"). Certain portfolios also contain terms whereby the net operating profits of the Properties are combined for the purpose of funding rental payments due under each lease ("Pooling" or "Pooled"). In addition, as of December 31, 2005, we held \$24.0 million in security deposits and rental support related to certain Properties.

We had the following remaining rental support and limited guarantees from certain tenants and operators at December 31, 2005 (dollars in thousands):

Cwamantaa

			_	Guarantee					
Guarantor	Number of Properties				Remaining Balance				
Horizon	21	\$ 17,500) \$	14,391	\$ 3,109				
Bay Aureus	11	10,000)	2,255	7,745				
ARC	8		(1)	9,416	(1	(

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

		Guarante	ee
	-	220	
6	(1)	329	(1)
17	(1)	791	(1)
5	(1)	2,493	(1)
2	(1)		(1)
17	(2)	6,281	(2)
3	(3)	2,809	(3)
	5 2 17	17 (1) 5 (1) 2 (1) 17 (2)	17 (1) 791 5 (1) 2,493 2 (1) 17 (2) 6,281

- (1) Unconditional guarantees
- Sunrise guaranteed the tenants' obligations to pay minimum rent and the FF&E reserve funds under the 17 leases until the later of (i) March 2006 or (ii) 18 months after the final development date of certain Properties, as defined in the lease agreement. The final development Property

commenced operations in January 2006; accordingly, the Sunrise guarantee will terminate in July 2007.

Sunrise guaranteed the tenants' rent obligations for these Seniors' Housing facilities that were acquired in 2004 and which commenced operations in 2004, until the later of (i) September 2006 or (ii) the Properties achieving predetermined rent coverage thresholds, which are not determinable at this time.

Although we acquire Properties located in various states and regions and screen our tenants in order to reduce risks of default, failure of certain lessees, their guarantors, or the Sunrise or Horizon Bay brands would significantly impact our results of operations.

19. Medical Facilities Acquisitions:

In April 2004, we acquired 22 Medical Facilities for an aggregate purchase price of \$272.0 million, including closing costs (the "MOP Acquisition").

In August 2004, we acquired ownership interests in entities that own 28 Medical Facilities and a 55% interest in DASCO for \$212.6 million, including closing costs. In November 2004, we acquired two additional Medical Facilities for \$19.4 million, including closing costs (collectively, the "DASCO Acquisition"). Included in the DASCO Acquisition were certain limited partnerships with finite lives. Therefore, the minority interests in these partnerships meet the definition of mandatorily redeemable noncontrolling interests as specified in Statement of Financial Accounting Standards No. 150, "Accounting for Certain Financial Instruments with Characteristics of both Liabilities and Equity." We estimate that the settlement value of these mandatorily redeemable noncontrolling interests at December 31, 2005 was \$13.7 million, based on the sale or disposition of all or substantially all of the assets of the partnerships and the repayments of outstanding liabilities as of that date.

In addition, certain partnerships that own Medical Facilities provide non-equity participation to various lessees or affiliates of lessees. Certain lessees in the Medical Facilities are entitled to receive a percentage of the pro rata net cash flow, as defined, for the term of their lease, calculated as the percentage of each lease with respect to the total leasable square footage. Such amounts are paid periodically, such as monthly or quarterly. Certain lessees are also entitled to a percentage of

their pro rata share of net capital proceeds, as defined, upon the occurrence of a capital transaction (including, but not limited to, the sale or refinancing of the property). Such pro rata share is calculated as the percentage of each lease with respect to the total leasable square footage.

The fair value of assets acquired and liabilities assumed at the date of the MOP and DASCO Acquisitions were based on independent appraisals and valuation studies from independent third-party

consultants. The aggregate value of the assets acquired, including closing costs, and liabilities assumed were as follows (in thousands):

Assets:	
Real estate investment	
properties:	
Accounted for using	\$ 455,194
the operating method	
Intangible lease costs	47,372
	502,560
Cash and cash	530
equivalents	
Restricted cash	2,483
Deferred costs, net	1,01
Other assets	1,698
Goodwill	5,79
Total assets acquired	514,08
Total assets acquired	311,00
Liabilities:	
Mortgages payable	94.80
Construction loan	48
payable	10
Accounts payable and	8,11
other liabilities	0,11
Intangible lease	4,46
liability	., 10.
Security deposits	2,01
, ₁	,,,,,,
Total liabilities	100.00
assumed	109,886
assumeu	
Minority interests	1,96
wimority interests	1,90
Net assets acquired	\$ 402,233

The amortization periods of the intangible lease costs acquired range from less than one year to 15 years.

The following condensed pro forma (unaudited) information assumes that the MOP and DASCO Acquisitions had occurred on January 1, 2003.

	Years December	
	2004	2003
Revenues	\$ 298,164	\$ 145,322
Expenses	178,495	92,623
Net income	118,396	51,903

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	Years Ended December 31,					
Basic and diluted						
income per share	\$	0.56 \$	0.52			
Weighted-average number of common shares outstanding (basic		242	00.015			
and diluted)	2	210,343	99,815			
F-	-47					

20. Subsequent Events:

In January 2006, we acquired seven Medical Facilities from Cirrus for \$84.5 million which we funded, in part, with proceeds from a new \$56.3, million ten-year mortgage loan that bears fixed-rate interest at 5.59%. Four of the acquired Properties are located in Texas, two are in Arizona and one is in Missouri, and in aggregate they contain approximately 255,000 square feet. Cirrus will manage the Properties.

In February 2006, we entered into a \$7.7 million construction loan for the development of a Medical Facility that we acquired in November 2005. The construction loan will mature in 2010 and bears interest at a rate of LIBOR plus 160 basis points.

In February 2006, we entered into a \$33.0 million mortgage loan and used the proceeds and cash on hand to pre-pay a \$48.0 million construction loan facility that had \$44.7 million outstanding at December 31, 2005. The new interest only, five-year loan bears interest at a rate equal to LIBOR plus 150 basis points.

In March 2006, we sold two Properties that were held for sale at December 31, 2005. The Properties were sold to an unrelated third party for \$6.0 million and we took back a purchase money mortgage with a three-year term secured by the Properties in the amount of \$4.8 million. Interest is payable annually at a rate of 6.0% and principle is due at maturity. We realized a net loss on the sale of the Properties of \$0.2 million in March 2006.

During the period January 1, 2006, through March 15, 2006, we received subscription proceeds for an additional 2.6 million shares (\$26.1 million) of common stock.

On January 1, February 1 and March 1, 2006, our Board of Directors declared distributions to stockholders of record on those dates, totaling \$45.5 million, or the aggregate of \$0.1776 per share of common stock, payable by March 31, 2006.

21. Selected Quarterly Financial Data (unaudited):

The following table presents selected unaudited quarterly financial data for each full quarter during the years ended December 31, 2005 and 2004 (in thousands, except per share amounts):

2005 Quarter First Second Third Fourth

Revenues(1) \$91,329 \$96,584 \$97,540 \$98,630

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

2005 Quarter	First	Second	Third	Fourth						
Income from continuing operations(1)	38,461	39,130	34,074	30,750						
Income (loss) from discontinued operations(1)	(5,826)	(1,173)	(264)	429						
Net income	32,635	37,957	33,810	31,179						
Income per share, basic and diluted:										
Continuing operations(1)	0.16	0.16	0.14	0.12						
Discontinued operations(1)	(0.02)	(0.01)								
Net income	0.14	0.15	0.14	0.12						
F-48										

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

2004 Quarter	First	Second	Third	Fourth
Revenues(1)	\$ 50,259	\$ 62,850	\$ 70,318	\$ 79,342
Income from continuing operations(1)	27,415	29,340	30,428	31,168
Income (loss)	386	351	(1,514)	344
from discontinued operations(1)				
Net income	27,801	29,691	28,914	31,512
Income per share, basic and diluted:				
Continuing operations(1)	0.16	0.14	0.14	0.13
Discontinued operations(1)			(0.01))
Net income	0.16	0.14	0.13	0.13

(1) The reoperat

The revenue, income from continuing operations and income (loss) from discontinued operations data in the table above has been restated from previously reported amounts to reflect the reclassification of the operating results from our real estate held for sale to discontinued operations (see Note 10).

CNL RETIRMENT PROPERTIES, INC.

Schedule II Valuation and Qualifying Accounts

Years Ended December 31, 2005, 2004, and 2003

(dollars in thousands)

		Additio	ns	Dedi	uctions	
Year Description	Balance at Beginnin of Year	Costs g and C	arg to thl	ged D Deemed		ed Balance at End
2003 Allowance for doubtfu accounts(a)	l	\$	\$	\$	\$	\$
2004 Allowance for doubtfu accounts(a)	l	\$ 3,900	\$	\$	\$	\$ 3,900
2005 Allowance for doubtfu accounts(a)	l	\$ 4,544	\$	\$ 161	\$ 1,083	\$ 7,200
			_			

(a) Deducted from receivables on the balance sheet.

CNL Retirement Properties, Inc. Schedule III Real Estate and Accumulated Depreciation December 31, 2005 (dollars in thousands)

Costs

Capitalized Gross Amount at Which

			Cost to pany(2)	Subseq	alized uent to isition		Amount a ried at Clo Period		
Enc	umbranc	:es [/a)nd l	Building, Fixtures and Equipmen	F	uilding, ixtures and juipmen		Building, Fixtures and Equipmen		ccun
Brighton Gardens of									
Orland Park,									
IL	\$	\$ 2,162	\$ 12,577	\$ 5	73	\$ 2,162	\$ 12,650	\$ 14,812	\$ 2
Broadway									
Plaza at Pecan Park,									
TX	3,600	1,344	9,425			1,344	9,425	10,769	1.
Homewood	,,,,,,	,-	. ,			,-	. , .	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
Residence at									
Boca Raton,	4,400	1 1 1 4 4	0.724			1 144	0.724	0.070	
FL Holley Court	4,400	1,144	8,734			1,144	8,734	9,878	1
Terrace, IL		2,144	16,850			2,144	16,850	18,994	1
Homewood									
Residence at									
Coconut Creek, FL	2,602	1,683	8,193			1,683	8,193	9,876	1
Heritage	2,002	1,003	0,193			1,003	0,193	9,670	1.
Club at									
Greenwood									
Village, CO		1,965	18,025		180	1,965	18,205	20,170	2
Mapleridge of									
Dartmouth,									
MA	4,403	920	8,799		68	920	8,867	9,787	
Mapleridge									
of Laguna	2 720	013	7.407			012	7.407	9 210	
Creek, CA Brighton	3,738	812	7,407			812	7,407	8,219	
Gardens of									
Towson, MD	6,706	990	14,109	(22)	163	968	14,272	15,240	1.
Brighton									
Gardens of Camarillo,									
CA	8,673	2,487	16,676	(1)	110	2,486	16,786	19,272	1
Vero Beach,	2,2.2	_,	,	(-)		_,	,	,	
FL	46,027	1,786	44,821			1,786	44,821	46,607	
Homewood									
Residence at Brookmont									
Terr., TN	1,931	464	8,652			464	8,652	9,116	
Mapleridge									
of Hemet,	2.440		2.00=				2.424	4.00=	
CA Brighton	3,110	1,176	3,087		44	1,176	3,131	4,307	
Gardens of									
Tulsa, OK	3,544	1,538	3,310	21	77	1,559	3,387	4,946	
Pleasant									
Hills, AR	8,050			241	281	764		11,472	
Brighton Gardens of	5,708	1,724	6,064		74	1,724	6,138	7,862	
Gardens of									

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

		Initial Cost to Company(2)		Capi Subsec	osts talized quent to isition		amount at ied at Clo Period		
Hoffman									
Estates, IL									
Mapleridge of									
Willoughby,									
ОН	3,731	1,091	4,032	84	60	1,175	4,092	5,267	2
Mapleridge of Plymouth,									
MA	3,466	1,090	3,667	8	73	1,098	3,740	4,838	4
Hearthside									
of Lynwood,	2.106	1.520	5.060	10	247	1.540	5 215	6.055	
WA Hearthside	3,196	1,530	5,068	10	247	1,540	5,315	6,855	2
of									
Snohomish,									
WA	4,362	645	8,364	3	71	648	8,435	9,083	6
Brighton									
Gardens of									
Vinings, GA	3,741	1,773	5,830	8	112	1,781	5,942	7,723	6
Brighton Gardens of									
Oklahoma									
City, OK	1,850	784	3,000	10	80	794	3,080	3,874	2
Brighton									
Gardens of									
Bellevue,									
WA	5,175	2,165	8,506		69	2,165	8,575	10,740	8
Brighton Gardens of									
Santa Rosa,									
CA	8,496	2,161	15,044	989	(2,530)	3,150	12,514	15,664	1,2
Brighton	-, -	,	- ,		, , /	, , ,	,-	- ,	,
Gardens of									
Denver, CO	10,936	1,084	17,245			1,084	17,245	18,329	1,3
		F-51							

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Brighton Gardens of Colorado										
	10,085	1,073	15,829			1,073	15,829	16,902	1,210	19
Lakewood, CO	11,512	1,073	18,221			1,073	18,221	19,294	1,385	19
Brighton Gardens of Rancho										
Mirage, CA The Fairfax,	7,017	1,716	12,482	5	120	1,721	12,602	14,323	1,192	20
VA The	43,894	17,641	60,643		9,795	17,641	70,438	88,079	4,991	1989/20
Quadrangle,	52,792	23.148	90.769	(37)	1.522	23.111	92.291	115,402	7.326	19
Brighton Gardens of Yorba Linda,	02,772	25,110	,,,,,,	(57)	1,022	20,111) <u>-,-</u>)1	110,102	7,020	
CA Brighton	10,203	2,397	11,410		86	2,397	11,496	13,893	954	20
Gardens of Salt Lake City,										
UT Brighton	11,372	392	15,013	5	89	397	15,102	15,499	1,296	19
Gardens of Northridge,	7.475	2.405	11.624	(1)	70	2.404	11.704	15 100	1 147	20
CA Mapleridge	7,475	3,485	11,634	(1)	70	3,484	11,704	15,188	1,147	20
of Palm Springs, CA	1,346	884	1,873		48	884	1,921	2,805	252	19
Brighton Gardens of Edgewood,										
KY Brighton Gardens of	1,347	886	1,876	6	36	892	1,912	2,804	299	20
Greenville, SC	2,097	352	3,938	4	75	356	4,013	4,369	498	19
Brighton Gardens of Saddle River,										
NJ Balmoral of	7,867	2,155	10,968			2,155	10,968	13,123	934	19
Palm Harbor, FL		1,002	11,493	2	333	1,004	11,826	12,830	901	19
Somerby at University Park, AL	37,322	2,633	49,166		3,603	2,633	52,769	55,402	3,489	19
Somerby at Jones Farm, AL	20.261	710	22 126		6 126	710	20.272	20.001	1 022	19
Brighton Gardens of	20,361		23,136		6,136		29,272	29,991		
Tampa, FL Greentree at Ft. Benjamin		1,670		4	117	1,674	117	1,791	12	19
Harrison, IL Greentree at Mt. Vernon,		469	4,761			469	4,761	5,230	296	19
IL Greentree at		225	7,244		1,830	225	9,074	9,299	531	20
Post, IN Greentree at		287	4,934			287	4,934	5,221	290	19
West Lafayette, IN		319	5,264		1,883	319	7,147	7,466	385	19
Sunrise of Arlington,	3,543	765	6,463	19	228	784	6,691	7,475	499	19

14,021	2,359	26,196	37	307	2,396	26,503	28,899	1,824	19
7,335	2,288	12,583	7	291	2,295	12,874	15,169	948	1945
4,341	1,221	7,631	3	73	1,224	7,704	8,928	600	19
4,690	1,212	8,414	18	56	1,230	8,470	9,700	695	19
	F-5	2							
	7,335 4,341	7,335 2,288 4,341 1,221 4,690 1,212	7,335 2,288 12,583 4,341 1,221 7,631	7,335 2,288 12,583 7 4,341 1,221 7,631 3 4,690 1,212 8,414 18	7,335 2,288 12,583 7 291 4,341 1,221 7,631 3 73 4,690 1,212 8,414 18 56	7,335 2,288 12,583 7 291 2,295 4,341 1,221 7,631 3 73 1,224 4,690 1,212 8,414 18 56 1,230	7,335 2,288 12,583 7 291 2,295 12,874 4,341 1,221 7,631 3 73 1,224 7,704 4,690 1,212 8,414 18 56 1,230 8,470	7,335 2,288 12,583 7 291 2,295 12,874 15,169 4,341 1,221 7,631 3 73 1,224 7,704 8,928 4,690 1,212 8,414 18 56 1,230 8,470 9,700	7,335 2,288 12,583 7 291 2,295 12,874 15,169 948 4,341 1,221 7,631 3 73 1,224 7,704 8,928 600 4,690 1,212 8,414 18 56 1,230 8,470 9,700 695

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

											_
Sunrise of Frederrick, MD	3,443	118	6,971	3	110	121	7,081	7,202	483	1991	S
Sunrise of Leesburg, VA	1,048				25		,,,,,	ŕ		1850/1989	
Sunrise of Mercer			ŕ				ŕ				
Island, WA Sunrise of	3,892	744	7,225	28	223	772	7,448	8,220	520	1990	S
Mills Basin, NY Sunrise of	12,075	2,596	22,134	25	63	2,621	22,197	24,818	1,623	2002	S
Poland, OH	4,291	742	8,044	21	33	763	8,077	8,840	544	1998	S
Sunrise of Raleigh, NC Sunrise of Sheepshead	3,143	457	5,935	3	109	460	6,044	6,504	503	1996	S
Bay, NY Sunrise of Beverly	12,823	3,856	22,395	24	24	3,880	22,419	26,299	1,512	2000	S
Hills, CA Sunrise of	19,806	3,950	24,230			3,950	24,230	28,180	156	2005	S
Cresskill, NJ Sunrise of	25,973	4,632	33,212			4,632	33,212	37,844	176	(3)	S
Edmonds, WA	10,072	968	12,681			968	12,681	13,649	457	2004	S
Sunrise at Five Forks, GA	8,126	997	11,161		132	997	11,293	12,290	683	2004	S
Sunrise of Madison, NJ	11,522	1,608	14,345			1,608	14,345	15,953	566	2004	S
Dogwood Forest of	1.,.	1,.	1 .,.			1,.	1 .,.	10,.			Ì
Dunwoody, GA		837	4,952		142	837	5,094	5,931	315	2000 1	N
EdenGardens of											
Gainesville, FL EdenBrook		436	7,789		47	436	7,836	8,272	484	2000 1	N
of Jacksonville, FL		1,114	6,112	14	312	1,128	6,424	7,552	490	1999 1	N
EdenBrook of		-,	-,			-,		,			
Tallahassee, FL		670	11,664		98	670	11,762	12,432	717	1999 1	N
EdenGardens of Aiken, SC EdenBrook	4,901	369	7,139	7	113	376	7,252	7,628	461	1995 1	N
of Alpharetta, GA	4,411	718	6,330		30	718	6,360	7,078	406	2000 1	Ŋ
EdenGardens of Arlington,	٦,	120	0,00			120	0,00	7,0.	Te .		
TX EdenTerrace		350	8,538	8	4	358	8,542	8,900	506	2000 1	N
of Arlington, TX		668	7,616	47	105	715	7,721	8,436	485	2000 1	N
EdenBrook of Buckhead,	4.411	702	C 071		11	702	(002	7.761	167	2000	
GA EdenBrook of	4,411	782	6,971		11	782	6,982	7,764	467	2000 1	Z
Champions, TX		530	11,581		54	530	11,635	12,165	713	2000 1	N
EdenBrook of Charleston,			11,				11,	12,			
SC	4,901	422	8,827	7	101	429	8,928	9,357	560	2000 1	N

EdenGardens of Columbia, SC		300	4,043	9 167	309	4,210	4,519	270	1996
EdenGardens of Concord, NC	2.419	393	3,548		393	3.548	3 941	228	1998
110	2,417	F-5			373	3,540	3,741	220	1770

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Edenbrook of Dunwoody,											
GA	4,629	368	4,559	7	208	375	4,767	5,142	357	1998	Nov-03
Edenbrook of Hunstville											
AL		605	8,900		86	605	8,986	9,591	579	2001	Nov-03
EdenGardens of Kingwood, TX		467	8,418		27	467	8,445	0.012	550	2001	Nov 02
EdenTerrace of Kingwood,		407	0,410		21	407	0,443	0,912	332	2001	Nov-03
TX EdenBrook		572	10,527		99	572	10,626	11,198	681	2001	Nov-03
of Louisville, KY	6,540	623	10,144	7	69	630	10,213	10,843	649	2001	Nov-03
EdenTerrace of Louisville, KY	7,769	886	11,897	5	15	801	11 012	12 803	746	2001	Nov-03
EdenGardens	1,109	000	11,097	5	13	071	11,912	12,603	740	2001	1404-03
of Marietta, GA		571	4,397		49	571	4,446	5,017	287	1998	Nov-03
EdenBrook of Plano, TX	6,273	464	12,004		28	464	12,032	12,496	730	2000	Nov-03
EdenGardens of Rock Hill,		255	ć 7 02			200	- 005	7.00 7	450	4005	
SC EdenBrook of The		277	6,783	23	222	300	7,005	7,305	459	1995	Nov-03
Woodlands, TX	4,901	395	13,490		40	395	13,530	13,925	822	2000	Nov-03
Summit at Park Hills,											
ОН		149	6,230			149	6,230	6,379	254	2001	Jun-04
Brighton Gardens of Carlsbad, CA	13,961	5,530	9,007			5,530	9,007	14,537	324	1999	Nov-04
Brighton	, ,	- ,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,			
Gardens of San Dimas,											
CA	12,535	3,390	19,788			3,390	19,788	23,178	632	1999	Nov-04
Brighton Gardens of Carmel											
Valley, CA	7,849	3,729	22,081			3,729	22,081	25,810	714	1999	Nov-04
Brighton Gardens of San Juan											
Capistrano,											
CA Dei-let-	4,380	3,009	5,144			3,009	5,144	8,153	234	1999	Nov-04
Brighton Gardens of Woodbridge,											
CT	3,777	1,624	5,457			1,624	5,457	7,081	192	1998	Nov-04
Brighton Gardens of Pikesville,											
MD	14,646	1,118	8,264			1,118	8,264	9,382	288	1999	Nov-04
Brighton Gardens of											
North Shore, MA	4,958	1,815	25,311			1,815	25,311	27,126	772	1999	Nov-04
Brighton Gardens of											
Dedham, MA Brighton Gardens of	11,055	1,806	18,682			1,806	18,682	20,488	613	1999	Nov-04
Paramus, NJ	12,226	2,826	20,012			2,826	20,012	22,838	653	1999	Nov-04
Brighton Gardens of Arlington,			13,907			4,658	13,907	18,565	458	1999	Nov-04
-8,											

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

VA									
Brighton									
Gardens of									
Richmond,									
VA	4,584	905	7,604	905	7,604	8,509	266	1999	Nov-04
Bickford									
Cottage of									
Davenport,									
IA	3,411	213	5,639	213	5,639	5,852	211	1999	Aug-04
Bickford									
Cottage of									
Marion, IA	2,794	224	5,711	224	5,711	5,935	213	1998	Aug-04
Bickford									
Cottage of									
Champaign,									
IL		54	2,501	54	2,501	2,555	100	2003	Aug-04
		F-:	54						

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

D. 10											
Bickford House of											
Bloomington,											
IL		514	6,866			514	6,866	7,380	262	2000	Aug-0
Bickford Cattage of											!
Cottage of Macomb, IL		54	4,315			54	4,315	4,369	164	2003	Aug-0
Bickford			1,010			اث	1,010	1,00,	10.	2002	Tiug C
Cottage of		27.5	7.650			27.5	7.650	0.004	200	2001	
Peoria, IL Courtyard		375	7,659			375	7,659	8,034	290	2001	Aug-0
Manor of											ŀ
Auburn Hills,											ŀ
MI		1,746	7,574		31	1,746	7,605	9,351	367	1999	Apr-04
Courtyard Manor at											
Sterling											
Heights, MI		1,076	7,834	5	11	1,081	7,845	8,926	375	1989	Apr-04
The Park at Olympia											ļ
Fields, IL	22,007	3,303	38,891			3,303	38,891	42,194	1,910	1999	Feb-0
East Bay											
Manor, RI Greenwich	9,372	686	12,752			686	12,752	13,438	652	1992	Feb-04
Bay Manor,											ļ
RI	6,540	180	11,401		103	180	11,504	11,684	573	1980	Feb-04
West Bay	10.002	1.000	15 401		70	1.000	15 560	17.460	760	1072	Esh O
Manor, RI Waterside	10,982	1,900	15,481		19	1,900	15,560	1/,400	/08	1972	Feb-04
Retirement											ļ
Estates, FL	26,033	1,820	32,645		56	1,820	32,701	34,521	1,575	1980	Feb-04
Carrington Pointe, CA	16.718	1.636	27,753	10		1 646	27 753	29,399	1 323	1988	Eeh-04
Cherry Hills	10,710	1,000	21,100	10		1,050	21,100	47,000	1,020	1760	Tto
Club, CA	9,815	1,428	23,814			1,428	23,814	25,242	1,178	1987	Feb-04
The Park at Golf Mills, IL	29 246	2.201	50 011		38	2.201	50 9/0	61,140	2 960	1080	Eab O
The Heritage	28,346	2,271	30,011		30	2,271	30,042	01,140	2,009	1909	F60-0-
Palmeras, AZ	32,319	1,556	45,622		428	1,556	46,050	47,606	2,246	1996	Feb-0
The Pointe at											
Newport Place, FL	4,696	900	6,453		37	900	6,490	7,390	376	2000	Feb-04
Newport	1,		0,				0,	,,,,,,			
Place, FL	28,938	5,265	41,850		110	5,265	41,960	47,225	2,063	1993	Feb-04
Prosperity Oaks, FL	21 039	5 415	59 690	63	519	5 478	60 209	65,687	2 903	1988	Feb-06
Pinecrest	21,000	3,413	37,070	05	317	3,410	00,207	05,007	2,703	1900	1000
Place											
Retirement											
Community, FL	31,815	893	60,674		149	893	60,823	61,716	2,968	1988	Feb-04
North Bay	,-		Í								
Manor, RI		464	19,402			464	19,402	19,866	973	1989	Feb-04
South Bay Manor, RI		654	16,606			654	16.606	17,260	821	1988	Feb-04
Emerald Bay		05.	10,000			05.	10,000	17,200	021	1700	1000
Manor, RI		1,382	18,237			1,382	18,237	19,619	925	1999	Feb-04
Treemont Retirement											
Community,											
TX		3,211	17,096		423	3,211	17,519	20,730	887	1974	Feb-04
The Park at											
Riverchase, AL		1,159	6,246			1,159	6,246	7,405	372	1997	Feb-04
Heron's Run,		1,100	0,210			1,107	0,210	7,703	J12	1///	1000
FL		446	1,798			446	1,798	2,244	89	1993	Feb-04
Sakonnet Bay Manor, RI		4 383	21,963			4 383	21 963	26,346	797	1998	Aug-0
Terrace at	19,000		33,496				33,482				Dec-04
Memorial											

City, TX

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Spring Shadows										
Place, TX	6,419	2,943	6,288		2,943	6,288	9,231	186	1973	Dec-04
Terrace at West										
University,										
TX Terrace at	17,281	3,650	24,976		3,650	24,976	28,626	763	1998	Dec-04
Willowbrook,										
TX	17,800	2,243	23,551		2,243	23,551	25,794	685	1996	Dec-04
Terrace at										
Clear Lake, TX	11.750	2.060	22.760		2.069	22.760	24 927	690	2000	Dag 04
Terrace at	11,/50	2,008	22,769		2,008	22,769	24,837	089	2000	Dec-04
First Colony,										
TX	17,750	2,160	22,871		2,160	22,871	25,031	691	2000	Dec-04
Sunrise of Des		4.420	46.004		4.420	16.001	20.442		2004	
Peres, MO Sunrise of		4,129	16,284		4,129	16,284	20,413	634	2004	Mar-04
Clayton, MO		3,565	14,819		3.565	14,819	18.384	735	2004	Mar-04
Sunrise of		, ,	,		,- ,-	,				
Wilmette, IL		2,640	7,053		2,640	7,053	9,693	296	2004	Mar-04
Boardwalk										
Medical Office, TX	7 587	1 665	11,366		1 665	11,366	13 031	707	1997	Apr-04
Las Colinas	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	1,000	11,500		1,000	11,000	10,001	, , ,	-,,,	11p1 0 1
Medical Plaza										
II, TX	6,863	1,763	8,801	1	1,763	8,802	10,565	665	2001	Apr-04
Independence Park-4204,										
NC	3.376	1,768	8,160	440	1,768	8.600	10,368	567	1994	Apr-04
Independence	-,	-,	0,200		-,, -	0,000	,		-,,	r- c
Park-4228,										
NC	1,070	888	2,483		888	2,483	3,371	243	1997	Apr-04
Independence Park-4233,										
NC	1,263	1,880	2,075		1,880	2,075	3,955	406	1996	Apr-04
Independence	,		,			,	,			•
Park-4323,			2 < 15			2 (1 =	2 2 4 4	40.5	400=	
NC Tampa	1,153	694	2,647		694	2,647	3,341	195	1997	Apr-04
Medical										
Tower, FL	6,069	2,648	7,243	736	2,648	7,979	10,627	1,355	1984	Apr-04
Yorktown 50,										
VA	14,722	2,089	22,618	337	2,089	22,955	25,044	1,553	1974	Apr-04
Sherman Oaks Medical										
Center, CA	9,545	9,024	5,272	158	9,024	5,430	14,454	1,127	1953	Apr-04
Valencia										
Medical	5 115	1 212	5.006	120	1 212	5 ACC	6.770		1002	. 01
Center, CA Encino	5,117	1,312	5,336	130	1,312	5,466	6,778	575	1983	Apr-04
Medical Plaza,										
CA	7,429	6,904	9,253	246	6,904	9,499	16,403	1,108	1973	Apr-04
Rocky										
Mountain										
Cancer Center, CO	4.601	1.069	7,801	45	1.069	7.846	8,915	466	1993	Apr-04
Aurora	.,001	1,00>	7,001		1,00>	7,010	0,710		1,,,,	прт о .
Medical										
Center II, CO	5,209	134	9,220	92	134	9,312	9,446	887	1994	Apr-04
Aurora Medical										
Center I, CO	4,653	123	8,485	142	123	8,627	8,750	921	1981	Apr-04
Dorsey Hall	.,		0,100			-,	-,			T-
Medical										
Chasanaska	3,833	1,324	4,020	51	1,324	4,071	5,395	504	1988	Apr-04
Chesapeake Medical										
Center, VA		2,087	7,520	11	2,087	7,531	9,618	793	1988	Apr-04

Randolph								
Medical								
Center, MD	2,575	6,453	644 2,575	7,097	9,672	656	1975	Apr-04
	F-56	5						

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Di 3 4 1' - 1											
Plano Medical Center, TX		2 519	12,190		140	2 519	12 330	14,849	1 125	1984	Apr-04
Medical Place		_,01)	12,170		. 10	_,51)	12,550	1 ,,077	1,123	1707	7.pr 0-
I, TX		19	24,746		402	19	25,148	25,167	2,568	1984	Apr-0
Northwest Regional Medical											
Center, TX		599	6,646		14	599	6,660	7,259	474	1999	Apr-04
The Diagnostic											
Diagnostic Clinic, FL BayCare		2,569	26,918		137	2,569	27,055	29,624	1,691	1972	Apr-04
Health											
Headquarters,		2.010	6.710			2.010	6.710	0.722	700	1000	
FL Southwest		3,019	6,713			3,019	6,713	9,732	/02	1988	Apr-0
General Birth Place, TX		990	12,308			990	12,308	13,298	559	1994	Aug-0
Baytown											
Plaza I & II, TX	1,200	337	1,096		1	337	1,097	1,434	237	1972	Aug-0
South	1,200	331	1,070		1	331	1,077	1,737	231	1712	Tug 0
Seminole											
Medical Office											
Building II,											
FL	2,710	709	4,063		51	709	4,114	4,823	486	1987	Aug-0
South											
Seminole Medical											
Office											
Building III,											
FL Outside	1,500	769	1,768		6	769	1,774	2,543	358	1993	Aug-0
Orlando Professional											
Center I, FL	800	384	788		29	384	817	1,201	162	1969	Aug-0
Orlando											J
Professional Center II, FL	1 600	1,258	1,704	322	20	1,581	1,733	3,314	274	1062	Δ11α ()
Oviedo	1,000	1,238	1,704	323	29	1,581	1,/33	3,314	214	1703	Aug-0
Medical											
Center, FL	4,500	1,712	6,484		439	1,452	6,923	8,375	1,101	1997	Aug-0
MedPlex B at Sand Lake											
Commons, FL	2,400	2,679	3,235		66	2,679	3,301	5,980	294	1988	Aug-0
Eagle Creek											_
Medical Plaza, KY	1 000	1.4	3 //11		250	1.4	3 660	3 602	501	1092	ΔησΩ
K Y Sand Lake	1,900	14	3,411		258	14	3,669	3,683	301	1982	Aug-0
Physicians											
Office							. =			400	
Building, FL North		23	1,748			23	1,748	1,771	161	1985	Aug-0
North Alvernon											
Medical, AZ	6,250	2,969	9,197		86	2,969	9,283	12,252	883	1986	Aug-0
St. Joseph's											
Medical Plaza, AZ	6,250	511	7,736		39	511	7,775	8,286	697	1985	Aug-0
Mercy	0,230	511	1,130		33	511	1,113	0,200	077	1703	rug-0
Medical											
Office	1.650		2.040		11		2.000	2.000	215	1006	A 0
Building Elgin Medical	1,650		3,049		11		3,060	3,060	315	1986	Aug-0
Office											
Building I, IL	4,000		6,291		74		6,365	6,365	581	1991	Aug-0
Elgin Medical											
Office Building II, IL	4,250		6,861		60		6,921	6,921	727	2001	Aug-0
Santa Rosa	.,250	13	8,111		196	13	8,307	8,320			Aug-0
Medical											·

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Office				
Building, GA				
Fannin				
Medical				
Office				
Building, GA	9 2,397	118	9 2,515 2,524	135 2002 Aug-0
Physicians				
East and				
West, TX	3 4,276	156	3 4,432 4,435	425 1991 Aug-0
Brentwood				
Medical				
Center, CA	10 26,331		10 26,331 26,341	453 2005 Aug-0
Heartland				
Regional				
Medical				
Office				
Building, IL	99 9,788	305	99 10,093 10,192	791 2002 Aug-0
	F-57			

Saint Joseph East		17	0.006		212	17	10 100	10 105	505	2002	
Office Park, KY Central Mississippi		17	9,896		212	1/	10,108	10,125	323	2003	4
Medical Center											
Building, MS		34	8,409		236	34	8,645	8,679	463	2002	
River Oaks Medical		27	0,407		230	57	0,043	0,077	703	2002	1
Building, MS		19	7,127		456	19	7,583	7,602	404	2003	
Parker Adventist		1)	7,127		730	1)	7,505	7,002	707	2003	-
Professional											
Building, CO		16	14,586	1	1,091	16	15,677	15 693	908	2004	
NASA Parkway		10	1 1,500		1,071	10	13,077	15,075	700	2001	Í
Medical Office											
Building, TX		460	7,478		19	460	7,497	7,957	453	2002	
Lake Granbury			.,				.,	.,,			
Medical Plaza, TX		63	6,197	1	1,690	63	7,887	7,950	330	2001	,
Durant Medical					,		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
Center, OK		1,133	7,914		170	1,133	8,084	9,217	467	1998	
Jackson Central II,		,	,,			,	.,	,,			_
MS	5,114		4,729				4,729	4,729	46	2005	1
McDowell	- /		,				,	, ,			
Mountain Medical											
Plaza, AZ	10,866	6,219	9,066	6	161	6,225	9,227	15,452	762	1999]
Lakeside		·	·				·	·			
Healthpark Medical											
Office Building, NE	11,750		12,024				12,024	12,024	200	2005]
Texarkana											
Professional											
Building, TX	7,585	1,061	7,620			1,061	7,620	8,681	255	1978	
The Park at Vernon											
Hills, IL	25,063	3,481	47,220			3,481	47,220	50,701	1,101	2001	
Oakbrook Terrace											
Medical Center I,											
IL		1,446	8,188			1,446	8,188	9,634	636	1989	
Oakbrook Terrace											
Medical Center II,											
IL		1,162	8,665			1,162	8,665	9,827	455	1986	1
Deaconess-Gateway											
Medical Office											
Building, IN	6,096		6,739				6,739	6,739	40	2005	
Canyon Hills Club,			•• ••				•• ••			1000	
CA	14,585	2,599	28,696			2,599	28,696	31,295	533	1989	l
Woodmont											
Retirement											
Residence, FL	4,986	388	9,120			388	9,120	9,508	172	1986]
Calaroga Terrace,											
OR	14,327	1,875	16,628			1,875	16,628	18,503	320	1968	1
Encore Senior											
Village at Naples,											
FL G :	786	1,005	1,280			1,005	1,280	2,285	31	1999	
Encore Senior											
Village at	2.555	505	1.500			505	1.500	E 115	0.1	1000	
Clearwater, FL	3,575	595	4,522			595	4,522	5,117	91	1999	1
	F-5	80									

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

E C '									
Encore Senior									
Village at Fort	2.476	1 400	4 417	1 400	4 417	5 017	0.5	1000	M 05
Myers, FL	2,4/6	1,400	4,417	1,400	4,417	5,817	85	1998	Mar-05
Encore Senior									
Village at	2.524	2 101	2.260	2 101	2.260	5 451		1000	
Greenacres, FL	3,534	2,191	3,260	2,191	3,260	5,451	66	1998	Mar-05
Encore Senior									
Village at									
Pensacola, FL	3,123	523	5,508	523	5,508	6,031	100	1997	Mar-05
Carpenter's									
Creek-Pensacola,									
FL	5,579	547	8,533	547	8,533				Mar-05
Valley Crest, CA	1,226	298	2,241	298	2,241	2,539	42	1986	Mar-05
Encore Senior									
Village at									
Riverside, CA	2,276	805	2,824	805	2,824	3,629	53	1997	Mar-05
Encore Senior									
Village at Peoria,									
AZ	7,216	1,241	8,810	1,241	8,810	10,051	158	1997	Mar-05
Encore Senior									
Village at									
Paradise Valley,									
AZ	3,181	1,649	3,357	1,649	3,357	5,006	61	1998	Mar-05
Encore Senior									
Village at									
Tucson, AZ	6,180		8,836		8,836	8,836	170	1999	Mar-05
Encore Senior									
Village at									
Portland, OR	7,308	889	10,491	889	10,491	11,380	203	1997	Mar-05
Millcreek									
Retirement									
Residence, UT	3,540	365	4,581	365	4,581	4,946	89	1996	Mar-05
Mary Washington									
Hospital, VA		1,403	2,675	1,403	2,675	4,078		(3)	Apr-05
Sierra Vista, CA	1,657	337	1,786	337	1,786	2,123	23	1990	Jun-05
Mission Surgery									
Center, TN			10,477		10,477	10,477	250	2003	Jun-05
Memorial Plaza,									
TN			36		36	36	341	1995	Jun-05
St. Vincent									
Clinic-South									
University, AR		653	3,008	653	3,008	3,661	45	1983	Jun-05
St. Vincent									
Clinic-Rodney									
Parham, AR		652	74	652	74	726	3	1972	Jun-05
St Anthony's, CO								(3)	Sep-05
Park Cities									•
Medical Plaza,									
TX	10.575	1,375	14,177	1,375	14,177	15,552	149	2002	Sep-05
Trophy Club	.,	,	,	,	,	- /			
Professional									
Office Building,									
TX	11.375	945	23,420	945	23.420	24.365	249	2004	Sep-05
Trophy Club	- 1,5 / 5	7.13	20,120	713	20,120	,505	/		2 c p 03
Medical Center,									
TX	16.380	1.444	16,161	1 444	16.161	17.605	202	2004	Sep-05
		F-59	10,101	1,777	10,101	17,003	202	200-F	30p 03
		1-39							

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Glen Lakes							
Health							
Plaza, TX	4,615	1,380	5,927			1,380	5,927
Valley							
View							
Medical							
Building,							
TX	3,315	1,122	4,010			1,122	4,010
Coppell							
Healthcare							
Center, TX	4,940	1,144	6,605			1,144	6,605
Meridian							
Medical							
Tower, OK	4,290	1,316	5,271			1,316	5,271
Meridian							
Medical							
Center, OK	2,165	657	3,634			657	3,634
St. Joseph							
Medical							
Center,							
MD	216		992				996
Memorial							
Hospital							
Cy-Fair,							
TX	189		538				538
Memorial							
Hospital							
Pearland,							
TX				0.0	1.001	0.0	1.000
Other				82	1,921	82	1,922

\$ 1,349,848 \$ 344,031 \$ 2,684,355 \$ 1,905 \$ 42,272 \$ 345,936 \$ 2,726,627

(1) Excludes encumbrances of \$111.9 million that are carried on Properties accounted for using the direct financing method.

(2) Includes Properties under construction.

(3) Property was under construction at December 31, 2005.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES

NOTES TO SCHEDULE III REAL ESTATE AND ACCUMULATED DEPRECIATION

December 31, 2005

(dollars in thousands)

(a)

Transactions in real estate and accumulated depreciation during 2003, 2004 and 2005 are summarized as follows:

	Accumulated
Cost(b)(d)	Depreciation

Property		
investments		
under operating		
leases:		
Balance,	\$ 241,200	\$ 3,765
December 31,		
2002	050 420	
Acquisitions	850,430	(222
Real estate	(6,602)	(232)
held for sale		16 245
Depreciation		16,345
expense(c)		
Balance,	1,085,028	19,878
December 31,		
2003		
Acquisitions	1,573,078	
Impairment	(1,883))
provisions	(1.550)	(506
Real estate held for sale	(1,559)	(526
mera rer sare		54,364
Depreciation		34,304
expense(c)		
D.I.	0.654.664	72.716
Balance,	2,654,664	73,716
December 31,		
2004	426 200	
Acquisitions	426,390 (7,740)	
Impairment provisions	(7,740)	
Real estate	(186)	(220
held for sale	(100)	(220)
Depreciation		84,250
expense(c)		01,230
I(-)		

\$3,073,128 \$ 157,746

Balance, December 31, 2005

393

Accumulated Cost(b)(d) Depreciation

(b)

As of December 31, 2005, 2004, and 2003 the aggregate cost of the Properties owned by the Company for federal income tax purposes, including Properties accounted for using the operating method and those accounted for using the direct financing method, was \$3.4 billion, \$3.0 billion and \$1.3 billion, respectively. Certain leases accounted for under the direct financing method are treated as operating leases for federal income tax purposes.

- (c)

 Depreciation expense is computed for buildings and equipment based upon estimated lives of 39 to 40 years, and 3 to 7 years, respectively.
- (d)

 Acquisition fees and miscellaneous closing costs of \$15.5 million, \$78.3 million and \$60.1 million are included in land, buildings, equipment and intangible lease costs at December 31, 2005, 2004 and 2003, respectively.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS

(in thousands, except per share data) (UNAUDITED)

December

	March 31, 2006	31, 2005
Assets		
Real estate		
investment		
properties:		
Accounted for		
using the		
operating		
method net of		
accumulated		
depreciation of		
\$180,953 and		
\$157,746	\$ 3,092,433	\$ 2,914,817
Accounted for		
using the direct		
financing	404 440	400 400
method	491,239	488,683
Intangible lease		
costs net of		
accumulated		
amortization of		
\$29,958 and	110 500	00 (11
\$26,021	110,580	99,611
	3,694,252	3,503,111
Cash and cash	102.201	0.4.000
equivalents	102,204	94,902
Restricted cash	22,767	21,920
Accounts and		
other	c	
receivables net of	I	
allowance for		
doubtful accounts		
of \$8,800 and \$7,200	19,848	22 496
Deferred costs,	19,040	23,486
net	23,283	24,705
Accrued rental	23,203	24,703
income	110,330	99,219
Other assets	64,043	52,935
Real estate held	0-1,0-13	32,733
for sale	6,921	12,692
Goodwill	5,791	5,791
500 G 11111	5,771	3,771
	¢ 4 0 40 420	¢ 2 929 761
	\$ 4,049,439	\$ 3,838,761

Liabilities and stockholders' equity

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	March 31, 2006	December 31, 2005
Liabilities:		
Mortgages		
payable	\$ 1,353,713	
Bonds payable	101,188	98,016
Construction loans payable	108,472	143,560
Line of credit	105,472	75,000
Due to related	100,000	75,000
parties	831	2,386
Accounts		
payable and	24.240	21.025
other liabilities	36,218	31,035
Intangible lease liability, net	4,744	4,505
Deferred income	8,055	6,607
Security deposits	23,386	23,954
Total liabilities	1,741,607	1,605,253
Commitments and contingencies		
Minority interests	8,270	5,701
Stockholders' equity:		
Preferred stock,		
without par		
value Authorized and unissued		
3,000 shares		
Excess shares,		
\$.01 par value		
per share		
Authorized and unissued		
103,000 shares		
Common stock,		
\$.01 par value		
per share		
Authorized one		
billion shares, issued 270,682		
and 260,293		
shares.		
respectively,		
outstanding		
264,193 and		
255,527 shares,	2 (42	0.555
respectively Capital in excess	2,642	2,555
of par value	2,373,850	2,295,307
Accumulated	, - , 0	, ,=
distributions in		
excess of net	(0.1.515)	(54.004)
income	(84,641) 7,711	(74,894) 4,839
	7,/11	4,839

	March 31, 2006	December 31, 2005		
Accumulated other comprehensive income				
Total stockholders' equity	2,299,562	2,227,807		
	\$ 4,049,439	\$ 3,838,761		
See accompanying notes to condensed consolidated financial statements.				
	F-62			

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED)

(in thousands, except per share data)

	Three months ended March 31,	
	2006	2005
Revenues:		
Seniors' Housing:		
Rental income		
from operating		
leases	\$ 62,265	55,578
Earned income		
from direct	4 = 0 < 0	
financing leases	15,969	15,312
FF&E reserve	1 002	1 500
income	1,992 101	1,582 1,721
Contingent rent Medical Facilities:	101	1,/21
Rental income		
from operating		
leases	19,057	13,136
Tenant expense	,	
reimbursements	4,583	2,710
Property		
management and		
development fees	294	1,290
Loan interest	610	
income	618	
	104,879	91,329
г		
Expenses:		
Seniors' Housing property expenses	283	255
Medical Facilities	203	233
operating expenses	7,484	5,486
General and	1,101	-,
administrative	4,744	4,028
Asset management		
fees to related		
party	5,098	4,325
Provision for	1 500	7.50
doubtful accounts	1,523	750
Depreciation and amortization	26.052	22,737
amortization	26,952	22,737
	46.004	27.501
	46,084	37,581
0		52.540
	58,795	53,748
Operating income Interest and other	58,795	,
		53,748 631 (15,539)

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	March 31,		
Interest and loan			
cost amortization			
expense			
Income before			
equity in earnings of			
unconsolidated			
entity, minority			
interests in income			
of consolidated			
subsidiaries and			
discontinued	26.20# 2	0.040	
operations	36,305 3	8,840	
Equity in earnings			
of unconsolidated			
entity	2	2	
Minority interests in			
income of			
consolidated			
subsidiaries	(86)	(381)	
Income from			
continuing			
operations	36,221 3	8,461	
Loss from	00,221	0, 101	
discontinued			
operations	(469)	5,826)	
operations	(402)	3,020)	
Net income	\$ 35,752 \$ 3	2,635	
Net income (loss)			
per share of			
common stock			
(basic and diluted)			
From continuing			
operations	\$ 0.14 \$	0.16	
From discontinued	φ 0.14 ψ	0.10	
operations		(0.02)	
operations		(0.02)	
	\$ 0.14 \$	0.14	
Weighted-average			
number of shares of			
common stock			
outstanding (basic			
and diluted)	257,507 24	0,699	
•			
Distributions			
declared per			
common share	\$ 0.1776 \$ 0	1776	
common snarc	Ψ 0.1770 Ψ 0	.1770	

Three months ended

See accompanying notes to condensed consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY

Three Months Ended March 31, 2006 (UNAUDITED)

(in thousands, except per share data)

	Commo	n stock				
	Number of shares	Par value		istributions in excession of net con income	s	
Balance at December 31, 2005 Net income	255,527	\$ 2,555	\$ 2,295,307	\$ (74,894) 35,752	\$ 4,839	\$ 2,227,807 35,752
Change in fair value of cash flow hedges				33,732	2,872	2,872
Total comprehensive income Subscriptions received for common stock through public offerings and						38,624
reinvestment plan	10,390	104	103,081			103,185
Redemption of common stock Stock issuance	(1,724)	(17)				(16,379)
costs Distributions declared (\$0.1776 per share)			(8,176)	(45,499)		(8,176)
Balance at March 31, 2006	264,193	\$ 2,642	\$ 2,373,850	\$ (84,641)	\$ 7,711 \$	\$ 2,299,562

See accompanying notes to condensed consolidated financial statements.

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

	Three Months Ended March 31,		
	2006	2005	
Increase (decrease) in cash and cash equivalents:			
Net cash provided by operating activities	\$ 55,511	\$ 52,814	
Investing			
activities:			
Investment in land, buildings and equipment	(177,370)	(194,641)	
Investment in direct financing			
leases Investment in	(300)		
intangible lease costs	(14,434)	(8,613)	
Investment in			
Senior Secured	(10 000)		
Term Loan Proceeds from	(18,000)		
sale of Properties	1,155		
Payment of	,		
acquisition fees and costs	(7,458)	(10,224)	
Payment of deferred leasing	(1.152)	(265)	
costs Decrease	(1,173)	(265)	
(increase) in restricted cash	(591)	12,251	
Net cash used in			
investing activities	(218,171)	(201,492)	
Financing			
activities: Proceeds from			
borrowings on			
mortgages			
payable	136,520	219,010	
Principal			
payments on mortgages			
payable	(2,896)	(1,767)	
· ·			

	Three Months Ended March 31,		
Proceeds from			
issuance of bonds	- 0-1	2 440	
payable Retirement of	5,371	2,449	
bonds payable	(2,199)	(2,481)	
Proceeds from	(=,1))	(2,101)	
construction			
loans payable	10,081	16,685	
Repayments of			
construction loans payable	(45,170)		
Proceeds from	(45,170)		
line of credit	45,000		
Repayment of			
line of credit	(15,000)		
Payment on term		(60,000)	
Refund of loan		(60,000)	
costs	2,657		
Payment of loan	,		
costs	(1,533)	(2,036)	
Contributions			
from minority interests	2,780	629	
Distributions to	2,700	02)	
minority interests	(296)	(11)	
Subscriptions			
received from stockholders	102 105	97 522	
Distributions to	103,185	87,533	
stockholders	(45,499)	(42,593)	
Redemption of			
common stock	(13,576)	(11,343)	
Payment of stock issuance costs	(9,463)	(7,468)	
issuance costs	(2,403)	(7,400)	
Net cash			
provided by			
financing			
activities	169,962	198,607	
Net increase in			
cash and cash			
equivalents	7,302	49,929	
Cash and cash			
equivalents at beginning of			
period	94,902	51,781	
		,	
Cash and cash			
equivalents at end			
of period	\$ 102,204 \$	101,710	
Supplemental			
schedule of			
non-cash investing			
and financing			

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Three Months Ended March 31, activities: Mortgage Loans issued in connection with the sale of Properties 4,800 \$ Mortgages assumed on properties purchased \$ 41,406 See accompanying notes to condensed consolidated financial statements. F-65

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Three months ended March 31, 2006 and 2005 (Unaudited)

1. Organizational and Basis of Presentation:

Organization CNL Retirement
Properties, Inc., a Maryland corporation, was organized in December 1997 to operate as a real estate investment trust (a "REIT") for federal income tax purposes. Throughout this document, CNL Retirement Properties, Inc. and each of its subsidiaries and several consolidated partnerships and joint ventures are referred to as "we", "us" and "our." Various other wholly owned or majority owned subsidiaries are expected to be formed in the future for the purpose of acquiring or developing additional real estate properties and holding other permitted investments.

We acquire primarily real estate properties related to seniors' housing and health care facilities (the "Properties") located primarily across the United States. The Properties may include independent living, assisted living and skilled nursing facilities, continuing care retirement communities ("CCRC") and life care communities (collectively "Seniors' Housing"), medical office buildings, specialty and walk-in clinics, free standing ambulatory surgery centers, specialty or general hospitals and other types of health care-related facilities (collectively "Medical Facilities"). Seniors' Housing facilities are generally leased on a long-term, triple-net basis and Medical Facilities are generally leased on a shorter-term, gross or triple-net basis. We may provide mortgage financing loans ("Mortgage Loans"), furniture, fixture and equipment financing ("Secured Equipment Leases") and other loans to operators or developers of Seniors' Housing and Medical Facilities. In addition, we may invest up to a maximum of 5% of total assets in equity interests in businesses, including those that provide services to or are otherwise ancillary to the retirement and health care industries. We operate in one business segment, which is the ownership, development, management and leasing of health care-related real estate. At March 31, 2006, we owned 185 Seniors' Housing facilities, 86 Medical Facilities, including 2 specialty hospitals and 2 walk-in clinics, and 2 Seniors' Housing facilities and a parcel of land that we hold for sale.

We retained CNL Retirement Corp. (the "Advisor") as our advisor to provide management, acquisition, advisory and administrative services

relating to our Properties, Mortgage Loans, Secured Equipment Lease program, other loans and other permitted investments pursuant to an advisory agreement dated May 14, 2004 (the "Advisory Agreement") that was renewed pursuant to a renewal agreement effective May 3, 2005 for a one-year term (the "2005 Renewal Agreement") and was amended by an amendment to the 2005 Renewal Agreement on July 13, 2005 (the "2005 Renewal Amendment" together with the 2005 Renewal Agreement, the "2005 Renewal Agreements"). On May 1, 2006, we entered into a renewal agreement (the "2006 Renewal Agreement") with the Advisor, pursuant to which the Advisory Agreement was renewed, as amended by the 2005 Renewal Agreements, for an additional one-year term commencing on May 3, 2006 and ending on May 3, 2007. The Advisory Agreement may be terminated at an earlier date upon 60 days prior written notice by either party or by mutual consent of the parties.

Strategic Alliances In 2005, we entered into an agreement with The Cirrus Group, LLC ("Cirrus"), a development and property management company, to acquire, at our election, Medical Facilities, some of which have yet to be developed. The acquisitions contemplated under this agreement are expected to occur over a five-year term, subject to certain conditions, or until \$1.0 billion is invested in Medical Facilities, including specialty hospitals. We will have minority interest partners in connection with the ownership of each of these Properties, including Cirrus principals, physicians and

other investors associated with Cirrus principals. As of March 31, 2006, we had acquired a majority equity interest in two Medical Facilities for \$52.6 million under this agreement, for which Cirrus and its affiliates made \$0.9 million in minority interest contributions.

At March 31, 2006, Cirrus managed 23 of our Medical Facilities, including two Properties that we acquired from third parties.

In 2005, we entered into an agreement to provide a Cirrus affiliate with an interest only, five-year senior secured term loan under which up to \$85.0 million (plus capitalized interest) may be borrowed to finance the acquisition, development, syndication and operation of new and existing surgical partnerships ("Senior Secured Term Loan"). At March 31, 2006, the balance outstanding under the Senior Secured Term Loan was \$34.0 million. In connection with the Senior Secured Term Loan, we received stock warrants which are exercisable into a 10% to 15% ownership interest of the borrower. The stock warrants are exercisable at the earlier of an event of default or the full repayment of the Senior Secured Term Loan and expire in September 2015.

We own a 55% controlling interest in The DASCO Companies, LLC ("DASCO"), a development and property management company. Our relationship with DASCO has provided and may continue to provide opportunities for us to participate in new Medical Facility development and acquisition opportunities as well as Medical Facilities management. DASCO may also provide development and property management services to third parties. At March 31, 2006, DASCO managed fifty-four of our Medical Facilities, including two walk-in clinics and was developing five of our Medical Facilities.

Basis of Presentation The accompanying condensed consolidated financial statements (the "consolidated financial statements") have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") for interim financial information and with the instructions to Form 10-O. Accordingly, they do not include all of the information and note disclosures required by GAAP for complete financial statements. The accompanying consolidated financial statements reflect all adjustments, consisting of normal recurring adjustments, which are, in the opinion of management, necessary for a fair statement of the results for the interim periods presented. Operating results for the three months ended March 31, 2006, may not be indicative of the results that may be

expected for the year ending December 31, 2006. Amounts included in the financial statements as of December 31, 2005, have been derived from the audited financial statements.

These consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Annual Report on Form 10-K of CNL Retirement Properties, Inc. and its subsidiaries for the year ended December 31, 2005. The accompanying consolidated financial statements include the accounts of our wholly owned subsidiaries, DASCO and other entities in which we own a majority and controlling interest. Interests of unaffiliated third parties in less than 100% owned and majority controlled entities are reflected as minority interests. All significant intercompany balances and transactions have been eliminated in consolidation.

Reclassifications Certain items in the prior periods' financial statements have been reclassified to conform to the 2006 presentation, including those related to our real estate held for sale (see Note 5). These reclassifications had no effect on reported equity or net income.

2. Public Offerings:

We completed our fifth public offering (the "2004 Offering") on March 26, 2006. During the three months ended March 31, 2006, we raised \$103.2 million in subscription proceeds from the 2004 Offering. Total subscription proceeds received from the 2004 Offering and the four prior public offerings amount to \$2.7 billion at March 31, 2006.

The price per share of all of the equity offerings of our common stock was \$10.00 per share, with the exception of (i) shares purchased pursuant to volume or other discounts and (ii) shares purchased through our reinvestment plan since the beginning of the 2004 Offering, which have been priced at \$9.50 per share.

3. Acquisitions:

In January 2006, we acquired majority equity interests in seven Medical Facilities for \$84.5 million which we funded, in part, with proceeds from a new \$56.3 million, ten-year mortgage loan. Four of the acquired Properties are located in Texas, two are in Arizona and one is in Missouri, and in aggregate they contain 323,000 square feet. Cirrus will manage the Properties.

In February 2006, we acquired a Seniors' Housing Property that is being developed. The project is expected to be completed in the fourth quarter of 2006 with an estimated cost of \$5.7 million. The 46-unit assisted living facility is located in Michigan.

In March 2006, we acquired majority equity interests in five Medical Facilities for \$72.6 million which we funded, in part, with proceeds from a new \$47.2 million, ten-year mortgage loan. Four of the Medical Facilities are located in Texas, and one is in Oklahoma, and in aggregate they contain 268,000 square feet. Cirrus will manage the Properties.

In March 2006, we acquired a majority equity interest in a Medical Facility for \$24.5 million. The Medical Facility is located in California and contains 55,000 square feet. Cirrus will manage the property.

4. Other Assets:

Other assets included the following (in thousands):

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	March 31, 2006	31, 2005
Senior Secured Term Loan	\$ 34,000	\$ 16,000
Property acquisition deposits		10,601
Deferred receivables	8,223	6,638
Fair value of cash flow hedges	7,711	4,839
Mortgage Loan receivable	4,800	
Prepaid expenses	2,999	4,950
Acquisition costs	4,055	7,633
Other	2,255	2,274
	\$ 64,043	\$ 52,935
F	-68	

5. Real Estate Held For Sale:

In March 2006, we sold two Properties which were classified as held for sale to an unrelated third party for \$6.0 million and recorded a net loss of \$0.5 million. We issued a Mortgage Loan receivable with a three-year term secured by the Properties in the amount of \$4.8 million. This amount is included in other assets on our consolidated balance sheet as of March 31, 2006. Interest is payable annually at a rate of 6.0% and principal is due at maturity. In January 2006, we entered into an agreement to sell one additional Property for an expected sales price of \$2.1 million.

As of March 31, 2006, real estate held for sale included two Seniors' Housing facilities and a parcel of land with an aggregate net carrying value of \$6.9 million.

The operational results associated with Properties classified as held for sale were presented as loss from discontinued operations in the accompanying consolidated statements of income. Summarized financial information was as follows (in thousands):

Thusa Months

	Three Months Ended March 31,		
	2006	2005	
Rental income from operating leases	\$ 148	3 \$ 528	
Provision for doubtful accounts	(124	4)	
Impairment provisions		(6,197)	
Net loss on disposal of Properties	(450	0)	
Loss from discontinued operations	(469	9) (5,826)	

6. Indebtedness:

Mortgages payable At March 31, 2006, we had \$1.4 billion in mortgage debt secured by Properties with an aggregate carrying amount of \$2.4 billion. Interest rates on the mortgage notes ranged from 4.85% to 8.42% with a weighted-average rate of 5.87% at March 31, 2006.

In January 2006, we entered into a \$56.3 million, ten-year mortgage loan that bears fixed-rate interest at 5.59%. Payments for the first

five years are interest only, with principal payments beginning in March 2011.

In February 2006, we entered into a \$33.0 million mortgage loan and used the proceeds and cash on hand to pre-pay a \$48.0 million construction loan facility with a principal balance of \$41.9 million. The new interest-only, five-year loan bears interest at a rate equal to LIBOR plus 150 basis points (6.13% all-in rate at March 31, 2006).

In March 2006, we entered into a \$47.2 million, ten-year mortgage loan that bears fixed-rate interest at 5.81%. Payments for the first five years are interest only, with principal payments beginning in May 2011.

Construction loans payable Total construction loans outstanding at March 31, 2006, were \$108.5 million, and total liquidity remaining was \$33.7 million. During the three months ended March 31, 2006, we prepaid a construction loan facility with a \$41.9 million balance, entered into a new construction loan facility of \$7.7 million and collectively drew a net of \$10.1 million under all of our

construction loans related to certain Properties in various stages of development. The loans are variable interest rate loans and mature from November 2006 through December 2013. We anticipate that we will obtain permanent financing to repay the construction loans as they become due.

Line of Credit At March 31, 2006, \$105.0 million was outstanding under our \$320.0 million two-year senior secured revolving line of credit (the "Revolving LOC"). The Revolving LOC requires interest-only payments at LIBOR plus a percentage that fluctuates depending on our aggregate amount of debt outstanding in relation to our total assets (6.52% all-in rate at March 31, 2006, which represents a pricing of LIBOR plus 170 basis points). The amount available for use under the Revolving LOC is subject to certain limitations based on the pledged collateral. As of March 31, 2006, the Revolving LOC was collateralized by 36 Properties with a carrying value of \$389.4 million that, in the aggregate, allowed us to draw up to \$283.0 million.

7. Financial Instruments: Derivatives and Hedging:

In May 2005, we entered into two interest rate swap agreements effective June 1, 2005, and one interest rate swap agreement effective July 1, 2005, for an aggregate notional amount of \$233.8 million to hedge against unfavorable fluctuations in interest rates on our variable interest rate mortgage notes payable. At March 31, 2006, the fair value of these contracts was \$7.7 million and was included in other assets in the accompanying consolidated balance sheets. The change in net unrealized gain of \$2.9 million as of March 31, 2006, for derivatives designated as cash flow hedges is disclosed separately in the accompanying consolidated statements of stockholders' equity as the change in fair value of cash flow hedges. The effective portion of gains and losses on these contracts are recognized in accumulated other comprehensive income whereas the ineffective portions are recognized in earnings. During the three months ended March 31, 2006, the ineffective portion of these hedges was not significant.

8. Related Party Transactions:

Pursuant to the Advisory Agreement, as amended and renewed, the Advisor and its affiliates earn certain fees and are entitled to receive reimbursement of certain expenses.

Acquisition fees During the three months ended March 31, 2006 and 2005, we incurred acquisition fees of \$7.3 million and \$13.5 million, respectively, for, among other things, identifying Properties and structuring the terms of the leases (equal to 3.0% of gross offering proceeds and loan proceeds from permanent financing under the 2004 Offering). These fees are included in other assets in the accompanying consolidated balance sheets prior to being allocated to individual Properties or intangible lease costs.

Management fees We incurred monthly asset management fees totaling \$5.1 million and \$4.5 million during the three months ended March 31, 2006 and 2005, respectively (0.05% of the amount actually paid or allocated to the purchase, development, construction or improvement of a property, exclusive of acquisition fees and acquisition expenses, and the outstanding principal balance of any Mortgage Loans as of the end of the preceding month).

Administrative services Our Advisor and its affiliates provide various administrative services, including, but not limited to, accounting; financial, tax, insurance administration and regulatory compliance reporting; stockholder distributions and reporting; due diligence and marketing; and investor relations. During the three months ended March 31, 2006 and 2005, we incurred \$0.6 million and \$2.0 million for these services, respectively.

Offering expenses Offering expenses incurred by the Advisor and its affiliates on our behalf, together with selling commissions, the marketing support fee and due diligence expense reimbursements were \$8.2 million during each of the quarters ended March 31, 2006 and 2005. These amounts are treated as stock issuance costs and charged to stockholders' equity.

9. Subsequent Events:

Distributions On April 1 and May 1, 2006, our Board of Directors authorized distributions to stockholders of record on those dates, totaling \$31.3 million, or \$0.0592 per share of common stock at each record date, payable by June 30, 2006.

Advisory Agreement On May 1, 2006, we entered into the 2006 Renewal Agreement with the Advisor, pursuant to which the Advisory Agreement was renewed, as amended by the 2005 Renewal Agreements, for an additional one-year term commencing on May 3, 2006 and ending on May 3, 2007.

Pending Merger On May 1, 2006, we entered into a definitive Agreement and Plan of Merger (the "Merger Agreement") with Health Care Property Investors, Inc., a Maryland corporation ("HCP") and Ocean Acquisition 1, Inc., a Maryland corporation and a wholly owned subsidiary of HCP ("Merger Sub"), pursuant to which we have agreed to merge (the "Merger") with and into Merger Sub, with Merger Sub continuing as the surviving corporation. Under the terms of the Merger Agreement, at the effective time of the Merger, each share of our common stock, par value \$0.01, issued and outstanding immediately prior to the effective time of the Merger (other than shares held by HCP, Merger Sub, us or any of their or our respective wholly owned subsidiaries, and any dissenting stockholders), will be converted into the right to receive consideration equivalent in value to approximately \$13.50 per share (without interest), consisting of approximately:

\$11.13 in cash (representing approximately 82% of the total consideration per share); and

0.0865 shares of HCP common stock, par value \$1.00 per share.

As of May 1, 2006, we had approximately 264.2 million shares of common stock outstanding. HCP will also assume approximately \$1.6 billion of our outstanding debt.

Simultaneously with the execution of the Merger Agreement, HCP entered into a merger agreement (the "Advisor Merger Agreement") with the Advisor and the stockholders of the Advisor, pursuant to which HCP has agreed to acquire the Advisor for shares of HCP common stock valued at approximately \$120.0 million (the "Advisor Merger"). The consummation of the Merger and the Advisor Merger are each conditioned upon the consummation of the other. There can be no assurances that the Merger and the Advisor Merger will be consummated.

Redemption Plan Our Board of Directors has determined that it is in the best interest of our company to suspend our redemption plan, beginning with the second quarter of 2006. The suspension of our redemption plan is effective as of June 15, 2006, and therefore no shares of our common stock will be redeemed for the second quarter of 2006.

Reinvestment Plan Our Board of Directors has also determined that it is in the best interest of our company to terminate our distribution reinvestment plan, beginning with the second quarter of 2006. The termination of our distribution reinvestment plan is effective as of June 15, 2006, and therefore no distributions to our stockholders will be reinvested in shares of our common stock pursuant to our distribution reinvestment plan for the second quarter of 2006.

Report of Independent Certified Public Accountants

To the Board of Directors and Stockholders of CNL Retirement Corp.

In our opinion, the accompanying consolidated balance sheets and the related consolidated statements of income, of stockholder's equity and of cash flows present fairly, in all material respects, the financial position of CNL Retirement Corp. and its subsidiary (the "Company") at December 31, 2005 and 2004, and the results of its operations and its cash flows for the years then ended in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

As discussed in Note 1, the Company provides management, advisory and administrative services, and assists in developing and identifying seniors' housing and health care-related properties and obtaining financing for CNL Retirement Properties, Inc. and subsidiaries.

/s/ PricewaterhouseCoopers LLP

Orlando, Florida April 7, 2006

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED BALANCE SHEETS December 31, 2005 and 2004

	2005	2004
ASSETS		
Current assets:		
Cash and cash		
equivalents	\$ 1,492,063	\$ 5,235,669
Due from related		
parties	2,129,615	1,397,956
Prepaid expenses	70,057	122,780
Total current assets Property and equipment, net Other assets, at cost	3,691,735 2,466,460 245,137	6,756,405 1,457,363 230,799
	\$ 6,403,332	\$ 8,444,567
LIABILITIES AND		

STOCKHOLDER'S **EQUITY**

Current liabilities:		
Accounts		
payable and		
accrued expenses	\$ 3,678,057	\$ 2,357,176
Due to related		1.500.150
parties		1,728,172
Total current		
liabilities	3,678,057	4,085,348
Deferred rent	202,133	165,279
expense	202,133	103,279
Total liabilities and deferred expense	3,880,190	4,250,627
Commitments		
and contingencies		
Stockholder's		
equity:	4.000	4.000
Class A common	1,000	1,000
stock; \$1 par value per share;		
10,000 shares		
authorized; 1,000		

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	2005	2004
shares issued and		
outstanding		
Class B common		
stock; \$1 par		
value per share;		
5,000 shares		
authorized; 1,027		
shares issued and		
outstanding	1,027	
Additional paid		
in capital	2,521,115	
Retained	, ,	
earnings		4,192,940
8		, , , , ,
 1		
Total		
stockholder's		
equity	2,523,142	4,193,940
	\$ 6,403,332	\$ 8,444,567
	,,	, .,

The accompanying notes are an integral part of these consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED STATEMENTS OF INCOME

For the years ended December 31, 2005 and $2004\,$

	2005	2004	
Revenue:			
Acquisition fees	\$ 6,349,471	\$ 38,146,953	
Debt acquisition fees	13,789,410	29,951,684	
Management fees leases	19,143,304	13,032,700	
Development fees	26,632	231,847	
Interest and other income	3,008,244	1,860,112	
	42,317,061	83,223,296	
T.			
Expenses: Corporate services	\$ 4,106,138	\$ 23,362,267	
provided by related parties			
Salaries and benefits	12,735,042	9,641,686	
General and administrative	4,222,131	3,642,123	
Rent	914,127	838,582	
Depreciation and	801,137	497,263	
amortization Interest		5,057	
	22,778,575	37,986,978	
Income before provision for	19,538,486	45,236,318	
Provision for income taxes	7,473,471	17,049,512	
Net Income	\$ 12,065,015	\$ 28,186,806	

The accompanying notes are an integral part of these consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED STATEMENTS OF STOCKHOLDER'S EQUITY

For the years ended December 31, 2005 and 2004

	Class A Common Stock		Class B Common Stock			
	Shares Outstanding	Par Value	Shares Outstanding	Par Value	Additional paid in Capital	Retained Earnings
Balance, December 31, 2003	1,000	\$ 1,000		\$	\$	\$ 402,3
Dividends paid						(24,396,1
Net income						28,186,8
Balance, December 31, 2004	1,000	1,000				4,192,9
Shares issued			1,027	1,027	8,626,674	
Dividends paid Net					(6,105,559) (16,257,9
income						12,065,0
Balance, December 31, 2005	1,000	\$ 1,000	1,027	\$ 1,027	\$ 2,521,115	\$

The accompanying notes are an integral part of these consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED STATEMENTS OF CASH FLOWS

For the years ended December 31, 2005 and $2004\,$

_	2005	2004	
Cash flows			
from			
operating			
activities:			
Net income \$	12,065,015	\$ 28,186,806	
Adjustments			
to reconcile			
net income			
to net cash			
provided by			
operating			
activities:			
Depreciation expense	801,137	497,263	
Loss on	801,137	497,203	
sale of			
assets	4,373		
Changes in	1,575		
operating			
assets and			
liabilities:			
Due from			
related			
parties	(731,659)	(1,373,812)	
Prepaid			
expenses	52,723	(45,564)	
Accounts			
payable			
and			
accrued	726,239	1,441,273	
expenses Due to	120,239	1,441,273	
related			
parties	(1,728,172)	1,728,172	
Deferred	(1,720,172)	1,720,172	
rent			
expense	36,854	57,345	
_			
Net cash			
provided			
by			
operating			
activities	11,226,510	30,491,483	
_			
Cash flows			
from			
investing			
activities:			
Purchases of			
property and			
equipment	(1,219,965)	(704,817)	

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	2005	2004
Increase in other assets	(14,338)	
Net cash used in investing activities	(1,234,303)	(704,817)
Cash flows from financing		
activities: Principal		
payments on capital lease obligations Issuance of		(157,358)
B shares Dividends paid to	8,627,701	
stockholders	(22,363,514)	(24,396,169)
Net cash used in financing activities	(13,735,813)	(24,553,527)
Net (decrease) increase in cash and cash equivalents Cash and cash equivalents at beginning of period	(3,743,606) 5,235,669	5,233,139
Cash and		
cash equivalents at end of period	\$ 1,492,063	\$ 5,235,669
Supplemental disclosure of cash flow information: Cash paid during the year for interest	\$	\$ 5,057
Cash paid during the year for income taxes	\$ 7,473,471	\$ 15,321,340

The accompanying notes are an integral part of these consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

For the years ended December 31, 2005 and 2004

Note 1 Significant Accounting Policies:

Organization and Nature of Business CNL Retirement Corp. (the "Company"), a Florida C corporation, was organized on July 1, 1997. The Company is owned by CNL Real Estate Group, Inc., (the "Parent"), a wholly-owned subsidiary of CNL Financial Group, Inc. ("CFG").

The Company and its wholly-owned subsidiary, CNL Retirement Development Company ("CRDC"), provide management, advisory and administrative services, and assists in identifying and acquiring seniors' housing and health care-related properties and obtaining financing for CNL Retirement Properties, Inc. and its subsidiaries ("CRP"). CRP is a Real Estate Investment Trust registered with the Securities and Exchange Commission.

Principles of Consolidation The accompanying consolidated financial statements include the accounts of the Company and CRDC. All significant intercompany balances and transactions have been eliminated.

Cash and Cash Equivalents The Company considers all highly liquid investments with a maturity of three months or less to be cash equivalents. Cash and cash equivalents consist of demand deposits at commercial banks and money market funds. Cash equivalents are stated at cost, which approximates fair value.

Cash accounts maintained on behalf of the Company in demand deposits at commercial banks and money market funds may exceed federally insured levels; however, the Company has not experienced any losses in such accounts.

Property and Equipment Property and equipment is stated at cost less accumulated depreciation. For financial statement purposes, the Company changed to the straight-line method of depreciation effective January 1, 2005, for all newly acquired property and equipment. Assets acquired before the effective date of the change continue to be depreciated principally by accelerated methods. The estimated useful lives of the property and equipment range from 3 to 15 years. The Company believes the new

straight-line method will more accurately reflect its financial results by better matching costs of new property over the useful lives of these assets. The effect of the change was not material to the 2005 financial results of operations. Leasehold improvements are amortized over the life of the improvement or the term of the lease, whichever is shorter.

Revenue Recognition The Company records revenue relating to services performed under the terms of its advisory agreement with CRP as follows:

Acquisition Fees As offering proceeds are received by CRP and available for investment at the rate of 3.0 to 4.0 percent of gross offering proceeds. (See Note 2.)

Debt Acquisition Fees Upon closing of permanent financing at the rate of 3.0 to 4.0 percent of loan proceeds. (See Note 2.)

Collectively, the acquisition fees on gross equity proceeds and the debt acquisition fees on loan proceeds are referred to as "Acquisition Fees".

Development Fees As the related services are performed in amounts equal to a negotiated percentage of anticipated project costs.

Management Fees On a monthly basis at ¹/₁₂ of .6 percent of CRP's real estate asset value and the outstanding principal balance of any mortgage loans as of the end of the preceding month.

Personnel Reimbursement Charges On a monthly basis as the related costs are incurred.

Income Taxes The Company's taxable income or loss is included in its Parent's consolidated federal and state income tax returns. The Company accounts for income taxes under a tax sharing arrangement that approximates the provision that would be recognized if it were filing tax returns on a stand-alone basis.

Use of Estimates Management of the Company has made a number of estimates and assumptions relating to the reporting of assets and liabilities and the disclosure of contingent assets and liabilities to prepare these consolidated financial statements in conformity with generally accepted accounting principles. Actual results could differ from those estimates.

Recent Accounting Pronouncements In May 2005, the Financial Accounting Standards Board (the "FASB") issued Statement of Financial Accounting Standards No. 154, Accounting Changes and Error Corrections ("FAS 154"), which replaces Accounting Principles Board Opinion No. 20, Accounting Changes, and Statement of Financial Accounting Standards Board No. 3, Reporting Changes in Interim Financial Statements An Amendment of APB Opinion No. 28. FAS 154 provides guidance on the accounting for and reporting of accounting changes and error corrections. It establishes retrospective application as the required method for reporting a change in accounting principle and the reporting of a correction of an error. FAS 154 is effective for accounting changes and corrections of errors made in fiscal years beginning after December 15, 2005.

In December 2004, the FASB issued Statement of Financial Accounting Standards No. 123R, *Share-Based Payment* ("FAS 123R"). FAS 123R is a revision of FASB Statement No. 123, *Accounting for Stock-Based Compensation*. FAS 123R supersedes APB Opinion No. 25, *Accounting for Stock Issued to Employees*, and its related implementation guidance. FAS 123 focuses primarily on accounting for transaction in which an entity

obtains employee services in share-based payment transactions. As the Company does not currently have share-based payment transactions, adoptions of FAS 123R is not expected to have a significant impact on the Company's results of operations.

Note 2 Related Party Transactions:

One of the principal shareholders of the Parent and CFG is a stockholder, director and officer of CRP and is an officer and director of the Company. Additionally, the President of CFG, who is an officer of the Company, is a director and officer of CRP.

The Company's fee revenue is primarily earned for services provided to CRP. The Company and CRP have entered into an advisory agreement pursuant to which the Company earns a monthly management fee equal to one-twelfth of 0.6 percent of CRP's real estate asset value and the outstanding principal amount of any mortgage loans as of the end of the preceding month. Management fees earned under the advisory agreement for the years ended December 31, 2005 and 2004 were \$19,143,304 and \$13,032,700, respectively.

The Company is also entitled to receive acquisition fees for services rendered in identifying and acquiring properties and structuring the terms of the related leases and negotiating mortgage loans equal to a percentage of gross equity proceeds from CRP stock offerings and loan proceeds from permanent financing. This percentage equaled 4.0 percent from January 1, 2005 to May 2, 2005 and 3.0 percent from May 3, 2005 to December 31, 2005. Total acquisition fees of \$20,138,881 were earned for the year ended December 31, 2005. This percentage equaled 4.5 percent from January 1, 2004 to May 13, 2004 and 4.0 percent from May 14, 2004 to December 31, 2004. Total acquisition fees of \$68,098,637 were earned for the year ended December 31, 2004.

The Company pays CFG, based on a verbal agreement, a fee equal to a percentage of gross equity proceeds from CRP stock offerings for branding, executive management and other corporate services. This percentage equaled 2.5 percent from January 1, 2005 to May 2, 2005 and 2.0 percent from May 3, 2005 to December 31, 2005. The Company paid \$4,106,138 to CFG for these services during the year ended December 31, 2005. This percentage equaled 2.75 percent from January 1, 2004 to May 13, 2004 and 2.5 percent from May 14, 2004 to December 31, 2004. The Company paid \$23,362,267 to CFG for these services during the year ended December 31, 2004.

The Company is entitled to receive fees in connection with the development, construction, and renovation of properties and earned \$26,632 and \$231,847 from related entities for the years ended December 31, 2005 and 2004, respectively.

The Company is required under the terms of its advisory agreement with CRP to reimburse CRP for operating expenses if CRP's operating expenses paid or incurred exceed, in any four consecutive fiscal quarters, the greater of 2 percent of CRP's average invested assets or 25 percent of net income. For the years ended December 31, 2005 and 2004, no reimbursement was required.

The Company provides accounting and administrative services to CRP for which it receives personnel expense reimbursements, in addition to the fees described above. For the years ended December 31, 2005 and 2004, such reimbursements amounted to \$2,935,373 and \$1,814,744, respectively and are included in interest and other income in the accompanying consolidated statements of income.

CFG provides marketing, administration, technology systems, human resources, accounting, tax and compliance services to the Company. Amounts paid to CFG for these services amounted to \$2,270,610 and \$2,294,340, for the years ended December 31, 2005 and 2004, respectively, and are included is general and administrative expense in the accompanying consolidated statements of income.

Amounts due from related parties at December 31, 2005 and 2004, consist of the following:

	2005	2004
-	\$ 309,191	\$ 523,709
fees Other	1,820,424	874,247
	\$ 2,129,615	\$ 1,397,956
	F-80	

Amounts due to related parties at December 31, 2005 and 2004, consist of the following:

	2005	2004	
Income taxes	\$ \$	5 1,728,172	

The Company maintains bank accounts in a bank in which certain officers and directors of the Company serve as directors and are significant stockholders. The amounts deposited with this bank were \$1,492,063 and \$5,235,069 at December 31, 2005 and 2004, respectively.

See Note 8 for Related Party Lease Obligations.

Note 3 Property and Equipment, net:

Property and equipment consist of the following at December 31, 2005 and 2004:

	2005	2004
Furniture and	\$ 1,522,253	\$ 930,015
fixtures		
Computer	1,859,959	747,637
equipment and		
software		
Leasehold	615,642	571,883
improvements		
	3,997,854	2,249,535
Less	(1,531,394)	(792,172)
accumulated		
depreciation		
	\$ 2,466,460	\$ 1,457,363

Note 4 Other Assets:

Other assets consist of the following at December 31, 2005 and 2004:

	2005	2004
Common stock CNL Retirement Properties, Inc., carried at cost which approximates fair market value (20,000	2005	2004
shares) Other	\$ 200,000	,
Other	45,137	30,799

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

2005	2004
* 2 1 7 1 2 7	* * * * * * * * * * * * * * * * * * *
\$ 245,137	\$ 230,799

Note 5 Income Taxes:

The provision for income taxes consisted of the following for the years ended December 31, 2005 and 2004:

	2005	2004
Current:		
Federal	\$ 6,496,547	\$ 14,557,556
State	976,924	2,491,956
	\$ 7,473,471	\$ 17,049,512
	F-81	

Note 6 Profit Sharing Plan:

Employees of the Company are included in the Parent's defined contribution profit sharing plan (the "Plan"). The Plan is designed in accordance with the applicable sections of the Internal Revenue Code, and is not subject to minimum funding requirements. The Plan covers all eligible employees of the Company and its subsidiary upon completion of six months of service. The employees may elect to contribute up to a maximum of 98 percent of their salary under Internal Revenue Service regulations. The Company matches 50 percent of the first 6 percent of each employee's contribution up to a maximum of 3 percent of their salary. For the years ended December 31, 2005 and 2004, the Company's contribution, including administration costs, amounted to approximately \$133,000 and \$110,000, respectively, and is included in salaries and benefits in the accompanying consolidated statements of income.

Note 7 Rights of Common Stockholders:

Each share of Class A common stock is entitled to one vote. The rights of the Class B common stock are as follows:

each share of Class B common stock is equivalent to 1/100th of a share of Class A common stock with regard to all matters, including voting rights, participation in payment of dividends, and distribution in liquidation of the Company; and

if the Company is a participant in a merger or consolidation that results in the conversion, exchange or cancellation of the outstanding shares of Class A common stock or the sale or transfer of all or substantially all of the assets of the Company, then in such event each holder of the Class B common stock shall be entitled to the same consideration as the holder of an equivalent number of shares of Class A common stock.

the Class B common shares issued are subject to employee stock purchase agreements that

give CFG the right to purchase varying percentages of the stock at the issuance price in the event that the holders leave employment of the Company or other CFG affiliates within 48 months after issuance. Upon change of control or merger, the repurchase option is voided.

Note 8 Obligations Under Operating Leases:

The Company was allocated a portion of lease rent obligations relating to office space leased from a related party. The lease provides for minimum monthly rental payments through October 2014. Deferred rent expense represents the difference between rent paid and the total cost of the lease recognized on a straight-line basis over the remaining term of the lease. Rent expense, including amortization of deferred rent, relating to the Company's allocated portion of the lease payments totaled \$468,466 and \$544,067 for the years ended December 31, 2005 and 2004.

The Company entered into a new lease with a related party for office space in November 2005. The lease provides for minimum monthly rent payments through November 2015 of \$86,198 per month commencing March 2006. This lease is replacing the allocation of the lease for office space noted above. As such, the future minimum lease payments have been adjusted to reflect the replacement of that lease allocation with the new lease.

The Company has been allocated a portion of a lease rent obligation relating to other office space. This lease provides for minimum monthly rental payments of \$30,881 through March 2006. Rent expense relating to the Company's allocated portion of the lease payments totaled \$409,712 and \$270,343 for the years ended December 31, 2005 and 2004.

The Company's allocation of future minimum lease payments is as follows:

Year ending		
December 31,		
2006		\$ 1,070,899
2007		1,051,770
2008		1,072,650
2009		1,094,145
2010		1,116,070
Thereafter		6,129,750
		\$ 11,535,290
	F-83	

CNL RETIREMENT CORP. AND SUBSIDIARY CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED)

December

	March 31, 2006	31, 2005
ASSETS		
Current assets: Cash and cash		
equivalents	\$ 1,426,809	\$ 1,492,063
Due from related parties	477,406	2,129,615
Prepaid expenses and other assets	66,221	70,057
Total current assets	1,970,436	3,691,735
Property and equipment, net	3,525,706	2,466,460
Other assets, at cost	245,137	245,137
	\$ 5,741,279	\$ 6,403,332

LIABILITIES AND STOCKHOLDERS' EQUITY

Current liabilities:		
Accounts payable		
and accrued		
expenses	\$ 1,164,527	\$ 3,678,057
Dividends		
payable	50,492	
Due to related		
parties	2,579,964	
Total current		
liabilities	3,794,983	3,678,057
Deferred rent		
expense		202,133
•		
Total liabilities		
and deferred		
una acremea	3,794,983	3,880,190
expense	3,194,983	3,000,190

Commitments and

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	March 31, 2006	December 31, 2005
contingencies		
Stockholders' equity: Class A common stock; \$1 par		
value per share; 10,000 shares authorized; 1,000 shares issued and		
outstanding	1,000	1,000
Class B common stock; \$1 par value per share; 5,000 shares authorized; 1,027 shares issued and		
outstanding	1,027	1,027
Additional paid-in capital Retained earnings	1,944,269	2,521,115
Total stockholders' equity	1,946,296	2,523,142
	\$ 5,741,279	\$ 6,403,332

The accompanying notes are an integral part of these condensed consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONDENSED CONSOLIDATED STATEMENTS OF EARNINGS (UNAUDITED)

Three Months Ended March 31,

	2006	2005
Revenue:		
Acquisition	\$ 2,581,268	\$ 3,490,727
fees	Ψ 2,301,200	Ψ 3,170,727
Debt	4,327,935	10,450,657
acquisition		
fees		
Management	5,126,778	4,498,507
fees leases		
Development	3,905	
fees		
Interest and	621,945	647,998
other income		
	12,661,831	19,087,889
		,001,001
Evmanaaa		
Expenses: Corporate	860,423	2,174,642
services	800,423	2,174,042
provided by		
related parties		
Salaries and	3,310,462	2,443,099
benefits	3,310,402	2,443,077
General and	920,069	933,239
administrative	,20,00	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
Rent	260,693	226,761
Depreciation	201,600	126,742
and	,,,,,,	
amortization		
	5,553,247	5,904,483
	3,333,247	3,904,463
Income before	7,108,584	13,183,406
provision for		
income taxes		
Provision for	(2,719,034)	(4,960,916)
income taxes		
Net income	\$ 4,389,550	\$ 8,222,490

The accompanying notes are an integral part of these condensed consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (DEFICIT) (UNAUDITED)

Three Months Ended March 31, 2006

	Class A Common Stock		Class B Common Stock			
	Shares Outstanding	Par Value	Shares Outstanding	Par Value	Additional Paid-In Capital	Retain Earnii (Defic
Balance at December 31, 2005	1 000	\$ 1,000	1 027	\$ 1 027	\$ 2,521,115	\$
Dividends paid	1,000	Ψ 1,000	1,027	Ψ 1,027	(576,846)	
Net income						4,389
Balance at March 31, 2006	1,000	\$ 1,000	1,027	\$ 1,027	\$ 1,944,269	\$

The accompanying notes are an integral part of these condensed consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

Three Months Ended March 31,

	2006	2005
Cook flows		
Cash flows		
from operating activities:		
Net income	\$ 4,389,550	\$ 8,222,490
Adjustments	,,505,550	φ 0,222,190
to reconcile		
net income to		
net cash		
provided by		
operating		
activities:		
Depreciation		
and		
amortization	201,600	126,742
Loss on	222.25	
retired assets	232,850	
Changes in		
operating		
assets and		
liabilities: Due from		
related		
parties	1,652,209	(3,609,342)
Prepaid	1,032,209	(3,009,342)
expenses		
and other		
assets	3,836	26,045
Accounts	2,020	20,0.0
payable and		
accrued		
expenses	(1,918,888)	(1,943,068)
Dividends		
payable	50,492	
Due to		
related		
parties	1,985,322	4,954,832
Deferred		
rent expense	(202,133)	9,564
Net cash		
provided by		
operating		
activities	6,394,838	7,787,263
Cash flows		
from investing		
activities:		
	(1,992,027)	(39,034)

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

	Three Months Ended March 31,		
Purchases of property and equipment			
Proceeds from sale of assets	498,331		
Net cash used in investing activities	(1,493,696) (39,034)		
Cash flow from financing activities: Dividends			
paid to stockholders	(4,966,396) (12,978,098)		
Net cash used in financing activities	(4,966,396) (12,978,098)		
Net increase in cash and cash equivalents Cash and cash	(65,254) (5,229,869)		
equivalents at beginning of period	1,492,063 5,235,669		
Cash and cash equivalents at end of period	\$ 1,426,809 \$ 5,800		
Supplemental disclosure of cash flow information:			
Cash paid during the year for income taxes	\$ 2,719,034 \$ 4,960,916		

The accompanying notes are an integral part of these condensed consolidated financial statements.

CNL RETIREMENT CORP. AND SUBSIDIARY NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)
Quarters Ended March 31, 2006 and 2005

1. Organization and Nature of Business

CNL Retirement Corp. (the "Company") and its wholly owned subsidiary, CNL Retirement Development Company, provide management, advisory and administrative services and assist in identifying and acquiring seniors' housing and health care-related properties and obtaining financing for CNL Retirement Properties, Inc. and it's subsidiaries ("CRP").

2. Interim Period Financial Statements

The interim statements have been prepared without audit. Certain information and footnote disclosures normally included in financial statements presented in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted. The Company's management believes the disclosures made are adequate to make the interim financial information presented not misleading.

In the opinion of management, the accompanying interim statements reflect all adjustments (which include only normal recurring adjustments) necessary to present fairly the financial position of the Company as of March 31, 2006 and the results of operations and cash flows for the three months ended March 31, 2006 and 2005. Interim results are not necessarily indicative of fiscal year performance because of seasonal and short-term variations.

Amounts included in the financial statements as of December 31, 2005, have been derived from the audited financial statements of the Company for the year then ended.

These unaudited condensed consolidated financial statements should be read in conjunction with the audited financial statements of CNL Retirement Corp. for the year ended December 31, 2005. The accompanying unaudited condensed consolidated financial statements include the accounts of CNL Retirement Corp. and its wholly owned subsidiary. All significant intercompany balances and transactions have been eliminated.

3. Related Party Transactions

One of the principal stockholders of the Company's parent is a stockholder, director and officer of CRP and is also an officer and director of the Company. Additionally, the President of the Company's parent is an officer and director of the Company as well as an officer and director of CRP.

The Company's fee revenue is primarily earned for services provided to CRP. In addition, the Company provides accounting and administrative services to CRP and other related companies for which it receives personnel reimbursements. For the three months ended March 31, 2006 and 2005 such reimbursements amounted to \$612,892 and \$630,359, respectively and are included in interest and other income.

Amounts due from related parties at March 31, 2006 and December 31, 2005 consist of the following:

	March 31, 2006	December 31, 2005
Acquisition fees	\$ (63,473)	\$ 309,191
Other	540,879	1,820,424
	\$ 477,406	\$ 2,129,615

Amounts due to related parties at March 31, 2006 and December 31, 2005 consist of the following:

	March 31, 2006	31, 2005
Amounts due for furniture deposits paid by Parent	\$ 2,579,964	\$
	\$ 2,579,964	\$

Amounts due are unsecured, non-interest bearing and due on demand.

The Company maintains bank accounts in a bank in which certain officers and directors serve as directors and are stockholders. The amounts deposited with this bank were \$1,426,209 and \$1,492,063 as of March 31, 2006 and December 31, 2005, respectively.

Annex A

AGREEMENT AND PLAN OF MERGER

by and among

HEALTH CARE PROPERTY INVESTORS, INC.,

OCEAN ACQUISITION 1, INC.

and

CNL RETIREMENT PROPERTIES, INC.

Dated as of May 1, 2006

TABLE OF CONTENTS

		Page
	ARTICLE I	
	DEFINITIONS	
	Definitions	
1.01.		A-1
	ADTICLE II	
	ARTICLE II THE MERGER	
	THE MERGER	
SECTION	The Merger	
2.01.	C	A-7
SECTION	Closing; Effective	
2.02.	Time	A-7
SECTION	Effect of the Merger	
2.03.	Cl · D l	A-7
SECTION	Charter; Bylaws	۸ 7
2.04. SECTION	Directors and	A-7
2.05.	Officers	A-7
SECTION		11,
2.06.	Securities	A-7
SECTION	Appraisal Rights	
2.07.		A-8
SECTION	Exchange	
2.08.		A-8
	ARTICLE III	
REPR	ESENTATIONS AN	D
	TIES OF THE COMI	
	Organization and	
SECTION		
3.01.	Subsidiaries	A-11
an amical	Articles of	
SECTION	Incorporation and	۸ 10
3.02. SECTION	Bylaws Capitalization	A-12
3.03.	Capitalization	A-12
SECTION	Authority Relative	11-12
3.04.	to this Agreement	A-12
	No Conflict;	
SECTION	Required Filings	
3.05.	and Consents	A-13
SECTION	Permits;	
3.06.	Compliance	A-14
CECTION	SEC Filings;	
SECTION 3.07.	Financial Statements	A-14
SECTION	Absence of Certain	A-14
3.08.	Changes or Events	A-15
	Absence of	10
SECTION	Litigation;	
3.09.	Liabilities	A-15
	Employees and	
SECTION	Employee Benefit	
3.10.	Plans	A-16

	Prospectus/Proxy	
	Statement and	
SECTION	Information	
3.11.	Supplied	A-1
SECTION		A-1
	Property and Leases	A 1.
3.12.	T 4 11 4 1 D 4	A-1
SECTION	Intellectual Property	A 1
3.13.	T	A-1
SECTION	Taxes	A 1
3.14.	.	A-1
SECTION	Environmental	
3.15.	Matters	A-2
SECTION	Material Contracts	
3.16.		A-2
SECTION	Insurance	
3.17.		A-2
SECTION	Affiliate	
3.18.	Transactions	A-2
SECTION	Brokers	
3.19.		A-2
SECTION	Opinion of	
3.20.	Financial Advisor	A-2
3.20. REPR	Financial Advisor ARTICLE IV RESENTATIONS ANI NTIES OF PARENT A	
3.20. REPR	Financial Advisor ARTICLE IV RESENTATIONS ANI)
3.20. REPR WARRA	Financial Advisor ARTICLE IV RESENTATIONS ANI NTIES OF PARENT A PURCHASER)
3.20. REPR	Financial Advisor ARTICLE IV RESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate) AND
3.20. REPR WARRA	Financial Advisor ARTICLE IV RESENTATIONS ANI NTIES OF PARENT A PURCHASER) AND
REPR WARRA SECTION 4.01.	Financial Advisor ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of) AND
REPR WARRA SECTION 4.01.	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and	O AND A-2
REPR WARRAL SECTION 4.01.	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws	O AND A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and	AND A-2 A-2
REPR WARRAL SECTION 4.01. SECTION 4.02. SECTION 4.03.	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization	AND A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative	A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04.	ARTICLE IV RESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative to This Agreement	A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04. SECTION	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative	A-2 A-2 A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04. SECTION 4.04.	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative to This Agreement No Conflict	A-2 A-2 A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04. SECTION 4.05. SECTION	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative to This Agreement No Conflict Consents and	A-2 A-2 A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04. SECTION 4.05. SECTION 4.06.	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative to This Agreement No Conflict Consents and Approvals	A-2 A-2 A-2 A-2
REPR WARRA SECTION 4.01. SECTION 4.02. SECTION 4.03. SECTION 4.04. SECTION 4.05. SECTION	ARTICLE IV ESENTATIONS ANI NTIES OF PARENT A PURCHASER Corporate Organization Articles of Incorporation and Bylaws Capitalization Authority Relative to This Agreement No Conflict Consents and)

Page

448

	SEC Filings;	
SECTION	Financial	
4.08.	Statements	A-25
SECTION		
4.09.	Changes or Events	A-26
CECTION	Absence of	
SECTION 4.10.	Litigation; Liabilities	A-26
SECTION	Employee Benefit	A-20
4.11.	Plans	A-26
SECTION	Property and Leases	A-20
4.12.	Troperty and Leases	A-27
SECTION	Taxes	11 27
4.13.		A-28
SECTION	Environmental	
4.14.	Matters	A-29
SECTION	Material Contracts	
4.15.		A-29
SECTION	Insurance	
4.16.		A-29
SECTION	Affiliate	
4.17.	Transactions	A-29
SECTION	Financing	
4.18.	T. C	A-29
SECTION	Information	A 20
4.19.	Supplied Ownership of	A-30
SECTION	Purchaser; No Prior	
4.20.	Activities	A-30
1.20.	No Ownership of	11 50
SECTION	Company Capital	
4.21.	Stock	A-30
SECTION	Brokers	
4.22.		A-30
	ARTICLE V	
CONDUCT	Γ OF BUSINESS PEN	DING
	THE MERGER	
	Can do at af	
	Conduct of Business by the	
SECTION	Company Pending	
5.01.	the Merger	A-30
0.01.	Conduct of	1100
	Business by Parent	
SECTION	and Purchaser	
5.02.	Pending the Merger	A-32
	ARTICLE VI	
ADDIT	IONAL AGREEMEN	ΓS
CECTION	0, 11 11 1	
SECTION	Stockholders'	A 22
6.01.	Meeting Proxy Statement;	A-33
SECTION	S-4 Registration	
6.02.	Statement	A-33
0.02.	Access to	11-33
SECTION	Information;	
6.03.	Confidentiality	A-33
SECTION	No Solicitation of	
6.04.	Transactions	A-34
		A-35

,		
SECTION	Employees and	
6.05.	Employees and Employee Benefits	
0.05.		
	Matters	
	Directors' and	
	Officers'	
SECTION	Indemnification and	
6.06.	Insurance	A-35
	Further Action;	
SECTION	Reasonable Best	
6.07.	Efforts	A-36
SECTION	Public	A-30
		A 27
6.08.	Announcements	A-37
SECTION	Affiliates	
6.09.		A-37
SECTION	Dividends	
6.10.		A-37
SECTION	Stock Exchange	
6.11.	Listing	A-38
SECTION		71 30
6.12.	1031 Exchange	A-38
0.12.		A-30
	. n.mr.or n. r.m	
	ARTICLE VII	
CONDIT	IONS TO THE MERO	GER
SECTION	Conditions to the	
7.01.	Merger	A-39
	Conditions to the	
	Obligations of	
SECTION	Parent and	
		1 20
7.02.	Purchaser	A-39
	Conditions to the	
SECTION	Obligations of the	
7.03.	Company	A-40
SECTION	Frustration of	
7.04.	Closing Conditions	A-40
	Ţ.	
	ARTICLE VIII	
TERMIN	NATION, AMENDME	ENT
12111111	AND WAIVER	
	THE WHITEK	
SECTION	Termination	
	Termination	A 40
8.01.	Ecc , c	A-40
SECTION	Effect of	
8.02.	Termination	A-42
SECTION	Fees and Expenses	
8.03.		A-42
	Deferral Provisions	
	to Insure	
	Compliance with	
CECTION	REIT Gross Income	
SECTION		1 12
8.04.	Tests	A-43
	ii	

ARTICLE IX GENERAL PROVISIONS

	Non-Survival of	
SECTION		
9.01.	Warranties	A-44
SECTION	Notices	
9.02.		A-44
SECTION	Severability	
9.03.	·	A-45
SECTION	Entire Agreement;	
9.04.	Assignment	A-45
SECTION	_	
9.05.		A-45
SECTION	Remedies	
9.06.		A-45
SECTION	Governing Law	
9.07.	Ü	A-46
SECTION	Waiver of Jury Trial	
9.08.		A-46
SECTION	Interpretation	
9.09.	F	A-46
SECTION	Performance	
9.10.	Guaranty	A-46
SECTION	•	
9.11.		A-46
SECTION	Waiver	
9.12.		A-47
	Counterparts	
9.13.		A-47
EXHIBIT	S	
A		
	Form of Articles of Mer	roer
В	roini of Africies of Mei	gci
_	Knowledge of the Comp	2027
C	Knowledge of the Comp	Daily
	Knowledge of Parent	
D	Knowledge of Falent	
_	Form of Affiliate Letter	
E	Porm of Arribate Letter	
_	Form of Greenberg Trau	ria I I D DEIT
	Opinion	ing LLI KEII
F	Ориноп	
1	Form of Latham & Wat	kins LLP REIT
•	Opinion	
	iii	
	111	

AGREEMENT AND PLAN OF MERGER, dated as of May 1, 2006 (this "Agreement"), by and among HEALTH CARE PROPERTY INVESTORS, INC., a Maryland corporation ("Parent"), OCEAN ACQUISITION 1, INC., a Maryland corporation and a wholly owned subsidiary of Parent ("Purchaser"), and CNL RETIREMENT PROPERTIES, INC., a Maryland corporation (the "Company") that operates as a real estate investment trust ("REIT") for federal income tax purposes.

WHEREAS, the Company Board (as defined herein), acting upon the recommendation of the Special Committee (as defined herein), has approved this Agreement and declared that it is advisable and in the best interest of the Company and its stockholders for the Company to merge with and into Purchaser, in accordance with the General Corporation Law of the State of Maryland (the "MGCL") and upon the terms and subject to the conditions set forth herein (the "Merger");

WHEREAS, the Boards of Directors of Parent and Purchaser have each determined that it is in their best interests and in the best interests of their respective stockholders to consummate the Merger upon the terms and subject to the conditions set forth herein, and have approved this Agreement and the Merger;

WHEREAS, simultaneously with the execution and delivery of this Agreement, Parent is entering into an agreement and plan of merger (the "Advisor Agreement") with CNL RETIREMENT CORP., a Florida corporation (the "Advisor") and OCEAN ACQUISITION 2, LLC, a Florida limited liability company and a wholly owned subsidiary of Parent ("Advisor Purchaser"), pursuant to which at the closing of the transactions contemplated by the Advisor Agreement, the Advisor shall merge with and into the Advisor Purchaser (the "Advisor Merger");

WHEREAS, immediately prior to the Merger, Purchaser and the Company shall execute Articles of Merger substantially in the form attached hereto as Exhibit A (the "Articles of Merger") and shall file the Articles of Merger in accordance with the MGCL to effectuate the Merger;

WHEREAS, Parent, Purchaser and the Company intend that the Merger not qualify as a "reorganization" for purposes of Section 368 of the Code; and

WHEREAS, Parent, Purchaser and the Company desire to make certain representations, warranties, covenants and agreements in connection with the Merger.

NOW, THEREFORE, in consideration of the foregoing and the mutual covenants and agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, Parent, Purchaser and the Company hereby agree as follows:

ARTICLE I DEFINITIONS

SECTION 1.01. *Definitions.* (a) For purposes of this Agreement:

"Acquisition Proposal" means (i) any proposal or offer from any Person (other than as contemplated by this Agreement) relating to any direct or indirect acquisition by any means of (A) more than 20% of the consolidated assets of the Company (including securities of any of the Company's Subsidiaries) or (B) more than 20% of the equity securities of the Company then outstanding; (ii) any tender offer or exchange offer, as defined pursuant to the Exchange Act, that, if consummated, would result in any Person beneficially owning more than 20% of the equity securities of the Company then outstanding; or (iii) any merger, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction involving the Company or any of its then Significant Subsidiaries, in each case other than the Merger and the other transactions contemplated hereby.

"Aggregate Limit" means \$25,000,000 in the aggregate for all asset acquisitions and all authorizations of and commitments for capital expenditures and capital expenditures made by the Company or any of its Subsidiaries after the date hereof.

"Business Day" means any day on which the principal offices of the SEC in Washington, D.C. are open to accept filings, or, in the case of determining a date when any payment is due, any day (other than a Saturday or Sunday) on which banks are not required or authorized to close in the City of New York.

"CERCLA" means the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (42 U.S.C. Section 9601 et seq.).

"*Company Board*" means the Board of Directors of the Company.

"Company Charter" means the Amended and Restated Articles of Incorporation of the Company, as amended.

"Company Common Stock" means the common stock of the Company, \$0.01 par value per share.

"Company Stockholder Approval" means the affirmative approval of the Merger by at least a majority of all the votes entitled to be cast on the matter by the holders of all outstanding shares of Company Common Stock as of the record date for the Company Stockholders' Meeting.

"Company Tax Subsidiary" means any entity in which the Company owns a direct or indirect equity interest for federal or state income tax purposes of at least 10%, determined by either voting power or value.

"Employee Plan" means any plan, program, policy, practice, contract, agreement or other arrangement providing for compensation, severance, termination pay, deferred compensation, performance awards, stock or stock-related awards, fringe benefits or other employee benefits or remuneration of any kind, whether written, unwritten or otherwise, funded or unfunded, including each "employee benefit plan," within the meaning of Section 3(3) of ERISA (whether or not ERISA is applicable to such plan), that is or has been maintained, contributed to, or required to be contributed to, by the Company or any of its Subsidiaries for the benefit of any

service provider of the Company or any of its Subsidiaries, or with respect to which the Company or any of its Subsidiaries has or may have any liability or obligation.

"Environmental Claim" means any and all administrative, regulatory, judicial or third-party claims, demands, notices of violation or non-compliance, directives, proceedings, investigations, orders, decrees, judgments or other allegations of noncompliance with or liability or potential liability relating in any way to any Environmental Law or any Company Environmental Permit, as the case may be, that the Company has received notice of directly or by its registered agent in writing.

"Environmental Laws" means all applicable federal, state, and local laws, rules and regulations, orders, judgments, decrees and other legal requirements including common law relating to the handling, use, presence, disposal, release or threatened release of any Hazardous Materials or the regulation and protection, investigation or restoration of human health, safety, the environment or natural resources or of noise, odor, indoor air, employee exposure, wetlands, pollution, contamination or any injury or threatened injury to persons or property relating to any Hazardous Materials, including, but not limited to, CERCLA; the Hazardous Materials Transportation Act, as amended (49 U.S.C. Section. 5101 et seq.); the Federal Insecticide, Fungicide, and Rodenticide Act, as amended (7 U.S.C. Section. 136 et seq.); the Resource Conservation and Recovery Act, as amended (42 U.S.C. Section 6901 et seq.); the Toxic Substances Control Act, as amended (42 U.S.C. Section. 7401 et seq.); the Clean Air Act, as amended (42 U.S.C. Section 7401 et seq.); the Federal Water Pollution Control Act, as amended (33 U.S.C. Section 1251 et seq.); the Occupational Safety and Health Act, as amended (29 U.S.C. Section 651 et seq.); the Safe Drinking Water Act, as amended (42 U.S.C. Section 300f et

seq.); and their state and local counterparts or equivalents and any transfer of ownership notification or approval statute and any requirements or standards applicable to the voluntary cleanup of environmental contamination.

"*ERISA*" means the Employee Retirement Income Security Act of 1974, as amended.

"Exchange Ratio" means 0.0865.

"Excess Shares" has the meaning set forth in Section 7.7 of the Company Charter.

"*Exchange Act*" means the U.S. Securities Exchange Act of 1934, as amended.

"Hazardous Materials" means all substances, pollutants, chemicals, compounds and wastes listed, classified or regulated pursuant to any Environmental Law, including "pollutants," "toxic substances," "contaminants," "hazardous substances," "regulated wastes," "special wastes," or "hazardous wastes" and petroleum and any fraction thereof or asbestos-containing material, lead-based paint or plumbing, biphenyls, radioactive material, radon or other substances otherwise potentially injurious to human health and the environment including mold or other toxic growth.

"Intellectual Property" means (i) United States, non-United States and international patents, patent applications and statutory invention registrations, (ii) trademarks, service marks, domain names, trade dress, logos, trade names, d/b/a's, corporate names and other source identifiers, and registrations and applications for registration thereof and all renewals thereof, (iii) published and unpublished works of authorship, whether copyrightable or not, copyrightable works, copyrights, and registrations and applications for registration thereof and all renewals or extensions thereof, (iv) computer software, programs and databases, and (v) confidential and proprietary information, including trade secrets and know-how.

"IRS" means the Internal Revenue Service of the United States.

"Knowledge of the Company" or "Company's Knowledge" (or words of similar import) means the actual knowledge of any of the individuals listed on Exhibit B.

"Knowledge of Parent" or "Parent's Knowledge" (or words of similar import) means

the actual knowledge of any of the individuals listed on Exhibit C.

"Liens" means any mortgages, liens, security interests, pledges, deeds to secure debt, charges or any easement, right of way, covenants, restrictions, rights of first refusal, assessments, or any other encumbrance to title.

"Material Adverse Effect" means, when used with reference to the Company or Parent, as the case may be, any event, circumstance, change or effect (any such item, an "Effect") that is materially adverse to the business, financial condition or results of operations of such party and its Subsidiaries taken as a whole; provided, however, that in no event shall any of the following be deemed, either alone or in combination, to constitute, nor shall any of the following be taken into account in determining whether there has been, a Material Adverse Effect: (i) any Effect that results from changes in general economic conditions or changes in securities markets in general, including any changes in interest rates, (ii) any Effect that results from general changes in the industries in which such party and its Subsidiaries operate, (iii) any Effect related to the public announcement or the pendency or consummation of the transactions contemplated by this Agreement, (iv) any Effect that results from any action taken at the specific request of the other party, (v) any change in the market price or trading volume of the Company Common Stock or the Parent Common Stock, as the case may be, after the date hereof, provided, that, the exception in this clause shall not prevent or otherwise affect a determination that any Effect(s) underlying such change has, individually or in the aggregate with any other Effect, resulted in a Material Adverse Effect, (vi) any Effect that results from natural disasters, acts of war, sabotage or terrorism, military actions or the escalation thereof, or (vii) any Effects resulting from any

change in applicable law or regulation in the geographic regions in which a party or any of its Subsidiaries operates; except in the case of clauses (i), (ii), (vi) and (vii), for any Effect that has a significantly disproportionate adverse impact on such party and its Subsidiaries compared to other companies of similar size operating in the principal industries in which such party and its Subsidiaries operate.

"Ordinary Course" means the ordinary course of business either of the Company and its Subsidiaries or of Parent, Purchaser and Parent's Subsidiaries, as applicable, consistent with past practice, and, when referring to the Company and its Subsidiaries, shall include (but not be limited to) any action taken by the Company or its Subsidiaries pursuant to the Company's existing Company Common Stock redemption program.

"Parent Board" means the Board of Directors of Parent.

"Parent Charter" means the Articles of Incorporation of Parent, as amended.

"Permitted Liens" means (i) Liens for Taxes not yet due or delinquent or, if delinquent, as to which there is a good faith dispute; provided that such Liens are reserved against (if such reserves are recognized pursuant to GAAP) or otherwise disclosed in the SEC Reports or the Parent SEC Reports, as the case may be, or the Company Disclosure Schedule or the Parent Disclosure Schedule, as the case may be, (ii) any matter disclosed in the Company Title Insurance Policies, or any title insurance policy insuring Parent's or any of its Subsidiaries' fee simple or leasehold title to any Parent Property or Parent Leased Property, (iii) Liens arising in accordance with the terms of the Company Material Contracts or the Parent Material Contracts, as the case may be (and not as a result of any breach thereunder), (iv) inchoate materialmen's, mechanics', carriers', workmen's and repairmen's liens and other liens created by Law, in each case arising in the Ordinary Course and not past due and payable or the payment of which is being contested in good faith by appropriate proceedings, (v) Liens arising under the Company Leases or the Parent leases, as the case may be, (vi) mortgages and deeds of trust granted as security for financings listed or described in, as applicable, the Company Title Insurance Policies, the Company Disclosure Schedule or the Company SEC Reports, or any title insurance policy insuring Parent's or any of its Subsidiaries' fee simple or leasehold title to any Parent Property or Parent Leased Property, the Parent Disclosure Schedule or the Parent SEC

Reports, as the case may be, (vii) Liens described in Section 3.12 of the Company Disclosure Schedule or Section 4.11 of the Parent Disclosure Schedule, as the case may be, and (viii) any other Lien not specifically addressed in clauses (i) (vii) of this sentence that does not materially diminish the value of the Company Property or the Parent Property, as the case may be, and does not materially impair or materially interfere with any permitted use of any Company Property or Parent Property, as the case may be, affected thereby.

"Person" means an individual, corporation, partnership, limited partnership, limited liability company, syndicate, person, trust, association or entity or government, political subdivision, agency or instrumentality of a government.

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the U.S. Securities Act of 1933, as amended.

"Special Committee" means the special committee of independent directors of the Company formed by the Company Board for the purpose of investigating a possible business combination involving the Company and its Subsidiaries.

"Subsidiary" or "Subsidiaries" of the Company, the Surviving Corporation, Parent or any other Person means any corporation, partnership, joint venture, limited liability company, business trust or other entity, of which such Person, directly or indirectly, owns or controls at least 50% of the securities or other interests entitled to vote in the election of directors or others performing similar functions with respect to such organization.

"Superior Proposal" means any unsolicited bona fide Acquisition Proposal to acquire more than 50% of the total voting power of the equity securities then outstanding, or of the assets (on a consolidated basis), of the Company that the Company Board determines in its good faith judgment upon the recommendation of the Special Committee (in each case after having received the advice of its or their financial advisor(s)), is reasonably capable of being consummated in accordance with its terms, taking into account all legal, financial and regulatory aspects of the proposal and the Person making the proposal and, if consummated, would result in a transaction more favorable to the Company's stockholders from a financial point of view than the Merger.

"Taxes" shall mean any and all taxes, fees, levies, duties, tariffs, imposts and other similar charges of any kind imposed by any Governmental Authority, including: taxes or other charges on or with respect to income, property, sales, use, payroll, employment, social security, workers' compensation, unemployment compensation or net worth; taxes or other charges in the nature of excise, withholding, ad valorem, stamp, transfer, value-added or gains taxes; license, registration and documentation fees; and customs' duties, tariffs and similar charges.

"Tax Returns" shall mean any federal, state, local or foreign report, return, document, declaration, election or any other information or filing required to be supplied to any taxing authority or jurisdiction (foreign or domestic) with respect to Taxes, including information returns, any document with respect to or accompanying payments or estimated Taxes, or with respect to or accompanying requests for the extension of time in which to file any such report, return document, declaration or other information.

(a) The following terms have the meaning set forth in the Sections set forth below:

Defined Term	Location of Definition
Action	§ 3.09
Advisor	Recitals
Advisor Agreement	Recitals
Advisor Merger	Recitals
Advisor Purchaser	Recitals
Affiliate Letter	§ 6.09
Agreement	Preamble
Articles of Merger	Recitals
Blue Sky Laws	§ 3.05(b)
Break-Up Fee	§ 8.02(b)
Cash Consideration	§ 2.06(a)
Certificate	§ 2.06(a)

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Defined Term	Location of Definition
Closing	§ 2.02
Closing Date	§ 2.02
Code	§ 2.08(e)
Company	Preamble
Company Disclosure Schedule	Article III
Company Lease	§ 3.12(h)
Company Material Contract	§ 3.16
Company Property	§ 3.12(a)(i)
Company	§ 6.01
Recommendation	
Company Stockholders' Meeting	§ 6.01
Company Title Insurance Policy	§ 3.12(c)
Confidentiality Agreement	§ 6.03(b)
Counsel's Gross Income Opinion	§ 8.04(a)
Counsel's Ruling Letter	§ 8.04(a)

Edgar Filing: KAR Auction Services, Inc. - Form 10-K

Danartmant	
Department	§ 2.02
Dissenting Stockholders	§ 2.06(a)
Drop Dead Date	§
Drop Bead Bate	
	8.01(b)(iii)
Effective Time	§ 2.02
Environmental Permits	§ 3.15(b)
Exchange Agent	§ 2.08(b)(i)
Exchange Fund	§ 2.08(b)(i)
Excluded Shares	§ 2.06(a)
GAAP	§ 3.07(c)
Governmental Authority	§ 3.05(b)
Indemnified Parties	§ 6.06(b)
Law	§ 3.05(a)
Lease Documents	§
	3.12(a)(ii)
Leased Properties	§
	3.12(a)(ii)
) /	
Merger	Recitals
Merger Consideration	§ 2.06(a)
MGCL	Recitals
NYSE	§ 2.08(g)
Order	§ 7.01(c)
Parent	Preamble
Parent Common Stock	§ 2.06(a)
Parent Disclosure	Article IV
	Afficiety
Schedule	
Parent Employee Plan	§ 4.11(a)
Parent Leased Properties	§ 4.12
Parent Preferred Stock	§ 4.03
	_
Parent Properties	§ 4.12
Parent Property Owners	§ 4.12
Parent Rights	§ 4.03
Parent SEC Reports	§ 4.08(a)
Parent Significant	§ 4.02
_	§ 4.02
Subsidiary	
Paying Party	§ 8.04(a)
PCB	§ 3.15(h)
Permits	§ 3.06
Permitted Encumbrances	-
	§ 3.12(a)(i)
Preferred Stock	§ 3.03(a)
Property Restrictions	§ 3.12(a)(i)
Prospectus/Proxy	§ 3.11
Statement	3 0.11
Statement	
	D 11
Purchaser	Preamble
	Preamble § 8.04(a)
Purchaser Qualifying Income	§ 8.04(a)
Purchaser Qualifying Income Receiving Party	§ 8.04(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT	§ 8.04(a) § 8.04(a) Preamble
Purchaser Qualifying Income Receiving Party REIT S-4 Registration	§ 8.04(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement	§ 8.04(a) § 8.04(a) Preamble
Purchaser Qualifying Income Receiving Party REIT S-4 Registration	§ 8.04(a) § 8.04(a) Preamble
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act	§ 8.04(a) § 8.04(a) Preamble § 3.11
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation Bylaws	\$ 8.04(a) \$ 8.04(a) Preamble \$ 3.11 \$ 3.07(a) \$ 3.07(a) \$ 3.02 \$ 2.06(a) \$ 8.04(a) \$ 2.03 \$ 2.04(b)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation Bylaws Surviving Corporation	§ 8.04(a) § 8.04(a) Preamble § 3.11 § 3.07(a) § 3.07(a) § 3.02 § 2.06(a) § 8.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation Bylaws Surviving Corporation Charter	\$ 8.04(a) \$ 8.04(a) Preamble \$ 3.11 \$ 3.07(a) \$ 3.07(a) \$ 3.02 \$ 2.06(a) \$ 8.04(a) \$ 2.03 \$ 2.04(b) \$ 2.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation Bylaws Surviving Corporation Charter Uncertificated Share	\$ 8.04(a) \$ 8.04(a) Preamble \$ 3.11 \$ 3.07(a) \$ 3.07(a) \$ 3.02 \$ 2.06(a) \$ 8.04(a) \$ 2.03 \$ 2.04(b) \$ 2.04(a)
Purchaser Qualifying Income Receiving Party REIT S-4 Registration Statement Sarbanes-Oxley Act SEC Reports Significant Subsidiary Stock Consideration Specified REIT Requirements Surviving Corporation Surviving Corporation Bylaws Surviving Corporation Charter	\$ 8.04(a) \$ 8.04(a) Preamble \$ 3.11 \$ 3.07(a) \$ 3.07(a) \$ 3.02 \$ 2.06(a) \$ 8.04(a) \$ 2.03 \$ 2.04(b) \$ 2.04(a)

ARTICLE II THE MERGER

SECTION 2.01. *The Merger*. Upon the terms and subject to the conditions set forth in this Agreement, at the Effective Time, the Company shall be merged with and into Purchaser in accordance with the MGCL.

SECTION 2.02. Closing; Effective Time. Unless this Agreement shall have been terminated and the transactions herein contemplated shall have been abandoned pursuant to Section 8.01, and subject to the satisfaction or waiver (where applicable) of the conditions set forth in Article VII, the closing of the Merger (the "Closing") shall take place (a) at the offices of Greenberg Traurig, LLP, 200 Park Avenue, New York, New York 10166, at 10:00 a.m., New York time, on such date as the Company and Parent shall mutually agree following the satisfaction or waiver of the conditions set forth in Article VII (other than those conditions that, by their nature, are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions), or if the parties do not so agree, on the third Business Day following satisfaction of such conditions or (b) at such other place, date or time as may be mutually agreed in writing by the parties. The date of the Closing is referred to herein as the "Closing Date." At the Closing, the parties hereto shall cause the Articles of Merger to be filed with, delivered in the manner required by the MGCL to, and accepted for record by, the Maryland State Department of Assessments and Taxation (the "Department") and shall make all other filings and recordings required under the MGCL. The "Effective Time" shall be the later of (a) the date and time of the acceptance for record of the Articles of Merger with the Department or (b) such later time as may be agreed by each of the parties hereto and specified in the Articles of Merger.

SECTION 2.03. Effect of the Merger. As a result of the Merger, the separate corporate existence of the Company shall cease and the Purchaser shall continue as the surviving corporation of the Merger (the "Surviving Corporation"). At the Effective Time, the effect of the Merger shall be as provided in the applicable provisions of the MGCL. Without limiting the generality of the foregoing, but subject to the terms and conditions of this Agreement, at the Effective Time, all the property, rights, privileges, powers and franchises of the Company and Purchaser shall vest in the Surviving Corporation, and all debts, liabilities, obligations, restrictions, disabilities and duties of the Company and Purchaser shall become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving

Corporation.

SECTION 2.04. *Charter; Bylaws.* (a) At the Effective Time, the Articles of Incorporation of the Purchaser, as in effect immediately prior to the Effective Time, shall be the Articles of Incorporation of the Surviving Corporation (the "Surviving Corporation Charter") until thereafter amended as provided by Law and such Articles.

(b) At the Effective Time, the Bylaws of Purchaser, as in effect immediately prior to the Effective Time, shall be the Bylaws of the Surviving Corporation (the "Surviving Corporation Bylaws") until thereafter amended as provided by Law, the Surviving Corporation Charter and such Bylaws.

SECTION 2.05. Directors and Officers. The directors and officers of Purchaser immediately prior to the Effective Time and such others as Parent shall have designated, if any, shall, from and after the Effective Time, be the directors and officers, respectively, of the Surviving Corporation until their successors shall have been duly elected or appointed and qualify or until their earlier death, resignation or removal in accordance with the Surviving Corporation Charter and the Surviving Corporation Bylaws.

SECTION 2.06. Conversion of Securities. At the Effective Time, as a result of the Merger and without any further action on the part of the Company, Parent, Purchaser or any holder of any capital stock of the Company, Parent or Purchaser:

- (a) Each share of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than shares owned by Parent, Purchaser or any other direct or indirect wholly-owned Subsidiary of Parent, shares that are owned by the Company or any direct or indirect wholly-owned Subsidiary of the Company, and in each case not held on behalf of third parties, and shares that are owned by stockholders ("Dissenting Stockholders") who have properly made and not properly withdrawn a demand for appraisal rights pursuant to Title 3, Subtitle 2 of the MGCL (each, an "Excluded Share" and collectively, "Excluded Shares")) shall be converted into, and become exchangeable for (i) \$11.1293 in cash (the "Cash Consideration") and (ii) a number of shares of common stock, par value \$1.00 per share, of Parent (the "Parent Common Stock") equal to the Exchange Ratio (the "Stock Consideration" and, together with the Cash Consideration, as the case may be, the "Merger Consideration"). At the Effective Time, all shares of Company Common Stock shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and (x) each share of Company Common Stock represented by a certificate (each, a "Certificate") (other than those representing Excluded Shares) shall thereafter represent only the right to receive, upon the surrender of such Certificate, the Merger Consideration and (y) each holder of shares of Company Common Stock not represented by a Certificate (each, an "Uncertificated Share") (other than those representing Excluded Shares) shall thereafter only have the right to receive the Merger Consideration, in each case, together with the right, if any, to receive pursuant to Section 2.08(g) cash in lieu of fractional shares into which such shares of Company Common Stock have been converted pursuant to this Section 2.06(a) and any dividends or other distributions pursuant to Section 2.08(i).
- (b) Each Excluded Share (other than shares owned by Dissenting Stockholders) outstanding immediately prior to the Effective Time shall be automatically canceled without any conversion thereof, and no payment or distribution shall be made hereunder with respect thereto, and each Certificate formerly representing shares of Company Common Stock held by a Dissenting Stockholder shall thereafter represent only (and each Dissenting Stockholder owning Uncertificated Shares shall thereafter only have) the right to receive the payment set forth in Section 2.07; and
- (c) Each share of common stock, par value \$0.01 per share, of Purchaser issued and outstanding immediately prior to the Effective Time, shall remain outstanding and unchanged after the Effective Time and shall constitute all of the issued and outstanding equity interests of the

Surviving Corporation after the Effective Time.

SECTION 2.07. Appraisal Rights. Stockholders of the Company objecting to the Merger will be entitled to demand and receive payment of the fair value of their shares of Company Common Stock from the Surviving Corporation, provided they comply with the requirements of Title 3, Subtitle 2 of the MGCL. No Person who has properly made a demand for appraisal rights pursuant to Title 3, Subtitle 2 of the MGCL shall be entitled to receive shares of Parent Common Stock or cash in lieu of fractional shares thereof or any dividends or other distributions pursuant to this Article II unless and until the holder thereof shall have effectively withdrawn or lost such holder's right to appraisal under the MGCL, and any Dissenting Stockholder shall be entitled to receive only the payment provided by Title 3, Subtitle 2 of the MGCL with respect to shares of Company Common Stock owned by such Dissenting Stockholder. The Company shall give Parent (i) prompt notice of any written demands for appraisal, attempted withdrawals of such demands, and any other instruments served pursuant to applicable law received by the Company relating to its stockholders' rights of appraisal and (ii) the opportunity to direct all negotiations and proceedings with respect to demand for appraisal by Dissenting Stockholders under the MGCL. The Company shall not, except with the prior written consent of Parent, voluntarily make any payment with respect to any demands for appraisals, offer to settle or settle any such demands or approve any withdrawal of any such demands.

SECTION

2.08. Exchange. (a) [Intentionally omitted.]

(b) Exchange Procedures.

- (i) At or prior to the Effective Time, Parent shall deposit, or shall cause to be deposited, with an exchange agent selected by Parent, with the Company's prior approval, which shall not be unreasonably withheld or delayed (the "Exchange Agent"), for the benefit of the holders of shares of Company Common Stock, certificates representing the shares of Parent Common Stock, cash in immediately available funds in an amount necessary to pay the Cash Consideration and any dividends or other distributions with respect to the Parent Common Stock to be issued or paid pursuant to Sections 2.06 and 2.08(i) in exchange for outstanding shares of Company Common Stock (upon due surrender of the Certificate(s) if applicable) pursuant to the provisions of this Article II (such cash and certificates for shares of Parent Common Stock. together with the amount of any dividends or other distributions payable with respect thereto, being hereinafter referred to as the "Exchange Fund"). The Exchange Fund shall not be used for any purpose that is not expressly provided for in this Agreement; provided that the Exchange Agent shall invest the Exchange Fund (if at all) only in obligations of or guaranteed by the United States of America and backed by the full faith and credit of the United States of America or in commercial paper obligations rated A-1 or P-1 or better by Moody's Investor Services, Inc. or Standard & Poor's Corporation, respectively, as directed by Parent. Any interest and other income resulting from such investment shall become a part of the Exchange Fund, and any amounts in excess of the amounts payable under Section 2.06(a) shall be promptly returned by the Exchange Agent to Parent.
- (ii) Promptly (but in no event later than five (5) Business Days) after the Effective Time, the Surviving Corporation shall if applicable cause the Exchange Agent to mail to each holder of record of shares of Company Common Stock (other than holders of Excluded Shares) notice advising such holders of the effectiveness of the Merger, including appropriate transmittal materials specifying that delivery shall be effected, and risk of loss and title to

the Certificates shall pass, only upon delivery of the Certificates (or affidavits of loss in lieu thereof, as provided in Section 2.08(d) or appropriate guarantee of delivery of such Certificate by a financial institution (including most commercial banks, savings and loan associations and brokerage houses) that is a participant in the Security Transfer Agents Medallion Program, the New York Stock Exchange Medallion Signature Guarantee Program or the Stock Exchange Medallion Program) and instructions for surrendering the Certificates to the Exchange Agent. Upon the surrender of a Certificate (or delivery of affidavits of loss in lieu thereof as provided in Section 2.08(d) or notices of guaranteed delivery as described above) to the Exchange Agent (if applicable) in accordance with the terms of such transmittal materials (or, in the case of Uncertificated Shares, at or promptly following the Effective Time) the holder of such Certificate (or of any Uncertificated Share) shall be entitled to receive in exchange therefore (x) a certificate representing that number of whole shares of Parent Common Stock that such holder is entitled to receive pursuant to this Article II and (y) a check in an amount (after giving effect to any required tax withholdings as provided in Section 2.08(e)) equal to the sum of (A) the Cash Consideration, (B) any cash in lieu of fractional shares and (C) any unpaid non-stock dividends and any other dividends or other distributions that such holder has the right to receive pursuant to the provisions of this Article II, and the Certificate so surrendered (or Uncertificated Shares) shall forthwith be cancelled. No interest will be paid or accrued on any amount payable upon due surrender of the Certificates or otherwise with respect to Uncertificated Shares.

(c) If the Merger Consideration is to be delivered to a Person other than the Person in whose name the surrendered Certificate formerly evidencing shares of Company Common Stock is registered on the stock transfer books of the Company, it shall be a condition of delivery that the Certificate so surrendered shall be endorsed properly or otherwise be in proper form for transfer and that the Person requesting such delivery shall have paid all transfer and other taxes required by reason of the payment of the Merger Consideration to a Person other than the registered holder of the Certificate surrendered, or shall have established to the satisfaction of the Surviving

Corporation that such taxes either have been paid or are not applicable.

- (d) If any holder of shares of Company Common Stock is unable to surrender such holder's Certificates because such Certificates have been lost, mutilated or destroyed, such holder may deliver in lieu thereof an affidavit in form and substance and with surety reasonably satisfactory to the Surviving Corporation.
- (e) The Exchange Agent on behalf of Parent and the Surviving Corporation shall be entitled to deduct and withhold from any amounts otherwise payable pursuant to this Agreement in respect of shares of Company Common Stock such amount as it reasonably determines is required to be deducted and withheld with respect to the making of such payment under the Internal Revenue Code of 1986, as amended through the date hereof (the "Code"). To the extent that amounts are so withheld, such withheld amounts shall be treated for purposes of this Agreement as having been paid to the holder of the shares of Company Common Stock in respect of which such deduction and withholding was made. The Exchange Agent shall promptly pay any amounts so withheld to the Internal Revenue Service or applicable state taxing authorities.
- (f) In the event of any reclassification, stock split, reverse split, stock dividend (including any dividend or distribution of securities convertible into Company Common Stock or Parent Common Stock, as the case may be), reorganization, recapitalization or other like change with respect to Company Common Stock or Parent Common Stock, as the case may be, occurring (or for which a record date is established) after the date hereof and prior to the Effective Time, the Exchange Ratio shall be proportionately adjusted to reflect fully such event.
- (g) No certificate or scrip representing fractional shares of Parent Common Stock shall be issued upon the surrender for exchange of Certificates or cancellation of Uncertificated Shares pursuant to Section 2.06(a), and such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a stockholder of Parent. Notwithstanding any other provision of this Agreement, each holder of shares of Company Common Stock converted pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Parent Common Stock (after taking into account all Certificates delivered by such holder and all Uncertificated Shares that have been cancelled pursuant to Section 2.06(a)) shall receive, in lieu thereof, cash (without interest) in an amount equal to such fractional part of a share of Parent Common Stock multiplied by the weighted average of the per share closing prices of Parent Common Stock on The New York Stock Exchange

(the "NYSE") Composite Transactions Reporting System during the ten consecutive trading days ending two trading days prior to the Effective Time.

(h) At any time following the sixth month after the Effective Time, the Surviving Corporation shall be entitled to require the Exchange Agent to deliver to it any funds which had been made available to the Exchange Agent and not disbursed to holders of shares of Company Common Stock (including all interest and other income received by the Exchange Agent in respect of all funds made available to it), and, thereafter, such holders shall be entitled to look to the Surviving Corporation (subject to abandoned property, escheat and other similar Laws) only as general creditors thereof with respect to any Merger Consideration that may be payable upon due surrender of the Certificates held by them or cancellation of Uncertificated Shares held by them pursuant to Section 2.06(a). Notwithstanding the foregoing, neither Parent, the Company, the Surviving Corporation nor the Exchange Agent shall be liable to any holder of a share of Company Common Stock for any Merger Consideration delivered in respect of such share of Company Common Stock to a public official pursuant to any abandoned property, escheat or other similar Law. Any amounts remaining unclaimed by such holders five years after the Effective Time (or such earlier date immediately prior to such time when the amounts would otherwise escheat to or become property of any Government Authority) shall become, to the extent permitted by applicable Law and public policy, the property of the Surviving Corporation free and clear of any claims or interest of any Person previously entitled thereto.

- (i) All shares of Parent Common Stock to be issued pursuant to the Merger shall be deemed issued and outstanding as of the Effective Time and whenever a dividend or other distribution is declared by Parent in respect of the Parent Common Stock, the record date for which is at or after the Effective Time, that declaration shall include dividends or other distributions in respect of all shares issuable pursuant to this Agreement. In the case of shares of Company Common Stock represented by Certificate(s) only, no dividends or other distributions in respect of the Parent Common Stock shall be paid to any holder of any unsurrendered Certificate until such Certificate (or affidavits of loss in lieu thereof as provided in Section 2.08(d)) is surrendered for exchange in accordance with this Article II. Subject to the effect of applicable laws, following surrender of any such Certificate (or affidavits of loss in lieu thereof as provided in Section 2.08(d) or notice of guaranteed delivery as described above), or in the case of Uncertificated Shares, the cancellation of such shares in accordance with Section 2.06(a), there shall be issued and/or paid to the holder of the certificates representing whole shares of Parent Common Stock issued in exchange therefor, without interest, (A) at the time of such surrender or cancellation, the dividends or other distributions with a record date at or after the Effective Time theretofore payable with respect to such whole shares of Parent Common Stock and not paid and (B) at the appropriate payment date, the dividends or other distributions payable with respect to such whole shares of Parent Common Stock with a record date at or after the Effective Time but with a payment date subsequent to surrender.
- (j) At the close of business on the day of the Effective Time, the stock transfer books of the Company shall be closed and thereafter there shall be no further registration of transfers of shares of Company Common Stock on the records of the Company. From and after the Effective Time, the holders of shares of Company Common Stock outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares of Company Common Stock except as otherwise provided herein or by applicable Law.
- (k) It is intended that, in the case of Uncertificated Shares, the Exchange Agent shall issue at or promptly following the Effective Time the consideration set forth in Section 2.08(b)(ii) with respect to such shares without action by the holders thereof. The parties hereto shall make appropriate adjustments to this Section 2.08 to the extent reasonably necessary to effect such issuance.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedule that has been prepared by the Company and delivered by the Company to Parent in connection with the execution and delivery of this Agreement (the "Company Disclosure Schedule") (it being agreed that disclosure of any item in any section of the Company Disclosure Schedule with respect to any section or subsection of this Article III shall be deemed disclosure with respect to any other section or subsection of this Article III to the extent such relationship is reasonably inferable, provided that nothing in the Company Disclosure Schedule is intended to broaden the scope of any representation or warranty of the Company made herein), or as disclosed in SEC Reports filed with, or furnished to, as applicable, the SEC prior to the date of this Agreement, the Company hereby represents and warrants to Parent that:

SECTION 3.01. Organization and Qualification; Subsidiaries. (a) Each of the Company and each Significant Subsidiary of the Company is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization and has the requisite corporate or other business entity power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted. With respect to each Subsidiary of the Company, other than a Significant Subsidiary, each such Subsidiary is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization and has the requisite corporate or other business entity power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, except for any such failure to be so

organized, validly existing or in good standing or to have such power or authority as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. The Company and each of its Subsidiaries is duly qualified or licensed as a foreign corporation, limited liability company or partnership, as the case may be, to do business, and is in good standing, in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its business makes such qualification or licensing necessary, except for such failures to be so qualified or licensed and in good standing that would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

(b) Section 3.01(b) of the Company Disclosure Schedules sets forth a list (true and complete in all material respects with respect to the following clauses (i) and (ii) and true and complete with respect to the following clause (iii) of: (i) each Subsidiary of the Company, (ii) the jurisdiction of incorporation or organization of such Subsidiary and (iii) the percentage of the outstanding capital stock or other equity interests of such Subsidiary owned directly or indirectly by the Company. Other than the Company's Subsidiaries, the Company does not directly or indirectly own any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any Person.

SECTION 3.02. Articles of Incorporation and Bylaws. The Company has heretofore made available to Parent a complete and correct copy of the Company Charter and the Company's Bylaws and the Articles of Incorporation and the Bylaws or equivalent organizational documents, each as amended to date, of (i) each Subsidiary of the Company that is a "significant subsidiary" of the Company as determined pursuant to Rule 1-02(w) of Regulation S-X of the SEC, but substituting 20% for each reference to 10% therein (a "Significant Subsidiary") and (ii) each Subsidiary of the Company that is a "Taxable REIT Subsidiary" under Section 856 of the Code. Such Company Charter, Articles of Incorporation, Bylaws or equivalent organizational documents are in full force and effect. Neither the Company nor any Significant Subsidiary is in violation (in the case of any Significant Subsidiary, in any material respect) of any of the provisions of the Company Charter, its Articles of Incorporation, Bylaws or equivalent organizational documents.

SECTION 3.03. *Capitalization*. (a) The authorized capital stock of the Company consists of 1,000,000,000 shares of Company Common

Stock, 3,000,000 shares of preferred stock, no par value ("Preferred Stock"), and 103,000,000 Excess Shares. As of April 30, 2006, 264,191,182 shares of Company Common Stock were issued and outstanding, all of which were validly issued, fully paid and nonassessable and no shares of Company Common Stock have been issued since such date. As of the date hereof, no shares of Preferred Stock and no Excess Shares are issued and outstanding. There are no options, warrants or other rights, agreements, arrangements or commitments of any character relating to the issued or unissued capital stock of the Company or any of its Subsidiaries or obligating the Company or any such Subsidiary to issue or sell any shares of capital stock of, or other equity interests in, the Company or such Subsidiary. There are no outstanding contractual obligations of the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of Company Common Stock or any capital stock of such Subsidiary or to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any such Subsidiary or any other Person.

(b) Each outstanding share of capital stock of, or other equity interest in, each Subsidiary of the Company is duly authorized, validly issued, fully paid and nonassessable, and each such share or other equity interest is owned, directly or indirectly, by the Company free and clear of any Liens, restrictions on transfer, preemptive rights and rights of first refusal.

SECTION 3.04. Authority Relative to this Agreement. (a) The Company has all necessary corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement by the Company and the consummation by the Company of the transactions contemplated

hereby have been duly and validly authorized by all necessary corporate action on the part of the Company, and no other corporate proceedings on the part of the Company are necessary to authorize this Agreement or to consummate the transactions contemplated hereby (other than, with respect to the Merger, the Company Stockholder Approval and the filing and recordation of appropriate merger documents as required by the MGCL). This Agreement has been duly and validly executed and delivered by the Company and, assuming the due authorization, execution and delivery by Parent and Purchaser, constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the effect of any applicable bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity (regardless of whether considered in a proceeding at Law or in equity).

(b) Each of the Company Board and the Special Committee, at a meeting duly called and held, (i) determined that the transactions contemplated hereby, including the Merger, are advisable and in the best interests of the Company and its stockholders, and approved this Agreement and the transactions contemplated hereby, including the Merger, (ii) directed that this Agreement, the Merger and the other transactions contemplated hereby be submitted to the stockholders of the Company for their approval and resolved to recommend that the stockholders of the Company vote in favor of the Merger and such other matters and (iii) if and to the extent necessary, adopted a resolution having the effect of causing, or have taken all other reasonable steps to cause, the parties hereto not to be subject to the Maryland Business Combination Act and Control Share Acquisition Act.

SECTION 3.05. No Conflict; Required Filings and Consents. (a) The execution and delivery of this Agreement by the Company do not, and the performance of this Agreement by the Company will not, and the consummation of the transactions contemplated hereby by the Company will not, (i) conflict with or violate the Company Charter or Bylaws or other equivalent organizational documents of the Company or any of its Subsidiaries, (ii) assuming that all consents, approvals and other authorizations described in Section 3.05(b) have been obtained and that all filings and other actions described in Section 3.05(b) have been made or taken, conflict with or violate any statute, law, regulation, judgment or decree ("Law") applicable to the

Company or any of its Subsidiaries or by which any property or assets of the Company or such Subsidiary is bound or affected, or (iii) result in any breach of or constitute a default (or an event which, with notice or lapse of time or both, would become a default) under, or give to others any right of termination, amendment, acceleration or cancellation of, or result in the creation of a lien or other encumbrance on any property or asset of the Company or such Subsidiary pursuant to, any note, bond, mortgage, indenture, contract, agreement, lease, license, permit, franchise or other instrument or obligation to which the Company or any such Subsidiary is a party or by which the Company or such Subsidiary or any property or asset of the Company or such Subsidiary is bound or affected, except, with respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults or other occurrences which would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company or prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement.

(b) The execution and delivery of this Agreement by the Company do not, and the performance of this Agreement by the Company will not, require any consent, approval, authorization or permit of, or filing with or notification to, any United States government, regulatory authority, or any court, tribunal or judicial body (a "Governmental Authority"), except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act and state securities or "blue sky" laws ("Blue Sky Laws"), (ii) as specified in Section 3.05(b) of the Company Disclosure Schedule, (iii) for the filing and recordation of appropriate merger documents as required by the MGCL and (iv) for where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to

the Company or prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement.

SECTION 3.06. Permits;

Compliance. Each of the Company and its Subsidiaries is in possession of all licenses, permits and approvals of any Governmental Authority necessary for each of the Company or such Subsidiary to own, lease and operate its properties or to carry on its business as it is now being conducted (the "Permits"), except where the failure to have, or the suspension or cancellation of, any of the Permits would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. No suspension or cancellation of any of the Permits is pending or, to the Knowledge of the Company, threatened, except where the failure to have, or the suspension or cancellation of, any of the Permits would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. The business of the Company and its Subsidiaries has not, since January 1, 2004, been, and is not being, conducted in conflict with, or in default, breach or violation of, (a) any Law applicable to the Company or such Subsidiary or by which any property or asset of the Company or such Subsidiary is bound or affected, or (b) any contract, Permit or other instrument or obligation to which the Company or such Subsidiary is a party or by which the Company or such Subsidiary or any property or asset of the Company or such Subsidiary is bound, except for any such conflicts, defaults, breaches or violations that would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

SECTION 3.07. SEC Filings; Financial Statements. (a) The Company has filed or furnished, as applicable, all forms, reports and documents required to be filed or furnished by it with the SEC since December 31, 2004 (the "SEC Reports"). The SEC Reports (i) were prepared or will be prepared in all material respects in accordance with either the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations promulgated thereunder, and (ii) did not or will not, at the time of filing or furnishing, or, if amended, as of the date of such amendment, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of the SEC Reports, at the time of its filing or being

furnished complied, or if not yet filed or furnished, when so filed or furnished, will comply in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), and any rules and regulations promulgated thereunder applicable to the SEC Reports.

- (b) Except as permitted by the Exchange Act, including Sections 13(k)(2) and (3) or rules of the SEC, since the enactment of the Sarbanes-Oxley Act, neither the Company nor any of its "Affiliates" (as defined in Rule 405 promulgated under the Securities Act) has made, arranged or modified (in any material way) any extension of credit in the form of a personal loan to any executive officer or director of the Company.
- (c) The Company maintains (i) disclosure controls and procedures (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act) designed to ensure that information required to be disclosed by the Company is recorded and reported on a timely basis to the individuals responsible for the preparation of the Company's filings with the SEC and other public disclosure documents and (ii) a system of internal control over financial reporting (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act) designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with United States generally accepted accounting principles ("GAAP") which includes policies and procedures that (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of the Company, (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that receipts and expenditures of the Company are being made only in accordance with

authorizations of management and directors of the Company, and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could have a material effect on its financial statements. The Company has disclosed, based on the most recent evaluation of its chief executive officer and its chief financial officer prior to the date hereof, to the Company's auditors and the audit committee of the Company Board (x) any significant deficiencies in the design or operation of its internal controls over financial reporting that are reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information (and has identified for the Company's auditors and audit committee of the Company Board any material weaknesses in internal control over financial reporting) and (y) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's internal control over financial reporting. The Company has made available to Parent (i) a summary of any such disclosure made by management to the Company's auditors and the audit committee since December 31, 2004 and (ii) any communication since December 31, 2004 made by management or the Company's auditors to the audit committee required or contemplated by the audit committee's charter or professional standards of the Public Company Accounting Oversight Board, Since December 31, 2004, no material written complaints from any source regarding questionable accounting, internal accounting controls or auditing matters have been received by the Company. No attorney representing the Company or any of its Subsidiaries, whether or not employed by the Company or any of its Subsidiaries, has reported evidence of a material violation of securities laws, material breach of fiduciary duty or similar material violation by the Company or any of its officers, directors or agents to the Company's chief legal officer, audit committee (or other committee designated for the purpose) of the Company Board or the Company Board pursuant to the rules adopted pursuant to Section 307 of the Sarbanes-Oxley Act.

(d) Each of the consolidated financial statements (including, in each case, any notes and schedules thereto) contained in, or incorporated by reference into, the SEC Reports was or will be prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except as may be indicated in the notes and schedules thereto) and each fairly presents, or in the case of SEC Reports filed after the date hereof, will fairly present in all material respects the

consolidated financial position, results of operations and cash flows of the Company and its consolidated Subsidiaries as at the respective dates thereof and for the respective periods indicated therein, except as otherwise noted therein (subject, in the case of unaudited statements, to normal and recurring year-end adjustments).

SECTION 3.08. Absence of Certain Changes or Events. Since December 31, 2005, except as contemplated by this Agreement, (a) the Company and its Subsidiaries have conducted their businesses in the Ordinary Course, (b) there has not been any Effect that, individually or in the aggregate, has had or would reasonably be likely to have a Material Adverse Effect with respect to the Company, and (c) none of the Company or any of its Subsidiaries has taken any action that, if taken after the date of this Agreement without the consent of Parent, would constitute a material breach of the covenants set forth in Section 5.01.

SECTION 3.09. Absence of Litigation; Liabilities. There is no litigation, action or proceeding (an "Action") pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries, or any property or asset of the Company or any of its Subsidiaries, before any Governmental Authority or arbitrator that (a) would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company or (b) seeks to materially delay or prevent the consummation of any of the transactions contemplated hereby. There are no obligations or liabilities of the Company or any of its Subsidiaries, whether or not accrued, contingent or otherwise and whether or not required to be disclosed or any other facts or circumstances of which the Company has Knowledge that could reasonably be expected to result in any claims against, or obligations or liabilities of, the Company or any of its Subsidiaries, including those relating to environmental and occupational safety and health matters, except for those that are not, individually or in the aggregate,

be reasonably likely to have a Material Adverse Effect with respect to the Company or prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement. Neither the Company nor any of its Subsidiaries nor any property or asset of the Company or such Subsidiary is subject to any continuing order of, or consent decree, settlement agreement or similar written agreement with, any Governmental Authority or arbitrator, or any order, judgment, injunction or decree of any Governmental Authority or arbitrator that would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company or prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement.

SECTION 3.10. Employees and Employee Benefit Plans. Neither the Company nor any of its Subsidiaries has any common law employees or has any independent contractors or consultants who may be reclassified as common law employees. Neither the Company nor any of its Subsidiaries maintains, sponsors, participates in, or contributes to or has any liability under or with respect to any Employee Plan.

SECTION 3.11. Prospectus/Proxy Statement and Information Supplied. The proxy statement to be sent to the stockholders of the Company in connection with the Company Stockholders' Meeting (as it may be amended or supplemented, the "Prospectus/Proxy Statement") shall not at the date the Prospectus/Proxy Statement (or any amendment or supplement thereto) is first mailed to stockholders of the Company or at the time of the Company Stockholders' Meeting, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. The information supplied by the Company for inclusion or incorporation by reference in the Registration Statement on Form S-4 registering the issuance of shares of Parent Common Stock in the Merger (the "S-4 Registration Statement") shall not, at the date when the S-4 Registration Statement (or any amendment or supplement thereto) becomes effective, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. Notwithstanding the foregoing, the Company makes no representation or warranty with respect to any information

supplied by Parent, Purchaser or any of Parent's or Purchaser's representatives for inclusion or incorporation by reference in the Prospectus/Proxy Statement or the S-4 Registration Statement. The Prospectus/Proxy Statement shall comply in all material respects as to form with the requirements of the Exchange Act and the rules and regulations thereunder.

SECTION 3.12. Property and

Leases. (a) (i) Section 3.12(a)(i) of the Company Disclosure Schedule sets forth a correct and complete list and address of all interests in real property owned by the Company and its Subsidiaries as of the date of this Agreement (all such interests in real property, together with all buildings, structures and other improvements and fixtures located on or under such real property and all easements, rights and other appurtenances to such real property, are individually referred to herein as "Company Property" and collectively referred to herein as the "Company Properties"). The Company and its Subsidiaries own fee simple title or leasehold title, as applicable, to each of the Company Properties, in each case free and clear of any Liens, or title defects, contractual restrictions, covenants or reservations of interests in title, restrictions, rights of first refusal, encroachments and any other burden or option (collectively, "Property Restrictions"), except for (i) Permitted Liens, (ii) Property Restrictions imposed or promulgated by Law or by any Governmental Authority and (iii) such other Property Restrictions that are shown in the Company Title Insurance Policies and as set forth in the Lease Documents, provided that such Permitted Liens and Property Restrictions are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect with respect to the Company (such matters in clauses (i), (ii) and (iii) above, collectively, "Permitted Encumbrances"). Neither the Company nor any of its Subsidiaries has Knowledge that the Company or the applicable Subsidiary has violated any covenants, conditions, easements or restrictions of record affecting any of the Company Properties, which violation has not been cured and, if not cured, would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

- (ii) Section 3.12(a)(ii) of the Company Disclosure Schedule lists each material parcel of real property leased (including ground leases) or subleased as of the date of this Agreement by the Company and any of its Subsidiaries (collectively, the "Leased Properties"). The Company and the applicable Subsidiary owns a valid leasehold interest in the Leased Properties, subject to and as disclosed in the applicable Company Title Insurance Policy, provided that nothing contained therein is, individually or in the aggregate, reasonably likely to have a Material Adverse Effect with respect to the Company. True, correct and complete copies of all leases and each material amendment or other material modifications thereto concerning the Leased Properties ("Lease Documents") have been made available to Parent. Each of the Lease Documents is valid, binding and in full force and effect as against the Company or the applicable Subsidiary and, to the Knowledge of the Company, as against the other party thereto and none of the Company or any of its Subsidiaries is in breach or default of any such Lease Document, except, in each case, as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. All material payments required to have been made under the Lease Documents by the Company or any of its Subsidiaries have been made. Except in accordance with the terms of the relevant Lease Document, neither the Company nor any of its Subsidiaries has exercised any option or right to terminate, renew or extend or otherwise affect the rights or obligations of the tenant under any Leased Properties, and no consent of any party is necessary for Parent and its Subsidiaries and affiliates to legally occupy each Leased Property.
- (b) The Company and its Subsidiaries have good, valid and sufficient title to all the material personal and non-real properties and assets reflected in their books and records as being owned by them, free and clear of all Liens, except for Permitted Liens.
- (c) Valid ALTA owner's policies of title insurance (each a "Company Title Insurance Policy") have been issued insuring, as of the effective date of each such Company Title Insurance Policy, the Company's or the applicable

Subsidiary's (whether directly or as a successor-in-interest) fee simple or leasehold title to the Company Properties, subject only to Permitted Encumbrances, and to the Company's Knowledge, such policies are, at the date hereof, valid and in full force and effect and no written claim has been made against any such policy. To the Company's Knowledge, no Company Title Insurance Policy contains what is commonly referred to as a standard survey exception (i.e. any state of facts which an accurate survey would show). A correct and complete copy of each Company Title Insurance Policy has been previously made available to Parent.

- (d) Section 3.12(d) of the Company Disclosure Schedule sets forth each contract to which the Company or its Subsidiaries are a party as of the date of this Agreement (i) for the acquisition, option to acquire, development or construction of any Company Property or any other real property that may result in total payments by or liability of the Company or any such Subsidiary in excess of \$10,000,000 or (ii) for the disposition or the option to sell (by merger, purchase, or sale of assets or stock or otherwise) of any Company Property or any other real property for consideration in excess of \$10,000,000.
- (e) The Company and its Subsidiaries have obtained all required certificates, permits and licenses from any Governmental Authority having jurisdiction over any of the Company Properties, all of which are in full force and effect, except for such failures to obtain, to have in full force and effect or to renew, which would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company; there is no pending written threat of modification or cancellation of any of same, which would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company; and there is no physical damage to any Company Properties to an extent which would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

- (f) (i) No condemnation or rezoning proceedings are pending or, to the Company's Knowledge, threatened with respect to any of the Company Properties, and (ii) no Laws including any zoning regulation or ordinance, building or similar Law, code, ordinance, order or regulation have been violated for any Company Property, or will be violated by the continued maintenance, operation or use of any buildings or other improvements on any of the Company Properties or by the continued maintenance, operation or use of the parking areas located thereon or appurtenant thereto or used in connection therewith, in the case of clauses (i) and (ii) above, which would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.
- (g) All work required to be performed, payments required to be made and actions required to be taken prior to the date of this Agreement pursuant to any application, submission or agreement the Company or any Subsidiary of the Company has entered into with a Governmental Authority in connection with a site approval, zoning reclassification or other similar action relating to any Company Properties (e.g., local improvement district, road improvement district, environmental compliance and environmental remediation, abatement and/or mitigation) have been and are being performed, paid or taken, as the case may be, in accordance with said application, submission or agreement and with applicable Laws, other than those where the failure to perform, pay or take would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.
- (h) Section 3.12(h) of the Company Disclosure Schedule sets forth a correct and complete list as of the date of this Agreement of each lease, ground lease or other occupancy agreement pursuant to which the Company or any of its Subsidiaries, as a landlord, leases any Company Property and which lease is for more than 100,000 square feet and is for a duration of six months or more (individually, "Company Lease" and collectively, "Company Leases"). Each Company Lease is in full force and effect and is valid, binding and enforceable in accordance with its terms against (a) the Company or a Subsidiary of the Company and (b) to the Knowledge of the Company, the other parties thereto, except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. Except as listed in Section 3.12(h) of the Company Disclosure Schedule, or as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company, the Company and its Subsidiaries have

performed all obligations required to be performed by it to date under each of the Company Leases and neither the Company nor any of its Subsidiaries is in default under any Company Lease, which default, individually or in the aggregate, is reasonably likely to have a Material Adverse Effect with respect to the Company (and to the Company's Knowledge, no event has occurred which, with due notice or lapse of time or both, would constitute such a default). The Company has made available to Parent a correct and complete copy of each Company Lease and all material amendments or material modifications thereto. Section 3.12(h) of the Company Disclosure Schedule includes any Company Lease which has been executed for which the term has not yet commenced. With respect to each Company Lease, Section 3.12(h) of the Company Disclosure Schedule sets forth (i) the name of the tenant, (ii) the expiration date and (iii) the base rent.

- (i) All rent and other amounts due under the Company Leases have been properly calculated and billed to tenants in all respects pursuant to the Company Leases, except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.
- (j) Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company and except as set forth in the Company Title Insurance Policies and the Lease Documents, as of the date of this Agreement neither the Company nor any of its Subsidiaries has granted any unexpired option agreements or rights of first refusal with respect to the purchase of a Company Property or any portion thereof or any other unexpired rights in favor of any third party to purchase or otherwise acquire a Company Property, which, in each case, would be triggered by the Merger.

- (k) The Company has not received any notices from lenders or insurance carriers requiring material repairs or other material alterations to Company Properties which have not been completed as of the date of this Agreement.
- (l) None of the Company Properties are managed by the Company or a Subsidiary of the Company.
- (m) The material improvements on each parcel of Company Property have legal and valid access to such sewer, water, gas, electric, telephone and other utilities as are necessary to allow the business of the Company and each of its Subsidiaries operated thereon to be operated in the Ordinary Course, except where the failure to have such access, individually or in the aggregate, would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. The major structural elements of the improvements comprising the Company Property, including, mechanical, electrical, heating, ventilation, air conditioning or plumbing systems, elevators or parking elements, are in sufficiently good condition (except for ordinary wear and tear) to allow the business of the Company and its Subsidiaries to be operated in the Ordinary Course, except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. No material Company Property is located in a special flood hazard area designated by any Governmental Authority. Each Company Property has sufficient parking that complies with all applicable Law (including variances and legal non-conforming approvals) except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. Each of the Company Properties is an independent property that does not rely on any facilities (other than public facilities, public roads and off-site parking) located in any real property which is not a Company Property to fulfill any requirement of any Governmental Authority or for the furnishing to such Company Property of any essential building system or utility, except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

SECTION 3.13. Intellectual

Property. (a) Other than with respect to software programs that are commercially available on a general basis, the Company and its Subsidiaries own, license or otherwise possess legally enforceable rights to use all Intellectual Property used in the conduct of the business of the

Company and its Subsidiaries, except where the failure to own, license or have rights to use such Intellectual Property would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company.

(b) Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company, (i) all patents, patent applications and registrations and applications for registered trademarks, service marks and copyrights which are held by the Company or any of its Subsidiaries are valid and subsisting and (ii) the Company and its Subsidiaries have taken reasonable measures to protect the proprietary nature of the Intellectual Property owned by the Company or any of its Subsidiaries. No other Person is infringing, violating or misappropriating any of the Intellectual Property owned or used by the Company or any of its Subsidiaries, except for infringements, violations or misappropriations which would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company. To the Company's Knowledge, the Company has not infringed or otherwise violated the Intellectual Property of any third party during the five (5) year period immediately preceding the date of this Agreement.

SECTION 3.14. Taxes.

(a) All Tax Returns (as hereinafter defined) and all other material Tax Returns required to be filed by or on behalf of the Company or any of its Tax Subsidiaries have been properly and timely filed with the appropriate taxing authorities in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings), and all such Tax Returns, as amended, are accurate and complete in all material respects. Except as and to the extent

publicly disclosed by the Company in the SEC Reports, (i) all Taxes payable by or on behalf of the Company or any of its Tax Subsidiaries (whether or not shown in a Tax Return) have been fully and timely paid or adequately provided for in accordance with GAAP, and (ii) adequate reserves or accruals for Taxes have been provided in accordance with GAAP with respect to any period for which Tax Returns have not yet been filed or for which Taxes are not yet due and owing or for which Taxes are being contested in good faith. Neither the Company nor any of its Tax Subsidiaries has executed or filed with the IRS or any other taxing authority any agreement, waiver or other document or arrangement extending or having the effect of extending the period for assessment or collection of Taxes (including, but not limited to, any applicable statute of limitation), and no power of attorney with respect to any Tax matter is currently in force and to the Knowledge of the Company, no request for any such waiver or extension is currently pending.

(b) (i) The Company for all taxable years commencing in 1999, the year in which the Company first made a REIT tax election, through the most recent December 31, has elected to be subject to taxation as a REIT within the meaning of Section 856 of the Code and has satisfied all requirements to qualify as a REIT for such years, (ii) the Company has operated, and intends to continue to operate, in such a manner as to qualify as a REIT for the taxable year of this Agreement and, if different, the taxable year in which the Merger becomes effective, (iii) if the taxable year of the Company were to close on the date of the Closing, the Company would have satisfied all requirements to qualify as a REIT for such year end and (iv) to the Company's Knowledge, no challenge to the Company's status as a REIT is pending or threatened. Each Company Tax Subsidiary that is a partnership, joint venture, or limited liability company (i) has been since its formation and continues to be treated for federal income tax purposes as a partnership or disregarded entity, as the case may be, and not as a corporation or an association taxable as a corporation and (ii) has not since the later of the date of its formation or the acquisition by the Company of a direct or indirect interest therein, owned any assets that would cause the Company to violate the requirements of Section 856(c)(4) of the Code. Each Company Tax Subsidiary that is a corporation has been since the later of the date of its formation or the date on which such Company Tax Subsidiary became a Company Tax Subsidiary a "qualified REIT subsidiary" pursuant to Section 856(i) of the Code a "Taxable REIT subsidiary" pursuant to Section 856(1) of the Code.

Neither the Company nor any of its Tax Subsidiaries holds any assets the disposition of which would be subject to rules similar to Section 1374 of the Code, including as a result of (A) an election under IRS Notice 88-19 or Treasury Regulations Section 1.337(d)-5 or Section 1.337(d)-6 or (B) the application of Treasury Regulations Section 1.337(d)-7.

- (c) No audit or other proceeding by any taxing authority is pending with respect to any Taxes due from or with respect to the Company or any Company Tax Subsidiary, nor is there any material dispute with respect to any liability for Taxes of the Company or any Company Tax Subsidiary either claimed or raised.
- (d) The Company and its Tax Subsidiaries (i) have complied in all material respects with all applicable Laws, rules and regulations relating to the payment and withholding of Taxes; and (ii) have duly and timely withheld from any compensation payable and from distributions to any stockholder or payments to any creditor and have paid over to the appropriate taxing authorities all amounts required to be withheld and paid over on or prior to the due date thereof under all applicable Laws.
- (e) Neither the Company nor any of its Tax Subsidiaries has received notice from any taxing authority in a jurisdiction in which the Company or such Company Tax Subsidiary does not file a Tax Return stating that the Company or such Company Tax Subsidiary is or may be subject to taxation by that jurisdiction.
- (f) Neither the Company nor any of its Tax Subsidiaries (i) is a party to any Tax sharing or similar agreement or arrangement, other than any agreement or arrangement between the Company and any of its Tax Subsidiaries, pursuant to which it will have any obligation to make any payments after the

Closing and (ii) has any material liability for the Taxes of any Person other than the Company and its Tax Subsidiaries (x) under Treasury Regulation §1.1502-6 (or similar provision of state, local or foreign law), (y) as transferee or successor or (z) by contract.

- (g) As of the date of this Agreement, neither the Company nor any of its Tax Subsidiaries has requested a private letter ruling from the IRS or comparable rulings from other taxing authorities.
- (h) Neither the Company nor any of its Tax Subsidiaries has engaged in any transaction that has given rise to or could be reasonably expected to give rise to a disclosure obligation as a "listed transaction" under Section 6011 of the Code and the regulations promulgated thereunder. The Company and all of its Tax Subsidiaries have complied with all obligations applicable to the Company or the relevant Company Tax Subsidiary under Sections 6111 and 6112 of the Code.
- (i) Neither the Company nor any of its Subsidiaries has entered into any Prohibited Transaction (as defined in Section 857 of the Code).

SECTION 3.15. *Environmental Matters.* Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to the Company:

- (a) each of the Company and its Subsidiaries is and has been, and at all times during the Company's and each of its Subsidiaries' ownership and operation of the Company Properties, the Company Properties are and have been in compliance with Environmental Laws;
- (b) each of the Company and its Subsidiaries has obtained and currently possesses and maintains in good standing, and in compliance with the terms and subject to the conditions thereof, all Permits required by Environmental Laws (collectively, "Environmental Permits") in connection with their ownership of the Company Properties or the development by the Company or its Subsidiaries of the Company Properties (provided that no representation or warranty is made with respect to any permit required to be obtained by any lessee of a Company Property or any Person other than the Company or its Subsidiaries with respect to the conduct of business on the Company Properties);
- (c) none of the Company or any of its Subsidiaries or real property currently owned,

leased or operated by the Company or its Subsidiaries is subject to any pending or, to the Knowledge of the Company, threatened Environmental Claim;

- (d) none of the Company or any of its Subsidiaries has generated, arranged for the disposal of or otherwise caused to be disposed of any Hazardous Material at any off-site location with respect to which they have received notice of any investigation, monitoring, cleanup, removal, remediation or other response action;
- (e) no release or disposal of Hazardous Material has occurred on any currently-owned Company Property or other property owned, leased or operated by the Company or any of its Subsidiaries that was in violation of or required investigation, monitoring, cleanup, removal, remediation or other response actions under Environmental Laws;
- (f) except as permitted by Law, and to the Knowledge of the Company, there are no wetlands (as that term is defined in Section 404 of the Federal Water Pollution Control Act, as amended, 33 U.S.C. Section 1254, and applicable state laws) at any of the Company Properties that would affect current development, nor is any Company Property subject to any current or, to the Knowledge of the Company, threatened environmental deed restriction, use restriction, institutional or engineering control or order or agreement with any Governmental Authority or any other restriction of record;
- (g) except as reflected in the Company's consolidated financial statements, no capital expenditures are presently required to maintain or achieve compliance with Environmental Laws;
- (h) to the Knowledge of the Company, except as permitted by Law, there are no underground storage tanks, polychlorinated biphenyls ("*PCB*") or PCB-containing equipment, except for PCB or

PCB-containing equipment owned by utility companies, or friable asbestos or asbestos-containing materials at any Company Property;

- (i) there have been no material incidents of water damage or visible evidence of mold growth at any of the Company Properties that have not been corrected or repaired;
- (j) except for customary terms in favor of lenders in mortgages and trusts, none of the Company or its Subsidiaries has assumed any liability of or duty to indemnify or pay contribution to any other party for any claim, damage or loss arising out of any Hazardous Material or pursuant to any Environmental Law;
- (k) no filing, notification or other submission to any Governmental Authority or any approval from any Governmental Authority is required under any Environmental Law for the execution of this Agreement or for the consummation of the Merger or any of the other transactions contemplated hereby; and
- (l) to the Knowledge of the Company, neither the Company nor any of its Subsidiaries has received any request for information from any Governmental Authority, pursuant to Section 104(e) of CERCLA or any similar Environmental Law.

The Company and its Subsidiaries have made available to Parent all material environmental audits, reports and other material environmental documents and reports in their possession or control relating to their current owned or operated properties, facilities or operations.

SECTION 3.16. Material Contracts. The Company has filed with the SEC copies of all material contracts that were required to be filed with the SEC Reports. Such contracts filed or required to be filed (other than the Company Leases) are herein referred to as "Company Material Contracts". Each Company Material Contract is valid and binding in all material respects on the Company or its applicable Subsidiary (as the case may be) that is a party thereto and, to the Company's Knowledge, each other party thereto, subject to the effect of any applicable bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity (regardless of whether considered in a proceeding

at Law or in equity). None of the Company and its Subsidiaries has received any claim of default of any material provision under or cancellation of any Company Material Contract and none of the Company and its Subsidiaries is in breach or violation of, or default of any material provision under, any Company Material Contract. To the Company's Knowledge, no other party is in breach or violation of, or default under, any material provision of any Company Material Contract; and neither the execution of this Agreement nor the consummation of the transactions contemplated by this Agreement shall constitute a default under, or give rise to cancellation rights under, or otherwise adversely affect any of the Company's material rights under any Company Material Contract.

SECTION 3.17. *Insurance*. The Company and its Subsidiaries maintain insurance coverage with reputable insurers in such amounts and covering such risks as are in accordance with normal industry practice for companies engaged in businesses similar to that of the Company and its Subsidiaries (taking into account the cost and availability of such insurance). All such insurance agreements are listed on Schedule 3.17 of the Disclosure Schedule.

SECTION 3.18. Affiliate

Transactions. There are no contracts, commitments, agreements, arrangements or other transactions between the Company or its Subsidiaries, on the one hand, and any (i) present or former officer or director of the Company or any of their immediate family members (including their spouses), (ii) affiliate of any such officer, director, family member or beneficial owner or (iii) the Advisor or any of its affiliates or any present or former officer or director of any such entity or any of their immediate family members (including their spouses), on the other hand.

SECTION 3.19. *Brokers*. No broker, finder or investment banker (other than Banc of America Securities, LLC and Houlihan Lokey Howard & Zukin) is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of the Company or any of its Subsidiaries. The Company has made available to Parent a complete and accurate copy of all agreements pursuant to which such entities are entitled to any fees and expenses in connection with any of the transactions contemplated by this Agreement.

SECTION 3.20. Opinion of Financial Advisor. The Special Committee and the Company Board have received an opinion of Banc of America Securities, LLC to the effect that, as of the date of this Agreement and subject to the various assumptions and limitations set forth therein, the Merger Consideration to be received by the holders of Company Common Stock is fair, from a financial point of view, to such holders. The Special Committee has received an opinion of Houlihan Lokey Howard & Zukin to the effect that, as of the date of this Agreement and subject to the various assumptions and limitations set forth therein, the Merger Consideration to be received by the holders of Company Common Stock is fair, from a financial point of view, to such holders.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT AND PURCHASER

Except as set forth in the disclosure schedule that has been prepared by Parent and delivered by Parent to the Company in connection with the execution and delivery of this Agreement (the "Parent Disclosure Schedule") (it being agreed that disclosure of any item in any section of the Parent Disclosure Schedule with respect to any section or subsection of this Article IV shall be deemed disclosure with respect to any other section or subsection of this Article IV to the extent such relationship is reasonably inferable, provided that nothing in the Parent Disclosure Schedule is intended to broaden the scope of any representation or warranty of Parent or Purchaser made herein), or as disclosed in Parent SEC Reports filed with, or furnished to, as applicable, the SEC prior to the date of this Agreement, each of Parent and Purchaser hereby, jointly and severally, represents and warrants to the Company that:

SECTION 4.01. Corporate Organization. Each of Parent and Purchaser and each Parent Significant Subsidiary is an entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization and has the requisite corporate or other business entity power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted. Parent, Purchaser and each Parent Significant Subsidiary is duly qualified or licensed as a foreign corporation, limited liability company or partnership, as the case may be, to do business, and is in good standing, in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its business makes such qualification or licensing necessary, except for such failures to be so qualified or licensed and in good standing that would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent.

SECTION 4.02. Articles of Incorporation and Bylaws. Each of the Parent Charter and Parent's Bylaws, the Articles of Incorporation and Bylaws of Purchaser and the Articles of Incorporation and the Bylaws or equivalent organizational documents, each as amended to date, of (i) each Subsidiary of Parent that is a "significant subsidiary" of Parent as determined pursuant to Rule 1-02(w) of Regulation S-X of the SEC, but substituting 20% for each reference to 10% therein (a "Parent Significant Subsidiary") and (ii) each Subsidiary of Parent that is a "Taxable REIT Subsidiary" under Section 856 of the Code is in full force and effect. Neither Parent nor Purchaser nor any Parent Significant Subsidiary is in violation (in the case of any Parent Significant Subsidiary, in any material respect) of any of the provisions of the Parent Charter, its Articles of Incorporation, Bylaws or equivalent organizational documents.

SECTION 4.03. Capitalization. The authorized capital stock of Parent consists of 750,000,000 shares of Parent Common Stock and 50,000,000 shares of preferred stock of Parent, \$1.00 par value ("Parent Preferred Stock"). As of April 24, 2006, 136,843,121 shares of Parent Common Stock were issued and outstanding, all of which were validly issued, fully paid and nonassessable and no shares of Parent Common Stock have been issued since such date other than pursuant to the exercise or conversion of Parent Rights. As of April 30, 2006, 4,000,000 shares of Parent Preferred Stock were issued and outstanding, all of which were validly issued, fully paid and nonassessable and no shares of Parent Preferred Stock have been issued since such date. All shares of Parent Common Stock subject to issuance pursuant to outstanding options, warrants or other rights, agreements, arrangements or commitments of any character relating to the issued or unissued capital stock of Parent ("Parent Rights"), including all shares of Parent Common Stock to be issued in the Merger, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable or this Agreement, as the case may be, will be duly authorized, validly issued, fully paid and non-assessable and not subject to, or issued in violation of, any preemptive right purchase option, call option, right of first refusal, subscription or any other similar right. Parent's Annual Report on Form 10-K for the year ended December 31, 2005 contains an accurate description in all material respects of all outstanding Parent Rights as of December 31, 2005, and no material amount of Parent Rights have been issued since such date other than compensation awards.

SECTION 4.04. Authority Relative to This Agreement. Each of Parent and Purchaser has all necessary corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement by Parent and Purchaser and the consummation by Parent and Purchaser of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of Parent and Purchaser, and no other corporate proceedings on the part of Parent or Purchaser are necessary to authorize this Agreement or to consummate the transactions contemplated hereby, including the issuance of the Parent Common Stock in the Merger (other than, with respect to the Merger, the filing and recordation of appropriate merger documents as required by the MGCL). This Agreement has been duly and validly executed and delivered by Parent and Purchaser and, assuming

due authorization, execution and delivery by the Company, constitutes a legal, valid and binding obligation of each of Parent and Purchaser enforceable against each of Parent and Purchaser in accordance with its terms, subject to the effect of any applicable bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium or similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity (regardless of whether considered in a proceeding at Law or in equity).

SECTION 4.05. No Conflict. The execution and delivery of this Agreement by Parent and Purchaser do not, and the performance of this Agreement by Parent and Purchaser will not, and the consummation of the transactions contemplated hereby by Parent and Purchaser will not, (i) conflict with or violate the Articles of Incorporation or Bylaws or other equivalent organizational documents of Parent or Purchaser or any of Parent's Subsidiaries, (ii) assuming that all consents, approvals and other authorizations described in Section 4.06 have been obtained and that all filings and other actions described in Section 4.06 have been made or taken, conflict with or violate any Law applicable to Parent or Purchaser or any of Parent's Subsidiaries or by which any property or assets of Parent or Purchaser or such Parent's Subsidiary is bound or affected, or (iii) result in any breach of, or constitute a default (or an event which, with notice or lapse of time or both, would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a lien or other encumbrance on any property or asset of Parent or Purchaser or such Parent's Subsidiary pursuant to, any note, bond, mortgage, indenture, contract, agreement, lease, license, permit, franchise or other instrument or obligation to which Parent or Purchaser or any such Parent's Subsidiary is a party or by which Parent or Purchaser or such Parent's Subsidiary or any property or asset of Parent or Purchaser or such Parent's Subsidiary is bound or affected, except, with

respect to clauses (ii) and (iii), for any such conflicts, violations, breaches, defaults or other occurrences which would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent or prevent, materially delay or materially impair the consummation of the transactions contemplated hereby.

SECTION 4.06. Consents and Approvals. The execution and delivery of this Agreement by Parent and Purchaser do not, and the performance of this Agreement by Parent and Purchaser will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Authority, except for (i) applicable requirements, if any, of the Securities Act, the Exchange Act or Blue Sky Laws, (ii) the filing and recordation of appropriate merger documents as required by the MGCL, (iii) the pre-merger notification requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as modified, to the extent applicable to the transactions contemplated by the Advisor Agreement, and (iv) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent or prevent, materially delay or materially impair the consummation of the transactions contemplated hereby. No vote of the holders of any class or series of Parent's capital stock is necessary to approve this Agreement or the Merger, or the Advisor Agreement or the Advisor Merger, and the other transactions contemplated hereby and thereby.

SECTION 4.07. Compliance with Applicable Laws. The business of Parent and its Subsidiaries has not, since January 1, 2004, been, and is not being, conducted in conflict with, or in default, breach or violation of, (a) any Law applicable to Parent or such Subsidiary or by which any property or asset of Parent or such Subsidiary is bound or affected, or (b) any contract, Permit or other instrument or obligation to which Parent or such Subsidiary is a party or by which Parent or such Subsidiary or any property or asset of Parent or such Subsidiary is bound, except for any such conflicts, defaults, breaches or violations that would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent.

SECTION 4.08. SEC Filings; Financial Statements. (a) Parent has filed or furnished, as applicable, all forms, reports and documents required to be filed or furnished by it with the SEC

since December 31, 2004 (the "Parent SEC Reports"). The Parent SEC Reports (i) were prepared or will be prepared in all material respects in accordance with either the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations promulgated thereunder, and (ii) did not or will not, at the time of filing or furnishing, or, if amended, as of the date of such amendment, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of the Parent SEC Reports, at the time of its filing or being furnished complied, or if not yet filed or furnished, when so filed or furnished, will comply in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act, and any rules and regulations promulgated thereunder applicable to the Parent SEC Reports.

(b) Parent maintains (i) disclosure controls and procedures (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act) designed to ensure that information required to be disclosed by Parent is recorded and reported on a timely basis to the individuals responsible for the preparation of Parent's filings with the SEC and other public disclosure documents and (ii) a system of internal control over financial reporting (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act) designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP which includes policies and procedures that (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of Parent, (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial

statements in accordance with GAAP, and that receipts and expenditures of Parent are being made only in accordance with authorizations of management and directors of Parent, and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of Parent's assets that could have a material effect on its financial statements. Parent has disclosed, based on the most recent evaluation of its chief executive officer and its chief financial officer prior to the date hereof, to Parent's auditors and the audit committee of the Parent Board (x) any significant deficiencies in the design or operation of its internal controls over financial reporting that are reasonably likely to materially adversely affect Parent's ability to record, process, summarize and report financial information (and has identified for Parent's auditors and audit committee of the Parent Board any material weaknesses in internal control over financial reporting) and (v) any fraud, whether or not material, that involves management or other employees who have a significant role Parent's internal control over financial reporting.

(c) Each of the consolidated financial statements (including, in each case, any notes and schedules thereto) contained in, or incorporated by reference into, the Parent SEC Reports was or will be prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated (except as may be indicated in the notes and schedules thereto) and each fairly presents, or in the case of Parent SEC Reports filed after the date hereof, will fairly present, in all material respects the consolidated financial position, results of operations and cash flows of Parent and its consolidated Subsidiaries as at the respective dates thereof and for the respective periods indicated therein, except as otherwise noted therein (subject, in the case of unaudited statements, to normal and recurring year-end adjustments).

SECTION 4.09. Absence of Certain Changes or Events. Since December 31, 2005, there has not been any Effect that, individually or in the aggregate, has had or would reasonably be likely to have a Material Adverse Effect with respect to Parent.

SECTION 4.10. Absence of Litigation; Liabilities. There is no Action pending or, to the Knowledge of Parent, threatened against Parent, Purchaser or any of Parent's Subsidiaries, or any property or asset of Parent, Purchaser or any of Parent's Subsidiaries, before any Governmental Authority or arbitrator that (a) would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent or

(b) seeks to materially delay or prevent the consummation of any of the transactions contemplated hereby. Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent, there are no obligations or liabilities of Parent, Purchaser or any of Parent's Subsidiaries, whether or not accrued, contingent or otherwise. Neither Parent nor Purchaser nor any of Parent's Subsidiaries nor any property or asset of Parent or such Subsidiary is subject to any continuing order of, or consent decree, settlement agreement or similar written agreement with, any Governmental Authority or arbitrator, or any order, judgment, injunction or decree of any Governmental Authority or arbitrator that would, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent or prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement.

SECTION 4.11. Employee Benefit Plans. (a) "Parent Employee Plan" means any plan, program, policy, practice, contract, agreement or other arrangement providing for compensation, severance, change in control payments, termination pay, deferred compensation, retirement, pension, performance awards, stock or stock-related awards, fringe benefits or other employee benefits or remuneration of any kind, whether written, unwritten or otherwise, funded or unfunded, including each "employee benefit plan," within the meaning of Section 3(3) of ERISA (whether or not ERISA is applicable to such plan), that is or has been maintained, contributed to, or required to be contributed to, by Parent, Purchaser or any Subsidiary of Parent for the benefit of any service provider of Parent, Purchaser or any Subsidiary of Parent, or with respect to which Parent, Purchaser or any Subsidiary of Parent has or may have any liability or obligation.

- (b) Each Parent Employee Plan complies with its terms and the requirements of applicable Law, including ERISA and the Code, except where the failure to so comply would not have a Material Adverse Effect with respect to Parent.
- (c) Each Parent Employee Plan that is intended to be qualified under Section 401(a) of the Code or Section 401(k) of the Code has received a favorable determination letter from the IRS and each trust established in connection with any Parent Employee Plan which is intended to be exempt from federal income taxation under Section 501(a) of the Code has received a determination letter from the IRS that it is so exempt, and no fact or event has occurred since the date of such determination letter or letters from the IRS to adversely affect the qualified status of any such Parent Employee Plan or the exempt status of any such trust.
- (d) None of Parent, Purchaser or any Subsidiary of Parent maintains, sponsors or contributes to (nor is required to contribute to) (i) any "multiemployer plan" (within the meaning of Section 3(37) or Section 4001(a)(3) of ERISA) either currently or for the past ten years; (ii) any a "defined benefit plan" (within the meaning of Section 3 (35) of ERISA); or (iii) Parent Employee Plan that is subject to the laws of a country other than the United States of America.
- (e) Except where any failure or liability with respect to any of the following has not had, and could not reasonably be expected to have a Material Adverse Effect with respect to Parent: (i) none of Parent, Purchaser or any Subsidiary of Parent has engaged in, and to Parent's Knowledge no fiduciary has engaged in, any "prohibited transaction" (within the meaning of Section 406 of ERISA or Section 4975(c) of the Code) that could subject Parent, Purchaser or any Subsidiary of Parent to a tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA; (ii) there are no audits, inquiries or proceedings pending or, to the best knowledge of the Parent, threatened by the IRS, Department of Labor, or any other governmental entity with respect to any Parent Employee Plan.
- (f) As of the date hereof, there is no pending or, to the Knowledge of Parent threatened, litigation (other than claims for benefits in the ordinary course) relating to the Parent Employee Plans that would reasonably be expected to have a Material Adverse Effect with respect to Parent;

(g) Except where any failure or liability with respect to any of the following has not had, and could not reasonably be expected to have a Material Adverse Effect with respect to Parent, Parent, Purchaser and any Subsidiary of Parent: (i) are in substantial compliance with all applicable legal requirements with respect to employment, employment practices, terms and conditions of employment and wages and hours, in each case, with respect to employees of the Parent, Purchaser or any Subsidiary of Parent; (ii) have withheld and reported all amounts required by law or by agreement to be withheld and reported with respect to wages, salaries and other payments to employees of the Parent, Purchaser or any Subsidiary of Parent; (iii) are not liable for any arrears of wages or any taxes or any penalty for failure to comply with the legal requirements applicable to the foregoing; and (iv) are not liable for any payment to any trust or other fund governed by or maintained by or on behalf of any governmental authority with respect to unemployment compensation benefits, social security or other benefits or obligations for employees of the Parent, Purchaser or any Subsidiary of Parent (other than routine payments to be made in the normal course of business and consistent with past practice).

SECTION 4.12. Property and Leases. Parent or one of its Subsidiaries (each a "Parent Property Owner") owns fee simple title to each of the real properties (or the applicable portion thereof) described in Parent's Annual Report on Form 10-K for the fiscal year ended December 31, 2005 as being owned in fee, as adjusted to reflect purchases and sales since such date (collectively, the "Parent Properties"), and a valid leasehold estate to each of the real properties described as being subject to a lease in Parent's Annual Report on Form 10-K for the fiscal year ended December 31, 2005, as adjusted to reflect

A-27

purchases and sales (collectively, the "Parent Leased Properties"), except as would

not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent. Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent, (i) the interests of the Parent Property Owners in the Parent Properties and the Parent Leased Properties are good and marketable, and the same are owned free and clear of Liens except for (A) Permitted Liens, (B) Property Restrictions imposed or promulgated by Law or by any Governmental Authority and (C) such other Property Restrictions that are shown in any title insurance policy insuring Parent's or any of its Subsidiaries' fee simple or leasehold title to any Parent Property or Parent Leased Property and as set forth in the leases and any material amendment or other material modifications thereto with respect to the Parent Leased Properties, provided that such Permitted Liens and Property Restrictions are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect with respect to Parent, (ii) except in the Ordinary Course, no Parent Property requires any extraordinary capital expenditure by Parent or any of its Subsidiaries, except as required by California Senate Bill 1953, (iii) the execution and delivery of this Agreement by Parent and Purchaser do not, and the performance of this Agreement by Parent and Purchaser will not, violate any Lien on any Parent Property, (iv) to Parent's Knowledge, no Person is in default under any lease, ground lease or other occupancy agreement pursuant to which Parent or any of its Subsidiaries, as a landlord, leases any Parent Property and (v) as of the date of this Agreement neither Parent nor any of its Subsidiaries has granted any unexpired option agreements or rights of first refusal with respect to the purchase of a Parent Property or any portion thereof or any other unexpired rights in favor of any third party to purchase or otherwise acquire a Parent Property, which, in each case, would be accelerated by the Merger.

SECTION 4.13. *Taxes.* (a) All Tax Returns and all other material Tax Returns required to be filed by or on behalf of Parent, Purchaser or any of Parent Tax Subsidiaries have been properly and timely filed with the appropriate taxing authorities in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings), and all such Tax Returns, as amended, are accurate and complete in all material respects. Except as and to the extent publicly disclosed by Parent in the Parent SEC Reports, (i) all Taxes payable by or on behalf of Parent, Purchaser or any of Parent Tax Subsidiaries (whether or not shown in a Tax Return) have been

fully and timely paid or adequately provided for in accordance with GAAP, and (ii) adequate reserves or accruals for Taxes have been provided in accordance with GAAP with respect to any period for which Tax Returns have not yet been filed or for which Taxes are not yet due and owing or for which Taxes are being contested in good faith. Neither Parent nor Purchaser nor any of Parent Tax Subsidiaries has executed or filed with the IRS or any other taxing authority any agreement, waiver or other document or arrangement extending or having the effect of extending the period for assessment or collection of Taxes (including, but not limited to, any applicable statute of limitation), and no power of attorney with respect to any Tax matter is currently in force and to the Knowledge of Parent, no request for any such waiver or extension is currently pending.

- (b) (i) Parent for all taxable years commencing in the year that the Parent first made a REIT tax election, through the most recent December 31, has elected to be subject to taxation as a REIT within the meaning of Section 856 of the Code and has satisfied all requirements to qualify as a REIT for such years, (ii) Parent has operated, and intends to continue to operate, in such a manner as to qualify as a REIT for the taxable year of this Agreement and, if different, the taxable year in which the Merger becomes effective, (iii) if the taxable year of Parent were to close on the date of the Closing, Parent would have satisfied all requirements to qualify as a REIT for such year end and (iv) to Parent's Knowledge, no challenge to Parent's status as a REIT is pending or threatened.
- (c) No audit or other proceeding by any taxing authority is pending with respect to any Taxes due from or with respect to Parent, Purchaser or any Parent Tax Subsidiary, nor is there any material dispute with respect to any liability for Taxes of Parent, Purchaser or any Parent Tax Subsidiary either claimed or raised.

- (d) Neither Parent nor Purchaser nor any of Parent Tax Subsidiaries has engaged in any transaction that has given rise to or could be reasonably expected to give rise to a disclosure obligation as a "listed transaction" under Section 6011 of the Code and the regulations promulgated thereunder. Parent, Purchaser and all of Parent Tax Subsidiaries have complied with all obligations applicable to Parent, Purchaser or the relevant Parent Tax Subsidiary under Sections 6111 and 6112 of the Code.
- (e) Neither Parent nor Purchaser nor any of its Subsidiaries has entered into any Prohibited Transaction (as defined in Section 857 of the Code).

SECTION 4.14. Environmental Matters. Except for such matters that, individually or in the aggregate, would not be reasonably likely to have a Material Adverse Effect with respect to Parent: (i) Parent and its Subsidiaries are in compliance with all applicable Environmental Laws; (ii) neither Parent nor any of its Subsidiaries has received since January 1, 2003 any written notice from any Governmental Authority or any third party indicating that Parent or any such Subsidiary is in violation of any Environmental Law, which matter has not been resolved; (iii) Parent and its Subsidiaries are not subject to any court order, administrative order or decree arising under any Environmental Law; (iv) no Hazardous Substance has been transported from any of the properties owned or operated by Parent or any of its Subsidiaries, other than as permitted under applicable Environmental Law; (v) to the Knowledge of Parent, except as permitted by Law, there are no underground storage tanks, PCB or PCB-containing equipment, except for PCB or PCB-containing equipment owned by utility companies, or friable asbestos or asbestos-containing materials at any of the properties owned or operated by Parent or any of its Subsidiaries; and (vi) to the Knowledge of Parent, there has been no material incidents of water damage or visible evidence of mold growth at any of the properties owned or operated by Parent or any its Subsidiaries that have not been corrected or repaired.

SECTION 4.15. Material

Contracts. Parent has filed with the SEC copies of all material contracts that were required to be filed with the SEC Reports. Such contracts filed or required to be filed, other than Parent leases, are herein referred to as "Parent Material Contracts". Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect with respect to Parent, (i) none of

Parent and its Subsidiaries has received any claim of default of any material provision under or cancellation of any Parent Material Contract and none of Parent and its Subsidiaries is in breach or violation of, or default of any material provision under, any Parent Material Contract, (ii) to Parent's Knowledge, no other party is in breach or violation of, or default under, any material provision of any Parent Material Contract and (iii) neither the execution of this Agreement nor the consummation of the transactions contemplated by this Agreement shall constitute a default under, or give rise to cancellation rights under, or otherwise adversely affect any of Parent's material rights under any Parent Material Contract.

SECTION 4.16. *Insurance*. Parent and Parent's Subsidiaries maintain insurance coverage with reputable insurers in such amounts and covering such risks as are in accordance with normal industry practice for companies engaged in businesses similar to that of Parent and Parent's Subsidiaries (taking into account the cost and availability of such insurance).

SECTION 4.17. Affiliate

Transactions. Neither Parent nor any of its Subsidiaries is a party to any relationships or transactions of the type described in Item 404 of Regulation S-K of the SEC that has not been disclosed in one or more Parent SEC Reports.

SECTION 4.18. *Financing*. Parent has, and Parent and Purchaser shall have, sufficient funds to permit Parent and Purchaser to perform all of their respective obligations under this Agreement and to consummate all the transactions contemplated hereby, including paying the Cash Consideration in the Merger and paying all fees and expenses of Parent and Purchaser related thereto.

SECTION 4.19. Information Supplied. The information supplied by Parent or Purchaser for inclusion or incorporation by reference in the Prospectus/Proxy Statement and the S-4 Registration Statement shall not, in the case of the Prospectus/Proxy Statement, at the date the Prospectus/Proxy Statement (or any amendment or supplement thereto) is first mailed to stockholders of the Company or at the time of the Company Stockholders' Meeting, and in the case of the S-4 Registration Statement, at the date when the S-4 Registration Statement (or any amendment or supplement thereto) becomes effective, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. Notwithstanding the foregoing, Parent makes no representation or warranty with respect to any information supplied by the Company or any of the Company's representatives for inclusion or incorporation by reference in the Prospectus/Proxy Statement or the S-4 Registration Statement. The S-4 Registration Statement shall comply in all material respects as to form with the requirements of the Securities Act and the rules and regulations thereunder.

SECTION 4.20. Ownership of Purchaser; No Prior Activities. Purchaser is a direct wholly-owned Subsidiary of Parent. Purchaser was formed solely for the purpose of engaging in the transactions contemplated by this Agreement and has not engaged in any business activities or conducted any operations other than in connection with the transactions contemplated by this Agreement. All the issued and outstanding shares of capital stock of Purchaser are owned of record and beneficially by Parent.

SECTION 4.21. *No Ownership of Company Capital Stock.* As of the date of this Agreement, neither Parent nor any of its Subsidiaries, including Purchaser, own any shares of Company Common Stock or other securities of the Company.

SECTION 4.22. *Brokers*. No broker, finder or investment banker (other than USB Securities LLC and Cohen & Steers Capital Advisors, LLC) is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of Parent or Purchaser.

ARTICLE V
CONDUCT OF BUSINESS PENDING THE
MERGER

SECTION 5.01. Conduct of Business by the Company Pending the Merger. The Company agrees that, between the date of this Agreement and the Effective Time, except as expressly set forth in this Agreement or Section 5.01 of the Company Disclosure Schedule, the businesses of the Company and its Subsidiaries shall be conducted in the Ordinary Course, and the Company shall use its reasonable efforts to preserve substantially intact the business organization of the Company and its Subsidiaries and to preserve the current relationships of the Company and its Subsidiaries with customers, suppliers and other Persons with which the Company or any of its Subsidiaries has significant business relations. Except as expressly set forth in this Agreement or set forth in Section 5.01 of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries shall, between the date of this Agreement and the Effective Time, directly or indirectly, do, or propose to do, any of the following without the prior written consent of Parent, which consent shall not be unreasonably withheld or delayed:

- (a) amend or otherwise change the Company
 Charter or Bylaws of the Company or any
 equivalent organizational documents of any
 Significant Subsidiaries of the Company;
- (b) issue, sell, pledge, dispose of, grant or encumber, or authorize the issuance, sale, pledge, disposition, grant or encumbrance of, any shares of any class of capital stock of the Company or any of its Subsidiaries, or any options, warrants, convertible securities or other rights of any kind to acquire any shares of such capital stock, or any other ownership interest, of the Company or any of its Subsidiaries;

- (c) except to the extent necessary to maintain its status as a REIT (provided that any dividend or distribution materially in excess of dividends or distributions paid prior to the date hereof shall require prior consultation with Parent), declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock except for (i) distributions by the Company or any wholly-owned Subsidiary of the Company or another wholly-owned Subsidiary of the Company and (ii) monthly dividends payable quarterly of \$0.0592 or quarterly dividends of \$0.1776 per share of Company Common Stock consistent with past practice;
- (d) reclassify, combine, split, subdivide or redeem, or purchase or otherwise acquire, directly or indirectly, any of its capital stock except in the Ordinary Course;
- (e) (i) acquire any corporation, partnership, other business organization or any division thereof in any transaction or series of related transactions (including by merger, consolidation or acquisition of stock or assets or any other business combination) other than acquisitions pursuant to agreements or commitments in effect as of the date hereof which agreements are listed on Schedule 5.01 hereto; (ii) acquire any asset having a value in excess of \$500,000 without first providing Parent five (5) Business Days' prior notice and consulting with Parent with respect thereto and acquire any asset with a value or for consideration in excess of \$5,000,000 or in excess of the Aggregate Limit in the aggregate, other than acquisitions pursuant to contracts or agreements in effect as of the date of this Agreement (including by merger, consolidation, or acquisition of stock or assets or any other business combination) and provided that the agreements with respect to any asset acquisitions entered into as permitted under this clause (i) are on an arm's-length basis; (iii) except for borrowings under that certain Amended and Restated Credit Agreement dated as of August 23, 2005 by and among the Company (and various CNL entities) and Bank of America, N.A. and various other financial institutions in an amount not to exceed \$25,000,000 and indebtedness for borrowed money assumed in connection with any asset acquisition permitted under Section 5.01(e)(ii) incur any indebtedness for borrowed money or issue any debt securities or assume, guarantee or endorse, or otherwise become responsible for, the obligations of any Person, or make any loans or advances, or grant any security interest in any of its assets; (iv) enter into or amend or modify in any material respect or terminate any material contract or agreement other

than (1) renewals or extensions of Company Leases on terms substantially equivalent to such Company Lease so renewed or extended, (2) leases with respect to real property entered into by the Company or a Subsidiary of the Company as landlord in connection with any asset acquisition permitted under Section 5.02(e)(ii) (provided that, in connection with any asset acquisition permitted under such Section without the prior consent of Parent, the Company shall also have consulted with Parent on such leases during the five (5) Business Day period referred to therein), (3) any other lease of real property entered into by the Company or a Subsidiary of the Company as landlord provided that the lease does not represent in excess of 10% of the gross leasable area with respect thereto and (4) any other material contract other than a lease provided that such contract is terminable by the Company or one of its Subsidiaries on no more than 90 days' notice without the payment of a penalty or premium, and provided that any material contract or amendment or modification of any material contract under any of clauses (1) through (4) is on an arm's-length basis; or (v) authorize, or make any commitment with respect to or make any single capital expenditure that is in excess of \$5,000,000 or capital expenditures that are, in the aggregate, in excess of the Aggregate Limit, except for capital expenditures in accordance with and on the timetable contemplated by the capital budget of the Company attached hereto as Schedule 5.01(e);

- (f) voluntarily create or incur any Lien material to it or any of its Subsidiaries on any of its assets or any assets of its Subsidiaries having a value in excess of \$500,000;
- (g) make any loans, advances or capital contributions to or investments in any Person (other than between itself and any of its direct or indirect wholly-owned Subsidiaries);

- (h) cancel, modify or waive any debts or claims held by it or waive any rights having in each case a value in excess of \$5,000,000 or settle any litigation or other proceedings before a Governmental Authority or arbitrator or make any payments in respect thereof in excess of \$100,000 for any individual litigation or proceeding or \$750,000 in the aggregate (it being understood that this clause shall not apply to any fees and expenses paid in connection with the defense of any such litigation or proceeding);
- (i) transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of any of its assets, including (with respect to assets of the Company) capital stock of any of its Subsidiaries, having a value in excess of \$500,000 individually or \$1,500,000 in the aggregate other than pursuant to agreements or contracts in effect as of the date of this Agreement and provided that all contracts entered into as permitted under this clause (i) shall be on an arm's-length basis;
- (j) take any material action with respect to accounting policies or procedures;
- (k) enter into an agreement with respect to any merger, consolidation, liquidation or business combination, or any acquisition or disposition of all or substantially all of the assets or securities of the Company or any of its Significant Subsidiaries;
- (l) except to the extent reasonably necessary to maintain the Company's status as a REIT, make or rescind any material Tax election, settle or compromise any material Tax liability or make any material amendment to any Tax Return;
- (m) enter into any Prohibited Transaction (as defined in Section 857 of the Code);
- (n) take any action (or fail to take any action) that would cause the Company to fail to qualify as a REIT;
- (o) fail to timely file any Tax Returns that are required to be filed by the Company or any Company Tax Subsidiary; or
- (p) announce an intention, enter into any agreement or otherwise make a commitment, to do any of the foregoing.

SECTION 5.02. Conduct of Business by Parent and Purchaser Pending the Merger.

Parent agrees that, between the date of this Agreement and the Effective Time, Parent shall use its reasonable efforts to preserve substantially intact the business organization of Parent, Purchaser and Parent's Subsidiaries, to retain the services of its executive officers and key employees and to preserve the current relationships of Parent, Purchaser and Parent's Subsidiaries with material customers, suppliers and other Persons with which Parent, Purchaser or any of Parent's Subsidiaries has significant business relations. Except as expressly set forth in this Agreement or set forth in Section 5.02 of the Parent Disclosure Schedule, Parent shall not, between the date of this Agreement and the Effective Time, directly or indirectly, do, or propose to do, any of the following without the prior written consent of the Company, which consent shall not be unreasonably withheld or delayed: (i) make any amendment to the Parent Charter that changes the fundamental attributes of Parent Common Stock (it being understood that the authorization and/or issuance of preferred stock shall not be a change in the fundamental attributes of Parent Common Stock); (ii) except to the extent necessary to maintain its status as a REIT (provided that any dividend or distribution materially in excess of dividends or distributions paid prior to the date hereof shall require prior consultation with the Company), declare, set aside, make or pay any extraordinary dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock; (iii) take any action (or fail to take any action) that would cause Parent to fail to qualify as a REIT; (iv) permit or cause any of Parent's Subsidiaries to do any of the foregoing or agree or commit to any of the foregoing (except for distributions by

Parent or any wholly-owned Subsidiary of Parent to Parent or another wholly-owned Subsidiary of Parent); or (v) agree in writing or otherwise to take any of the foregoing actions.

ARTICLE VI ADDITIONAL AGREEMENTS

SECTION 6.01. Stockholders'

Meeting. Subject to Section 6.04, the Company shall, (i) use its reasonable best efforts to duly call, give notice of, convene and hold, in accordance with applicable Law and the Company Charter and Bylaws, no later than the 60th calendar day following the commencement of mailing of the Prospectus/Proxy Statement to its stockholders, an annual or special meeting of its stockholders for the purpose of considering and taking action on this Agreement and the transactions contemplated hereby, including the Merger, and obtaining the Company Stockholder Approval (the "Company Stockholders' Meeting") and

(ii) (A) include in the Prospectus/Proxy Statement the recommendation of the Company Board and the Special Committee that the stockholders of the Company approve the Merger (the "Company Recommendation") and (B) use its reasonable efforts to obtain such approval and adoption; provided that, the Company may adjourn or postpone the Company Stockholders' Meeting to the extent necessary to ensure that any required supplement or amendment to the Prospectus/Proxy Statement is provided to the Company's stockholders or if, as of the time for which the Company Stockholders' Meeting is originally scheduled (as set forth in the Prospectus/Proxy Statement), there are insufficient shares of Company Common Stock represented (either in person or by proxy) to constitute a quorum necessary to conduct the business of the Company Stockholders' Meeting.

SECTION 6.02. Proxy Statement; S-4
Registration Statement. As promptly as
reasonably practicable after the date hereof, the
Company shall prepare and file with the SEC the
Prospectus/Proxy Statement, and Parent shall
prepare and file with the SEC the S-4 Registration
Statement. The Company and Parent shall each use
its reasonable efforts to have the Prospectus/Proxy
Statement cleared and the S-4 Registration

Statement declared effective, respectively, by the SEC as promptly as practicable after filing thereof, and the Company shall as promptly as practicable thereafter mail the Prospectus/Proxy Statement to its stockholders. Parent, Purchaser and the Company shall cooperate with each other in the preparation of the Prospectus/Proxy Statement and the S-4 Registration Statement, and each party shall notify the other of the receipt of any comments of the SEC with respect thereto and of any requests by the SEC for any amendment or supplement thereto or for additional information and shall provide to the other party copies of all correspondence between them or any of their representatives and the SEC with respect thereto. Each party shall give the other party and its counsel a reasonable opportunity to review and comment on the S-4 Registration Statement and the Prospectus/Proxy Statement, including all amendments and supplements thereto, prior to such documents being filed with the SEC or disseminated to stockholders of the Company and shall give the other party and its counsel a reasonable opportunity to review and comment on all responses to requests for additional information and replies to comments prior to their being filed with, or sent to, the SEC. Each of the Company, Parent and Purchaser agrees to use its reasonable efforts, after consultation with the other parties hereto, to respond promptly to all such comments of and requests by the SEC and to cause the Prospectus/Proxy Statement and all required amendments and supplements thereto to be mailed to the holders of shares of Company Common Stock entitled to vote at the Company Stockholders' Meeting at the earliest practicable time.

SECTION 6.03. Access to Information; Confidentiality. (a) Subject to applicable Law and confidentiality agreements, from the date hereof until the Effective Time, the Company shall (i) cause its Subsidiaries and the officers and representatives of the Company and its Subsidiaries and (ii) use reasonable efforts to cause the officers and employees of the Advisor to, afford the officers, employees and agents of Parent and Purchaser reasonable access during normal business hours (and in such a way

as to not unduly interfere with the operation of the businesses of the Company or its Subsidiaries) to the personnel, properties, offices, plants and other facilities, books and records of the Company and its Subsidiaries, and shall furnish Parent and Purchaser with such financial, operating and other data and information as Parent or Purchaser may reasonably request.

- (b) Subject to applicable Law and confidentiality agreements, from the date hereof until the Effective Time, Parent shall (i) cause Purchaser, Parent's Subsidiaries and the officers and representatives of Parent, Purchaser and Parent's Subsidiaries to afford the officers, employees and agents of the Company reasonable access during normal business hours (and in such a way as to not unduly interfere with the operation of the businesses of Parent, Purchaser or Parent's Subsidiaries) to the personnel, properties, offices, plants and other facilities, books and records of Parent, Purchaser and Parent's Subsidiaries, and shall furnish the Company with such financial, operating and other data and information as the Company may reasonably request; provided that in no event shall Parent be obligated to afford to the Company or such officers, employees and agents greater access to such personnel, properties, offices, plants and other facilities, books and records than was afforded to the Company and such officers, employees and agents prior to the date hereof.
- (c) All information obtained by Parent, Purchaser or the Company pursuant to this Section 6.03 shall be kept confidential in accordance with the confidentiality agreement, dated March 10, 2006, as amended on April 14, 2006 (the "Confidentiality Agreement"), among Parent, the Company and the Advisor.

SECTION 6.04. *No Solicitation of Transactions.* (a) The Company will, and will cause each of its Subsidiaries to, and its and their respective officers, directors and representatives to, immediately cease any existing solicitations, discussions or negotiations with any Person that has made or indicated an intention to make an Acquisition Proposal.

(b) The Company shall not, and shall cause its Subsidiaries and any of their respective directors, officers and representatives not to, (i) solicit, initiate, knowingly encourage or facilitate (including by way of furnishing non-public information) any inquiries with respect to an Acquisition Proposal, or (ii) initiate, participate in or knowingly encourage any discussions or negotiations regarding an

Acquisition Proposal; provided, however, that, at any time prior to the Company Stockholder Approval, if the Company receives a bona fide Acquisition Proposal or a proposal that may reasonably be expected to lead to an Acquisition Proposal that was not solicited after the date of this Agreement or that did not otherwise result from a breach of this Section 6.04, the Company may furnish, or cause to be furnished, non-public information with respect to the Company and its Subsidiaries to the Person who made such proposal and may participate in discussions and negotiations regarding such proposal if (A) each of the Company Board and the Special Committee determines in good faith, after consultation with its financial advisor and outside counsel, that such action is necessary in order for such directors to comply with the directors' statutory duties under Section 2-405.1 of the MGCL and fiduciary duties under the Company Charter, (B) each of the Company Board and the Special Committee determines in good faith, after consultation with its financial advisor, that such Acquisition Proposal either constitutes a Superior Proposal or is reasonably likely to result in a Superior Proposal, and (C) prior to taking such action, the Company enters into a confidentiality agreement with respect to such proposal that contains confidentiality provisions no less restrictive than the Confidentiality Agreement. The Company shall promptly (and, in any event, within 24 hours) notify Parent after receipt by the Company of any Acquisition Proposal, including the material terms and conditions thereof, to the extent known, the identity of the third party making any proposal and any material change in the status of discussions or negotiations (including any material amendments to the proposal) between the Company and the Person making such proposal.

(c) Prior to the Effective Time the Company Board and the Special Committee shall not(i) withdraw, qualify or modify or publicly propose to withdraw, qualify or modify in a manner adverse

to Parent or Purchaser, the Company Recommendation or (ii) approve, adopt or recommend, or publicly propose to approve, adopt or recommend, an Acquisition Proposal to holders of the Company Common Stock; provided, however, that in the event an Acquisition Proposal is made prior to the Company Stockholder Approval, each of the Company Board and the Special Committee may take such action if the Company has complied with the notice provisions set forth in the last sentence of Section 6.04(b) and each has determined in good faith (x) after consultation with outside legal counsel that such action is necessary in order for such directors to comply with the directors' statutory duties under Section 2-405.1 of the MGCL and fiduciary duties under the Company Charter and (y) after consultation with its financial advisor and outside legal counsel, that such Acquisition Proposal is a Superior Proposal. In the event that the Company Board and the Special Committee comply with their respective obligations in the preceding sentence, the Company may enter into a definitive agreement to effect a Superior Proposal, but not prior to such time as the Company has provided Parent with written notice that the Company has elected to terminate this Agreement pursuant to Section 8.01(e) and otherwise complied with the Company's obligations in the preceding sentence and in Section 8.01(e).

(d) Nothing contained in this Section 6.04 shall prohibit the Company from at any time taking and disclosing to its stockholders a position contemplated by Rule 14d-9 or Rule 14e-2(a) promulgated under the Exchange Act or making any disclosure required by Rule 14a-9 promulgated under the Exchange Act or Item 1012(a) of Regulation M-A or from making any other disclosure to its stockholders or in any other regulatory filing if, in the good faith judgment of the Company Board and the Special Committee, based on the advice of their respective outside counsel, failure to so disclose would be inconsistent with their or the Company's obligations under applicable Law (it being understood that nothing in this Section 6.04(d) shall affect Parent's termination rights under Section 8.01(f)).

SECTION 6.05. Employees and Employee Benefits Matters. Neither the Company nor any of its Subsidiaries shall hire any natural person as an employee, nor will the Company or any of its Subsidiaries enter into any agreement with a natural person for the provision of services to the Company or any of its Subsidiaries in the capacity of an employee. Neither the Company nor any of its Subsidiaries shall establish, adopt or enter into

any Employee Plan.

SECTION 6.06. Directors' and Officers' Indemnification and Insurance. (a) For a period of six years from the Effective Time, the Surviving Corporation Charter and the Surviving Corporation Bylaws shall contain provisions for the indemnification to the full extent permitted by Law of individuals who, at or prior to the Effective Time, were directors, officers, fiduciaries or agents of the Company or any of its Subsidiaries, which provisions shall not be amended, repealed or otherwise modified for a period of six years from the Effective Time in any manner that would affect adversely the rights thereunder of those individuals, unless such modification shall be required by Law and then only to the minimum extent required by Law.

(b) After the Effective Time, the Surviving Corporation and Parent shall, jointly and severally, to the fullest extent permitted under applicable Law or any applicable contract or agreement in effect on the date hereof and made available to Parent, indemnify and hold harmless, and provide advancement of expenses to, each present and former director and officer of the Company and each of its Subsidiaries (collectively, the "Indemnified Parties") against all costs and expenses (including attorneys' fees), judgments, fines, penalties, losses, claims, damages, liabilities and settlement amounts paid in connection with any claim, action, suit, proceeding or investigation (whether arising before or after the Effective Time), whether civil, criminal, administrative or investigative, arising out of or pertaining to any action or omission, or alleged act or omission, in their capacity as an officer or director, whether occurring before or at the Effective Time, for a period of six years after the date hereof. In the event of any such claim, action, suit, proceeding or investigation, any Indemnified Party wishing to claim indemnification hereunder shall promptly notify Parent upon learning of same; provided that the failure to so notify shall not relieve Parent of any liability it may have hereunder except to the extent such

failure materially prejudices Parent. In the event of any such claim, action, suit, proceeding or investigation, (i) the Surviving Corporation shall pay the reasonable fees and expenses of counsel selected by the Indemnified Parties, which counsel shall be reasonably satisfactory to the Surviving Corporation, promptly after statements therefor are received and (ii) the Surviving Corporation shall cooperate in the defense of any such matter; provided, however, that the Surviving Corporation shall not be liable for any settlement effected without its written consent (which consent shall not be unreasonably withheld or delayed); and provided, further, that neither the Company nor the Surviving Corporation shall be obligated pursuant to this Section 6.06(b) to pay the fees and expenses of more than one counsel (in addition to local counsel) for all Indemnified Parties in any single action except to the extent that two or more of such Indemnified Parties shall have conflicting interests in the outcome of such action; and provided, further, that, in the event that any claim for indemnification is asserted or made within such six year period, all rights to indemnification in respect of such claim shall continue until the disposition of such claim.

- (c) The Surviving Corporation shall maintain in effect for six years from the Effective Time, if available, the current directors' and officers' liability insurance policies maintained by the Company (provided that the Surviving Corporation may substitute therefor policies of at least the same coverage with respect to matters occurring prior to the Effective Time containing terms and conditions that are not in the aggregate less favorable; provided, however, that in no event shall the Surviving Corporation be required to expend pursuant to this Section 6.06(c) more than an amount per year equal to 250% of current annual premiums paid by the Company for such insurance; provided, however, that in the event of an expiration, termination or cancellation of such current policies, Purchaser or the Surviving Corporation shall use their reasonable best efforts to obtain as much coverage as is possible under substantially similar policies for such maximum annual amount in aggregate annual premiums.
- (d) In the event the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in each such case, Parent shall, and shall cause the Surviving Corporation or its successors or assigns, to maintain the policies and honor the obligations provided for in this Section 6.06.

(e) Parent shall cause the Surviving Corporation to perform all of the obligations of the Surviving Corporation under this Section 6.06. The provisions of this Section 6.06 are intended for the benefit of and shall be enforceable by, each of the Indemnified Parties and their respective heirs and representatives.

SECTION 6.07. Further Action;

Reasonable Best Efforts. (a) Upon the terms and subject to the conditions of this Agreement, each of the parties hereto shall use its reasonable efforts to take, or cause to be taken, all appropriate action, and to do, or cause to be done, all things necessary, proper or advisable under applicable Law to consummate and make effective the transactions contemplated hereby, including using its reasonable best efforts to obtain all Permits, consents, approvals, authorizations, qualifications and orders of Governmental Authorities and parties to contracts with the Company and its Subsidiaries as are necessary for the consummation of the transactions contemplated hereby and to fulfill the conditions to the Merger. In case, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement, the proper officers and directors of each party to this Agreement shall use their reasonable best efforts to take all such action. Without limiting the generality of the foregoing, Parent shall discuss in good faith and agree to increase the amount of the Cash Consideration (subject to a corresponding decrease in the amount of the Stock Consideration) in the event that it is determined that the number of shares of Parent Common Stock to be issued in the Merger and the Advisor Merger would require the approval of the stockholders of Parent under the listing standards of the NYSE.

- (b) The parties hereto agree to cooperate and assist one another in connection with all actions to be taken pursuant to Section 6.07(a), including the preparation and making of any filings referred to therein and, if requested, amending or furnishing additional information thereunder, including, subject to applicable Law and the Confidentiality Agreement, providing copies of all related documents to the non-filing party and their advisors prior to filing, and to the extent practicable and permissible under applicable Law neither of the parties will file any such document or have any communication with any Governmental Authority without prior consultation with the other party. Each party shall keep the other apprised reasonably promptly of the content and status of any communications with, and communications from, any Governmental Authority with respect to the transactions contemplated hereby. To the extent practicable and permitted by a Governmental Authority, each party hereto shall permit representatives of the other party to participate in meetings and calls with such Governmental Authority.
- (c) Notwithstanding any other provision contained herein, the Company shall, with respect to any U.S. federal tax filing relating to the REIT status of the Company that is filed by the Company between the date hereof and Closing, give Parent a reasonable opportunity to comment on such filing before it is filed with the IRS and shall accept any reasonable comments (the reasonableness of which shall be determined by the Company in its sole discretion) that are provided by Parent with respect to the content of such filing.
- (d) Each of the parties hereto agrees to cooperate and use its reasonable efforts to vigorously contest and resist any Action, including administrative or judicial Action, and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order (whether temporary, preliminary or permanent) that is in effect and that restricts, prevents or prohibits consummation of the transactions contemplated hereby, including by vigorously pursuing all available avenues of administrative and judicial appeal.

SECTION 6.08. Public

Announcements. Parent, Purchaser and the Company agree that no public release or announcement concerning the transactions contemplated hereby or the Merger shall be issued by either party without the prior consent of the other party (which consent shall not be unreasonably withheld), except as such release or announcement may be required by Law or the

rules or regulations of any securities exchange, in which case the party required to make the release or announcement shall use its reasonable best efforts to allow the other party reasonable time to comment on such release or announcement in advance of such issuance. The parties have agreed upon the form of a joint press release announcing the execution of this Agreement.

SECTION 6.09. Affiliates. Prior to the date of the Company Stockholders' Meeting, the Company shall deliver to Parent a list of names and addresses of those Persons who are, in the Company's reasonable judgment, as of the time of the Company Stockholders' Meeting, "affiliates" of the Company within the meaning of Rule 145 under the Securities Act. There shall be added to such list the names and addresses of any other Person subsequently identified by the Company (in consultation with Parent) as a Person who may be deemed to be such an affiliate of the Company. The Company shall exercise its reasonable efforts to deliver or cause to be delivered to Parent, prior to the Closing, from each affiliate of the Company identified in the foregoing list (as the same may be supplemented as aforesaid), a letter dated as of the Closing Date substantially in the form attached as Exhibit D (the "Affiliate Letter").

SECTION 6.10. Dividends. The Company and Parent shall coordinate the declaration, setting of record dates and payment dates of dividends on shares of Company Common Stock and Parent Common Stock so that holders of shares of Company Common Stock do not receive dividends on both shares of Company Common Stock and shares of Parent Common Stock received in the Merger in respect of the calendar quarter in which the Closing occurs or fail to receive a dividend on either shares of Company Common Stock or shares of Parent Common Stock received in the Merger in

respect of such calendar quarter; provided, however, that nothing herein is intended to prevent holders of shares of Company Common Stock from receiving a dividend on (i) shares of Company Common Stock with respect to the period from the beginning of the calendar quarter in which the Closing occurs to the Closing Date, and (ii) on shares of Parent Common Stock received in the Merger in respect of shares of Company Common Stock with respect to the period from the Closing Date to the end of the calendar quarter in which the Closing occurs.

SECTION 6.11. Stock Exchange
Listing. Parent shall use its reasonable best
efforts to cause the shares of Parent Common
Stock to be issued in the Merger to be approved for
listing on the NYSE subject to official notice of
issuance, prior to the Closing Date. Prior to the
Closing Date, the Company shall cooperate with
Parent and use reasonable efforts to take, or cause
to be taken, all actions, and do or cause to be done
all things, reasonably necessary, proper or
advisable on its part under applicable Laws to
enable the deregistration of the shares of Company
Common Stock under the Exchange Act as
promptly as practicable after the Effective Time.

SECTION 6.12. 1031 Exchange. The Company agrees to cooperate, subject to the restrictions and conditions stated herein, should Parent elect to acquire, immediately before and in connection with the closing of the Merger, one or more of the Company's assets as part of a like-kind exchange under Section 1031 of the Code (such assets thus acquired hereinafter referred to as "Replacement Properties"), including the execution of such documents and the taking of such actions as Parent may reasonably request to complete the acquisition of the Replacement Properties immediately prior to and in connection with the closing of the Merger, provided, however, (i) that any such like-kind exchange shall be consummated through the use of a qualified intermediary or exchange accommodation titleholder (the "Assignee"); (ii) Parent's contemplated exchange shall not impose upon the Company any cost, additional liability or financial obligation, and Parent agrees to hold the Company harmless from any liability that might arise in connection with such exchange. The Company agrees to cooperate with Parent in attempting to close the acquisition of the Replacement Properties two days before the closing for the Merger; provided however that the Company may refuse to consent to an acquisition of the Replacement Properties that does not close immediately before and in connection with the Merger if Company in its sole discretion believes that an earlier closing could have adverse

consequences to the Company. Without limiting the generality of the immediate preceding sentence, any real property transfer, sales, use, transfer, value added, stock transfer and stamp taxes, stamp duties, and any transfer, recording, registration and other fees, charges, premiums and any similar taxes shall be borne solely by Parent; and Parent shall unconditionally guarantee the full and timely performance by the Assignee of each and every one of the representations, warranties, indemnities, obligations and undertakings of the Parent under this Agreement (and any amendments or modifications hereto) that relate to the Replacement Properties. As such guarantor, Parent shall be treated as primary obligor with respect to those representations, warranties, indemnities, obligations and undertakings, and, in the event of breach, the Company may proceed directly against Parent on this guarantee without the need to join Assignee.

In the event any exchange contemplated by Parent should fail to occur, for whatever reason, the Merger shall nonetheless be consummated as provided herein. Notwithstanding anything in this Agreement to the contrary, the Company makes no representation or warranty, and undertakes no other obligation, in each case with respect to the tax consequences of the transactions referred to in this Section 6.12 (including whether the transactions qualify under Section 1031 of the Code) and shall not be responsible for any loss or liability relating to any tax liability to Parent from such transactions.

ARTICLE VII CONDITIONS TO THE MERGER

SECTION 7.01. Conditions to the Merger. The obligations of each party to effect the Merger shall be subject to the satisfaction, at or prior to the Effective Time, of the following conditions:

- (a) Stockholder Approval. The Company shall have obtained the Company Stockholder Approval.
- (b) NYSE Listing. The shares of Parent Common Stock issuable to the Company stockholders pursuant to this Agreement shall have been authorized for listing on the New York Stock Exchange (or in the event Parent has delisted from the NYSE prior to Closing and listed on another national securities exchange or on the Nasdaq National Market, such other national securities exchange or the Nasdaq National Market, as applicable) upon official notice of issuance.
- (c) *No Order*. No Governmental Authority of competent jurisdiction shall have enacted or issued an order, decree, judgment, injunction or taken any other action (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits or makes illegal the consummation of the Merger (collectively, an "*Order*"); *provided, however*, that the party claiming the failure of the condition set forth in this Section 7.01(b) shall have used reasonable best efforts to have such Order vacated.
- (d) S-4. The S-4 Registration Statement shall have become effective under the Securities Act. No stop order suspending the effectiveness of the S-4 Registration Statement shall have been issued, and no proceedings for that purpose shall have been initiated or be threatened by the SEC.
- SECTION 7.02. Conditions to the Obligations of Parent and Purchaser. The obligations of Parent and Purchaser to consummate the Merger are subject to the satisfaction or waiver (where permissible) of the following additional conditions:
- (a) Representations and Warranties. The representations and warranties of the Company contained in this Agreement shall be true and correct as of the date hereof and as of the Effective Time as though made on and as of the Effective Time (except to the extent expressly made as of an earlier date, in which case as of such earlier date),

except for failures to be true and correct that individually or in the aggregate would not reasonably be likely to have a Material Adverse Effect with respect to the Company; *provided*, that for purposes of determining whether the condition in this Section 7.02(a) is satisfied, references to "Material Adverse Effect" and any other materiality qualification contained in such representations and warranties shall be ignored.

- (b) Agreements and Covenants. The Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Effective Time.
- (c) Officer Certificate. The Company shall have delivered to Parent a certificate, dated the date of the Closing, signed by the Chief Executive Officer or Chief Financial Officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 7.02(a) and 7.02(b).
- (d) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Effect that, individually or in the aggregate, has had, or is reasonably likely to have, a Material Adverse Effect with respect to the Company.
- (e) REIT Opinion. Parent shall have received the opinion, in the form attached hereto as Exhibit E, of Greenberg Traurig, LLP, dated the Closing Date, regarding the Company's tax status as a REIT. In rendering such opinion, Greenberg Traurig, LLP may rely on customary assumptions, qualifications and representations as to factual matters, each reasonably acceptable to Parent. Such opinion may be conditioned on representations made by the Company's management (including the Advisor in its capacity as advisor to the Company) regarding its organization, assets, sources of gross

income and other matters related to the conduct of the Company's business operations, each reasonably acceptable to Parent.

(f) Advisor Agreement. The Advisor Merger shall have simultaneously closed.

SECTION 7.03. Conditions to the Obligations of the Company. The obligations of the Company to consummate the Merger are subject to the satisfaction or waiver (where permissible) of the following additional conditions:

- (a) Representations and Warranties. The representations and warranties of Parent and Purchaser contained in this Agreement shall be true and correct in all material respects as of the date hereof and as of the Effective Time, as though made on and as of the Effective Time (except to the extent expressly made as of an earlier date, in which case as of such earlier date), except for failures to be true and correct that individually or in the aggregate would not reasonably be likely to have a Material Adverse Effect with respect to Parent; provided, that for purposes of determining whether the condition in this Section 7.03(a) is satisfied, references to "Material Adverse Effect" and any other materiality qualification contained in such representations and warranties shall be ignored.
- (b) Agreements and Covenants. Parent and Purchaser shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Effective Time.
- (c) Officer Certificate. Parent shall have delivered to the Company a certificate, dated the date of the Closing, signed by the Chief Executive Officer or Chief Financial Officer of Parent, certifying as to the satisfaction of the conditions specified in Sections 7.03(a) and 7.03(b).
- (d) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Effect that, individually or in the aggregate, has had, or is reasonably likely to have, a Material Adverse Effect with respect to Parent.
- (e) REIT Opinion. The Company shall have received the opinion, in the form attached hereto as Exhibit F, of Latham & Watkins LLP, counsel to Parent, dated the Closing Date, regarding Parent's tax status as a REIT. In rendering such opinion, Latham & Watkins LLP may rely on customary

opinions of counsel which opinions are reasonably acceptable to the Company on the date of this Agreement and on customary assumptions, qualifications and representations as to factual matters, each reasonably acceptable to the Company. Such opinion may be conditioned on representations made by Parent's management regarding its organization, assets, sources of gross income and other matters related to the conduct of Parent's business operations, each reasonably acceptable to the Company.

SECTION 7.04. Frustration of Closing Conditions. No party may rely on the failure of any condition set forth in this Article VII to be satisfied if such party's failure to comply with any provision of this Agreement in a material respect has been the proximate cause of, or resulted in, the failure of the condition.

ARTICLE VIII TERMINATION, AMENDMENT AND WAIVER

SECTION 8.01. *Termination*. This Agreement may be terminated and abandoned prior to the Closing Date only as follows:

- (a) by the mutual written consent of Parent and the Company;
- (b) by either of the Company or Parent by written notice to the other:
 - (i) if at the Company Stockholders' Meeting (or at any adjournment or postponement thereof) the Company Stockholder Approval is not obtained;

- (ii) if any Governmental Authority of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Order that is in effect and restrains, enjoins or otherwise prohibits or makes illegal the consummation of the Merger, and such Order shall have become final and non-appealable, *provided, however*, that the party terminating this Agreement pursuant to this Section 8.01(b)(ii) shall have used reasonable best efforts to have such Order vacated; or
- (iii) if the consummation of the Merger shall not have occurred on or before October 31, 2006 (the "Drop Dead Date"); provided, however, that the right to terminate this Agreement under this Section 8.01(b)(iii) shall not be available to either party if such party's breach of any representation, warranty or covenant herein in a material respect has been the proximate cause of, or resulted in, the failure of the Merger to occur on or before the Drop Dead Date.
- (c) by written notice from Parent to the Company, if the Company breaches or fails to perform in any material respect any of its representations, warranties or covenants contained in this Agreement, which breach or failure to perform would give rise to the failure of a condition set forth in Section 7.02(a) or 7.02(b), and such condition is incapable of being satisfied by the Drop Dead Date or such breach has not been cured by the Company within 30 Business Days after the Company's receipt of written notice of such breach from Parent; provided that Parent's right to terminate under this paragraph shall lapse and be of no effect if not exercised in writing within 15 Business Days following the date upon which the right to terminate hereunder first accrued:
- (d) by written notice from the Company to Parent if Parent or Purchaser breaches or fails to perform in any material respect any of its representations, warranties or covenants contained in this Agreement, which breach or failure to perform would give rise to the failure of a condition set forth in Section 7.03(a) or 7.03(b) and such condition is incapable of being satisfied by the Drop Dead Date or such breach has not been cured by Parent or Purchaser within 30 Business Days after Parent's receipt of written notice of such breach from the Company; provided that Company's right to terminate under this paragraph shall lapse and be of no effect if not exercised in writing within 15 Business Days following the date upon which the right to

terminate hereunder first accrued;

- (e) by written notice from the Company to Parent, in connection with entering into a binding written agreement to effect a Superior Proposal; provided, however, that prior to terminating this Agreement pursuant to this Section 8.01(e), (i) the Company (with the authorization of the Company Board and the Special Committee) shall have (1) notified Parent in writing of the Company's receipt of such Superior Proposal (and the material terms and conditions thereof) and its intention to enter into such binding written agreement Proposal, and (2) otherwise complied in all material respects with its obligations under Section 6.04, (ii) Parent does not make, within three Business Days after receipt of the Company's written notice pursuant to clause (i) above, an offer that the Company Board and the Special Committee shall have concluded in good faith (following consultation with their respective financial advisors and outside counsel) is at least as favorable, from a financial point of view, to the stockholders of the Company as such Superior Proposal, and (iii) the Company shall have paid to Parent the Break-up Fee as provided in Section 8.02(b). The Company agrees (x) that it will not enter into the binding agreement referred to in this Section 8.01(e) until at least the fourth Business Day after it has provided the notice to Parent required hereby and (y) to notify Parent promptly if its intention to enter into such agreement referred to in its notice to Parent will change at any time after giving such notice.
- (f) by written notice of Parent to the Company, if the Company Board and the Special Committee shall, prior to the Company Stockholder Approval, (A) fail to include the Company Recommendation in the Prospectus/Proxy Statement, (B) publicly withdraw or knowingly modify, in a manner adverse to Parent or Purchaser, the Company Recommendation, (C) approve any Acquisition Proposal or publicly recommend that the holders of the Company Common Stock accept or approve any other Acquisition Proposal, (D) at any time after the end of fifteen (15) Business Days following

receipt of an Acquisition Proposal, have failed to reaffirm the Company Recommendation as promptly as practicable (but in any event within five (5) Business Days) after receipt of any written request to do so from Parent (which five Business Day period will be extended for an additional five (5) Business Days if the Company certifies to Parent prior to the expiration of the initial five (5) Business Day period that the Company Board and the Special Committee are in good faith seeking to obtain additional information regarding their decision to reaffirm the Company Recommendation), or (E) if a tender offer or exchange offer for no less than twenty percent (20%) of the outstanding shares of Company Common Stock has been publicly disclosed (other than by Parent or an affiliate of Parent) and the Company Board recommends that the stockholders of the Company tender their shares in such tender or exchange offer; provided, further, that Parent's right to terminate under this Section 8.01(f) shall lapse and be of no effect if not exercised in writing within 15 Business Days following the date upon which the right to terminate hereunder first accrued: or

(g) by Parent in the event the Advisor Agreement is terminated in accordance with its terms; *provided* that Parent's right to terminate under this Section 8.01(g) shall lapse and be of no further force and effect if not exercised in writing within 15 Business Days following the date upon which the right to Terminate hereunder first accrued.

SECTION 8.02. Effect of

Termination. (a) Subject to the remainder of this Section 8.02 and to Section 8.03, in the event of the termination of this Agreement pursuant to Section 8.01, this Agreement shall forthwith become null and void and have no effect, without any liability on the part of Parent, Purchaser or the Company and each of their respective directors, trustees, officers, employees, partners, stockholders or stockholders and all rights and obligations of any party hereto shall cease, except for the agreements contained in Sections 6.03 (Confidentiality), 6.08 (Public Announcements), 8.02 (Effect of Termination), 8.03 (Fees and Expenses) and Article IX (General Provisions), which shall remain in full force and effect and survive any termination of this Agreement; provided, however, that nothing contained in this Section 8.02(a) shall relieve any party hereto from liabilities or damages arising out of any fraud or willful breach by such party of any of its representations, warranties, covenants or other agreements contained in this Agreement.

- (b) The Company shall pay to Parent an amount in cash equal to \$107,000,000 (the "*Break-Up Fee*") if:
 - (i) this Agreement is terminated by the Company pursuant to Section 8.01(e);
 - (ii) this Agreement is terminated by Parent pursuant to Section 8.01(f); or
 - (iii) this Agreement is terminated by the Company or Parent pursuant to Section 8.01(b)(i); at any time after the date of this Agreement and before such termination, a bona fide Acquisition Proposal shall have been publicly announced and remains outstanding; and within 12 months of such termination the Company shall have entered into a definitive agreement with respect to or shall have consummated an Acquisition Proposal (substituting 50% for 20% in the definition thereof).
- (c) Any fee due under Section 8.02(b) shall be paid by the Company by wire transfer of same-day funds:
 - (i) in the case of Section 8.02(b)(i), concurrently with such termination;
 - (ii) in the case of Sections 8.02(b)(ii), within two Business Days of such termination; and
 - (iii) in the case of Section 8.02(b)(iii), concurrently with the earlier of entering into such definitive agreement or consummation of such Acquisition Proposal.

SECTION 8.03. Fees and

Expenses. (a) Except as set forth in Sections 8.02 and this Section 8.03, whether or not the Merger is consummated, all fees, costs and expenses incurred in

connection with the preparation, execution and performance of this Agreement and the transactions contemplated hereby, including all fees, costs and expenses of agents, representatives, financial advisors, counsel and accountants shall be paid by the party incurring such fees, costs or expenses.

- (b) If this Agreement is terminated by Parent pursuant to Section 8.01(c), the Company shall pay to Parent within three Business Days after the date of termination all documented, reasonable out-of-pocket costs and expenses, including the reasonable fees and expenses of lawyers, accountants, financial advisors and investment bankers, incurred by Parent in connection with the entering into of this Agreement and the carrying out of any and all acts contemplated hereunder, provided that such fees and expenses to be paid by the Company hereunder shall not exceed \$3,000,000.
- (c) If this Agreement is terminated by the Company pursuant to Section 8.01(d), Parent shall pay to the Company within three Business Days after the date of termination all documented, reasonable out-of-pocket costs and expenses, including the reasonable fees and expenses of lawyers, accountants, financial advisors and investment bankers, incurred by the Company in connection with the entering into of this Agreement and the carrying out of any and all acts contemplated hereunder, provided that such fees and expenses to be paid by Parent hereunder shall not exceed \$3,000,000.

SECTION 8.04. Deferral Provisions to Insure Compliance with REIT Gross Income Tests. (a) Notwithstanding any other provision in this Agreement, any payments otherwise to be made by the Parent to the Company pursuant to Section 8.03(c) or by the Company to Parent under Section 8.02 or Section 8.03(b) for any calendar year shall not exceed the sum of (a) the amount that it is determined should not be gross income of the party receiving such payment (the "Receiving Party") for purposes of the requirements of Sections 856(c)(2) and (3) of the Code, with such determination to be set forth in an opinion of outside tax counsel selected by the Receiving Party (the "Counsel's Gross Income Opinion") plus (b) such additional amount that it is estimated can be paid to the Receiving Party in such taxable year without creating a risk that the payment would cause the Receiving Party to fail to meet the requirements of Section 856(c)(2) and (3) of the Code, determined as if the payment of such amount did not constitute income described in Section 856(c)(2)(A) -(H) and 856(c)(3)(A) (I) of the Code ("Qualifying Income"), which determination shall be made by independent tax

accountants to the Receiving Party and (c) in the event the Receiving Party receives a letter from outside tax counsel to the Receiving Party ("Counsel's Ruling Letter") indicating that Receiving Party has received a ruling from the Internal Revenue Service holding that the Receiving Party's receipt of the additional amount otherwise to be paid under Section 8.03(c) either would constitute Qualifying Income or would be excluded from gross income of the Receiving Party for purposes of Sections 856(c)(2) and (3) of the Code (the "Specified REIT Requirements"), the aggregate payments otherwise required to be made under this Agreement (determined without regard to this Section 8.04) less the amount otherwise previously paid under clauses (a) and (b) above. The obligation of the party required to make such payment (the "Paying Party") to pay any unpaid portion of the payment otherwise required under this Agreement that remains unpaid solely by reason of this Section 8.04 shall terminate five years from the date such payment otherwise would have been made but for this Section 8.04.

(b) The Paying Party shall place the full amount of any payments otherwise to be made by the Paying Party to the Receiving Party under Section 8.02 or 8.03(c) or 8.03(d), as the case may be, in a mutually agreed escrow account upon mutually acceptable terms which shall provide that any portion thereof shall not be released to the Receiving Party unless and until the Paying Party receives any of the following: (x) a letter from the Receiving Party's independent tax accountants indicating the amount that it is estimated can be paid at the time to the Receiving Party without creating a risk that the payment would cause the Receiving Party to fail to meet the Specified REIT Requirements for the taxable year in which the payment would be made, which determination shall be made by such independent tax accountants, (y) Counsel's Ruling Letter or (z) an opinion of outside tax counsel selected by the Receiving Party, to the effect that, based upon a change in Law or interpretation after the date on which payment was first deferred hereunder, receipt of the additional amount otherwise to

be paid under this Agreement either would be excluded from gross income of the Receiving Party for purposes of the Specified REIT Requirements or would constitute Qualifying Income, in any of which events the Paying Party shall pay to the Receiving Party the lesser of the unpaid amounts due under this Agreement (determined without regard to this Section 8.04 or the maximum amount stated in the letter referred to in clause (x) above). At the end of the five year period referred to above in this Section 8.04 with respect to any amount placed in such escrow, if none of the events referred to in clauses (x), (y) or (z) or the preceding sentences shall have occurred, such amount shall be released from such escrow to be used as determined by the Paying Party in its sole and absolute discretion.

ARTICLE IX GENERAL PROVISIONS

SECTION 9.01. *Non-Survival of Representations and Warranties*. The representations and warranties in this Agreement shall terminate at the Effective Time or upon the termination of this Agreement pursuant to Section 8.01, as the case may be.

SECTION 9.02. *Notices*. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by overnight courier, by facsimile or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 9.02):

if to Parent or Purchaser:

Health Care Property Investors, Inc. 3760 Kilroy Airport Way, Suite 300 Long Beach, CA 90806 Telecopier No: (562) 733-5204 Telephone No: (562) 733-5100 Attention: Edward J. Henning General Counsel, Senior Vice President and Corporate Secretary

with a copy to:

Sullivan & Cromwell LLP, 1888 Century Park East, Suite 2100

Los Angeles, CA 90292 Telecopier No.: (310)

712-8800

Telephone No.: (310)

712-6600

Attention: Alison S. Ressler,

Esq.

Patrick S. Brown, Esq.

if to the Company:

CNL Retirement Properties, Inc.

420 South Orange Avenue

Orlando, FL 32801

Telecopier No: (407) 835-3235 Telephone No: (407) 650-1000 Attention: Clark Hettinga Chief Financial Officer and Senior Vice President

with a copy to:

Greenberg Traurig, LLP 200 Park Avenue New York, New York 10166 Telecopier No: (212) 801-6400 Telephone No.: (212) 801-9200 Attention: Judith D. Fryer, Esq. Lorenzo Borgogni, Esq.

and to:

Akin Gump Strauss Hauer & Feld, LLP 590 Madison Avenue New York, NY 10022-2524 Telecopier No: (212) 872-1002 Telephone No.: (212) 872-1000 Attention: Patrick J. Dooley, Esq.

SECTION 9.03. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

SECTION 9.04. Entire Agreement; Assignment. This Agreement constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes, except as set forth in Section 6.03(c), all prior agreements and undertakings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof. This Agreement and any of the rights or obligations hereunder shall not be assigned (whether pursuant to a merger, by operation of Law or otherwise) by any party hereto without the prior written consent of the other parties and any such purported assignment shall be null and void, except that Parent and Purchaser may assign all or any of their rights and obligations hereunder to any wholly owned Subsidiary of Parent, provided that no such assignment shall relieve the assigning party of its

obligations hereunder and such assignment does not alter the type or amount of the Merger Consideration and would not and would not reasonably be likely to prevent, materially delay or materially impair the consummation of the transactions contemplated by this Agreement. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the parties hereto and their respective successors and permitted assigns.

SECTION 9.05. Parties in Interest. This Agreement shall be binding upon and inure solely to the benefit of each party hereto, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement, other than Section 6.06 (which is intended to be for the benefit of the Indemnified Parties and may be enforced by such parties).

SECTION 9.06. *Remedies*. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party shall be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy shall not preclude the exercise of any other remedy. The parties hereto agree that irreparable

A-45

damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which the parties are entitled at Law or in equity.

SECTION 9.07. Governing Law. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of New York applicable to contracts executed in and to be performed in that State (other than the Merger and those provisions set forth herein that are required to be governed by the MGCL). All actions and proceedings arising out of or relating to this Agreement shall be heard and determined exclusively in any New York state or federal court sitting in New York, New York (Borough of Manhattan). The parties hereto hereby (a) submit to the exclusive jurisdiction of any such court for the purpose of any Action arising out of or relating to this Agreement brought by any party hereto, and (b) irrevocably waive, and agree not to assert by way of motion, defense, or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any of the above-named courts. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 9.02 as to giving notice hereunder shall be deemed effective service of process on such party.

SECTION 9.08. Waiver of Jury Trial. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

SECTION 9.09. *Interpretation*. All references in this Agreement and either Disclosure Schedule to Articles, Sections and Exhibits refer to Articles and Sections of, and Exhibits to, this Agreement unless the context requires otherwise. The headings contained in this Agreement are for reference purposes only and shall not affect in any

way the meaning or interpretation of this Agreement. Wherever used herein, a pronoun in the masculine gender shall be considered as including the feminine gender unless the context clearly indicates otherwise and the word "including" shall mean "including without limitation." The phrases "herein," "hereof," "hereunder" and words of similar import will be deemed to refer to this Agreement as a whole, including the Exhibits and Schedules hereto, and not to any particular provision of this Agreement. The word "or" will be inclusive and not exclusive unless the context requires otherwise. All references in this Agreement to any particular Law will be deemed to refer also to any rules and regulations promulgated under that Law. References to a Person also refer to its predecessors and successors and permitted assigns.

SECTION 9.10. Performance

Guaranty. Parent hereby guarantees the due, prompt and faithful performance and discharge by, and compliance with, all of the obligations, covenants, terms, conditions and undertakings of Purchaser under this Agreement in accordance with the terms hereof.

SECTION 9.11. Amendment. This Agreement may be amended by the parties hereto by action taken by or on behalf of their respective Boards of Directors at any time prior to the Effective Time; provided, however, that, after the approval and adoption of this Agreement and the transactions contemplated hereby by the stockholders of the Company, no amendment may be made that would, under applicable Law, require further approval by such stockholders without such further approval. This Agreement may not be amended except by an instrument in writing signed by each of the parties hereto.

A-46

SECTION 9.12. Waiver. At any time prior to the Effective Time, any party hereto may
(a) extend the time for the performance of any obligation or other act of any other party hereto,
(b) waive any inaccuracy in the representations and warranties of any other party contained herein or in any document delivered pursuant hereto and
(c) waive compliance with any agreement of any other party or any condition to its own obligations contained herein. Any such extension or waiver shall be valid if set forth in an instrument in writing signed by the party or parties to be bound thereby.

SECTION 9.13. Counterparts. This Agreement may be executed and delivered (including by facsimile transmission) in one or more counterparts, and by the different parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.

[Remainder of page intentionally left blank]

A-47

IN WITNESS WHEREOF, Parent, Purchaser and the Company have caused this Agreement to be executed as of the date first written above by their respective officers thereunto duly authorized.

HEALTH CARE PROPERTY INVESTORS, INC.

By /s/ EDWARD J. HENNING

Name: Edward J. Henning
Title: Senior Vice
President, General
Counsel and
Corporate Secretary

OCEAN ACQUISITION 1, INC.

By /s/ EDWARD J. HENNING

Name: Edward J. Henning
Title: Authorized
Representative

CNL RETIREMENT PROPERTIES, INC.

By /s/ STUART J. BEEBE

Name: Stuart J. Beebe
Title: Chief Executive
Officer and
President

Annex B

Banc of America Securities Opinion

May 1, 2006

Special Committee of the Board of Directors and Board of Directors CNL Retirement Properties, Inc. CNL Center at City Commons 450 South Orange Avenue Orlando, Florida 32801

Members of the Special Committee and the Board of Directors:

You have requested our opinion as to the fairness, from a financial point of view, to the holders of Company Common Stock (as defined below) of the consideration proposed to be received by such holders pursuant to the terms of the Agreement and Plan of Merger, dated as of May 1, 2006 (the "Agreement"), among CNL Retirement Properties, Inc. (the "Company"), Health Care Property Investors, Inc. (the "Purchaser") and Ocean Acquisition 1, Inc., a wholly owned subsidiary of the Purchaser ("Merger Sub"). As more fully described in the Agreement, the Company will be merged with and into Merger Sub (the "Merger"), with Merger Sub continuing as the surviving corporation in the Merger, and each share of common stock, par value \$0.01 per share, of the Company (the "Company Common Stock") will be converted into the right to receive (i) \$11.1293 in cash (the "Cash Consideration") and (ii) 0.0865 shares (such number of shares, the "Stock Consideration", and together with the Cash Consideration, the "Merger Consideration") of common stock, par value \$1.00 per share, of the Purchaser (the "Purchaser Common Stock"). In addition, the Agreement provides that if the number of shares of Purchaser Common Stock to be issued in the Merger and the Advisor Merger (as defined below) would require the approval of the stockholders of the Purchaser under the listing standards of the New York Stock Exchange, then the Purchaser will discuss in good faith and agree to increase the amount of Cash Consideration, subject to a corresponding decrease in the amount of Stock Consideration (the "Consideration Adjustment"). The terms and conditions of the Merger are more fully set out in the Agreement.

We understand that, concurrently with the execution of the Agreement, the Purchaser, CNL Retirement Corp., the advisor to the Company (the "Advisor"), and Ocean Acquisition 2, LLC, a wholly owned subsidiary of the Purchaser (the "Advisor Merger Sub"), will enter into a definitive agreement pursuant to which the Advisor will be

merged with and into the Advisor Merger Sub (the "Advisor Merger") and all of the outstanding common stock of the Advisor will be converted into the right to receive aggregate consideration of approximately \$120 million in shares of Purchaser Common Stock. We also understand that the closing of the Merger will be conditioned upon, among other things, the closing of the Advisor Merger.

For purposes of the opinion set forth herein, we have:

- (i)
 reviewed certain publicly
 available financial statements
 and other business and
 financial information of the
 Company and the Purchaser,
 respectively;
- reviewed certain internal financial statements and other financial and operating data concerning the Company and the Purchaser, respectively;
- (iii)

 reviewed certain financial
 forecasts relating to the
 Company prepared by the
 management of the Company
 (the "Company Forecasts");
- reviewed with Purchaser's management certain financial forecasts relating to the Purchaser prepared by the management of the Purchaser for calendar year 2006 (the "Purchaser")

B-1

Forecasts") and discussed with Purchaser's management publicly available research analysts' estimates of the future financial performance of the Purchaser, which included a certain research analyst's estimates of the future financial performance of the Purchaser for calendar years 2006 and 2007 (the "Research Estimates");

- (v)

 reviewed and discussed with
 senior executives of the
 Company and the Purchaser
 information relating to cost
 savings estimated by the
 management of the Company
 to result from the Merger (the
 "Cost Savings");
- (vi)

 discussed the past and current operations, financial condition and prospects of the Company and the Purchaser with senior executives of the Company and the Purchaser;
- (vii)

 reviewed the potential pro
 forma financial impact of the
 Merger on the future financial
 performance of the Purchaser,
 including the potential effect
 on the Purchaser's funds from
 operations, cash flow,
 consolidated capitalization and
 financial ratios;
- (viii)

 reviewed the reported prices
 and trading activity for the
 Purchaser Common Stock;
- compared the financial performance of the Company and the Purchaser and the prices and trading activity of the Purchaser Common Stock with that of certain other publicly traded companies we deemed relevant;
- (x)

 compared certain financial
 terms of the Merger to
 financial terms, to the extent
 publicly available, of certain

other business combination transactions we deemed relevant:

participated in discussions and negotiations among representatives of the Company and the Purchaser and their respective advisors;

(xii)
reviewed the Agreement and
certain related documents; and

(xiii)

performed such other analyses
and considered such other
factors as we have deemed
appropriate.

We have assumed and relied upon, without independent verification, the accuracy and completeness of the financial and other information reviewed by us for the purposes of this opinion. With respect to the Company Forecasts, we have assumed, at the direction of the Company, that they have been reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of the Company as to the future financial performance of the Company. With respect to the Purchaser Forecasts, we have assumed, upon the advice of the Purchaser, that they have been reasonably prepared on bases reflecting the best currently available estimates and good faith judgments of the management of the Purchaser as to the future financial performance of the Purchaser for calendar year 2006. As you are aware, we have not been provided with, and did not have any access to, any financial projections of the Purchaser prepared by management of the Purchaser for periods beyond calendar year 2006. Accordingly, after discussions with the Purchaser's management concerning publicly available research analysts' estimates of the future financial performance of the Purchaser generally and with the Company's consent, we have assumed that the Research Estimates are a reasonable basis upon which to evaluate the future financial performance of the Purchaser for calendar years 2006 and 2007 and, we have relied, at the direction of the Company, on the Research Estimates for purposes of our opinion. With respect to the Cost Savings estimated by the management of the Company to result from the Merger, we have assumed that such Cost Savings will be realized substantially in accordance with such estimates. We have not made any independent valuation or appraisal of the assets or liabilities of the Company, nor have we been furnished with any such valuations or appraisals. For purposes of our opinion we have

assumed, at the direction of the Company, that no Consideration Adjustment will be made and that the Merger will be consummated as provided in the Agreement, with full satisfaction of all covenants and conditions set forth in the Agreement and without any waivers thereof. We also have assumed, with your consent, that all

B-2

governmental, regulatory or other consents and approvals necessary for the consummation of the Merger will be obtained without any adverse effect on the Company, the Purchaser or the Merger. We have been advised by the managements of the Company and the Purchaser that each of the Company and the Purchaser has operated in conformity with the requirements for qualification as a real estate investment trust ("REIT") for federal income tax purposes since its formation as a REIT and further have assumed, with your consent, that the Merger will not adversely affect the status or operations of the Company or the Purchaser as a REIT.

We express no view or opinion as to any terms or aspects of the transactions contemplated by the Agreement other than the Merger Consideration (assuming no Consideration Adjustment is made) to the extent expressly specified herein (including, without limitation, the form or structure of the Merger, any terms or aspects of the Advisor Merger or the fairness, from a financial point of view, of the consideration to be received by the Advisor's stockholders in the Advisor Merger). In addition, no opinion is expressed as to the relative merits of the Merger in comparison to other transactions available to the Company or in which the Company might engage or as to whether any transaction might be more favorable to the Company as an alternative to the Merger, nor are we expressing any opinion as to the underlying business decision of the Special Committee of the Board of Directors of the Company to recommend, or the Board of Directors of the Company to proceed with or effect, the Merger. In addition, we are not expressing any opinion as to the value of the Advisor or what the value of Purchaser Common Stock actually will be when issued or the prices at which Purchaser Common Stock may trade at any time.

We have acted as sole financial advisor to the Board of Directors of the Company in connection with the Merger for which services we have received and will receive fees, a portion of which was payable upon our engagement, a portion of which is payable upon the rendering of this opinion and a significant portion of which is payable upon the consummation of the Merger. We or our affiliates have provided and in the future may provide financial advisory and financing services to the Company, the Purchaser, certain of their respective affiliates and to certain entities owned or controlled by, or affiliated with, James Seneff, the Chairman of the Board of the Company and the Chairman of the Board and a significant stockholder of the Advisor (the "Chairman"), and have received and in the future may receive fees

for the rendering of these services, including, among other things, acting as (i) lead arranger, book manager, book runner, administrative agent and lender for certain credit facilities of the Company and certain of its and the Chairman's affiliates, (ii) financial advisor to certain affiliates of the Company and the Chairman in connection with certain acquisitions and dispositions by such affiliates, (iii) provider of acquisition financing to certain affiliates of the Company and the Chairman in connection with certain acquisitions by such affiliates, (iv) lead manager and sole book runner for certain debt and equity offerings by certain affiliates of the Company and the Chairman, (v) co-lead arranger, joint book manager, administrative agent and lender for a credit facility of the Purchaser, (vi) manager for certain debt offerings by the Purchaser and (vii) agent for certain medium term note issuances by the Purchaser. In addition, a member of the board of directors of one of our affiliates is also a member of the board of directors of the Purchaser. In the ordinary course of our businesses, we and our affiliates may actively trade the debt and equity securities or loans of the Company, the Purchaser, certain of their respective affiliates and certain entities owned or controlled by, or affiliated with, the Chairman for our own account or for the accounts of customers, and accordingly, we or our affiliates may at any time hold long or short positions in such securities or loans.

It is understood that this letter is for the benefit and use of the Special Committee of the Board of Directors of the Company and the Board of Directors of the Company in connection with and for purposes of their respective evaluations of the Merger. Our opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof. It should be understood that subsequent developments may affect this opinion, and we do not have any obligation to update, revise, or reaffirm this opinion. In addition, we express no opinion or

B-3

recommendation as to how the stockholders of the Company should vote or act in connection with the Merger.

Based upon and subject to the foregoing, including the various assumptions and limitations set forth herein, we are of the opinion on the date hereof that the Merger Consideration to be received by the holders of Company Common Stock in the proposed Merger is fair, from a financial point of view, to such holders.

Very truly yours,

/s/ Banc of America Securities LLC

BANC OF AMERICA SECURITIES LLC

B-4

Annex C

Houlihan Lokey Howard & Zukin Financial Advisors, Inc. Opinion

May 1, 2006

The Special Committee of the Board of Directors CNL Retirement Properties, Inc. 450 South Orange Avenue Orlando, FL 32801

Dear Members of the Special Committee of the Board of Directors:

We understand that CNL Retirement Properties, Inc., a Maryland corporation, (the "Company") is a publicly held REIT that owns approximately 263 assets consisting primarily of medical office buildings and retirement communities. The Company's day to day operations are performed by an affiliate of the Company, CNL Retirement Corp., a Florida corporation (the "Advisor"). Specifically, the Advisor or its affiliates provide management, acquisition, advisory, and administrative services to the Company.

We further understand that the Company intends to enter into an Agreement and Plan of Merger (the "Merger Agreement") by and among Healthcare Property Investors Inc. ("HCP"), a subsidiary of HCP ("Purchaser"), and the Company pursuant to which the Company shall merge into Purchaser (the "Merger") and in connection therewith each of the outstanding shares of the Company's common stock shall be converted into, and become exchangeable for consideration consisting of (i) \$11.1293 in cash (the "Cash Consideration") and (ii) .0865 shares of HCP's common stock (with such shares of HCP common stock being referred to herein as the "Stock Consideration" and, together with the Cash Consideration, the "Merger Consideration"). The Merger and the associated exchange of Company shares for the Merger Consideration is referred to herein as the "Transaction."

Finally, we understand that simultaneously with the execution of the Merger Agreement, HCP and a wholly owned merger subsidiary of HCP are entering into an agreement and plan of merger (the "Advisor Agreement") with the Advisor pursuant to which at the closing of the transactions contemplated by the Advisor Agreement, the Advisor shall merge with and into such subsidiary of HCP (the "Advisor Merger") and the owners of the Advisor shall receive shares of HCP common stock in accordance with the terms of the Advisor

Agreement and, to the extent contemplated by the Advisor Agreement, certain other consideration (collectively, the "Advisor Consideration" and together with the Merger Consideration, the "Aggregate Consideration").

You have requested that Houlihan Lokey Howard & Zukin Financial Advisors, Inc. ("Houlihan Lokey") provide an opinion (the "Opinion") as to whether, as of the date hereof, the Merger Consideration to be received by the stockholders of the Company in the Transaction is fair to them from a financial point of view and as to whether the portion of the Aggregate Consideration that consists of the Advisor Consideration is fair to the stockholders of the Company from a financial point of view.

In giving our opinion as to the fairness of the Advisor Consideration to the stockholders of the Company from a financial point of view, we considered (i) whether the Advisor Consideration is less than or within a reasonable range of fair market values of the Advisor, (ii) whether the Advisor Consideration relative to the fair market value of the Advisor is less than or approximately equivalent to the Merger Consideration relative to the fair market value of the Company, and (iii) whether the portion of the Aggregate Consideration that consists of the Advisor Consideration is within the range indicated by other transactions that we deemed relevant to our opinion.

C-1

In connection with this Opinion, we have made such reviews, analyses and inquiries as we have deemed necessary and appropriate under the circumstances. Among other things, we have:

- 1.
 reviewed the Company's
 annual report to shareholders
 on Form 10-K for the fiscal
 year ended December 31,
 2005;
- 2. reviewed the Advisor's audited financial statements for the fiscal years ended December 31, 2004 and December 31, 2005;
- 3.
 reviewed HCP's annual report
 to shareholders on Form 10-K
 for the fiscal year ended
 December 31, 2005;
- 4.

 held conversations with
 representatives of the
 Company and the Advisor
 regarding the operations,
 financial condition, future
 prospects and projected
 operations and performance of
 the Company and the Advisor
 and regarding the Transaction;
- 5.

 held conversations with
 representatives of the
 Company's counsel regarding
 the Company, the Transaction,
 and related matters;
- 6.

 held conversations with the representatives of the Advisor's investment bankers regarding the Advisor, the Transaction and related matters;
- 7.
 reviewed the draft Merger
 Agreement by and among the
 Purchaser and the Company,
 dated May 1, 2006;

8.

reviewed the draft Advisor Agreement by and among the Purchaser and the Advisor dated May 1, 2006;

9.

reviewed the Offering Memorandum pertaining to the Company and prepared by Banc of America Securities on behalf of the Company dated March 2006;

10.

reviewed the Offering Memorandum pertaining to the Advisor, and prepared by Stifel Nicolaus & Company, Incorporated dated March 2006;

11.

held conversations with Banc of America Securities, CRP's financial advisor, regarding CRP's sale process;

12.

reviewed the fiscal year ended December 31, 2006 Budget for the Advisor prepared by the Advisor's management;

13.

reviewed forecasts and projections prepared by the Company's management with respect to the Company for the fiscal years ending December 31, 2006 through 2010;

14.

reviewed forecasts and projections prepared by the Advisor's management with respect to the Advisor for the fiscal years ending December 31, 2006 through 2010;

15.

reviewed the HCP Financial Review and 2006 Annual Operating Plan prepared by HCP management and dated February 10, 2006;

16.

reviewed the historical market prices and trading volume for HCP's publicly traded

securities and those of certain publicly traded companies which we deemed relevant;

17.

reviewed certain publicly available financial data for certain companies that we deemed relevant and publicly available transaction prices and premiums paid in other change of control transactions that we deemed relevant for companies in related industries to the Company and the Advisor; and

18.

conducted such other financial studies, analyses and inquiries, as we deemed appropriate.

C-2

We have relied upon and assumed, without independent verification, the accuracy and completeness of all data, material and other information furnished, or otherwise made available, to us, discussed with or reviewed by us, or publicly available, and do not assume any responsibility with respect to such data, material and other information. In addition, management of the Company and the Advisor have advised us, and we have assumed, without independent verification, that the financial forecasts and projections (including without limitation any synergies) have been reasonably prepared on bases reflecting the best currently available estimates and judgments of the future financial results and condition of the Company and the Advisor, and we express no opinion with respect to such forecasts and projections or the assumptions on which they are based. We have relied upon and assumed, without independent verification, that there has been no material change in the assets, liabilities, financial condition, results of operations, business or prospects of the Company, or the Advisor, or the Purchaser and HCP since the date of the most recent financial statements provided to us, and that there is no information or facts that would make the information reviewed by us incomplete or misleading. We have also assumed that neither the Company, the Advisor, the Purchaser nor HCP is a party to any material pending transaction, including, without limitation, any external financing, recapitalization, acquisition or merger, divestiture or spin-off (other than the Transaction and the Advisor Merger).

We have relied upon and assumed, without independent verification, that (a) the representations and warranties of all parties to the agreements identified in items 7 - 8 above and all other related documents and instruments that are referred to therein are true and correct, (b) each party to all such agreements will perform all of the covenants and agreements required to be performed by such party, (c) all conditions to the consummation of the Transaction will be satisfied without waiver thereof, and (d) the Transaction will be consummated in a timely manner in accordance with the terms described in the agreements provided to us, without any amendments or modifications thereto or any adjustment to the aggregate consideration (through offset, reduction, indemnity claims, post-closing purchase price adjustments or otherwise). We have also relied upon and assumed, without independent verification, that all governmental, regulatory, and other consents and approvals necessary for the consummation of the Transaction will be obtained and that no delay, limitations, restrictions or conditions will be imposed that would result in the disposition of any material portion of the assets of the Company, or the Advisor, or the Purchaser or HCP, or otherwise have an adverse effect on the

Company, or the Advisor, or the Purchaser or HCP, or the expected benefits of the Transaction. In addition, we have relied upon and assumed, without independent verification, that the final forms of the draft documents identified above will not differ in any material respect from such draft documents.

Furthermore, we have not been requested to make, and have not made, any physical inspection or independent appraisal or evaluation of any of the assets, properties or liabilities (contingent or otherwise) of the Company, the Advisor, the Purchaser or HCP, or any other party, nor were we provided with any such appraisal or evaluation. We express no opinion regarding the liquidation value of any entity. Furthermore, we have undertaken no independent analysis of any potential or actual litigation, regulatory action, possible unasserted claims or other contingent liabilities, to which the Company, the Advisor, the Purchaser or HCP is a party or may be subject, or of any governmental investigation of any possible unasserted claims or other contingent liabilities to which the Company, the Advisor, the Purchaser or HCP is a party or may be subject. With your consent, this Opinion makes no assumption concerning, and therefore does not consider, the potential effects of any such litigation, claims or investigations or possible assertions of claims, outcomes or damages arising out of any such matters.

We have not been requested to, and did not,
(a) initiate any discussions with, or solicit any
indications of interest from, third parties with
respect to the Transaction or any alternatives to the
Transaction, (b) negotiate the terms of the
Transaction, or (c) advise the Board of Directors of
the Company or the Special Committee of the
Board of Directors of the Company or any other
party with

C-3

respect to alternatives to the Transaction. This Opinion is necessarily based on financial, economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof. We have not undertaken, and are under no obligation, to update, revise, reaffirm or withdraw this Opinion, or otherwise comment on or consider events occurring after the date hereof. We have not considered, nor are we expressing any opinion herein with respect to, the prices at which the Company's common stock or HCP's common stock has traded or may trade subsequent to the disclosure or consummation of the Transaction. We have assumed that HCP's common stock to be issued in the Transaction to the Company's common stockholders will be freely tradable and listed on the New York Stock Exchange or any other securities exchange as set forth in the Merger Agreement.

This Opinion is furnished for the use and benefit of the Special Committee of the Board of Directors in connection with its consideration of the Transaction, and is not intended to be used, and may not be used, for any other purpose, without our express, prior written consent. This Opinion is not intended to be, and does not constitute, a recommendation to any security holder as to how such security holder should vote in connection with the Transaction.

In the ordinary course of business, certain of our affiliates may acquire, hold or sell, long or short positions, or trade or otherwise effect transactions, in debt, equity, and other securities and financial instruments (including bank loans and other obligations) of the Company, the Advisor, HCP and any other party that may be involved in the Transaction.

Houlihan Lokey, or its affiliates, have provided certain other financial advisory and investment banking services for the Company, and have received fees for rendering such services, and we may continue to do so in the future. Certain of Houlihan Lokey's affiliates engaged in the lending business is a lender to the Company with respect to a certain mortgage relating to the Company's asset known as Prime Care in an amount of approximately \$19.340 million.

We have not been requested to opine as to, and this Opinion does not address: (i) the underlying business decision of the Company or its security holders or any other party to proceed with or effect the Transaction, (ii) the fairness of any portion or aspect of the Transaction not expressly addressed in this Opinion, (iii) the fairness of any portion or aspect of the Transaction to the holders of any class of securities, creditors or other constituencies of the Company, or any other party

other than those set forth in this Opinion, (iv) the relative merits of the Transaction as compared to any alternative business strategies that might exist for the Company or any other party or the effect of any other transaction in which the Company or any other party might engage, (v) the tax or legal consequences of the Transaction to the Company or its security holders or any other party, or (vi) whether or not the Company or its security holders or any other party is receiving or paying reasonably equivalent value in the Transaction. Furthermore, no opinion, counsel or interpretation is intended in matters that require legal, regulatory, accounting, insurance, tax or other similar professional advice. It is assumed that such opinions, counsel or interpretations have been or will be obtained by the Company and other parties from the appropriate professional sources.

Based upon and subject to the foregoing, and in reliance thereon, it is our opinion that, as of the date hereof, (i) the Merger Consideration to be received by the stockholders of the Company in the Transaction is fair to them from a financial point of view and (ii) the portion of the Aggregate Consideration that consists of the Advisor Consideration is fair to the stockholders of the Company from a financial point of view.

/s/ HOULIHAN LOKEY HOWARD & ZUKIN FINANCIAL ADVISORS, INC.

C-4

Annex D

Sections 3-201 et. Seq. of Maryland General Corporation Law

§ 3-201. "Successor" defined.

- (a) Corporation amending charter. In this subtitle, except as provided in subsection (b) of this section, "successor" includes a corporation which amends its charter in a way which alters the contract rights, as expressly set forth in the charter, of any outstanding stock, unless the right to do so is reserved by the charter of the corporation.
- (b) Corporation whose stock is acquired. When used with reference to a share exchange, "successor" means the corporation the stock of which was acquired in the share exchange.

§ 3-202. Right to fair value of stock.

- (a) General rule. Except as provided in subsection (c) of this section, a stockholder of a Maryland corporation has the right to demand and receive payment of the fair value of the stockholder's stock from the successor if:
 - (1) The corporation consolidates or merges with another corporation;
 - (2) The stockholder's stock is to be acquired in a share exchange;
 - (3) The corporation transfers its assets in a manner requiring action under § 3-105(e) of this title;
 - (4) The corporation amends its charter in a way which alters the contract rights, as expressly set forth in the charter, of any outstanding stock and substantially adversely affects the stockholder's rights, unless the right to do so is reserved by the charter of the corporation; or
 - (5) The transaction is governed by § 3-602 of this title or exempted by § 3-603(b) of this title.
 - (b) Basis of fair value.
 - (1) Fair value is determined as of the close of business:
 - (i) With respect to a merger under § 3-106 of this title of a 90 percent or more

owned subsidiary with or into its parent corporation, on the day notice is given or waived under § 3-106; or

- (ii) With respect to any other transaction, on the day the stockholders voted on the transaction objected to.
- (2) Except as provided in paragraph (3) of this subsection, fair value may not include any appreciation or depreciation which directly or indirectly results from the transaction objected to or from its proposal.
- (3) In any transaction governed by § 3-602 of this title or exempted by § 3-603(b) of this title, fair value shall be value determined in accordance with the requirements of § 3-603(b) of this title.

D-1

- (c) When right to fair value does not apply. Unless the transaction is governed by § 3-602 of this title or is exempted by § 3-603(b) of this title, a stockholder may not demand the fair value of the stockholder's stock and is bound by the terms of the transaction if:
 - (1) The stock is listed on a national securities exchange, is designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc., or is designated for trading on the NASDAQ Small Cap Market:
 - (i) With respect to a merger under § 3-106 of this title of a 90 percent or more owned subsidiary with or into its parent corporation, on the date notice is given or waived under § 3-106; or
 - (ii) With respect to any other transaction, on the record date for determining stockholders entitled to vote on the transaction objected to;
 - (2) The stock is that of the successor in a merger, unless:
 - (i) The merger alters the contract rights of the stock as expressly set forth in the charter, and the charter does not reserve the right to do so; or
 - (ii) The stock is to be changed or converted in whole or in part in the merger into something other than either stock in the successor or cash, scrip, or other rights or interests arising out of provisions for the treatment of fractional shares of stock in the successor;
 - (3) The stock is not entitled, other than solely because of § 3-106 of this title, to be voted on the transaction or the stockholder did not own the shares of stock on the record date for determining stockholders entitled to vote on the transaction;

- (4) The charter provides that the holders of the stock are not entitled to exercise the rights of an objecting stockholder under this subtitle; or
- (5) The stock is that of an open-end investment company registered with the Securities and Exchange Commission under the Investment Company Act of 1940 and the value placed on the stock in the transaction is its net asset value.

§ 3-203. Procedure by stockholder.

- (a) Specific duties. A stockholder of a corporation who desires to receive payment of the fair value of the stockholder's stock under this subtitle:
 - Shall file with the corporation a written objection to the proposed transaction:
 - (i) With respect to a merger under § 3-106 of this title of a 90 percent or more owned subsidiary with or into its parent corporation, within 30 days after notice is given or waived under § 3-106; or
 - (ii) With respect to any other transaction, at or before the stockholders' meeting at which the transaction will be considered or, in the case of action taken under § 2-505(b) of this article, within 10 days after the corporation gives the notice required by § 2-505(b) of this article;
 - (2) May not vote in favor of the transaction; and
 - (3) Within 20 days after the Department accepts the articles for record, shall make a written demand on the successor for payment for the stockholder's stock, stating the number and class of shares for which the stockholder demands payment.
- (b) Failure to comply with section. A stockholder who fails to comply with this section is bound by the terms of the consolidation, merger, share exchange, transfer of assets, or charter amendment.

§ 3-204. Effect of demand on dividend and other rights.

A stockholder who demands payment for his stock under this subtitle:

- (1) Has no right to receive any dividends or distributions payable to holders of record of that stock on a record date after the close of business on the day as at which fair value is to be determined under § 3-202 of this subtitle; and
- (2) Ceases to have any rights of a stockholder with respect to that stock, except the right to receive payment of its fair value.

§ 3-205. Withdrawal of demand.

A demand for payment may be withdrawn only with the consent of the successor.

§ 3-206. Restoration of dividend and other rights.

- (a) When rights restored. The rights of a stockholder who demands payment are restored in full, if:
 - (1) The demand for payment is withdrawn;
 - (2) A petition for an appraisal is not filed within the time required by this subtitle;
 - (3) A court determines that the stockholder is not entitled to relief; or
 - (4) The transaction objected to is abandoned or rescinded.
- (b) Effect of restoration. The restoration of a stockholder's rights entitles him to receive the dividends, distributions, and other rights he would have received if he had not demanded payment for his stock. However, the restoration does not prejudice any corporate proceedings taken before the restoration.

§ 3-207. Notice and offer to stockholders.

- (a) Duty of successor.
 - (1) The successor promptly shall notify each objecting stockholder in writing of the date the articles are

accepted for record by the Department.

- (2) The successor also may send a written offer to pay the objecting stockholder what it considers to be the fair value of his stock. Each offer shall be accompanied by the following information relating to the corporation which issued the stock:
 - (i) A balance sheet as of a date not more than six months before the date of the offer;
 - (ii) A profit and loss statement for the 12 months ending on the date of the balance sheet; and
 - (iii) Any other information the successor considers pertinent.
- (b) Manner of sending notice. The successor shall deliver the notice and offer to each objecting stockholder personally or mail them to him by certified mail, return receipt requested, bearing a postmark from the United States Postal Service, at the address he gives the successor in writing, or, if none, at his address as it appears on the records of the corporation which issued the stock.

§ 3-208. Petition for appraisal; consolidation of proceedings; joinder of objectors.

(a) Petition for appraisal. Within 50 days after the Department accepts the articles for record, the successor or an objecting stockholder who has not received payment for his stock may petition a

D-3

court of equity in the county where the principal office of the successor is located or, if it does not have a principal office in this State, where the resident agent of the successor is located, for an appraisal to determine the fair value of the stock.

- (b) Consolidation of suits; joinder of objectors.
 - (1) If more than one appraisal proceeding is instituted, the court shall direct the consolidation of all the proceedings on terms and conditions it considers proper.
 - (2) Two or more objecting stockholders may join or be joined in an appraisal proceeding.

§ 3-209. Notation on stock certificate.

- (a) Submission of certificate. At any time after a petition for appraisal is filed, the court may require the objecting stockholders parties to the proceeding to submit their stock certificates to the clerk of the court for notation on them that the appraisal proceeding is pending. If a stockholder fails to comply with the order, the court may dismiss the proceeding as to him or grant other appropriate relief.
- (b) Transfer of stock bearing notation. If any stock represented by a certificate which bears a notation is subsequently transferred, the new certificate issued for the stock shall bear a similar notation and the name of the original objecting stockholder. The transferee of this stock does not acquire rights of any character with respect to the stock other than the rights of the original objecting stockholder.

§ 3-210. Appraisal of fair value.

- (a) Court to appoint appraisers. If the court finds that the objecting stockholder is entitled to an appraisal of his stock, it shall appoint three disinterested appraisers to determine the fair value of the stock on terms and conditions the court considers proper. Each appraiser shall take an oath to discharge his duties honestly and faithfully.
- (b) Report of appraisers Filing. Within 60 days after their appointment, unless the court sets a longer time, the appraisers shall determine the fair value of the stock as of the appropriate date and file a report stating the conclusion of the majority as to the fair value of the stock.
- (c) Same Contents. The report shall state the reasons for the conclusion and shall include a

transcript of all testimony and exhibits offered.

- (d) Same Service; objection.
 - (1) On the same day that the report is filed, the appraisers shall mail a copy of it to each party to the proceedings.
 - (2) Within 15 days after the report is filed, any party may object to it and request a hearing.

§ 3-211. Action by court on appraisers' report.

- (a) Order of court. The court shall consider the report and, on motion of any party to the proceeding, enter an order which:
 - (1) Confirms, modifies, or rejects it; and
 - (2) If appropriate, sets the time for payment to the stockholder.

D-4

- (b) Procedure after order.
 - (1) If the appraisers' report is confirmed or modified by the order, judgment shall be entered against the successor and in favor of each objecting stockholder party to the proceeding for the appraised fair value of his stock.
 - (2) If the appraisers' report is rejected, the court may:
 - (i) Determine the fair value of the stock and enter judgment for the stockholder; or
 - (ii) Remit the proceedings to the same or other appraisers on terms and conditions it considers proper.
- (c) Judgment includes interest.
 - (1) Except as provided in paragraph (2) of this subsection, a judgment for the stockholder shall award the value of the stock and interest from the date as at which fair value is to be determined under § 3-202 of this subtitle.
 - (2) The court may not allow interest if it finds that the failure of the stockholder to accept an offer for the stock made under § 3-207 of this subtitle was arbitrary and vexatious or not in good faith. In making this finding, the court shall consider:
 - (i) The price which the successor offered for the stock;
 - (ii) The financial statements and other information furnished to the stockholder; and
 - (iii) Any other circumstances it considers relevant.
- (d) Costs of proceedings.
 - (1) The costs of the proceedings, including reasonable compensation and expenses of the appraisers, shall be set

by the court and assessed against the successor. However, the court may direct the costs to be apportioned and assessed against any objecting stockholder if the court finds that the failure of the stockholder to accept an offer for the stock made under § 3-207 of this subtitle was arbitrary and vexatious or not in good faith. In making this finding, the court shall consider:

- (i) The price which the successor offered for the stock;
- (ii) The financial statements and other information furnished to the stockholder; and
- (iii) Any other circumstances it considers relevant.
- (2) Costs may not include attorney's fees or expenses. The reasonable fees and expenses of experts may be included only if:
 - (i) The successor did not make an offer for the stock under § 3-207 of this subtitle;
 - (ii) The value of the stock determined in the proceeding materially exceeds the amount offered by the successor.
- (e) Effect of judgment. The judgment is final and conclusive on all parties and has the same force and effect as other decrees in equity. The judgment constitutes a lien on the assets of the successor with priority over any mortgage or other lien attaching on or after the effective date of the consolidation, merger, transfer, or charter amendment.

D-5

§ 3-212. Surrender of stock.

The successor is not required to pay for the stock of an objecting stockholder or to pay a judgment rendered against it in a proceeding for an appraisal unless, simultaneously with payment:

- The certificates representing the stock are surrendered to it, indorsed in blank, and in proper form for transfer; or
- (2) Satisfactory evidence of the loss or destruction of the certificates and sufficient indemnity bond are furnished.

§ 3-213. Rights of successor with respect to stock.

- (a) General rule. A successor which acquires the stock of an objecting stockholder is entitled to any dividends or distributions payable to holders of record of that stock on a record date after the close of business on the day as at which fair value is to be determined under § 3-202 of this subtitle.
- (b) Successor in transfer of assets. After acquiring the stock of an objecting stockholder, a successor in a transfer of assets may exercise all the rights of an owner of the stock.
- (c) Successor in consolidation, merger, or share exchange. Unless the articles provide otherwise, stock in the successor of a consolidation, merger, or share exchange otherwise deliverable in exchange for the stock of an objecting stockholder has the status of authorized but unissued stock of the successor. However, a proceeding for reduction of the capital of the successor is not necessary to retire the stock or to reduce the capital of the successor represented by the stock.

D-6

QuickLinks

YOUR VOTE ON OUR PROPOSED MERGER

IS VERY IMPORTANT

Additional Information

Authorizing Proxies Electronically Or By

Telephone

Table of Contents

OUESTIONS AND ANSWERS

SUMMARY

SELECTED HISTORICAL FINANCIAL DATA

OF HCP

SELECTED HISTORICAL FINANCIAL DATA

OF CRP

SUMMARY UNAUDITED PRO FORMA

CONDENSED CONSOLIDATED FINANCIAL

DATA

UNAUDITED COMPARATIVE PER SHARE

DATA

COMPARATIVE PER SHARE MARKET PRICE

DATA AND DIVIDEND INFORMATION

RISK FACTORS

INFORMATION REGARDING

FORWARD-LOOKING STATEMENTS

THE SPECIAL MEETING OF CRP

STOCKHOLDERS

THE MERGER

THE MERGER AGREEMENT

MATERIAL UNITED STATES FEDERAL

INCOME TAX CONSIDERATIONS

INFORMATION ABOUT HCP AND MERGER

SUB

INFORMATION ABOUT CRP

DESCRIPTION OF HCP CAPITAL STOCK

ANTI-TAKEOVER CONSIDERATIONS

COMPARISON OF STOCKHOLDER RIGHTS

VALIDITY OF COMMON STOCK

EXPERTS

STOCKHOLDER PROPOSALS

WHERE YOU CAN FIND MORE

INFORMATION

INDEX TO FINANCIAL STATEMENTS

HEALTH CARE PROPERTY INVESTORS, INC.

<u>UNAUDITED PRO FORMA CONDENSED</u>

CONSOLIDATED FINANCIAL STATEMENTS

HEALTH CARE PROPERTY INVESTORS, INC.

UNAUDITED PRO FORMA CONDENSED

CONSOLIDATED BALANCE SHEET March 31,

2006 (In thousands)

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF INCOME For the year ended December 31, 2005 (In

thousands, except per share data)

HEALTH CARE PROPERTY INVESTORS, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF INCOME For the three months ended March 31, 2006 (In

thousands, except per share data)

<u>Health Care Property Investors, Inc. Notes to</u> <u>Unaudited Pro Forma Condensed Consolidated</u>

Financial Statements

Report of Independent Registered Certified Public

Accounting Firm

<u>CRP'S MANAGEMENT'S REPORT ON</u> <u>INTERNAL CONTROL OVER FINANCIAL</u> REPORTING

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE

SHEETS (in thousands, except per share data)

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF INCOME (in thousands,

except per share data)

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED

STATEMENTS OF STOCKHOLDERS' EQUITY Years Ended December 31, 2005, 2004 and 2003

(in thousands, except per share data)

<u>CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONSOLIDATED</u>

STATEMENTS OF CASH FLOWS (in thousands)

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS Years Ended

December 31, 2005, 2004 and 2003

CNL RETIRMENT PROPERTIES, INC. Schedule II Valuation and Qualifying Accounts Years Ended December 31, 2005, 2004, and 2003 (dollars in thousands)

CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES NOTES TO SCHEDULE
III REAL ESTATE AND ACCUMULATED
DEPRECIATION December 31, 2005 (dollars in thousands)

<u>CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED</u>

CONSOLIDATED BALANCE SHEETS (in

thousands, except per share data) (UNAUDITED)

<u>CNL RETIREMENT PROPERTIES, INC. AND SUBSIDIARIES CONDENSED</u>

CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED) (in thousands, except per share

<u>data)</u>

CNL RETIREMENT PROPERTIES, INC. AND

SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY Three Months Ended March 31, 2006 (UNAUDITED) (in

thousands, except per share data)

CNL RETIREMENT PROPERTIES, INC. AND

SUBSIDIARIES CONDENSED

CONSOLIDATED STATEMENTS OF CASH

FLOWS (UNAUDITED) (in thousands)

CNL RETIREMENT PROPERTIES, INC. AND

SUBSIDIARIES NOTES TO CONDENSED

CONSOLIDATED FINANCIAL STATEMENTS

Three months ended March 31, 2006 and 2005

(Unaudited)

Report of Independent Certified Public

Accountants

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED BALANCE SHEETS

December 31, 2005 and 2004

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED STATEMENTS OF INCOME

For the years ended December 31, 2005 and 2004

CNL RETIREMENT CORP. AND SUBSIDIARY

CONSOLIDATED STATEMENTS OF

STOCKHOLDER'S EQUITY For the years ended

December 31, 2005 and 2004

CNL RETIREMENT CORP. AND SUBSIDIARY CONSOLIDATED STATEMENTS OF CASH FLOWS For the years ended December 31, 2005

and 2004

CNL RETIREMENT CORP. AND SUBSIDIARY NOTES TO CONSOLIDATED FINANCIAL

STATEMENTS For the years ended December 31,

2005 and 2004

CNL RETIREMENT CORP. AND SUBSIDIARY CONDENSED CONSOLIDATED BALANCE

SHEETS (UNAUDITED)

CNL RETIREMENT CORP. AND SUBSIDIARY

CONDENSED CONSOLIDATED

STATEMENTS OF EARNINGS (UNAUDITED)

CNL RETIREMENT CORP. AND SUBSIDIARY

CONDENSED CONSOLIDATED

STATEMENTS OF STOCKHOLDERS' EQUITY

(DEFICIT) (UNAUDITED) Three Months Ended

March 31, 2006

CNL RETIREMENT CORP. AND SUBSIDIARY

CONDENSED CONSOLIDATED

STATEMENTS OF CASH FLOWS

(UNAUDITED)

CNL RETIREMENT CORP. AND SUBSIDIARY

NOTES TO CONDENSED CONSOLIDATED

FINANCIAL STATEMENTS (Unaudited)

Quarters Ended March 31, 2006 and 2005

AGREEMENT AND PLAN OF MERGER by and

among HEALTH CARE PROPERTY

INVESTORS, INC., OCEAN ACQUISITION 1,

INC. and CNL RETIREMENT PROPERTIES.

INC. Dated as of May 1, 2006

TABLE OF CONTENTS

ARTICLE I DEFINITIONS

ARTICLE II THE MERGER

ARTICLE III REPRESENTATIONS AND

WARRANTIES OF THE COMPANY

ARTICLE IV REPRESENTATIONS AND

WARRANTIES OF PARENT AND

PURCHASER

ARTICLE V CONDUCT OF BUSINESS

PENDING THE MERGER

ARTICLE VI ADDITIONAL AGREEMENTS

ARTICLE VII CONDITIONS TO THE MERGER

ARTICLE VIII TERMINATION, AMENDMENT

AND WAIVER

ARTICLE IX GENERAL PROVISIONS

Banc of America Securities Opinion

Houlihan Lokey Howard & Zukin Financial

Advisors, Inc. Opinion

Sections 3-201 et. Seq. of Maryland General

Corporation Law